Annotation in nurse education: Towards a hermeneutic understanding

By Paul Regan

Submitted in Partial Fulfilment of the Requirements of the

Degree of Doctor of Philosophy, October 2016

School of nursing, midwifery, social work and social sciences,

University of Salford

Table of Contents

Acknowledgements	
Preface	13
Abstract	14
Chapter 1: Introduction	16
-	
1.1 Aim of the research study	
1.1.1 Introduction	17
1.1.2 Setting the scene	17
1.1.3 The task of hermeneutic phenomenology	
1.1.4 Personal reasons for the study	19
1.2 Annotation	22
1.2.1 Annotation and nurse education: An initial exploration	24
1.2.2 Moral sense of duty	26
1.2.3 Reflective practice: NHS and higher education institution	
1.2.4 Reflection for nurses in the NHS	29
1.2.5 Reflection for nurse lecturers in HEIs	
1.2.6 Exploratory questions	
1.2.7 Evidence of annotation use in nurse education	
1.3 A guide to the chapters	
1.3.1 Chapter Two: Background	
1.4 Chapter Three: Literature review	

1.5 Chapter Four: Theoretical and philosophical positioning	34
1.6 Chapter Five: Study design and methods	34
Table 1: Index of study design and methods	35
1.7 Chapter Six: Initial findings and four research themes	35
1.8 Chapter Seven: "The hermeneutic self"	
1.8.1 Chapter Eight: "Rhetoric and saying it well with proof"	36
1.8.2 Chapter Nine: "Individualism"	36
1.8.3 Chapter Ten: The "reflective consciousness and slippage"	36
1.8.4 Chapter Eleven and Twelve	36
1.9 Conclusion	36

Chapter Two: Background	
2.1 Introduction	
2.1.1 Annotation as an interpretation	
2. 2 Assessment of learning in nurse education	40
2.3 Annotation and nurse education	42
2.4 History of annotation	45
2.5 Social meaning of annotation	46
2.6 The academic essay	
2.7 The benefits of reading	49
2.7.1 Be suspicious of the text	51
2.7.2 Suspicion of the nursing essay	
2.8 Standardisation of essays	53

2.9 Conclusion

Chapter Three: Literature review	57
3.1 Introduction	57
3.2 Method	
3.3 Search strategy	58
Figure 1: Search strategy key words	59
Figure 2: Inclusion and exclusion criteria	59
3.3.1 Hermeneutic process and the literature review	60
3.3.2 Structured literature review: An adaption	60
3.3.3 Reduced reading bias through the hermeneutic circle	61
Figure 3: Adaptation of Boell & Cecez-Kecmanovic (2010) model to the hermo	eneutic circle and
literature review	63
3.4 Hermeneutic circle and choice of publications reviewed	63
Figure 4: An example of the hermeneutic circle at work in the literature re-	eview: Feito and
Donahue's (2008) study (see appendix 1)	65
3.4.1 Entry and exit points: Saturation	67
3.4.2 Retrieved search 1965 to 2016	
Figure 5: Literary sources from retrieved search 1965 to 2016	69
Figure 6: Research studies 1965 to 2016	72
3.5 Retrieved literature	77
3.5.1 Annotation in education	79
3.5.2 A lack of evidence	79

3.5.3 Annotation research studies	80
3.5.4 Four dimensions: Competence, tone, appearance and fitness	80
3.5.5 Print based and digital annotation: Capture and share	81
3.5.6 Imitation.	87
3.5.7 Forgetting thoughts and feelings when annotating	
3.5.8 Self-efficacy and ownership	
3.6 Consistency building	92
3.6.1 Criticism- not hermeneutic and knowing colleagues practice	93
3.6.2 Experienced lecturers and reflection	94
3.6.3 Presence and tone	
3.7 Tone and evoked emotion	98
3.7.1 Tone neutrality	
3.8 Discussion	101
3.8.1 Hidden from the annotator	102
3.9 Conclusion	
Chapter Four: Theoretical and philosophical positioning	107
4.1 Introduction	107
4.1.1 Aim of research study	107
4.2 The theoretical approach	108
4.3 The phenomenological influences on Ricoeur	109
4.4 Hermeneutics and phenomenology	109
4.5 Phenomenology versus empirical science	111

113
114
116
117
120
120
123
124
128
130
132
134
135
135
135
135 136
135 136 137
135 136 137 137
135 136 137 137 138
135 136 137 137 138 139
135 136 137 137 138 139 139
-

Figure 10: Summary of interviews with students and lecturers at the second university	143
5.5.2 Interview participants: Student focus group and individual interview	143
Figure 11: Demographic details of interviewed students	144
5.5.3 Interview participants: Lecturers	144
Figure 12: Demographic details of interviewed lecturers	145
5.5.4 Interview questions: Follow the lead	145
Figure 13: My issues during the interviews with lecturers and students	149
5.5.5 Post interview analysis	150
5.5.6 Transcriptions	151
5.6 Method of data analysis- to use software or not	152
5.6.1 The hermeneutic circle	154
5.7 Three analytical processes within the hermeneutic circle	157
5.7.1 First order interpretation	157
5.7.2 Second order interpretation	158
5.8 Third order interpretation	159
5.9 Conclusion	161
Chapter Six: Initial findings and four identified themes	163
6.1 Introduction	163
6. 2 Emergent themes leading from first and second order interpretation	163
Table 2: Index of figures and notes	164

6.3 Student focus group interviews (n=20).....165

6.3.1 Explanation of focus group research notes and grid	171
6.3.2 Pyramid and triangle significance	171
6.4 Individual student interviews (n=5)	172
Figure 15: Extracts from researcher notes of individual student interview (n=5)	172
6.4.1 Student 1: Explanation of the research notes and grid	175
6.4.2 Student 2: Explanation of the research notes and grid	186
6.4.3 Student 3: Explanation of the research notes and grid	189
6.4.4 Student 4: Explanation of the research notes and grid	192
6.4.5 Student 5: Explanation of the research notes and grid	194
6.5 Lecturer interviews (n=8)	195
Figure 16: Extracts from researcher notes of individual lecturer interviews (n=8)	195
6.5.1 Lecturer 1: Explanation of the research notes and grid	197
6.5.2 Lecturer 2: Explanation of the research notes and grid	199
6.5.3 Lecturer 3 and 4: Explanation of the research notes and grid	
6.5.4 Lecturer 5: Explanation of the research notes and grid	207
6.5.5 Lecturer 6: Explanation of the research notes and grid	212
6.5.6 Lecturer 7: Explanation of the research notes and grid	217
6.5.7 Lecturer 8: Explanation of the research notes and grid	220
6.6 First and second order interpretation from digital annotation extracts	
Figure 17: Extracts from researcher notes of digital annotation	221
6.6.1 Explanation of digital annotation research notes and grid	224
6.7 First and second order interpretation from handwritten annotation extracts	225
Figure 18: Extracts from researcher notes of handwritten annotation	

6.7.1 Explanation of handwritten annotation research notes and grid	.228
6.8 Second order interpretation and sub-themes	.228
Table 3: Figures 14 to 18 collated in three steps.	.229
Figure 19: Third order interpretation: Four identified research themes	.232
6.9 Conclusion	.233

Chapter Seven: "The hermeneutic self"	.234
7.1 Introduction	.235
7.2 Identifying the theme	.237
7.3 Defining "it"	.238
7.4 The annotator	.239
7.5 Temporal understanding	.241
7.6 The student and embodiment	.242
7.7 The student self in an essay	244
7.8 Moving beyond itself	.246
7.9 Conclusion	.247

Chapter Eight: Rhetoric and "saying it well" with proof	
8.1 Introduction	250
8.2. From level two to level three analysis: Naming through tropes	250
Figure 20: Ricoeur's new rhetoric	251
Table 4: Tropes entitled as research data extracts	251
8.3 Plot	252

8.3.1 Plot and proof	255
8.4. Metaphor and the productive imagination	
8.4.1 Metaphor and the embodied experience	
8.5 Mimesis and temporal action	
Figure 21: Ricoeur's model of temporal action through mimesis1-3	
8.5.1 Ricoeur's model of temporal action through mimesis1-3	
8.5.2 An analogy of mimesis: The glass of water	
8.6 Mimesis1	
8.6.1 Dual role	
8.7 Mimesis2 and plot: As if	
8.7.1 My glass is full	273
8.8 Mimesis3	275
8.9 Conclusion	

Chapter Nine: "Individualism"	283
9.1 Introduction	284
9.2 Working out the theme: Setting the scene	284
9.3 A working definition of "individualism"	287
9.4 Responsibility and the "just"	288
9.5 The annotator as a citizen: Promoting equality	290
9.6 In defence of nursing: Self and others	293
9.7 The use of imperative language	294
9.8 Technology	229

9.9 Conclusion	
Chapter Ten: "Reflective consciousness and slippage"	
10.1 Introduction	
10.2 Identifying the theme	
10.3 Reflective consciousness	
10.4 Single word tropes	
10.5 Imperative and proof	
10.6 Memory and the absence of the thing remembered	
10.7 An ethico-moral level of memory	
10.7.1 A practical level of memory	
10.7.2 The wounded memory	
10.8 The reflective unconscious	
10.8.1 Perceived tone	
10.8.2 The student as a child	
10.9 Conclusion	
Chapter Eleven: Discussion	
11.1 Introduction	
11.2 The hermeneutic self	
11.3 Rhetoric	
11.4 Individualism	
11.5 Reflective consciousness and slippage	
11.6 Part 2: Limitations of the study and research methodology	

11.7 Conclusion	
-----------------	--

Chapter Twelve: Recommendations and implication for practice	
12.1 Introduction	351
12.2 Recommendations	
12.3 Training	351
12.4 Transferability	
12.5 Formative and summative annotation	
12.6 Promoting student reflection	
12.7 Discursive versus scientific essays	
12.8 Promoting annotator reflection	
12.9 Technology	
12.9.1 Implications for further research	355
12.9.2 Recommendations and implication for practice	

References	
Appendix 1: An example of the hermeneutic circle in the literature review	374
Appendix 2: An example of an edited transcript of interview: Lecturer interview 8	
Appendix 3: Research notes from chapter seven	
Appendix 4: Research notes from chapter eight	
Appendix 5: Research notes from chapter nine	
Appendix 6: Research notes from chapter ten	401

Acknowledgements

I would like to extend my grateful thanks to my primary supervisor Dr Elaine Ball who has given me unstinting academic support. I would like to thank Professor Martin Johnson who started out as co-supervisor and Professor Gregory Smith who supervised me to completion as cosupervisor. In particular, considering the subject matter of this thesis, their use of annotation proved to be inspirational and exemplary. On a personal note I would like to thank all of my family, my extraordinary wife Florence and our sons Daniel, Luke and Thomas for their love, patience and understanding.

Preface

My nursing career began in 1982 in adult nursing and 1986 in mental health nursing until 1996 when I trained as a health visitor, practising until 2010. I have worked in higher education part time between 2004 and 2007 and full time as a lecturer since 2010 in pre- and post-registration adult nursing. Since 1996, I have continued to work as a bank staff nurse in acute mental health nursing and psychiatric intensive care.

Abstract

Aim: To explore the meaning of annotation in nurse education within higher education

Background

Annotation is a common practice in higher education pedagogy aimed at communicating the lecturer's comments about an assignment back to the student. A literature review identified a dearth of research available to inform annotation and its use in nurse education was generally inductive and learnt from experiences of giving and receiving annotation feedback.

Method/methodology

The research methods included one focus group interview with nursing students (n=20), individual interviews with nursing students (n=5), individual interviews with lecturers (n=8) and a selection of annotation extracts from one hundred essays, with digital annotation (n=50) and handwritten annotation (n=50) from two universities. The research data was analysed using Ricoeur's textual hermeneutics.

Findings: Research themes

Four research themes explore the meaning of annotation in nurse education. The first theme, the "hermeneutic self" explores the hermeneutic process of reading and writing, and making sense of discourse. The second research theme, "rhetoric" explores Ricoeur's new rhetoric in the form of temporal action called mimesis1-3. The third research theme called "individualism" explores social justice, negotiating the political labyrinth, and the annotator's sense of moral autonomy to

act on behalf of society. The fourth research theme, the "reflective consciousness and slippage" develops the transference hypothesis and memory recall (Ricoeur, 2006).

The original contribution to current knowledge

A Ricoeurean textual hermeneutic contributes to a better understanding of the gaps in current nurse education knowledge. Ricoeur's organising principle of temporal action informs the processes of student misrecognition, misunderstanding and the reading self interpreting the work of an-other. Ricoeur's new rhetoric can be seen in the instinctive use of suasory discourse that shapes annotation in nurse education. Annotation is advisory, judgemental and powerful. The annotator as a citizen aims to promote a "defence of nursing" against the effects of the political labyrinth, disembodiment and technology. However, with an essay considered a safe space to think in preparation for the rigours of clinical nursing practice, the recall of past events refigured for the present may lead to something useful or not being communicated to the student. **Chapter One**

Introduction

1.1 Aim of the research study

The aim of this hermeneutic phenomenological research study is to explore the meaning of annotation in nurse education situated within the United Kingdom's (UK) higher education institutions. In particular, the study is an opportunity to focus on the dynamics of reading, writing and interpretation for student essays. The act of annotation refers to the communication of one person's thoughts about text communicated back to the author in written form (DiYanni, 2002). This makes annotation a form of discourse involving the act of interpretation (Derrida, 1982).

1.1.1 Introduction

This chapter prepares a grounding for the thesis. I discuss the assessment processes within nurse education, define what annotation is and introduce Paul Ricoeur's textual hermeneutics selected as a method from which to explore the meaning of annotation in nurse education. I discuss the personal reasons for writing the thesis and explore how nursing, annotation and the protection of the public form the basis for this study. I outline nurse education reforms over the last twenty years, discuss their impact and explore reflection, evidence and language in nurse education. Finally, the chapter finishes with a brief summary of each chapter.

1.1.2 Setting the scene

The *Quality Assurance Agency for Higher Education* (QAA) produced the *UK Quality Code for higher education* (2012a) to support students achieve learning through a variety of opportunities, assessment methods and learning styles. The report suggested that feedback to students should aim to give constructive advice to build confidence on what they have done well and develop further (QAA, 2012a). The QAA (2012b) guidelines entitled *Understanding assessment: Its role in safeguarding academic standards and quality in higher education* identifies the module leader being responsible for choosing appropriate assessment methods, such as an essay on a nursing topic with guidelines aligned to the module content. One of the feedback and assessment methods in nurse education is annotation on student essays from lecturers to students.

First, let me discuss the known processes involved. The student reads the essay guidelines to understand how to meet the module learning outcomes. The student searches, finds and reads nursing literature and uses their judgement to interpret its relevance to the essay. Despite feedback being offered in different textual ways, such as by email or directly on the draft essay using something like "Microsoft[®] Word" document track changes, students may or may not seek academic supervision. The student continues to write, make revisions and finally the work is marked (DiYanni, 2002). The annotator reads the essay, interpret its contents, assesses its relevance in meeting the module learning outcomes and annotates on the essay for the student to read (Regan, 2010). The student then receives their mark and reads the annotator's comments about their essay (Regan, 2010). Some students may not read the annotation comments but for the most that do, they need to understand the annotation in order to further improve the essay for re-submission or learn for the next essay (QAA, 2012a).

I have so far discussed the known processes involved citing QAA guidelines, yet it already appears annotation and essay writing is far from a simple task than at first glance. Even though annotation is held in check by QAA guidelines, it is also a singular interpretive activity. All of these processes involve acts of reading, interpreting and writing from at least three different perspectives; sometimes more (Connors & Lunsford, 2006). Yet all these layers do not take into consideration the fact that the author's literary work itself may be an act of interpreting someone else's work (Ricoeur, 2003). The act of interpretation, in this situation, needs to take into consideration there may be a second marker or external moderator reading the essay for evaluative purposes, layering on more interpretation. In the context of this thesis, the annotator is an experienced nurse, reader, and senior lecturer giving feedback to an inexperienced nurse, reader and essay writer to add to interpretive difficulty (Regan, 2010). If the student, annotator, second marker or the external moderator have not read the same literature (which is unlikely), interpretation may become problematic because individuals interpret discourse in relation to the meaning it has for them (Derrida, 1982; Ricoeur, 2003). The next section examines why a hermeneutic approach to exploring the meaning of annotation in nurse education is worthy of study.

1.1.3 The task of hermeneutic phenomenology

The word "meaning" in the title of this thesis is purposeful. Ricoeur (2008) suggests the word "meaning" is making sense of something in order to "understand" it, to recognise and "grasp" meaning, which I examine conceptually in chapter 4.7 to 4.7.1. Ricoeur (1998a) suggests in achieving meaning and understanding of discourse, the reader has to negotiate the polysemy of language. Polysemy refers to the multiple meanings words have which result in imperceptible

skips and assumptions being made in order to understand (Ricoeur, 1998a). Ricoeur (1998a) refers to the polysemic of multiple meanings as a hermeneutic problem of language addressed only by revealing the hidden meaning of language. The difficulties I negotiate in this thesis first lie in the fact that the significance words have for a person may not carry the same significance for another (Ricoeur, 2003). Interpretation of discourse, whether read or written, changes in time as new knowledge is assimilated into the world view of the reader and what Ricoeur (1998a) called temporal action. I examine temporal action in three of the four research themes of "rhetoric" and "saying it well" with proof in chapter eight and "reflective consciousness and slippage" in chapter ten. Therefore, a nuanced form of pedagogy and the choice of Paul Ricoeur's textual hermeneutics as a research methodology to explore the meaning of annotation in nurse education, is a great resource for the analytical task ahead.

1.1.4 Personal reasons for the study

There are three key influences, which initiated this study. First, my experience of giving and receiving annotation feedback led me to believe there was "something more" communicated back to the addressee. The essay content that may have an effect on the annotator are: mistakes, lack of academic rigour or effort, poorly written text, content with attitude, prejudice and errors of judgement, lack of evidence and phraseology, to name but a few causing frustrations (Connors & Lunsford, 2006). As a nurse lecturer, I use annotation and knew at times, instead of my immediate thoughts; I had to phrase my annotation sensitively if aspects of the essay content had an impact on me. I realised my experience of clinical nursing practice had an impact on thinking and emotional processing which I brought with me into nurse education (see section 1.2.7). I had

monthly voluntary clinical supervision on my National Health Service (NHS) work which I had valued highly but did not have the same opportunities to reflect on practice in nurse education.

The second influence was when undertaking a post graduate certificate in higher education at the University of Salford, and I chose to focus on annotation using action research which was later published (Regan, 2010) and critiqued (Ball & Regan, 2013). I then became aware of the term pedagogy. Knowles (1973; 1984) first brought adult education to the attention of educationalists to suggest that adults learn differently from children, calling his theory andragogy. Rather than a child being perceived to "soak up" information, to seek approval from their teachers, to be told what to do and what to learn, adults in contrast were viewed as being self-directed learners (Knowles, 1984). Adult learning allows for practical learning applied to real life. However, in order to progress in nurse education towards registration I realised there was an overlap between the old ideas about pedagogy and andragogy, in that adults are taught what someone else has decided they needed to learn (Knowles, 1973; 1984). Therefore, when told what to do and what to learn. What to learn, the conditions are ripe for mis-interpretation due to a lack of motivation to learn. What the evidence said about annotation became more pertinent to my practice.

When performing a literature review I found there was a lack of research in the higher education literature about annotation in nurse education (Regan, 2010). I found 12 articles between 1997 and 2009 on annotation (Regan, 2010) and a significant paper by McColly in 1965. By 2016, the number had expanded to 13. Five of the research articles were from nurse education (Ball, Franks, Jenkins, McGrath & Leigh, 2009; Ball, 2009; Ball 2010; Regan, 2010; Ball & Regan, 2013) with one literature review (Ball, 2009). The general themes identified annotation being a

nurturing process, the benefit of formative feedback, annotator's reflection, students influenced by annotation, annotators lacking training, annotation being an intuitive, inductive process and having a negative tone. With nursing embracing evidence based practice wholeheartedly to inform clinical decision making (Skorga & Young, 2014), it came as a surprise that nurse education had a poor base from which to inform practice. Therefore, I believed there was something extraordinary about annotation in its interpretation, reading, writing, linguistics and discourse.

The third influence on the study was to trust that Paul Ricoeur's textual hermeneutics could offer an appropriate methodology to explore the meaning of annotation for nursing education. I say trust, because before a wider reading of Ricoeur, I had to trust what little I had read of his work to inform the meaning of annotation in nurse education. I was searching for meaning; understanding, depth and a method that would help develop an intellectual repertoire. Conversely, I wanted to make sense of the experience for those who gave and received annotation and an analysis of consciousness, with me described as the explorer within the hermeneutic phenomenological process (Husserl, 2014; Ricoeur, 1996b). The below reflection was written after receiving editorial annotation on a paper I was trying to get published and illustrates a key motivation for me to start to write this study:

"I knew first-hand what it felt like to receive annotation feedback from an editorial reviewer which I perceived to be negative, painful to read but useful nonetheless. I began to wonder what informed annotation, was it reading and writing experience, what was its theoretical and evidence base for nurse education and if I were to learn about its dynamics how could annotation be improved for educational purposes? Nursing experience had not equipped me to understand the nuances of the text, let alone understanding annotation and due to the tone and irritation in the annotation; I sensed it was a complex form of feedback. Whether the annotation had any underlying hidden communications it was clear that my text had irritated the annotator who then went on to irritate me, in a repetition of the act of reading, I wondered therefore what impact hidden discourse had on nursing students. What was being projected and why?"

The issues appear to include making sense of reading, writing, promoting a constructive learning experience for students, being aware of the nursing context and its purpose and understanding the part I and fellow academics, play within it. The reflective extract identifies a strong sense of empathy and motivation to understand the processes involved. A brief introduction to annotation in different genres follows to prepare an analysis for nurse education and annotation.

1.2 Annotation

Annotation is a widespread educational practice used for assessment of student learning within higher education (Feito & Donahue, 2008). Lecturers use annotation to assess the content of student essays to improve the writing process and marks awarded (Feito & Donahue, 2008; QAA, 2012a). Annotations are brief notes, single words, and phrases about a text questioning or identifying a point of interest (DiYanni, 2002). Yet, annotations are a varied and idiosyncratic phenomenon, made in the margins or on the page to indicate the reader understands and reacts to the message of the text (Derrida, 1982; DiYanni, 2002). Annotations are therefore atypical and refer to any additional telegraphical marks made on a page such as underlining, circling,

highlighting text and drawing arrows, lines or symbols within the rubric of annotation (DiYanni, 2002). Annotations therefore act as an *aid memoire* (DiYanni, 2002). However, a significant issue is that they are often the reader's initial thoughts about the text and so are more than likely reactionary statements rather than measured comments (Jackson, 2001). This issue will be discussed later in relation to the theory and the research theme of "rhetoric" and "saying it well" with proof I examined in chapter eight (see chapter 8.5 onwards). The main point to remember is that this temporal action is immediate and an initial and reactive understanding of the text (Ricoeur, 2007). The meaning text has on the reader is constructed from the inferences of the text and immediacy of interpretation affected by the amount of reading, time constraints and temporal action as will be demonstrated in the later chapters (Ricoeur, 1985). Due to the lack of research and articles available to inform nurse education, literary disciplines offer a rich insight into the dynamic culture of annotation, which is of relevance to non-linguists using an inductive annotation method (Di Yanni, 2002).

Annotation in humanities subjects is more common than other disciplines, yet even here there is little consensus about the presentation of annotation (Jackson, 2001). Annotation is found in all forms of literature from antiquity, with most found in archived books (Jackson, 2001). Henige (2002) explored annotation's impact in literary circles at length but it remains ambiguous. Annotation, as telegraphic signs, can appear like rough jottings, disorderly semiotic signs and, lying in the margins, they may often be ignored (Jackson, 2001). The reader-annotated book is often scruffy, ink stained, with unintelligible scribbles defacing the page. This may remind the reader of second-hand books with pages littered with cartoons, circles, arrows, witticisms and cryptic comments from a previous time. What annotation bring is the presence of the previous

reader to the current one, which can be either off-putting or welcome, offering an opportunity to benefit from the others' insight and point of view (Jackson, 2001).

When the British Library in 1998 bought Galileo's letters he wrote in 1613 on sunspots, entitled *Isatoria e di-mostrazioni intorno alle macchie solari* (Galilei, 1613/1932), it demonstrated that annotation could increase the worth of a book. This second edition book was of particular interest because of the addition of annotation written in the margins of the book by unknown Italian contemporaries in the seventeenth century (Jackson, 2001). The key word here is "translation" because a contemporary of Galileo at the time had added his or her own translations in Italian making translation more complete (Henige, 2002). Translation gives the text a future because it activates in the mind of the reader, a new consciousness (Ricoeur, 1998a; 2009). The process may be either passive or active. For example, the translation of the text using annotation in scholarly, editorial and critical writing appears to be combative, political, personal and sometimes written to confront and undermine the author (Benstock, 1983). In educational research and practice, as chapter three's literature review demonstrates, the annotation phenomenon is largely unexplored and idiosyncratic and this thesis aims to address this issue by exploring annotation conceptually.

1.2.1 Annotation and nurse education: An initial exploration

This next section explores how nursing, annotation and the protection of the public fit together to form the basis for this study. Various guidelines from the UK's Quality Assurance Agency for Higher Education (QAA) and the regulating body Nursing and Midwifery Council (NMC) aim to set educational standards for higher and nurse education. Nurse education falls under the QAA remit and has a statutory duty to ensure that nursing students complete a higher education programme and are fit to practice prior to registration (NMC, 2010; 2011; 2013; 2015a; 2015b). The translation of nursing knowledge through essay feedback aims to ensure the protection of the public through pre and post registration standards for nurse education (NMC, 2010; 2013). The NMC (2015a) *Professional standards of practice and behaviour for nurses and midwives* called *The code* suggests nurses should act on their understanding of how people's lifestyles, environments and care delivery influence health and wellbeing. These guidelines promote the need for clear communication, understanding, moral and ethical practice and reflective practice. I will develop the need for reflective practice in section 1.2.3 to 1.2.5 and later in the research theme of "reflective consciousness and slippage" in chapter ten.

The QAA (2012b) guide entitled *Understanding assessment: Its role in safeguarding academic standards and quality in higher education*, promotes assessment in a number of ways and one of them is through feedback written on the page of the essay. Annotation given on formative and summative essays has an expectation that students make the necessary changes to their essay (QAA, 2012a; 2012b). However, the QAA guide concedes that changes made on the essay remain largely unknown because the student assesses whether annotation helped them revise their essay content whilst not recognising its benefits (QAA, 2012b, p. 14). What we do know is that the content of the student's essay has an effect on the annotator who in turn has an effect on the student (Knoblauch & Brannon, 2006).

The literature on annotation however, suggests promoting clear communication in examples of nursing annotation (Regan, 2010). Nor is reflective practice on annotation promoted universally

(Ball et al., 2009). If annotation fails to be clearly understood, it fails to convey the principles and standards about nursing promoted by the NMC (NMC, 2015b; Regan, 2010). This is of concern because of the potential to misunderstand knowledge into practice reinforced in all feedback processes especially annotation, which I examine in the next section and later in the research theme entitled "individualism" in chapter nine.

1.2.2 Moral sense of duty

Four issues appear to be relevant. First, the empathic nature of nursing relates to the concept of duty, beneficence, good will and the preservation of life. Nursing is a practical profession, which engenders a strong sense of public approval and professional identity (Henderson, 1966; NMC, 2015a). There are many reasons for entering a profession with a sense of duty and the social contract that binds people together (Ricoeur, 2003). A strong moral compass is rewarding until organisational constraints potentially impede the quality of care given to the nurse's satisfaction, and then a form of moral distress occurs when knowing the right thing to do cannot be achieved (Smith & Allan, 2010). This is an important issue because when a nurse is recruited into nurse education they remain guided by a moral code (Smith & Allan, 2010) which I examine in the research themes of "individualism" in chapter nine and the "reflective consciousness and slippage" in chapter ten.

Second, Project 2000 made changes to nurse education provided by the hospital (Clifford, 1993). Nurse education then changed to come under the umbrella of HEIs in the mid 1990s and nurse teachers then became lecturers and nursing students became supernumerary (Allan, Smith & O'Driscoll, 2011). There were initially two roles, the nurse tutor and clinical nurse teacher (Clifford, 1993); however, the separation of nurse education from the National Health Service (NHS) led to both being combined in the nursing lecturer (Price, Hastie, Duffy, Ness, & McCallum, 2011). New priorities of academic achievement for nurse education (diploma, degree) and research activities diverted lecturers away from their clinical role to be academic. Smith & Allan (2010) suggest this separation led to lecturer uncertainty and feeling in between nurse education and the NHS and feeling somewhere in the middle. Hence, working in higher education appeared to devalue teaching of nursing care over theory and research (Smith & Allan, 2010). These issues are discussed in the research themes of "rhetoric" and saying it well" with proof in chapter eight, "individualism" in chapter nine, and the "reflective consciousness and slippage" in chapter ten.

Third, due to the above changes to nurse education, students changed their perception of lecturers to be far removed from clinical practice and having an idealised view of nursing (Steven, Magnusson, Smith, & Pearson, 2014). This critical view of lecturers in nurse education is perhaps naive when considering the amount of clinical experience a lecturer may have had before entering nurse education, and their commitment is evident by entering nurse education (Price et al., 2011). Lastly, one significant issue to discuss for the preparation of this thesis is the transition from clinical practice to higher education (Wilson, 2013). An imbalance occurs because the nurse lecturer "knows" more about clinical nursing than nurse education and this transition can take years to change (Wilson, 2013). This is evident when new nurse lecturers experience issues of self-identity, anxiety and academic identity (Wilson, 2013).

The bridge to overcome this theory-practice gap is through reflective practice as suggested by the QAA's (2012b) *Understanding assessment: Its role in safeguarding academic standards and quality in higher education. A guide for early career staff* (indicator 2, p. 14 and indicator 9, p. 21) which I discuss as a grounding for the thesis.

1.2.3 Reflective practice: NHS and higher education institutions

The QAA (2012b), the NMC (2011) *Standards to support learning and assessment in practice* and NMC (2010b) *Standards for pre-registration nursing* support the suggestion that reflection is relevant to annotation practice. Reflection, according to Dewey (1997) is an experimental process involving the mind leaping ahead to form connections, hypothesis and trial and error processes. However, the potential for understanding first relates to a nurse's capacity to reflect on their clinical practice (pre and post registration) and becoming nurse lecturers (Wilson, 2013). This is a critical point I examine as a research theme in chapter ten entitled the "reflective consciousness and slippage" and so I will discuss the importance of reflection a little more in relation to the NHS and HEIs. Nurses have a moral duty to understand their own actions and reflection is concerned not with the actions that are obvious to others but with personal motivations (NMC, 2015a; Regan, 2008).

The NMC *The code* (2015a) suggest nurses must actively be involved in clinical supervision and learn from experience, reflection and evaluation. The moral motivation for reflection addresses a nurse's need to think about the intentions of their actions to meet the public and professions expectations of fitness for practice (NMC, 2015a). However, the evidence internationally and in the UK suggests there is an inadequate uptake of reflective practice in models such as clinical

supervision and mandatory models of peer observation of teaching (King, Garcia-Perez, Graham, Jones, Tickle & Wilson, 2014; McMahon, Barrett, & O'Neill, 2007). I will first discuss reflection for NHS nurses in practice before linking it to when nurses become lecturers in higher education.

1.2.4 Reflection for nurses in the NHS

Nurses receive training and supervised practice through mentorship (Butterworth, Bell, Jackson & Pajnkihar, 2008). For qualified nurses and health visitors, clinical supervision aims to promote reflective nursing practice (Butterworth et al., 2008). Clinical supervision is an opportunity for time, guided reflection, support and learning and is a developmental and incremental process promoting insight, self-efficacy and reflective capacity (Hawkins & Shohet, 2012). However, one research study identified that one third of nurses have experienced some form of clinical supervision during their careers with no universal model being identified (Bishop, 2008). If clinical supervision does occur in nursing, it is likely to be management led, outcome driven and superficial due to a restrictive view of autonomous learning (Yip, 2006).

The evidence for the uptake of clinical supervision however is poor and the majority of UK nurses do not receive or engage in it (Butterworth et al., 2008). Research identifies poor access to clinical supervision for health visitors, other than safeguarding children supervision (Regan, 2012a; Regan, 2012b) and hospital nurses (Koivu, Hyrkäs & Saarinen, 2011). The variations regarding the uptake of clinical supervision are widespread with 18% of practice nurses in Leicestershire engaging in it, compared to 85.9% of mental health nurses in Northern Ireland (Butterworth et al., 2008). The findings indicate cultural differences between disciplines and

adult nurses less likely to experience guided forms of reflection from clinical supervision (Butterworth et al., 2008). Such findings are of concern for nurse education because this is the pool that higher education recruit from and reflective practice commenced in the NHS should then continue in higher education. As I demonstrate later in the research themes, reflective practice is of concern because it is likely there is the layer of meaning communicated in annotation than may be otherwise intended.

1.2.5 Reflection for nurse lecturers in HEIs

Generally, there is a lack of opportunity for reflective practice in HEIs (Gelter, 2003) and the annotation tone I locate in chapter 3.7 and research themes seven to ten, suggest this impacts on the meaning of annotation. In educational practice, a need for reflective practice is well known but there are few examples of guided and supportive reflective models (Gelter, 2003). The lack of reflective opportunities is surprising considering the amount of reading commented upon every day, not to mention other aspects of lecturing practice involving discourse and its analysis (Hays & Gay, 2011). Despite the NMC (2015a) code promoting reflective practice, a failure to reflect on higher education practice can lead to poor decision making, poor practice, insight and judgment (Hays & Gay, 2011). Institutional obstacles to reflective practice include business models for higher education, reduced lecturer autonomy, a culture of failing to challenge existing educational practice and a lack of peer support (Loughran, 2002). Other constraints are unreasonable curricular demands, a lack of time, and a lack of observation skills training (Loughran, 2002). These obstacles therefore potentially reduce professional autonomy and increase lecturers' critical feelings of anxiety, fear, helplessness, isolation and sense of being

valued (Raj, 2013). Again, I suggest these issues can affect the meaning of annotation and the research themes indicate this too.

1.2.6 Exploratory questions

A few exploratory questions appear to be relevant at this stage. What can a textual hermeneutic of annotation add to nursing practice, what is the meaning of annotation and what is communicated? This last question is significant in that the meaning of annotation relates to understanding discourse in general which is related to linguistic education and training. Therefore, a hermeneutic of annotation offers a critical reflective space to explore the language-in-use.

1.2.7 Evidence of annotation use in nurse education

As stated earlier, there is evidence of what annotation is, but little research on the meaning of annotation within higher education. In précis, and for the purpose of this introduction, between 1965 and 2016, there were only 13 research studies published on the use of annotation in education. As previously discussed only five related to any healthcare profession and all were from nursing. There was consensus that annotation is generally based on experience and instinct and rarely taught within educational practice yet nurse education aims to work from a strong evidence-base. How annotation helps students understand the text as an *aid memoire* for the reader or as a comment for other readers is relatively unknown.

More critically, in a nursing context, if students struggle to interpret annotation, then there are implications for applying knowledge in clinical practice. This is because if the translation of academic study is incomplete there may be an impact on nursing practice at some point in the future. Preventing nurses doing more harm than good relate to standards for professional practice and evidence-based healthcare within the NHS (NMC, 2015a). For example, the QAA (2012b, p.15) report *Understanding assessments: Its role in safeguarding academic standards and quality in higher education*, states that lecturers should demonstrate clearly to students where an essay could improve in terms of content, quality and grading marks. This requires principles of good communication and mutual expectation between lecturer and student about what constitutes a good essay. The report does not indicate how widespread the use of annotation is in higher education however, there is some way to go before higher education practice is fully research informed.

Annotation in nurse education, indicated by the low numbers of research studies, lacks an appropriate evidence base despite its widespread practice. This means a lack of evidence for annotation practice will have a direct impact on students' understanding of knowledge and evidence based clinical practice. However, if one were to parallel the acceptance of a lack of evidence in nurse education with clinical nursing, and in an instant one would likely conclude there was a risk of superficiality, poor engagement, outcomes and assessment with more opportunities missed than identified (Regan, 2010). Therefore, student learning is potentially inadequate because the relationship between lecturer and student are both important and complex, and a lack of research increases the potential for a disintegrated learning environment (Allan et al., 2011). If academic teaching and learning is to be effective, it is important that annotation practice is more than an equivocal process. Therefore, this thesis is important because it provides a deeper exploration of the meaning of annotation in nurse education and higher

education. I will now briefly summarise the contents of each chapter and introduce the research themes in more detail.

1.3 A guide to the chapters

1.3.1 Chapter two defines the background to annotation, assessment and feedback in more detail. Although already discussed briefly in chapter one, I use some of the research extracts from the literature to locate annotation in relation to its history impacting on people's lives through legal, theological and socio-cultural dissemination. I discuss the importance of reading, writing, memory, language and the essay as a predominant form of assessment.

1.4 Chapter three outlines a hermeneutic approach to literature reviews and identifies the hermeneutic circle's importance in choosing publications and interpreting their contents (Boell & Cecez-Kecmanovic, 2010, see table 1, chapter 1.6). The latter is important because the literature and interpretation help to challenge my pre-conceived ideas about annotation through a wider search of literature, other than educational annotation. The inclusion and exclusion criteria and database searches map the publication findings through a flexible, intuitive and hermeneutic approach to reading, which optimised the scope of the search (Boell & Cecez-Kecmanovic 2010). This approach, in my view, balances the literature review being personalised, structured and allowing a degree of flexibility to follow new leads. The chapter outlines the main issues related to annotation fully and using the search terms "annotation" and a combination of terms. I found 13 research studies in total with five from nursing and the rest from educational research. A summary of the literature provides a rationale for further research on the meaning of annotation practice within higher and nurse education.

1.5 In chapter four, I outline the work of Paul Ricoeur, in order to provide an understanding of his textual hermeneutics and its application to studying annotation. I define the terms hermeneutics and phenomenology and in combination examine Ricoeur's textual hermeneutics in some detail. The hermeneutic circle is introduced which is in constant motion to develop the researcher's phenomenological reflection, and adding more theory to the practical application of the circle, which I discuss in chapter five in relation to research methodology.

Section 4.6 is an important section because it outlines the plan for the remainder of the chapter's content and what I believe are Ricoeur's key concepts applicable to annotation. These key concepts include: the act of "recognition" of language allowing for phenomenal reflection (self-hood), making sense of words and signs (meaning, understanding), research through writing (discourse), the act of persuasion (metaphor and rhetoric) and how meaning and resonance of language inevitably change over time (temporality, distanciation) to alter perceptions. Lastly, annotation and its emotion in relation to nurse education is discussed (Ricoeur, 2012).

1.6 Chapter five introduces the research design and methodology. First, I contextualise the research study in nursing and reasons for choosing Ricoeur's textual hermeneutic phenomenology before giving an overview of the research design. van Manen's (1997) hermeneutic analysis model helped analysis and involves three stages of interpretation. The first order interpretation stays with the actual words used, and the second starts with a collection of naive meaning. The third order interpretations aim to analyse the hidden, ontological and deeper research themes to emerge (van Manen, 1997). Table 1 entitled *Index of study design and*

methods identifies the research methodology and use of Ricoeur's textual hermeneutics as a systematic discourse analysis.

Literature review	Research methodology and data collection	Thematic analysis
Informed by Boell &	Focus group interview (n=20), individual	Ricoeur's textual
Cecez-Kecmanovic's	interviews with students (n=5) and lecturers (n=8).	hermeneutics, use
(2010) hermeneutic	Digital and handwritten annotation research	of the hermeneutic
model (figure 3)	extracts. van Manen's (1997) three stages of	circle
	interpretation. First (initial), second (naïve) and	
	third (depth).	

Table 1: Index of study design and methods

1.7 Chapter six examines emergent themes leading from the first to third order interpretations, using van Manen's (1997) model to analyse the results of each research method. Table 2 summarises the chapter which briefly comprise student focus group interview (n=20), student individual interviews (n=5), lecturer interviews (n=8), digital annotation research extracts and finally, handwritten annotation research extracts. The collation of the research data using key words were collated in table 3 to identify four research findings which I explore in chapters seven to ten.

1.8 Chapter seven introduces the first theme of the "hermeneutic self." The theme was identified by the word "it" starting a sentence which is a trope for the authorial "I." Marcel and Ricoeur's work are utilised to analyse the research extracts. 1.8.1 Chapter eight's theme of "rhetoric" was found to organise the findings from eclectic annotation practises. Ricoeur's new rhetoric encompasses a theory of argumentation, "saying it well" with proof, composition, imitation of writing styles, the use of metaphor, productive imagination and temporal action called mimesis1-3 to aid teaching and learning.

1.8.2 Chapter nine examines the theme of "individualism," a moral predicate for justice, fairness and doing no harm in relation to the nursing profession. Rawls and Ricoeur's work are used to analyse the research extracts. The research theme identifies that annotators aim to maintain person centred care in nursing as a defence against the damaging influences on nursing and person centred care, such as the political labyrinth and technology.

1.8.3 Chapter ten develops the theme of the "reflective consciousness and slippage" projected in annotation, with analysis drawn from Freud and Ricoeur. The conscious thoughts of the annotator are evident in the themes examined so far, but, in general, comments given back to the student have an undeveloped unconscious communication, either from the annotator or the student, and subsequently result in evoked emotion.

1.8.4 Chapter eleven discusses the original contribution to current knowledge, and the research themes. Chapter twelve discuss the recommendations and implication for practice.

1.9 Conclusion

This thesis explores the meaning of annotation for nurse education within higher education. I outlined the aim of the thesis and the choice to use Ricoeur's textual hermeneutics. I discuss the
history of the study and anecdotal and personal reflection on what led me to initiate exploring annotation within nurse education. I define annotation as the addition of signs, symbols and language onto the page related to literary disciplines and feedback processes in nurse education. An initial examination of the evidence base underlying annotation concluded there was a lack of evidence of its use in higher education, and professional education. In particular, annotation seen as translations of knowledge is dependent on the lecturers' view of nursing and students' perceptions of being up to date. Issues of lecturers knowing more about practice than can be articulated through the written word affects the meaning of language when reading and writing annotation. Therefore, the notion of a disintegrated learning environment poses a potential risk for nurse education. The promotion of reflective practice for students and annotators on annotation relates to the four identified research themes. **Chapter Two**

Background

2.1 Introduction

In this chapter, I introduce a number of issues to prepare an exploration of the meaning of annotation in nurse education. Despite a lack of empirical evidence about annotation, I define and summarise the key principles of annotation in more depth through linguistic, historical and cultural theory. I explore annotation's social meaning in more detail through the phenomenon of assessment, measuring learning, reading, writing, essays and memory from annotation. I use some of the literary sources I found in chapter three's literature review to inform meaning for the purpose of advancing a more detailed understanding and scope of annotation to inform the research findings. This chapter therefore, continues to prepare the context for an exploration of annotation in nurse education.

2.1.1 Annotation as an interpretation

The literature suggests annotation may range from being benign and well-meaning to purposefully provocative and mean-spirited (Benstock, 1983; Derrida, 1982). Yet, in simple terms, annotation is the response of the reader to the text (Benstock, 1983; Derrida, 1982). All responses therefore, are an engagement with the text and its addenda (Derrida, 1988). As a communication device, annotation is circular and open to mis-interpretation because, as an

extension to the text annotation feeds on the text to add nourishment as a paraphrased gloss. This extension of the text challenges the pre-supposed notion of a homogenous space on the page for the author's communication (Derrida, 1982; 1988; 1991).

The relational aspect of annotation involves at least two audiences: first, the annotator to themselves, second, the authors' work (Derrida, 1982; Jackson, 2001). In education, a third audience is for the benefit of the student (Juwah, Macfarlane-Dick, Matthew, Nicol, Ross, & Smith, 2004). To readers who use it, annotation might represent a routine commentary alongside the text (Juwah et al., 2004). The problem of annotation appears to be that "something else" is perceived, whether knowingly or not, whether real or not and the evidence suggests annotation can communicate tone, attitude, irritation, authority, care, praise and questioning (Knoblauch & Brannon, 2006). According to Jackson (2001), the annotator stops reading long enough to make an annotated comment and this interactive relationship allows for the minds of two people to communicate in a scholastic contract. This also depends on the texture of experience, which affects the interpretation of text, even before we can consider how reading shapes a guiding preference to interpretation (Derrida, 1982; Ricoeur, 2006).

Due to annotation's potential for hermeneutic analysis, comments can always be elaborated on further, and the self-conscious motive of annotation in a professional sense, is important. What is meant, why, and for whose benefit annotations are made, are good questions to ask (Derrida, 1982; Jackson, 2001) and the above issues are just a selection of what I explore in this thesis in relation to annotation and nurse education. The assessment of learning in nurse education will now be discussed.

2.2 Assessment of learning in nurse education

Assessment is the main reason for annotation feedback because assessment is an opportunity to read and diagnose learning through the essay content (DiYanni, 2002). Assessment and student learning are at the heart of an integrated approach to student education, with a diverse methodology available to the educator, student and UK institutions (Knight, 2004). The QAA (2012b) guidelines suggest a more detailed definition of assessment from Angelo (1995, p.7) with assessment requiring set specific criteria, making explicit expectations for learning and, systematically gathering data for evidence of learning (QAA, 2012a; QAA, 2012b). The resultant evidence, matched against set criteria and standards, are used to explain, analyse student performance and articulate any shared assumptions to improve the quality of higher education (QAA, 2012a; QAA, 2012b).

Guidelines from the QAA (2012a; 2012b) promote the notion that good assessment methods aim to identify student learning in order to obtain information about student performance. A variety of assessment methods serve three purposes: the first strand is providing feedback to improve student learning. The second strand aims to measure student knowledge, and the third provides a grade to establish a student's performance (QAA, 2012a). In particular, guidelines suggest the return of "...*assessed work with written comments*..." would benefit from the use of annotation in the marginalia or end comments (QAA, 2012b, p. 39). Feedback should therefore, be a continual and timely dialogue and engagement between the lecturer and student, but in a format that offers necessary challenge to develop critical thinking (QAA, 2012b). All three strands of assessment are expected to enhance the potential for the student and lecturer to make progress judgements based on information received (QAA, 2012a; QAA, 2012b). However, the process is problematic for three main reasons: first, assessment is defined by valuing "what" and "how" learning has occurred. When reinforcing "what" to assess, its importance is declared in terms of quality assurance in the form of mission statements, course and module handbooks (QAA, 2012b) and what Knight (2004) called the "…*DNA evidence of the learning experience*…" (p.13). Prescribing "what" and "how" learning needs to be achieved, is problematic for adult learners (Boud, 2007; Boud, 2010; QAA, 2012a) because it parallels the obstacles for reflective practice in being told "what" to learn which I discussed in chapter one.

Second, a note of caution is necessary in light of the effectiveness of feedback within higher education (Knoblauch & Brannon, 2006a; 2006b; Sommer, 2006). With parallel assurances of quality standards in the NHS leading to damning reports such as the Parliamentary and Health Service Ombudsman *Care and Compassion* (2011) and *The Mid Staffordshire NHS Foundation Trust Public Inquiry* (2013), what is chosen "not" to be assessed may be more indicative of what organisations value more highly (Ball, 2012). As an antidote to this issue I suggest later in this chapter that discursive essays are more useful to identify issues of self-disclosure, for example, critical thinking, prevailing attitudes, care, compassion, and morality, instead of the move towards research orientated essays (Berg, Brämberg, Carlström, Öhlén, 2014; Maloney, Tai, Lo, Molloy, & Ilic, 2013). Empirical evidence on whether students understand lecturer feedback comments and apply feedback to improve essay work is also lacking (Price et al., 2010; 2011; Sommers, 2006).

Third, claims by educators that their annotation feedback to students on their essay content led to improvement, also lacks empirical evidence (Knoblauch & Brannon, 2006). Knoblauch & Brannon (2006) suggest such claims are equivalent to the "...*Emperor's new clothes*..." (p. 2) because, to paraphrase them, "powerful incentives" such as a belief gained from experience ensures that teachers maintain the "illusion" (p. 2) that their feedback responses change essay performance. Hence, Knoblauch and Brannon's (2006) use of the phrase concluded "...*the Emperor (still) has no clothes*..." (p. 2). This lack of evidence when combined with the literature review in chapter three and finding only 13 research studies between 1965 and 2016 on annotation in higher and nurse education, ensures this thesis gains more relevance. I now define annotation in terms of its purpose, history and social impact on nurse education to open up annotation to a fuller analytical perspective.

2.3 Annotation and nurse education

Annotation for assessment purposes in nurse education relates to the engagement to learn from reading (Lunsford & Straub, 2006). Annotation from the lecturer to the student on the essay is suggested to carry with it modes of commentary ranging from negative, imperative, problem centred, reflective, advice giving and positive praise (Knoblauch & Brannon, 2006). The language-in-use in annotation therefore, signifies a reference to something deeper and hermeneutic (Derrida, 1982). To paraphrase Changeaux and Ricoeur's (2000) conversation between a neuro-scientist and Ricoeur the philosopher, human consciousness relates intrinsically to learnt behaviours, actions and memories and various emotional tones, which are hard wired into the human mind (p. 136). I examine this further in the research theme the "reflective consciousness" and the transference hypothesis in chapter ten.

Briefly introduced in chapter one, annotation refers to the notes made on the page, whether in the marginalia, between the lines, or the base of a page by the reader, author, or editor (DiYanni, 2002). Annotation is a method of expanding and reinforcing the meaning of the text because it offers a unique commentary on the social dynamic at the time of writing (Slights, 1992; Nohrnberg, 1991). A definition I paraphrase in chapter one by DiYanni (2002) identifies annotation being:

"... brief notes written about a text during the process of reading. Underlining, circling words and phrases, highlighting passages, drawing arrows to link related points ... using question marks to indicate confusion are what a reader can do to signal importance. Marginal comments are also used to reflect the reader's understanding of and attitude toward the text ..." (p. 20).

This broad definition suggests annotation is extensive and varied, and in reference to the text, could include "anything" on the page that improves the reader's understanding (Henige, 2001; 2002). Therefore, if Henige is right that "anything" added onto the page promotes understanding, annotation would also include; editorial instructions, footnotes and glossaries, titles, contents pages, graphs and charts, Microsoft[®] Word, track changes showing revisions and semantic software for the digital age, and anything that captures the reader's attention (Henige, 2001; 2002). Apart from the text being subjected to "anything," annotation is found in all kinds of literature and is generally located discreetly as an appendage or supplement to meaning (Derrida, 2002). Benstock (1983) suggests therefore, annotation fulfils a human need to personalise and engage with the host text by using familiar words to make more sense of it. Annotation

reinforces the need for the clarity of discourse in order to satiate the reader's curiosity, by questioning, commenting and improving the text by the reader's use of short notes (Liu, 2006). So, if anything is added to the text in order to help shape understanding for those reasons, then other people reading who had not made those annotations, may also misunderstand what they mean because it is not personalised by them (Henige, 2002).

Annotation is viewed by literary theorists such as Derrida (1982) as a dislocation and disruption of the text because another person is commenting on an individual's work. The presence of "another" is suggested to reduce the synchronous meaning of the text and it is then changed. This powerful position is not generally of concern however, in educational circles, because annotation is viewed less critically, is well intentioned and more instructive and *showing* students where the parts and the whole of the essay can be improved (Lunsford & Straub, 2006). Timely feedback also promotes independence of the student as a writer and teaches valuable writing skills to understand levels of analysis (Liu, 2006; Lunsford & Straub, 2006; QAA, 2012a). The phenomenon of annotation therefore, remains a persistent form of educational feedback not only for a student in nurse education to meet professional statutory and regulatory requirements of learning, but their future learning (QAA, 2012a).

As Derrida (1982) suggests, annotation, as part of a language system, appears to carry with it undertones of emotion, experience and authority and may be perceived as mean-spirited (Lunsford & Straub, 2006; Sommers, 2006). Annotation therefore, is not suggested to be benign as it can have a direct impact upon the thinking and feelings of another person (Hyland & Hyland, 2006; Lunsford & Straub, 2006). In this manner, the annotator reaches out to the reader to influence the social and professional contract. Hence, the contract is cultural and anthropological (Strauss, 1972) and perhaps annotation's historical past parallels current higher education practice as an adjunct to the social phenomenon of education.

2.4 History of annotation

Annotation has been in existence since ancient civilisations developed their writing systems (Henige, 2001; 2002). Annotation in Chinese calligraphy dates back to the Six Dynasties period (AD 220 to 589) when responses, known as colophons, were added to text whether artistic, political, poetic or philosophical, and valued as much for their colophons as the primary text (Liu, 2006). The colophon in Chinese is *ti* meaning "...*to lift pen in response*..." and offers a poignant image of what actually occurs physically in the act of annotation (Liu, 2006, p. 194). Historical literature identifies the scope and depth of annotation as an *agent provocateur* (Nohrnberg, 1991) and a combative commentary challenging the thoughts of the reader (Benstock, 1983).

The power dynamics of Roman and medieval annotation suggest the practice was a powerful social force shaping the lives of those people ruled by dictators (Slights, 1992). Annotation in this format held power over the life or death of citizens as the Emperor's edicts were enforced (Slights, 1992). A parallel here is that in nurse education, the annotator too has a comment to make in the student's essay about life and death issues. History reinforces the contextual nature of annotation and the power dynamic of institutions promoting authorised interpretation of material and discouraging alternative ideas (Woodbridge, 1922). The history of theological annotation recognises a lack of freedom to make "glosses" (p. 255) on religious text because they

would in a literal sense "spoile" the word of God (Slights, 1992). "Glosses" or meanings in medieval annotation are in the form of short authoritative clerical comments to aid the reader (Slights, 1992). Adding glosses to clerical text by the laity was discouraged because only clerical authority had permission to interpret the word of God (Slights, 1992). The clerical annotation is a sustained attempt by the clergy to maintain authority over clerical text and preferring copying, translating and doctrinal interpretation over individual expression by the laity (Slights, 1992). Medieval European scrolls pointed to annotation as a commentary, which helped a reader's comprehension within hermeneutic literary criticism based on marginal commentary (Liu, 2006). Eventually, lay owners of rare biblical text added glosses based on their own interpretation of religious text against the best efforts of the clergy (Slights, 1992). In time, annotation had reduced its social impact to such an extent that it was viewed in narrow terms or not at all (Barney, 1991; Jackson, 2001; Nicholls, 1991; Nohrnberg, 1991).

2.5 Social meaning of annotation

The individual personal glosses appear to have become more valued as a contemporary analysis by future generations, because it is someone's direct experience and involvement with the host text from another time (Nicholls, 1991). Historical glosses are therefore an interpretation of the time giving the future reader a clear indication of past social perspectives (Nicholls, 1991). Nohrnberg in Barney's *Annotation and its texts* (1991) suggests when reading a text; the reader is exposed to a saturation of glosses attached to its textual form. The reader can learn from anonymous annotators without recrimination other than further annotation (Derrida, 1982; Nohrnberg, 1991). In contrast, readers' experienced in reading previous text can produce their own glosses and individual reaction to the host text (Derrida, 1982; Nohrnberg, 1991). Nohrnberg (1991) suggests the term annotation itself began as a gloss imprinted onto the mind first. The text starts out as silence then becomes physical in the form of marks, words, phrases, even sounds that are externalised onto the page, to act as a gloss that make sense from the prior experience of the reader. When marks and words become readable and begin to make sense, they then become comprehensible and comfortable, taking shape to become coherent from the text to the margins (Nohrnberg, 1991). Internalised meanings are imprinted onto the mind because as the text is engaged, "everything" becomes a gloss in the addition of meaning on what had been previously understood from the original text (Nohrnberg, 1991). Therefore, it is at the margins that the reader reaches a sense of the text's meaning, reached through a body of glosses or meanings intrinsically individual and personal (Nohrnberg, 1991). Add to the notion "anything" being considered annotation, and interpretation of text too is opened up for exploration (Henige, 2002).

Text constitutes just one mode of communication to offer a framework that structures ideas onto the page (Iser, 2006). The structure of text enables comprehension and is largely dependent on how successful it consciously resonates with the reader (Iser, 2006). This is what Iser refers to as the transfer of the text onto the reader's conscious awareness, dependent on the reader's capacity to perceive and process the message of the text (Iser, 2006). The reader literally "grasps" the meaning of the text directed by the clues to its content. These clues are useful in uncovering the hidden meaning of the signifier: a sign, symbol, word or image and what is signified to each person (Barthes, 1964; 1973; Ricoeur, 2006). To paraphrase Barthes, "nothing" can be exempt from meaning, because language is conceptual not referential. Text, according to Ricoeur (1977) is an external re-shaping of the thinking process allowing internal thoughts to become tangible and meaning grasped by a reader's perception. Hence, what the text signifies, is the potential for multiple realities in discourse (Barthes, 1964; 1973; Ricoeur, 2006).

2.6 The academic essay

I specifically discuss the essay format here but it can be substituted for other forms of assessment and discourse where annotation comments are added onto, such as storyboard and poster presentation feedback. Nursing essays should reveal the "art and science of nursing" and searching the archives for meaningful ideas to become externalised in writing (Gardner & Rolfe, 2013). So if annotation is to be explored further, what annotation is added on to, the essay, requires a brief exploration as an assessment method for nurse education.

First, I outline what an essay is and discuss its purpose as an assessment method in nurse education. Essays are a form of expression since Greco-Roman antiquity with the stoic philosopher Seneca (c.1 BCE to CE 65) writing on subjects such as asthma, noise and dealing with one's slaves (Seneca, 2004). In the tenth century, the Japanese writer Sei Shonagon captured her auto-biographical experiences of the Royal Court (DiYanni, 2002). The seventeen century western essayists Francis Bacon and Michel de Montaigne's early works were often revised and expanded on and re-published, and Montaigne coined the word "essay" from the French "essaie" derived from the verb "essayer," meaning to try or attempt (DiYanni, 2002, p. 7). Montaigne's own essays changed over time to develop from using large amounts of quotations to an internalised approach demonstrating his mind in the act of thinking through writing (DiYanni, 2002). This same style of writing can be seen in novice writers when comparing their earlier to later writing.

Essays can be either informal with the use of the pronoun "I" or formal, which allows the writer the freedom to experiment with ideas on the page and see where their thinking can lead them (DiYanni, 2002). An essay is an anthology of ideas, observations, speculations from fiction and non-fiction with the capacity to influence the reader's thoughts, feelings and insight (DiYanni, 2002). Essays have therefore, long been considered the dominant form of assessment in HEIs and disciplines such as the humanities who still hold the essay up as the gold standard for developing critical thinking through critical writing (Gardner & Rolfe, 2013). The lecturer being mindful of the art and science of nursing may view educational constraints sceptically and consider a preference for facts and certainty, such as science over the unquantifiable aspects of nursing with concern (Gardner & Rolfe, 2013). This leads me on to the benefits of reading for academic purposes.

2.7 The benefits of reading

The benefits of reading involve processes of observing, connecting, inferring, concluding and questioning (DiYanni, 2002). Observing the choice of language, tone, style and content occurs consciously and unconsciously to connect the words being read. This entails seeing how the words relate to each other, noticing the detail, phraseology and observing the inferences made. Identifying the imperceptible gaps of words used in the essay are helped by the inferences made by an author's choice of words to drive the interpretive process, in other words, what the text means to the writer and to the reader (Ricoeur, 2003). Inferences are therefore a statement based on what has been observed when read and they consider the emphasis on words which only an individual interpretation can create. For example; if the text states "who knows what he thinks?" it could be interpreted a number of ways according to DiYanni: who "knows" what he thinks,

who knows "what" he thinks, who knows what he "thinks?" On the other hand, the emphasis on the "who" or "he" all depend on how the reader hears it and sees it to resonate its meaning to them (DiYanni, 2002, p. 10). Therefore, interpretation is a process of explaining the meaning of the text through various stages to make explanation conclusive. The writer, when developing their discursive style, does so through the various processes of reading and writing. Good essays have certain characteristics such as clarity, coherence, organisation, accuracy and correctness, sufficiency and style and the perceptive reader looks for structure and connections (DiYanni, 2002). Readers should be able to follow the text, and the text should flow from sentence to sentence and paragraph to paragraph, text which has a clear introduction, middle and an end, is accurate in terms of grammar, spelling, punctuation and has sufficient scope. These characteristics however, need to be developed and this thinking-through-writing in the revision stage ensures a complex, circuitous process of reading, writing, interpreting, thinking again, writing again and then repeating.

Ricoeur (2007) suggests this thinking-through-writing phenomenon starts again when re-reading and with almost immediate effect, errors suddenly becoming obvious. I found this to be particularly pertinent to the research process because this meta-cognitive process is both experimental and temporal and to read more ensures that a different direction may be taken when the text activates previously forgotten or unknown information (Ricoeur, 2007). The text and its reading therefore, become hermeneutic (Ricoeur, 2007). This is because reading and revising are helped with the addition of annotation and any initial thoughts the reader has, and free writing the expansion of those thoughts into an interpretation. Annotation therefore moves from immediacy to be contemplative, and once the latter is engaged, any manner of interpretations may follow dependent on the resonance of the text and further reading (DiYanni, 2002).

2.7.1 Be suspicious of the text

I found that Ricoeur's own essays were a process of comprehensive immersion into a body of work, often from Greek Antiquity with the express purpose of developing his own ideas from the host text (Clark, 1990; Kemp, 1996b). Ricoeur's essays aimed to entice the reader to a certain view, and then de-construct that view with his own re-conceptualisations of it. Ricoeur wanted to encourage the reader to be "suspicious" of the text and be aware of its re-contextualisation (Ricoeur, 2008). Unlike literary essays, which can take many dynamic forms, in universities the student is bound by formal rules of citation, style and content which bind the essay to the textuality of the discipline (Gardner & Rolfe, 2013; Good, 1988).

The academic essay is different again to other forms of literary essays because it aims to inform and persuade the reader-lecturer that the student-author has demonstrated learning (Gardner & Rolfe, 2013). This purpose is even more important in nurse education because assessment of student knowledge and competencies has a direct impact on public safety and a poor essay, or attitude, indicates a need to assess their fitness to practice (NMC, 2015a). In that sense the academic essay in nurse education is considered to have a social and professional influence. However, the essay's purpose for nursing offers more than a planned assessment opportunity because, it is the externalisation of thought which in itself is otherwise hard to quantify in the absence of acting out those thoughts (Ricoeur, 2003). To wait until a nursing student makes an error based on a lack of skill or knowledge or poor attitude would be professionally unethical (NMC, 2011; 2013). Therefore, an essay is a timely opportunity for a nursing student to "experiment" on the page without harming a patient, leaving the annotator to guide, advise and reflect on their own views and interpretation.

2.7.2 Suspicion of the nursing essay

Gardner and Rolfe (2013) suggest the essay has become increasingly unfashionable in nursing and refer to the "hegemony of the laboratory," which relates to science's apparent suspicion of self-disclosure. Perhaps, this may be because the latter has a degree of uncertainty? The authors suggest the language of certainty, which research perpetuates, is a worrying nursing trend because the essay, when imagined, appears to be seen in a lifeless form and therefore, different to literary essays which are inspirational, observational, and experimental to challenge the reader to think differently (Gardner & Rolfe, 2013). When a search for literature is restricted by disciplinary constraints, the answers found will inevitably be restricted too (Boell & Cecez-Kecmanovic, 2010). Therefore, Gardner and Rolfe (2013) suggest that the essay should perform three tasks in universities: claims to knowledge, rigour and resistance. The essay makes claims to knowledge, is less structured, more discursive, unlike research reports. In contrast, the scientific research report is structured (abstract, introduction, methods, findings, discussion, recommendations), but restricted by criteria and search constraints which ensure a degree of bias and reproducibility, conventions, rigour and ethics processes (Gardner & Rolfe, 2013).

The discursive method, on the other hand, relies on rhetoric and the persuasive power of the argument to overcome resistance (Connors & Lunsford, (2006). In this format, I also include reflective essays on clinical experience and developing new perspectives through the third person

narrative (Benner, Sutphen, Leonard, & Day, 2010). Benner et al (2010) found this kind of essay to be most beneficial because the nursing student could hermeneutically open up the possibilities of understanding by applying theory and evidence to clinical situations. The discursive essay follows the twists and turns of the critical reading process and this process enables reflection on everyday clinical activities and trends (Gardner & Rolfe, 2013; Maloney, Tai, Lo, Molloy, & Ilic, 2013). The discursive essay allows for critique and a loose form to write and think differently. This is important if nurse education aims to address the theory-practice gaps illuminated by the *The Mid Staffordshire NHS Foundation Trust Public Inquiry* (2013).

Gardner and Rolfe's (2013) concern about the rise of the scientific aspects of nurse education are therefore, noteworthy. Nurse education came late into universities in the 1990s and so research papers are viewed in contrast to the humanities, as the gold standard for publications enhanced by the growth of evidence based practice (Gardner & Rolfe, 2013). The expansion of research and evidence based practice within universities therefore, threaten scholarly writing and appear to heighten the theory practice gap, which is an obstacle to understanding nursing in more depth (Benner et al., 2010; Gardner & Rolfe, 2013).

2.8 Standardisation of essays

The standardisation of nurse education and an apparent preference for evidence based practice essays is problematic, not only because it reinforces Gardner and Rolfe's (2013) concern, but when standardising nursing, it can at best be considered a restrictive model (Maloney et al., 2013). Standardisation, however, offers little hope of maintaining the known benefits of discursive essays because of the standardisation of quantity, content, programmes and qualifications promoted to ensure UK nurse education is comparable with other European degree programmes through the Bologna process (1999). In particular, tensions have arisen from the implementation of standardisation in nursing specialism programmes due to the focus on research projects (Berg, Brämberg, Carlström, Öhlén, 2014). Rather than nursing essays identifying issues of self-disclosure and critical reflective thinking, the evidence suggests that research application into clinical practice is tokenistic and superficial and actually "talked up" when little change has actually occurred in practice (Foxcroft & Cole, 2003; Regan, 2012a). Berg et al's (2014) Swedish study reported nursing students frustrated by the scientific orientation to acquiring nursing knowledge through research projects, which appear to them to be paper exercises. Therefore, facts not ideas devalue nursing and reduce a sense of ownership.

Contrary to essays considered a paper exercise, in promoting research appreciation essays should be an expression of the higher values of the academic process, which considers all sides of an argument (Maloney et al., 2013; QAA, 2012a). Let me detail this point. Discursive essays promote good nursing values and principles through critique, rigour and resistance to contemporary pressures (Gardner & Rolfe, 2013). When one of the contemporary pressures is the promotion of evidence based practrice and scientific research reports being preferred over discursive essays, the risks therefore, are a sense of depersonalisation and a disintegrated nurse education programme (Benner et al., 2010; Gardner & Rolfe, 2013). How this occurs is because writing allows for the writer to not only write differently but also to think differently (Gardner & Rolfe, 2013). This issue is more significant when considering that nursing attitudes, care and compassion are all found in the nursing essay and in the received annotation (NMC, 2011; 2013; 2015a). Therefore, in giving feedback on the discursive essay, nursing lecturers can engage authentically with nursing students and assess the essay content meets professional standards (NMC, 2011; 2013). Being part of a range of assessment methods, the nursing essay allows the lecturer, as annotator, the opportunity to assess a variety of issues, notably fitness to practice, knowledge, integrity, character and meeting the module learning outcomes (QAA, 2012a; 2012b). Sometimes the choice of language in an essay indicates a perception such as depersonalisation of the patient and this means the student is at risk of betraying actual beliefs rather than professional expectations (Schön, 1983; 1987). The nursing essay therefore, identifies hard to quantify issues such as spirituality, holism, emotion, professionalism, a lack of time to care, lack of skills, privacy and fear of what may be uncovered when talking to a patient (Keall, Clayton & Butow, 2014; Lopez, Fischer, Leigh, Larkin & Webster, 2014; Walker, 2014).

2.9 Conclusion

This chapter explored a variety of issues in relation to the practice of annotation within various literary genres and nurse education. Notably, annotation is defined in detail, its history, its purpose, benefits and potential influences reinforcing annotation as a social construct (Ricoeur, 2006). Annotations include the addition of marks, signs, symbols onto the page, made by the annotator to share their understanding of reading with others (DiYanni, 2002). Therefore, its social, linguistic and cultural connotation indicates annotation is considered either benign or provocative and depending on the context of which it is given and for what purpose (Benstock, 1983; Ricoeur, 2006). Assessment feedback and the academic essay is the most predominant format in promoting student meta-cognition (Gardner & Rolfe, 2013). This contrasts with the promotion of standardisation, which allow the student to hide in the text. Therefore, the Bologna process (1999) preference for standardisation and scientific research reports place nurse

education in a difficult position indicated by the annotator's presence in the margins of the essay. The next chapter examines the literature and evidence underpinning annotation and nurse education. **Chapter Three**

Literature review

"All practices of annotation are riddled with paradox. They are designed to convince both doubters and believers and, while they aim at achieving argumentative invulnerability, actually manage to open their authors to other kinds of vulnerability..." (Henige, 2002, p. 87)

3.1 Introduction

The quote by Henige (2002) indicates a view in the literature that some annotator's feedback comments make them vulnerable to criticism and the choice of language exposes the possibility of hidden meaning. In nurse education the choice of language used for annotation feedback can be interpreted a number of ways and has an impact on student learning (see figure 6). The kind of language which informs annotation is examined in chapter nine (see section 9.7) in relation to the meaning and effect of value laden words in NMC policy standards. What is interpreted is relevant to nurse education and emerges as a theme from the literature review findings from ten literary sources (four books and six articles) and 13 research studies which I examine to identify relevant evidence and knowledge (see figures 5 and 6). Annotation has been defined in chapter one and in more detail in chapter two. I will now outline the rationale and method for a

hermeneutic approach to the literature and explore the meaning of annotation in nurse education through the literature review findings.

3.2 Method

Literature reviews are an important part of the research process (Cresswell, 2007; Grbich, 2010). Cresswell (2007) suggests the strongest academic rationale for a qualitative research study is derived from a review of the literature to identify gaps and additions to knowledge. A literature review aims to inform both the researcher, the reading audience and to promote interest and engagement (Cresswell, 2007). Smythe, Ironside, Sims, Swenson, & Spence (2008) suggest that hermeneutic phenomenological research should search for as much information as possible to reduce the risk of researcher bias in the application of theory to research data. The researcher's immersion within the research process is central to the interpretive process and when reading the available literature new perspectives develop to inform the research study and what is known and unknown (Smythe et al., 2008)

3.3 Search strategy

The time period 1965 to 2016 was chosen to include McColly's key 1965 seminal research study on annotation in education (see figure 6 and a brief appraisal in section 3.5.4). The search terms were "annotation" and a combination of terms used (see figure 1). Inclusion and exclusion criteria (see figure 2) identify a variety of sources from scholastic and literary genres.

Figure 1: Search strategy key words

Data bases searched 1965-2013	Search terms	Retrieved papers
Cinahl Plus Fulltext, AMED, Nursing Index,	Annotation	10,256
ERIC, Psychinfo, PsychArticles, JSTOR, Google		
Scholar, Library catalogue		
+Humanities International complete	Annotation+learning	n=1803
	Annotation+higher education	n=673
	Annotation+feedback	n=210
	Annotation+student education	n=91
	Annotation+feedback+higher	n=31
	education	
EThOS electronic thesis online		n=2

Figure 2: Inclusion and exclusion criteria

Inclusion criteria	Exclusion criteria
Scholarly and research studies on annotation	Studies with the word annotation, for example genome annotation, the description of an individual gene
Studies on written feedback to students with annotation the primary focus	Annotation is subordinate to technical discussion. e.g. development of digital software, rather than adding anything new to annotation studies
Studies in English	Studies not in English
Nursing studies	
Research studies from computer science on annotation as a primary focus for discussion before secondary discussion of technical programming	Studies on annotation from computer sciences with a primary focus on technological programming and limited discussion on annotation

The retrieved publications were reduced from the initial results of 10,256 using a hermeneutic approach to selection which I will explain more about in sections 3.3.3. The search term "nurse education" was not used because it would have restricted the number of retrieved articles found and lessen the scope to understand annotation in its wider literary influence. This was important because I knew little about annotation at the time. However, when searching the retrieved papers, I was mindful of nurse education literature. The university online e-database was used to search CINAHL Plus Fulltext, AMED, ERIC (teaching and assessment), Psychinfo, PsychArticles (psychological processes) and Nursing Index, JSTOR (historical papers). Tracking references

from retrieved publications provided me with valuable sources of information presented in figures 5 and 6 (Boell & Cecez-Kecmanovic, 2010).

3.3.1 Hermeneutic process and the literature review

In the human and social sciences research questions are rarely fixed at the literature review stage (Boell & Cecez-Kecmanovic, 2010). The researcher needs to situate a research study within the wider existing literature depicted by the findings presented in figures 5 and 6. Boell & Cecez-Kecmanovic (2010) as professional librarians, suggest there should be structure in qualitative research literature reviews in the use of search terms, databases, inclusion and exclusion criteria and appraisal of publications. However, they suggest the primary rationale for structure is to enable a review to be complete, reproducible and unbiased. Following normal conventions of a literature review is fine up to a point but reading and interpreting the relevance of publications makes a literature review hermeneutic. This realisation helped to clarify for me a conundrum of interpretation for the remainder of the chapters before the research findings were identified and informed by Ricoeur's textual hermeneutic phenomenology. I will now explain how Boell & Cecez-Kecmanovic's (2010) model applies to this literature review process.

3.3.2 Structured literature review: An adaption

The aim of a structured literature review is to identify a relevant body of publications (Cresswell, 2007). A reproducible process occurs when search criteria are used to find the body of publications and evaluating each publication using the paper's abstract and key words to identify relevant sections (Cresswell, 2007). However, this may lead to irrelevant keywords and research being "designed out" of the process (Boell & Cecez-Kecmanovic, 2010). The choice of

keywords helps identify the relevant publications (see figure 1) and where a hermeneutic approach is different is that the reading process is central to making decisions about the search strategy and research findings.

Boell & Cecez-Kecmanovic (2010) suggest structured literature reviews are derived from medical research since the mid 1990's which require analysis of research to be rigorous in identifying lifesaving evidence, knowledge and treatments. To ensure high standards of rigour, structured literature reviews need to start from a position of certainty where the research question is fixed and keywords for inclusion and exclusion criteria develop from a fixed research question. Therefore, the results are reproducible, complete and unbiased (Kitchenham, 2004). Key words however may inhibit finding publications that do not match the research question or key words (Boell & Cecez-Kecmanovic, 2010). This means there is less opportunity to deviate from a structured and deeper review path.

3.3.3 Reduced reading bias through the hermeneutic circle

Boell & Cecez-Kecmanovic (2010) question the claim that a structured literature review can be unbiased, reproducible and complete, and this parallels my experience of the search and reading process. Structured literature reviews inevitably emphasise certainty over uncertainty which qualitative research emphasises through experiential narrative (Grbich, 2010). Boell & Cecez-Kecmanovic suggest knowledge acquired for a doctoral thesis does not fit conveniently into a review process, because the research question itself has to be first fixed, and then the method designed to inform the research question. In this doctoral thesis, the exploration of the meaning of annotation is unfolding over time with many working titles, diversions, dead ends, and false leads dissolving as the writing process is revised constantly. This contrasts with a search of the literature with a fixed research question and new information shaping, guiding and refining the research focus. A doctoral thesis should be about reading relevant publications to furnish the researcher with the right kind of questions that may then lead to unexpected avenues, unexpected because they were probably unknown (Boell & Cecez-Kecmanovic, 2010). When little is known by the researcher, placing restrictions on a literature review before a degree of understanding is achieved, ensures a difficult position to start from and may frustrate enquiry (Boell & Cecez-Kecmanovic, 2010). I realised an overly structured literature review had inhibited my immersion into annotation and, despite the conventions of systematic enquiries; I thought such an approach was anti-thetical to a hermeneutics enquiry. Figure 3 overleaf presents Boell & Cecez-Kecmanovic's (2010) hermeneutic model to my review of the literature.

Figure 3: Adaptation of Boell & Cecez-Kecmanovic's (2010) model to the hermeneutic circle and literature review



3.4 The hermeneutic circle and choice of publications reviewed (see figure 3)

The interpretation of content makes a literature review hermeneuti*c* because there can be no final understanding of the literature (Boell & Cecez-Kecmanovic, 2010). Thereferore, I discuss Ricoeur's approach to the hermeneutic circle (see chapters 4.8, 5.6.1, 5.7 to 5.7.3, 8.5.1) which relates to the above figure 3 stage of literature appraisal. Ricoeur (1991) suggests reading

produces a constant re-interpretation of what has been read, understood, and the depth of any understanding and its resonance changes over time through a concept called temporality (See chapters 4.8.4, 8.5.1). The following discussion demonstrates a temporal approach to the literature review. When choosing the subject of annotation for this literature review I searched publications to help me understand annotation first. I was also mindful of my previous review of the literature in 2010 when I used the same search terms and databases (Regan, 2010) but I found few familiar studies due to different databases and journal subscriptions. Since I started this thesis, I had worked in three North West of England universities and found reproducing the results from the same search criteria problematic and so, in the end, I started again and used one university's online database. Therefore, the literature review follows roughly the same process presented in figure 3 in which to reduce the 10,256 to 31 retrieved papers found in figure 1. The process of checking for meaning and relevance commences by reading literature content relates to a movement back and forth in order to promote understanding and choice of publication relevance. This is initiated in the hermeneutic circle and figure 3, which involves a constant iterative process of reading, re-reading, and interpreting with an awareness of what one actually thinks and then eventually thinks after reading (Ricoeur, 2003). An example of this reading process can be found in figure 4 entitled An example of the hermeneutic circle at work in the literature review of annotations made on Feito and Donahue's (2008) research study, which I present more fully in appendix 1. Feito and Donahue's (2008) study led to discussion in section 3.6 to 3.6.1. The numbers one to three added to the paper in figure 4 (see chapter 6.3.1 for further explanation and appendix 1) indicates how time improves an understanding of reading, with "1" relating to an initial reading, "2" and "3" subsequent readings indicated on the page by the addition of more comments.

Figure 4: An example of the hermeneutic circle at work in the literature review (see appendix1).

Same from , naming prenomenar - (what is this?) Description, naming prenomenar - (what is this?) Description beauties, reception theory - de not face DIFFaulty quote - (P.1) - the difficulty is we do not face difficulty con eraph." Calvatori verme--3)KE' Intral annotation Minding the Gap 2) Re-reading annotation slaving readup Annotation as preparation for discussion of relevent text (150, Riccew, Salvapris Darahue) JOSÉ ALFONSO FEITO Futurer notes made Saint Mary's College of California, USA oftext PATRICIA DONAHUE Lafayette College, USA elsator + ABSTRACT ane This research project examines classroom discussion in its relationship to reading as made visible, through the practice of textual annotation. In order to develop a (2005 rich description of student reading/discussion processes, we targeted multiple Difficitions? undergraduate seminars at a liberal arts college as they encountered the first two Acts of Shakespeare's King Lear. We collected triangulated data from these class Strangeness sessions including targeted reading surveys, student reading annotations, naturalof text reader istic observation of real-time seminar discussion behavior, and student reflections. Not bra Our analysis of the students' annotation strelies upon reading theorist Worgang Aser's conceptions of interpretive gap, consistency building, and individual repertogetstar toile. Our discussion considers the theoretical implications of this local, in-depth - andigut data for the broader analysis of student reading and discussion practices. Barres. - tedium annotation, difficulty, discussion, pedagogy, reading, student readers KEYWORDS - connotentia Mark - historiccel ficoen remoteness on than of MOST DISCUSSION-BASED COURSES - first-year seminars, capstone (resonance courses, writing courses, numerous courses in the Humanities and Social Sciences, the occasional course in the Natural Sciences or Engineering --Mounds Gad saci make significant use of assigned texts. Such 'readings' create opportunities for about two -Mis-Interpr students to confront provocative and difficult ideas, to build upon them, reader provs unation challenge them. Of course, dialogical work of this kind does not just happen, but must be carefully prepared for. And one way in which students prepare about South readup themselves for intellectual engagement - or so the story goes - is by writing, Arlas for chategies hemselves into texts through annotation. By circling words, registering -Annatate-bradget/highlight chiscuss pulding on - play with words Arts & Humanities in Higher Education © 2008, SAGE PUBLICATIONS, Los Angeles, London, New Delhi, Singapore as on DC. ISSN 1474-0222 - id pro runder and p VOL 7(3) 295-307 DOI: 10.1177/147403 - Review readupr 22009 gutil dees [296] God tex -trusties a sosparia frestoes the -testing draspaus 1.87 reference - Slow dethin read Simonstra - difficulty Seen as an apprhunity - try not to do it allatare down on November 18, 2010 led from ahh sagepus com by gu atuato -Hyseidity - two knidsor Conforments

When reading Feito and Donahue's (2008) research study, I became immediately aware of their concept of "difficulty" which I did not understand fully, but I presumed it referred to struggling to understand a text, then stopping to think or continuing to read on with the awareness that not all of the text was making sense. I had experienced that same phenomenon when reading philosophy and not understanding the technical details of its terminology. I could simply have an impression of the meaning of a word and when reading the sentence have a "sense" of what it may mean. For example, the reader is helped by their mental capacity to deal with more than one idea at a time before the most likely idea is made clear (Ricoeur, 2008, chapter 4.8).

Chladenius in Gadamer (2004a) suggested that the hermeneutic circle involves the reader's anticipation of meaning running along the text like a rhythm and at the same time, they need to be aware of their pre-understanding, their pre-conceptions, prejudices and judgement whilst the reading rhythm is in motion. As I read Feito and Donahue's (2008) study over the years, my understanding of the paper grew in depth as it directed me to other useful text to read and then returning to the paper to be better informed (see appendix 1 for an example of the hermeneutic circle and reading). As previously stated, I have indicated the temporal dimensions of my annotation by using the numbers one to three (1-3) in figure 4 and research notes (see chapter 6.3.1 for explanation). Therefore, my interpretations changed over time from using minimal annotations often with single words underlined or made in the margins such as "buy the book" in figure 4. These short annotations indicate the stage of reading and engagement with the text. This phenomenon relates to Ricoeur's mimesis1-3 and the stages of the hermeneutic circle when reading and bridging the gap between initial and superficial engagement with the text and a deeper engagement later on (see chapter 8.5 to 8.8). For example, my initial reading concluded

that the first page of their study demonstrates the hermeneutic circle in progress as I went from simply circling key words to make more extensive annotations notes. The later annotations could only be made after reading Salvatori and Donahue's (2005) book entitled The elements (and *pleasures*) of difficulty which the paper refers to and then I made notes on their research study to make links with the contents of both. Although I had the book in my possession within six months of initially reading the research study, it took a further year or two to finally read it and then develop a deeper understanding of the text, as indicated by marks made in figure 4 and the numbers 2 and 3. This was paralleled by what Feito and Donahue (2008, see appendix 1) state, that sentences are the basic components of the text in order to make statements, claims or observations. Sentences need to be "plumbed" for what they appear to say (or not) and as such, they do not constitute the whole text but text as text. The authors quote Iser (2006) and in relation to the promotion of understanding the text as it "...begins to emerge when a reader "climbs aboard," when readers employ their imagination in discerning how linguistic elements connect or relate to each other..." (p. 299). Therefore, connections were made by my own expectations about the text, which undoubtedly changed to evolve with additional texts. The later stages of the hermeneutic circle are discussed further in sections 3.6 to 3.6.1. This hermeneutic circle process was repeated for all of the literature cited, to a lesser or greater degree matching the examples given in appendix 1.

3.4.1 Entry and exit points: Saturation

Understanding the meaning and relevance of the available literature is a central focus to the hermeneutic circle, which changes over time (Ricoeur, 2003). A good starting point is reading literature reviews, which identify the concepts, themes, technical language, and relevant authors

(Boell & Cecez-Kecmanovic, 2010). This hermeneutic and iterative approach suggests a constant process of revision, which only ends when the process of encirclement reaches saturation point and when a paper offers nothing new to what has been discussed already (Smythe et al., 2008). This approach to a literature review identifies the overlap between disciplines such as technology, literary criticism and education and identifying the usefulness of the literature for the purpose of this thesis. First, let me explain in the next paragraph why some digital annotation papers from the technological literature were useful to annotation exploration and why others were excluded.

3.4.2 Retrieved search 1965 to 2016 (see figures 5 and 6)

One finding from the search was that emerging educational technology developed many methods for modifying the content of digital resources for sharing (Novak, Razzouk, & Johnson, 2012). A number of retrieved studies relate to digital annotation, a term referring to hypertext technology, which allows textual additions to online text. Annotation is also linked to gene research, which constituted the bulk of research papers found and therefore excluded from this thesis. This left thirty (n=30) retrieved results which I divided into two categories: first, literary sources (see figure 5) and second, research papers (see figure 6). First, the literary sources presented in figure 5 found four (n=4) literary books (Barney, 1991; Derrida, 1982; Jackson, 2001; Straub, 2006) and six (n=6) literary papers (Benstock, 1983; Henige, 2001, 2002; Lyons, 1967; Mandel 1965; Slights, 1992). These literary sources acted as the background to the thesis because of their scholastic expertise on annotation. The literary sources gave a rich contextual commentary by placing annotation in a combative position, as annotators' engage with author's work somewhat personally at times (Mandel, 1965).

Books n=4	Summary		
Authors			
Barney (1991)	Essays from numerous authors on annotation from biblical (Nohrnberg), Early English (Too		
	medieval manuscripts and glosses (Nicols; Lawler), undermining footnotes (Cosgrove;		
A collection of	Derrida), and lastly, annotation as social and political practice (Hanna; Mayali).		
essays on			
annotation	Annotation is less a relationship of meaning and more exercising power. Annotation is a		
(background)	mechanism of the political appropriation of knowledge (Mayali, p. 186). Derrida suggested the annotator subordinates themselves to the hierarchy of the host text. As a result, annotation has		
	its own sense of autonomy and space (p. 193).		
Derrida (1982)	De-construction of major philosophical traditions. In his book Derrida refers to writing and its		
201100 (1902)	space as the disruption of presence in the mark. Writes about difference, dislocation and		
	displacement and the word in the margin is the presence of another displacing the word of the		
	other. The parasitic nature of annotation may surround language like a ditch, distorting the text		
	by the influence of another.		
Jackson (2001)	Annotation from books dated 1700-2000, not articles or studies but there is similarity. The		
	physical characteristics of a book means the margins are free to make notes. Annotation		
Annotation	history is vital in understanding its impact on the present.		
history (background)	The English Civil war used pamphlets for communicating and turning annotator's from		
(background)	"docile supports into contesting readers" (p. 52). The anonymous personal comments		
	make annotation important as a contemporary commentary on the text.		
	r i f i f i f i f i f i f i f i f i f i		
	Why? Because, annotators are readers and not all readers are annotators. The anonymity of the		
	annotator gives a refreshing view of note taking because the author gets the first word in and		
	the annotator the last word. Some authors therefore see annotation as invasive due to its		
	parasitical and often phlegmatic relationship to the host text		
Straub (2006)	On teacher feedback annotation is a form of commentary on student essay composition. A number of authors' essays and research are reported. There are four overlapping themes of the		
A collection of	literature: First, proposing what teachers should do when commenting on students' essays.		
studies and	Second, seeking ways to identify what teachers actually do. Third, identifying the effect of a		
discussion	response of any kind on actual essay performance (p. 6). Fourth, there is a lack of empirical		
papers on	evidence students typically understand teacher feedback, nor evidence of use to modify their		
educational	essays (p. 69).		
composition			
studies	Straub suggests from 1984 to 2001 there has been a lack of evidence that comments on a draft		
	made quality improvements to the next draft other than complying with overt instruction.		
	Acquiring such evidence is affected by the imperfect assessment tools available to teachers. Teachers claims of changed performance equals the " <i>The Emperor's new clothes</i> " (Knoblauch		
	& Brannon, 2006b, p. 2 <i>In</i> Straub, 2006). Cosmetic and mechanical behaviour change pass for		
	a student's willingness to make error corrections. Therefore, the myth is exposed due to a lack		
	of in depth evidence of behaviour change. Students need time to grow and their writing to		
	mature.		
	Teachers' annotations are subjective with socio-political baggage, such as assessment		
	mandates, quantifying teaching and learning and identifying measurable improvement (p. 5): they think they are evaluating the students essay but in fact are "looking at a textual mirror		
	of themselves" (p. 4) and risk unknowing bias.		
Literary	Discursive papers from literary criticism, translation and editors. The use of direct		
Papers n=6	quotes in this section aims to capture a sense of the emotion discussion.		
Mandel (1965)	Citing A Soviet View of the American Past: An annotated translation of the section on		
	American history of the Great Soviet Encyclopedia (Bolshaia) with a preface by Adlai		
Translation, re-	Stevenson. Discussed need for error free translations, re-translation, annotation and footnotes		

Figure 5: Literary sources retrieved from search 1965 to 2016

translation and annotation	from one language to another e.g. Russian to English.
annotation	Russian scholarship and additional annotation comments made by distinguished academics. American annotators appeared to distort the meaning of the original. Distortions become meaningless at times; translation is lazy and ignorant of Marxism, Soviet history and history of labour and socialism.
	For example: "The War for Independence was a bourgeois revolution. Popular masses won the victory over England, playing a decisive role in the revolutionary war against England" (p. 716).
	The footnote reads: "The writer seems to be somewhat confused about his own terminology. In Marxist theory a 'bourgeois' and a 'mass' revolution are not the same thing. It is the annotator, unfortunately, who was confused on terminology" (p. 716).
	Notably, when annotating academics make errors based on national attitudes and perspectives, a degree of partisan bias is evident.
Lyons (1967) Literary criticism, translation,	Scholastic translation, annotation and footnotes have a long tradition of negating the message of the host text in favour of their own perspectives. Lyons suggested attitude and tone come through from the annotation footnotes. These old styles suggest literary annotators/ footnotes are dissenting exegetes (interpreters) who aim to defend their views:
footnotes and annotation	"with whatever weapons they have: pedantry, hauteur, slap-stick, acrimony, paranoia, ridicule, a sincere wrath, and an insincere urbanity" (Lyons, 1967, p. 243).
	The use of annotation footnotes poke fun at the text and adds a tone of worldly nonchalance and the footnotes <i>facetiously</i> oppose text (p. 245). The annotating " <i>exegete is caught between the work of art and life, and his mediation</i> <i>(annotation) is awkward He is afraid of murderingto dissect and yet out of reverence</i> <i>wishes to become the work of art Its every detail and implication are paraded, and yet there</i> <i>is a suspicion of a more perfect work that is savoured in private</i> " (p. 245). In other words, the annotator may fail to publicly and objectively appraise the text but in private admire it.
Benstock (1983)	Footnotes serve as commentaries and references to the parts of text they are keyed to. Preface annotations are referential and marginal, reflecting and engaging on the text. Whilst also negotiating an extra-textual world between the author and others.
Discursive paper on critical writing	Annotation aims to co-operate with the text but not be intrinsic to it by maintaining comments that are inner (to the text) and outer (to other text) directed resulting in a critical addition that may contradict the host text. The marginal positioning of annotation means it is a closed space asking the annotator to make explicit assumptions and after words, after thoughts, questions in a voice different to the host text. Annotation's extra textual nature leads to this situation.
	Sometimes there is a breakdown between the third and first person which identifies an argumentative breakdown of the " <i>carefully controlled voice</i> " (p. 204) reflecting a general <i>ambivalence</i> to the text, the speaker in the text and to the audience.
	Authority is reinforced by the annotator's marginal presence, their first person voice (personal address) and the third person (impersonal) discourse. In other words, they belie their authority. A dis-locution is often apparent with footnote comments " <i>The author ought to consider</i> " Later changed to the " <i>reader, before we continue I intend to digress</i> " becomes direct and personal. Changes in narrative (3 rd to 1 st person) suggest a shift in the annotator's tone revealing ambivalence, a see-sawing that makes the reader uncertain of the commentary implications.
Slights (1992)	Biblical annotation, interpretation and theological authorities discouraging marginal notes that "spoile" the text. Slights discussed Derrida's theory in the margins of de-construction where

	the whole of the host text is incomplete and there is no frame, no boundary, and no marginal border.
	Each addition to the host text becomes a corpus, a body of writing that is evolving. From the title, to the borders, there is a politicisation and therefore an over-run of the text. It is no longer a finished corpus but extra-textual. Derrida's habit of reading is to "settle ona peripheral fragment in the work - a footnote, a recurrent minor term orand work itthrough to the point where it threatens to dismantle the oppositions which govern the text as a whole" (Eagleton, 2008, p. 133-34). Therefore, Derrida is saying do not focus on my text, but the margins where the essentials will be grasped.
Henige (2001)	Editorial annotation is defined as "anything that improves access to, or understanding of, a given text without changing it" (p. 97). Henige therefore includes editorial introductions,
Literary text editors	glossaries and bibliographies.
	Normal editorial practice has some short annotation for example information about the author, his work, the importance of the article. Translated articles need to be checked and annotated
	for their reliability, authenticity, and textual accuracy. The use of annotation with references is
	necessary otherwise the source alone is the annotator and an assertion. Annotation should
	include both reasons for believing and reasons for doubting in a realistic context. The sizes of foot/ end notes are suggested to be no more than the host text.
Henige (2002)	"All practices of annotation are riddled with paradox" (p.87) designed to convince both
	doubters and believers. At their core, irrespective of the surrounding annotation, they aim to
Literary editors	"access" textual understanding. Access to the mind of the author, other thoughts on the matter,
and publishers	or the properties of the text (p. 87). More forms of annotation are the use of quotations, seductive titles, tables of contents, prefaces (to capture the attention of the reader), epigraphs,
	graphs and charts, maps, quoted matter (to shape his own text), facsimiles, appendices,
	glossaries, bibliographies, and indexes (p. 63). Even up to date hyperlinks can be used to check
	for reliability. Most importantly sound scholarship needs exact, accurate and reliable
	annotations. Annotation of text refers to different observers and different angles of perspective.
	A writer's intention is always to have some influence on others and to bring the reader with them based on courtesy, rigour, logic and hard work.
	them bused on courtes, ingout, logic and hard work.

The second category of retrieved sources found 13 research studies which I present in figure 6 next, two of which were literature reviews primarily focussed on annotation (Ball, 2010; Neuwirth & Wolfe, 2001). A number of papers were written from original studies so the number of original research studies could in fact be reduced to seven (Ball et al., 2008; Feito & Donahue, 2008; Liu, 2006; Marshall, 1997; McColly, 1965; Wolfe 2002; Regan, 2010). For example, following the norms of research dissemination, Marshall's (1997) original research study led to further re-writing for publication in Marshall (1998a; 1998b) and Marshall and Brush (2004). Ball et al (2008) was also disseminated through Ball et al (2009)

and Ball (2009; 2010). However, due to the dissemination, and critical development of the research findings, all 13 will be discussed. A few seminal papers focussed on annotation, rather than technology, are included in this literature review because of their comprehensive discussion on annotation (Marshall, 1997; 1998a; Marshall & Brush, 2004; Wolfe & Neuwirth, 2001). For example, Wolfe and Neuwirth (2001) and Wolfe's (2002) papers focus initially on the sociohistorical aspects of annotation before locating digital annotation within this tradition. These earlier studies focus primarily on annotation processes and offer comprehensive information about annotation at a time when research studies were few (see figure 6). Therefore, the studies were instrumental in initially developing my knowledge base on annotation for contemporary use and discussion in this thesis (Marshall, 1997; 1998a; 1998b; Wolfe & Neuwirth, 2001; Wolfe, 2002). One research study by Hyland and Hyland (2006), was later excluded because only end text and written feedback on separate sheets were analysed. Of the 13 research studies, five came from nursing with no research studies from other healthcare disciplines found (see figure 6). Therefore, it appears that annotation studies for nurse education practice means that annotation has a unique nursing focus, which requires further exploration. The following figure 6 summarises the findings of the research studies.

Figure 6: Research studies 1965 to 2016

Research papers n=13 two of which are literature reviews			
Author,	Study design & methods	Findings and conclusion	
context & location			
McColly	n=32 English teachers'	Findings: Identified four internalised dimensions which	
(1965)	handwritten annotation of four compositions of 9-13 grade	affects the impact of annotation; content- style factor, scope- depth and purposefulness; tone- reflecting feeling	
United States	students' in secondary school,	and attitude and lastly; visual impact and	
(US) research	totalling 313 compositions.	appropriateness.	
study. 9-13			
grade students			
Marshall	Annotated samples obtained	Explored the use of annotation in a digital library. All	
-----------------------	--	--	
(1997)	from students' annotation in	annotations are not equal and are created for different	
US research	textbooks (n=150).	purposes.	
study		Findings: two categories-form and function. Form- Within text telegraphic statements e.g. underlining, highlight, circles, boxes. In the margins- brackets, circles, stars. Explicit annotations are textual and in the text in the form of highlighting, removal or permanent, written between lines, long or short words, phrases in margins.	
		Functions: interpretation, aid memoire, attention for future use, tracing progress through difficult narrative, incidental reflection. Implications for the digital library; Annotation limited by current technology, following others annotation may prove difficult, readers like to write on the physical material. Private to public annotation sharing is less expected but useful to future readers.	
Marshall (1998a;	Long term study of a community of annotators	Systems develop hypertext not annotations of hypertext. Taking advantage of individual practices in order to	
(1998a, 1998b)	(n=50) books from the	increase hypertext value for future readers.	
	earlier 1997 study, examined	Findings:	
US research study	(n=410) books representing 39 titles in	Reinforced the 1997 findings and discussion. Author suggests it is important to view annotation practices (on	
study	21 different subject areas.	paper and digital) as continuous not a dichotomy.	
		Annotation is a practice that develops over time.	
		Experience and disciplinary expectations change the marks people make (p. 42).	
		Annotations on paper are hyper-textual, non-linear in relationship to the linear line.	
		They interrupt, connect, are playful, informal, serious	
		and a direct reflection of the reader's engagement with	
		the host text. Digital annotation needs to reflect this (p. 43). The research article later develops technological	
		43). The research article later develops technological discussion that becomes less relevant to this review.	
Wolfe (2002)	n=122 undergraduate students.	Underlining text does not mean the student will	
US research study	Students assigned to 1 of 4	understand what has been highlighted.	
Study	annotation hand-written	The annotators assume the student will understand.	
	feedback styles.		
	Students were asked to write an	Students are outside the lecturer's interpretive community and need support to understand annotation.	
	essay, followed- by a	Tone can be identified through the choice of language	
	questionnaire, analysed for	and how it is projected. Annotators' presence can	
	recall of annotated information	influence the student's engagement with the text.	
	and views about annotation.	Annotation invites the student into the academic community they aspire to join. Receiving evaluative	
		annotation means students are more likely to engage with	
XX7-10 1		the text compared to students not receiving annotation.	
Wolfe and Nuewirth	Review of research and technical advances and benefits	Historical annotation and research on annotation benefits of digital technologies are more interactive, accessible	
1,00,01111	teeninear auvances and benefits		

(2001)	of annotation to aid learning.	and leave traces that can be built upon.
US Review of the literature technology and higher		Medieval annotators copied and shared their text: in contrast printed technologies are bought individually and are not shared unless sold on. Digital annotation resources promote this tradition. There is a distinction between private and public annotation; the former is more widespread and personal, the latter intended for sharing with other readers.
education		When reading annotation occurs 25% of the time and for many different reasons: acquiring knowledge, developing new insights, aiding re-reading and memory recall (p. 338).
		Annotation promotes re-reading, taking notice, interpretation, and " <i>eavesdropping in the insights of</i> <i>others</i> " (p.347), shared annotation allows development of consensus and controversy.
		Four purposes of annotation are currently known:
		To facilitate reading and writing (e.g. through self- directed annotations whilst reading.
		To benefit from the insights of other readers' annotations.
		To provide feedback to writers and promote collaboration with others.
		To call attention to topics and important passages (annotations made by the author for the benefit of the reader).
Marshall and Brush (2004)	Three sources of data: semi- structured interviews, a collection of annotated readings on paper and	Develops further Marshall's previous work (1997; 1998a; 1998b) and focuses on manipulation and re-use of annotations for online document discussion.
US technology and sharing annotation	participants' contributions to online discussions.	Annotations findings: Anchor - underlining, highlighting, circle, margin bar. Content - Notes, marks (e.g. *), other (doodles). Compound- anchor and the above (p. 351). Of 1700 annotations only 7.8% were made public. What was shared was dramatically revised.
Liu (2006) Writing and comprehension studies for	n=40 students taught how to use annotation strategies in order to help them develop critical writing composition	The study of learning strategies is ultimately aimed at understanding how to help students improve their ability to learn.
freshman Composition classes in	skills. Two questions were asked;	Learning strategies help understanding and improve their ability to learn. Examine the qualitative differences between individual learning strategies among students
college	1/Do you feel that the strategies we went over affected the way you revised your first draft and	whose ability to write critical and analytical essays varied.
US research study	prepared the second draft?	Findings: Students' producing a lack of annotation demonstrates surface learning and a weak essay. More

	2/When you read which	skilful student annotators produced more critical and
	2/ When you read, which strategy or strategies do you	skilful student annotators produced more critical and analytical writing than less engaged annotators.
	prefer to use to help you	Too show such as its such found to be how to the discussion of
	comprehend the material and later write about it?	Teacher authority was found to be key to the diagnosis of learning abilities, providing instructions and learning strategies.
Feito and Donahue (2008)	n=32 undergraduate arts students at a liberal arts college	Textual annotation divided into purposes using Wolfgang Iser's (2006) taxonomy; trackings, gap finding, personal and literary repertoire towards consistency building familiarity:
US Art		fainnairty.
research study		Lecturers' know very little about their students' annotative practice. Annotation is often subordinated by other pedagogical issues. Annotation by students reinforces the concept of owning other peoples' language. Annotation is in widespread use but lecturers do not know about students' practises. Teaching annotation to students helped reading awareness.
		"annotations are never neutral or arbitrary but represent interpretive decisions" (p. 298).
		Apply Iser's (2006) work. Causal nature of textual inter- subjectivity. Student ownership of the host essay through annotation. Annotation involves internalised signification. Annotation and 4 categories: trackings, identifying gaps, individual and literary repertoire. Difficult reading is a resource for linking other literary work. Every new read refines what is known.
Ball, Franks,	A Teaching and Learning	Research methods; literature review, student, staff
Jenkins, McGrath &	quality improvement scheme funded research study allowed	questionnaires, staff focus group, random sample of level 3 post qualifying annotated student essays ($n = 40$). The
Leigh (2008)	for a small team from a school of nursing to explore the impact of annotation on student learning.	findings are published in Ball et al (2009), Ball, (2009; 2010).
Ball et al	Handwritten annotation	Students found annotated feedback helpful, identified
(2009)	feedback for higher education at the summative stage and its	strengths and weaknesses when annotation feedback understood. There was no consensus that the lecturer
UK research study Nursing	impact on students and lecturers.	projected an underlying tone or attitude to the students. Despite this some annotation had a tone which students' felt was critical and unhelpful.
	(n=249) students and lecturers (n=74) perceptions of annotation research.	Annotation feedback is more demanding when using a feedback grid. Second markers were influenced by the presence of the first marker. Lecturers' felt annotation feedback was clear: ticks and underlining were helpful to the student.
Ball (2009)	Participatory action $n-4$ lecturers analysing 40	Study focussed on annotation and tone.
UK research	n=4 lecturers analysing 40 marked and handwritten	Some questions about annotation practice in higher
study Nursing	annotated essays	education remain largely unanswered. The literature is unconcerned with the content of comments and their effect on the reader. However, if there is a negative tone
<u> </u>		in the annotation students consider it disparaging and

Ball (2010) Review of the literature Ball (2010) Review of the literature Focus of the search evidence of pedagos where and students' learning? A participating learning and the students' learning and the student and the stude student and the student and the student and th			
Ball (2010) Review of the literature Focus on handwriten pecilianotation is given by the same of second withing and the same second within the same second with the same seco			
Ball (2010) Review of the literature For the search evidence of pedagogy with practice form annotation is students and writing is quite shocking really; it looks aggressive" (p. 120). **Semiotic signs and practices form annotation's textual content" (p. 112). Annotation is given by lecturers but lacks training. Annotation comments are rarely moderated. Tone and irritability are noticeable in lecturers' annotation by lecturers themselves. 2 ^m marker annotators are affected by 1 ^m marker annotators are affected by 1 ^m marker annotators are not taught nor moderated. A close reading by the annotator means students have direct reference points identifying strengths and weaknesses Ball (2010) Review of the literature Focus on handwritten pencil annotations are not taught nor moderated. A close reading by the annotation lacks research evidence of pedagogy with practice largely being inductive and individual. Feedback needs to be transparent, nurturing and given in the context of summative and formative. Discussed 8 research studies found between 1965 to 2009 (Feito & Donahue, 2008; Liu, 2006; McColly, 1965; Marshall, 1998b; Porter-Oronenl, 2004; Wolfe 2002; Wolfe & Nuewirth, 2001). Students' borrow the perceptive skills of the lecturer. It is a nurturing process. Lecturers should reflect on their own development. Visual marks triggers meaning (semiotics-signifier and signified) and resonance of language as a whole. Annotation may telegraph authority, tone, can be hasty and misunderstood. *Because of the mixture of comments, phrases and signs, annotation is more annuaction tha annotation tha annotation tha annotation tha annotation tha annotation the student. *Because of the mixt			
scribbles and writing is quite shocking really; it looks aggressive" (p. 120). **Semiotic signs and practices form annotation's textual content" (p. 112). Annotation is given by lecturers but lacks training. Annotation comments are rarely moderated. Tone and irritability are noticeable in lecturers' annotation by lecturers themselves. 2 nd marker annotations are not taught in marker annotation. Lengthy annotation is unhelpful. Brief, balanced and constructive comments helped the student make links. Annotations are not taught nor moderated. A close reading by the annotator means students have direct reference points identifying strengths and weaknesses Ball (2010) Review of the literature Focus on handwritten pencil annotation. Annotation lacks research evidence of pedagogy with practice largely being inductive and individual. Feedback needs to be transparent, nutruring and given in the context of summative and formative. Discussed 8 research studies found between 1965 to 2009 (Feito & Donahue, 2008; Liu, 2006; McColly, 1965; Marshall, 1998; Porter-O'Donnell, 2004: Wolfe 2002; Wolfe & Nuewirth, 2001). Students' borrow the perceptive skills of the lecturer. It is a nutruring process. Lecturers should reflect on their own development. Visual marks triggers meaning (semiotics-signifier and signified) and resonance of language as a whole. Annotation may telegraph authority, tone, can be hasty and misunderstood. *'Because of the mixture of comments, phrases and signs, annotation is more annunciation than announcement (and for this reason perhaps, annotation can escape meaning* (citing Derrida, 1982 in Ball, 2010, p. 142). For annotation the meaningful, it has to signify meaning to the student.			students' learning? A participating lecturer stated:
scribbles and writing is quite shocking really; it looks aggressive" (p. 120). **Semiotic signs and practices form annotation's textual content" (p. 112). Annotation is given by lecturers but lacks training. Annotation comments are rarely moderated. Tone and irritability are noticeable in lecturers' annotation by lecturers themselves. 2 nd marker annotations are not taught in marker annotation. Lengthy annotation is unhelpful. Brief, balanced and constructive comments helped the student make links. Annotations are not taught nor moderated. A close reading by the annotator means students have direct reference points identifying strengths and weaknesses Ball (2010) Review of the literature Focus on handwritten pencil annotation. Annotation lacks research evidence of pedagogy with practice largely being inductive and individual. Feedback needs to be transparent, nutruring and given in the context of summative and formative. Discussed 8 research studies found between 1965 to 2009 (Feito & Donahue, 2008; Liu, 2006; McColly, 1965; Marshall, 1998; Porter-O'Donnell, 2004: Wolfe 2002; Wolfe & Nuewirth, 2001). Students' borrow the perceptive skills of the lecturer. It is a nutruring process. Lecturers should reflect on their own development. Visual marks triggers meaning (semiotics-signifier and signified) and resonance of language as a whole. Annotation may telegraph authority, tone, can be hasty and misunderstood. *'Because of the mixture of comments, phrases and signs, annotation is more annunciation than announcement (and for this reason perhaps, annotation can escape meaning* (citing Derrida, 1982 in Ball, 2010, p. 142). For annotation the meaningful, it has to signify meaning to the student.			······································
aggressive" (p. 120). "Semiotic signs and practices form annotation's textual content" (p. 112). Annotation is given by lecturers but lacks training. Annotation comments are rarely moderated. Tone and irritability are noticeable in lecturers' annotation by lecturers themselves. 2 ^{ed} marker annotators are affected by 1 st marker annotation. Lengthy annotators are affected by 1 st marker annotation. Lengthy annotation is unhelpful. Brief, balanced and constructive comments helped the student make links. Annotations are not taught nor moderated. A close reading by the annotator means students have direct reference points identifying strengths and weaknesses Ball (2010) Review of the literature Focus on handwritten pencil annotation. Annotation lacks research evidence of pedagogy with practice largely being inductive and individual. Feedback needs to be transparent, nurturing and given in the context of summative and formative. Discussed 8 research studies found between 1965 to 2009 (Feito & Donahue, 2008; Liu, 2006; McColly, 1965; Marshall, 1998a; 1998b; Porter-O'Donnell, 2004: Wolfe 2002; Wolfe & Nuewirth, 2001). Students' borrow the perceptive skills of the lecturer. It is a nurturing process. Lecturers should reflect on their own development. Visual marks triggers meaning (semiotics-signifier and signified) and resonance of language as a whole. Annotation may telegraph authority, tone, can be hasty and misunderstood. "Because of the mixture of comments, phrases and sign; annotation is more annunciation than announcement (and for this reason perhaps, annotation can escape meaning" (citing Derrida, 1982 in Ball, 2010, p. 142). For annotation to be meaningful, it has to signify meaning to the student.			
Ball (2010) Review of the literature Ball (2010) Review of the literature Fessor Focus on handwritten pencil annotation. Annotation is gurmative and formative. Discussed 8 research review Wirsing Nursing Nursing Nursing Review of the literature Focus on handwritten pencil annotation. Annotation is unmative and formative. Discussed 8 research review Nursing Nursing Visual marks triggers meaning (semiotics-signifier and signified) and resonance of language as a whole. Annotation may relegraph authority, tone, can be hasty and misunderstood. "Semiotic is one anning for this reason perhaps, annotation can escape meaning," (citing Derrida, 1982 in Ball, 2010)			
Image: strange in the straight of the straight is strange in the straight is straight is straight is straight is straight in the straight is s			<i>aggressive</i> (p. 120).
Image: strange in the straight of the straight is strange in the straight is straight is straight is straight is straight in the straight is s			" Semiotic signs and practices form appotation's
Annotation is given by lecturers but lacks training. Annotation comments are rarely moderated. Tone and irritability are noticeable in lecturers' annotation by lecturers themselves. 2 nd marker annotators are affected by 1 st marker annotation. Lengthy annotation is unhelpful. Brief, balanced and constructive comments helped the student make links. Annotation means students have direct reference points identifying strengths and weaknessesBall (2010)Review of the literatureFocus on handwritten pencil annotation. Annotation lacks research evidence of pedagogy with practice largely being inductive and individual. Feedback needs to be transparent, nurturing and given in the context of summative and formative. Discussed 8 research studies found between 1965 to 2009 (Feito & Donahue, 2008; Liu, 2006; McColly, 1965; Marshall, 1998a; 1998b; Porter-O'Donnell, 2004: Wolfe 2002; Wolfe & Nuewirth, 2001). Students' borrow the perceptive skills of the lecturer. It is a nurturing process. Lecturers should reflect on their own development.Visual marks triggers meaning (semiotics-signifier and signified) and resonance of language as a whole. Annotation may telegraph authority, tone, can be hasty and misunderstood."Because of the mixture of comments, phrases and signif, annotation is more annunciation than announcement (and for this reason perhaps, annotation can escape meaning" (citing Derrida, 1982 in Ball, 2010, p. 142). For annotation to be meaningful, it has to signify meaning to the student.			
Annotation comments are rarely moderated. Tone and irritability are noticeable in lecturers' annotation by lecturers themselves. 2nd marker annotators are affected by 1st marker annotation. Lengthy annotation is unhelpful. Brief, balanced and constructive comments helped the student make links. Annotatiom means students have direct reference points identifying strengths and weaknessesBall (2010)Review of the literatureFocus on handwritten pencil annotation. Annotation lacks research evidence of pedagogy with practice largely being inductive and individual. Feedback needs to be transparent, nurturing and given in the context of summative and formative. Discussed 8 research studies found between 1965 to 2009 (Feito & Donahue, 2008; Liu, 2006; McColly, 1965; Marshall, 1998a; 1998b; Porter-O'Donnell, 2004: Wolfe 2002; Wolfe & NursingNursingVisual marks triggers meaning (semiotics-signifier and signified) and resonance of language as a whole. Annotation may telegraph authority, tone, can be hasty and misunderstood."Because of the mixture of comments, phrases and signif, annotation is more annunciation than announcement (and for this reason perhaps, annotation can escape meaning" (citing Derrida, 1982 in Ball, 2010, p. 142). For annotation to be meaningful, it has to signify meaning to the student.			(p. 112).
Annotation comments are rarely moderated. Tone and irritability are noticeable in lecturers' annotation by lecturers themselves. 2nd marker annotation is unhelpful. Brief, balanced and constructive comments helped the student make links. Annotatiom raeans students have direct reference points identifying strengths and weaknessesBall (2010)Review of the literatureFocus on handwritten pencil annotation. Annotation lacks research evidence of pedagogy with practice largely being inductive and individual. Feedback needs to be transparent, nurturing and given in the context of summative and formative. Discussed 8 research studies found between 1965 to 2009 (Feito & Donahue, 2008; Liu, 2006; McColly, 1965; Marshall, 1998a; 1998b; Porter-O'Donnell, 2004: Wolfe 2002; Wolfe & NursingNursingVisual marks triggers meaning (semiotics-signifier and signified) and resonance of language as a whole. Annotation may telegraph authority, tone, can be hasty and misunderstood."Because of the mixture of comments, phrases and signify meaning to the student.			Annotation is given by lecturers but lacks training.
irritability are noticeable in lecturers' annotation by lecturers themselves. 2 nd marker annotators are affected by 1 st marker annotation. Lengthy annotation is unhelpful. Brief, balanced and constructive comments helped the student make links. Annotations are not taught nor moderated. A close reading by the annotator means students have direct reference points identifying strengths and weaknessesBall (2010)Review of the literatureFocus on handwritten pencil annotation. Annotation lacks research evidence of pedagogy with practice largely being inductive and individual. Feedback needs to be transparent, nurturing and given in the context of summative and formative. Discussed 8 research studies found between 1965 to 2009 (Feito & Donahue, 2008; Liu, 2006; McColly, 1965; Marshall, 1998a; 1998b; Porter-O'Donnell, 2004: Wolfe 2002; Wolfe & NursingNursingVisual marks triggers meaning (semiotics-signifier and signified) and resonance of language as a whole. Annotation may telggraph authority, tone, can be hasty and misunderstood."Because of the mixture of comments, phrases and signs, annotation is more annunciation than announcement (and for this reason perhaps, annotation can escape meaning" (citing Derrida, 1982 in Ball, 2010, p. 142). For annotation to be meaningful, it has to signify meaning to the student.			
by 1st marker annotation. Lengthy annotation is unhelpful. Brief, balanced and constructive comments helped the student make links. Annotations are not taught nor moderated. A close reading by the annotator means students have direct reference points identifying strengths and weaknessesBall (2010)Review of the literatureFocus on handwritten pencil annotation. Annotation lacks research evidence of pedagogy with practice largely being inductive and individual. Feedback needs to be transparent, nurturing and given in the context of summative and formative. Discussed 8 research studies found between 1965 to 2009 (Feito & Donahue, 2008; Liu, 2006; McColly, 1965; Marshall, 1998a; 1998b; Porter-O'Donnell, 2004: Wolfe 2002; Wolfe & NursingNursingVisual marks triggers meaning (semiotics-signifier and signified) and resonance of language as a whole. Annotation may telegraph authority, tone, can be hasty and misunderstood."Because of the mixture of comments, phrases and signs, annotation is more annuciation than announcement (and for this reason perhaps, annotation can escape meaning." (citing Derrida, 1982 in Ball, 2010, p. 142). For annotation to be meaningful, it has to signify meaning to the student.			
unhelpful. Brief, balanced and constructive comments helped the student make links. Annotations are not taught nor moderated. A close reading by the annotator means students have direct reference points identifying strengths and weaknessesBall (2010)Review of the literatureFocus on handwritten pencil annotation. Annotation lacks research evidence of pedagogy with practice largely being inductive and individual. Feedback needs to be transparent, nurturing and given in the context of summative and formative. Discussed 8 research studies found between 1965 to 2009 (Feito & Donahue, 2008; Liu, 2006; McColly, 1965; Marshall, 1998a; 1998b; Porter-O'Donnell, 2004: Wolfe 2002; Wolfe & Nuewirth, 2001). Students' borrow the perceptive skills of the lecturer. It is a nurturing process. Lecturers should reflect on their own development.Visual marks triggers meaning (semiotics-signifier and signified) and resonance of language as a whole. Annotation may telegraph authority, tone, can be hasty and misunderstood."Because of the mixture of comments, phrases and signs, annotation is more annunciation than announcement (and for this reason perhaps, annotation ccan escape meaning" (citing Derrida, 1982 in Ball, 2010, p. 142). For annotation to be meaningful, it has to signify meaning to the student.			
helped the student make links. Annotations are not taught nor moderated. A close reading by the annotator means students have direct reference points identifying strengths and weaknessesBall (2010)Review of the literatureFocus on handwritten pencil annotation. Annotation lacks research evidence of pedagogy with practice largely being inductive and individual. Feedback needs to be transparent, nurturing and given in the context of summative and formative. Discussed 8 research studies found between 1965 to 2009 (Feito & Donahue, 2008; Liu, 2006; McColly, 1965; Marshall, 1998a; 1998b; Porter-O'Donnell, 2004: Wolfe 2002; Wolfe & NursingNursingVisual marks triggers meaning (semiotics-signifier and signified) and resonance of language as a whole. Annotation may telegraph authority, tone, can be hasty and misunderstood."Because of the mixture of comments, phrases and signs, annotation is more annunciation than announcement (and for this reason perhaps, annotation can escape meaning" (citing Derrida, 1982 in Ball, 2010, p. 142). For annotation to be meaningful, it has to signify meaning to the student.			
nor moderated. A close reading by the annotator means students have direct reference points identifying strengths and weaknessesBall (2010)Review of the literatureFocus on handwritten pencil annotation. Annotation lacks research evidence of pedagogy with practice largely being inductive and individual. Feedback needs to be transparent, nurturing and given in the context of summative and formative. Discussed 8 research studies found between 1965 to 2009 (Feito & Donahue, 2008; Liu, 2006; McColly, 1965; Marshall, 1998a; 1998b; Porter-O'Donnell, 2004: Wolfe 2002; Wolfe & Nuewirth, 2001). Students' borrow the perceptive skills of the leturer. It is a nurturing process. Lecturers should reflect on their own development.Visual marks triggers meaning (semiotics-signifier and signified) and resonance of language as a whole. Annotation may telegraph authority, tone, can be hasty and misunderstood."Because of the mixture of comments, phrases and signs, annotation is more annunciation than announcement (and for this reason perhaps, annotation can escape meaning" (citing Derrida, 1982 in Ball, 2010, p. 142). For annotation to be meaningful, it has to signify meaning to the student.			
Ball (2010)Review of the literaturestudents have direct reference points identifying strengths and weaknessesBall (2010)Review of the literatureFocus on handwritten pencil annotation. Annotation lacks research evidence of pedagogy with practice largely being inductive and individual. Feedback needs to be transparent, nurturing and given in the context of summative and formative. Discussed 8 research studies found between 1965 to 2009 (Feito & Donahue, 2008; Liu, 2006; McColly, 1965; Marshall, 1998a; 1998b; Porter-O'Donnell, 2004: Wolfe 2002; Wolfe & Nuewirth, 2001). Students' borrow the perceptive skills of the lecturer. It is a nurturing process. Lecturers should reflect on their own development.Visual marks triggers meaning (semiotics-signifier and signified) and resonance of language as a whole. Annotation may telegraph authority, tone, can be hasty and misunderstood."Because of the mixture of comments, phrases and signs, annotation is more annunciation than announcement (and for this reason perhaps, annotation can escape meaning" (citing Derrida, 1982 in Ball, 2010, p. 142). For annotation to be meaningful, it has to signify meaning to the student.			1
Ball (2010)Review of the literatureStrengths and weaknessesBall (2010)Review of the literatureFocus on handwritten pencil annotation. Annotation lacks research evidence of pedagogy with practice largely being inductive and individual. Feedback needs to be transparent, nurturing and given in the context of summative and formative. Discussed 8 research studies found between 1965 to 2009 (Feito & Donahue, 2008; Liu, 2006; McColly, 1965; Marshall, 1998a; 1998b; Porter-O'Donnell, 2004: Wolfe 2002; Wolfe & Nuewirth, 2001). Students' borrow the perceptive skills of the lecturer. It is a nurturing process. Lecturers should reflect on their own development.Visual marks triggers meaning (semiotics-signifier and signified) and resonance of language as a whole. Annotation may telegraph authority, tone, can be hasty and misunderstood."Because of the mixture of comments, phrases and signs, annotation is more annunciation than announcement (and for this reason perhaps, annotation can escape meaning" (citing Derrida, 1982 in Ball, 2010, p. 142). For annotation to be meaningful, it has to signify meaning to the student.			
Ball (2010)Review of the literatureFocus on handwritten pencil annotation. Annotation lacks research evidence of pedagogy with practice largely being inductive and individual. Feedback needs to be transparent, nurturing and given in the context of summative and formative. Discussed 8 research studies found between 1965 to 2009 (Feito & Donahue, 2008; Liu, 2006; McColly, 1965; Marshall, 1998a; 1998b; Porter-O'Donnell, 2004: Wolfe 2002; Wolfe & NursingNursingVisual marks triggers meaning (semiotics-signifier and signified) and resonance of language as a whole. Annotation may telegraph authority, tone, can be hasty and misunderstood."Because of the mixture of comments, phrases and signs, annotation is more annunciation than announcement (and for this reason perhaps, annotation can escape meaning" (citing Derrida, 1982 in Ball, 2010, p. 142). For annotation to be meaningful, it has to signify meaning to the student.			· · ·
UK literature & research reviewlacks research evidence of pedagogy with practice largely being inductive and individual. Feedback needs to be transparent, nurturing and given in the context of summative and formative. Discussed 8 research studies found between 1965 to 2009 (Feito & Donahue, 2008; Liu, 2006; McColly, 1965; Marshall, 1998a; 1998b; Porter-O'Donnell, 2004: Wolfe 2002; Wolfe & Nuewirth, 2001). Students' borrow the perceptive skills of the lecturer. It is a nurturing process. Lecturers should reflect on their own development.Visual marks triggers meaning (semiotics-signifier and signified) and resonance of language as a whole. Annotation may telegraph authority, tone, can be hasty and misunderstood."Because of the mixture of comments, phrases and signs, annotation is more annunciation than announcement (and for this reason perhaps, annotation can escape meaning" (citing Derrida, 1982 in Ball, 2010, p. 142). For annotation to be meaningful, it has to signify meaning to the student.	$B_{0}11(2010)$	Davian of the literature	
UK literature & research reviewlargely being inductive and individual. Feedback needs to be transparent, nurturing and given in the context of summative and formative. Discussed 8 research studies found between 1965 to 2009 (Feito & Donahue, 2008; Liu, 2006; McColly, 1965; Marshall, 1998a; 1998b; Porter-O'Donnell, 2004: Wolfe 2002; Wolfe & Nuewirth, 2001). Students' borrow the perceptive skills of the lecturer. It is a nurturing process. Lecturers should reflect on their own development.Visual marks triggers meaning (semiotics-signifier and signified) and resonance of language as a whole. Annotation may telegraph authority, tone, can be hasty and misunderstood."Because of the mixture of comments, phrases and signs, annotation is more annunciation than announcement (and for this reason perhaps, annotation can escape meaning" (citing Derrida, 1982 in Ball, 2010, p. 142). For annotation to be meaningful, it has to signify meaning to the student.	Dall (2010)	Review of the interature	
& research reviewto be transparent, nurturing and given in the context of summative and formative. Discussed 8 research studies found between 1965 to 2009 (Feito & Donahue, 2008; Liu, 2006; McColly, 1965; Marshall, 1998a; 1998b; Porter-O'Donnell, 2004: Wolfe 2002; Wolfe & Nuewirth, 2001). Students' borrow the perceptive skills of the lecturer. It is a nurturing process. Lecturers should reflect on their own development.Visual marks triggers meaning (semiotics-signifier and signified) and resonance of language as a whole. Annotation may telegraph authority, tone, can be hasty and misunderstood."Because of the mixture of comments, phrases and signs, annotation is more annunciation than announcement (and for this reason perhaps, annotation can escape meaning" (citing Derrida, 1982 in Ball, 2010, p. 142). For annotation to be meaningful, it has to signify meaning to the student.	UK literature		
review Nursing Nursing Summative and formative. Discussed 8 research studies found between 1965 to 2009 (Feito & Donahue, 2008; Liu, 2006; McColly, 1965; Marshall, 1998a; 1998b; Porter-O'Donnell, 2004: Wolfe 2002; Wolfe & Nuewirth, 2001). Students' borrow the perceptive skills of the lecturer. It is a nurturing process. Lecturers should reflect on their own development. Visual marks triggers meaning (semiotics-signifier and signified) and resonance of language as a whole. Annotation may telegraph authority, tone, can be hasty and misunderstood. "…Because of the mixture of comments, phrases and signs, annotation is more annunciation than announcement (and for this reason perhaps, annotation can escape meaning…" (citing Derrida, 1982 in Ball, 2010, p. 142). For annotation to be meaningful, it has to signify meaning to the student.			
Nursingfound between 1965 to 2009 (Feito & Donahue, 2008; Liu, 2006; McColly, 1965; Marshall, 1998a; 1998b; Porter-O'Donnell, 2004: Wolfe 2002; Wolfe & Nuewirth, 2001). Students' borrow the perceptive skills of the lecturer. It is a nurturing process. Lecturers should reflect on their own development.Visual marks triggers meaning (semiotics-signifier and signified) and resonance of language as a whole. Annotation may telegraph authority, tone, can be hasty and misunderstood."Because of the mixture of comments, phrases and signs, annotation is more annunciation than announcement (and for this reason perhaps, annotation can escape meaning" (citing Derrida, 1982 in Ball, 2010, p. 142). For annotation to be meaningful, it has to signify meaning to the student.			
NursingLiu, 2006; McColly, 1965; Marshall, 1998a; 1998b; Porter-O'Donnell, 2004: Wolfe 2002; Wolfe & Nuewirth, 2001). Students' borrow the perceptive skills of the lecturer. It is a nurturing process. Lecturers should reflect on their own development.Visual marks triggers meaning (semiotics-signifier and signified) and resonance of language as a whole. Annotation may telegraph authority, tone, can be hasty and misunderstood."Because of the mixture of comments, phrases and signs, annotation is more annunciation than announcement (and for this reason perhaps, annotation can escape meaning" (citing Derrida, 1982 in Ball, 2010, p. 142). For annotation to be meaningful, it has to signify meaning to the student.			
 Porter-O'Donnell, 2004: Wolfe 2002; Wolfe & Nuewirth, 2001). Students' borrow the perceptive skills of the lecturer. It is a nurturing process. Lecturers should reflect on their own development. Visual marks triggers meaning (semiotics-signifier and signified) and resonance of language as a whole. Annotation may telegraph authority, tone, can be hasty and misunderstood. <i>"Because of the mixture of comments, phrases and signs, annotation is more annunciation than announcement (and for this reason perhaps, annotation can escape meaning"</i> (citing Derrida, 1982 in Ball, 2010, p. 142). For annotation to be meaningful, it has to signify meaning to the student. 	Nursing		
of the lecturer. It is a nurturing process. Lecturers should reflect on their own development.Visual marks triggers meaning (semiotics-signifier and signified) and resonance of language as a whole. Annotation may telegraph authority, tone, can be hasty and misunderstood."Because of the mixture of comments, phrases and signs, annotation is more annunciation than announcement (and for this reason perhaps, annotation can escape meaning" (citing Derrida, 1982 in Ball, 2010, p. 142). For annotation to be meaningful, it has to signify meaning to the student.	C C		
reflect on their own development. Visual marks triggers meaning (semiotics-signifier and signified) and resonance of language as a whole. Annotation may telegraph authority, tone, can be hasty and misunderstood. "Because of the mixture of comments, phrases and signs, annotation is more annunciation than announcement (and for this reason perhaps, annotation can escape meaning" (citing Derrida, 1982 in Ball, 2010, p. 142). For annotation to be meaningful, it has to signify meaning to the student.			Nuewirth, 2001). Students' borrow the perceptive skills
 Visual marks triggers meaning (semiotics-signifier and signified) and resonance of language as a whole. Annotation may telegraph authority, tone, can be hasty and misunderstood. <i>"Because of the mixture of comments, phrases and signs, annotation is more annunciation than announcement (and for this reason perhaps, annotation can escape meaning"</i> (citing Derrida, 1982 in Ball, 2010, p. 142). For annotation to be meaningful, it has to signify meaning to the student. 			
 signified) and resonance of language as a whole. Annotation may telegraph authority, tone, can be hasty and misunderstood. <i>"Because of the mixture of comments, phrases and signs, annotation is more annunciation than announcement (and for this reason perhaps, annotation can escape meaning"</i> (citing Derrida, 1982 in Ball, 2010, p. 142). For annotation to be meaningful, it has to signify meaning to the student. 			reflect on their own development.
 signified) and resonance of language as a whole. Annotation may telegraph authority, tone, can be hasty and misunderstood. <i>"Because of the mixture of comments, phrases and signs, annotation is more annunciation than announcement (and for this reason perhaps, annotation can escape meaning"</i> (citing Derrida, 1982 in Ball, 2010, p. 142). For annotation to be meaningful, it has to signify meaning to the student. 			Visual marks triggers meaning (semiotics signifier and
 Annotation may telegraph authority, tone, can be hasty and misunderstood. <i>"…Because of the mixture of comments, phrases and signs, annotation is more annunciation than announcement (and for this reason perhaps, annotation can escape meaning…"</i> (citing Derrida, 1982 in Ball, 2010, p. 142). For annotation to be meaningful, it has to signify meaning to the student. 			
and misunderstood. "…Because of the mixture of comments, phrases and signs, annotation is more annunciation than announcement (and for this reason perhaps, annotation can escape meaning…" (citing Derrida, 1982 in Ball, 2010, p. 142). For annotation to be meaningful, it has to signify meaning to the student.			
"Because of the mixture of comments, phrases and signs, annotation is more annunciation than announcement (and for this reason perhaps, annotation can escape meaning" (citing Derrida, 1982 in Ball, 2010, p. 142). For annotation to be meaningful, it has to signify meaning to the student.			
signs, annotation is more annunciation than announcement (and for this reason perhaps, annotation can escape meaning" (citing Derrida, 1982 in Ball, 2010, p. 142). For annotation to be meaningful, it has to signify meaning to the student.			
signs, annotation is more annunciation than announcement (and for this reason perhaps, annotation can escape meaning" (citing Derrida, 1982 in Ball, 2010, p. 142). For annotation to be meaningful, it has to signify meaning to the student.			"Because of the mixture of comments, phrases and
<i>can escape meaning</i> " (citing Derrida, 1982 in Ball, 2010, p. 142). For annotation to be meaningful, it has to signify meaning to the student.			
2010, p. 142). For annotation to be meaningful, it has to signify meaning to the student.			
signify meaning to the student.			
Regan (2010) I Higher education students and -1.12 studies tound between 1965 and 2010 using the search	D (2010)	TT 1 1 1	
	Regan (2010)	Higher education students and	12 studies found between 1965 and 2010 using the search
UK researchlecturers' views about annotation use for formativeterms "annotation" and "formative assessment" (McColly, 1965; Marshall, 1998a; 1998b; Wolfe 2002;	UK research		
studyfeedback on draft essayWolfe & Nuewirth, 2001; Porter-O'Donnell, 2004; Liu, 2006; Handley et al., 2007; Feito & Donahue, 2008; Ball,		iceutack on trait essay	
Action research, with open 2009; Ball et al., 2007, Fello & Donande, 2008, Ball, 2010).	Tursing	Action research with open	•
ended sections		-	2007, Dan et al., 2007, Dan, 2010).
(n=22 lecturers/n=13 students) Some are now considered to be unhelpful as do not meet			Some are now considered to be unhelpful as do not meet
the inclusion criteria:		(-
Students report passive lecturer tone lacked clarity and			
transparency.			
Tone neutrality was an issue, being so neutral students			Tone neutrality was an issue, being so neutral students

		had to "read between the lines" to the perceived "real" message from the lecturer. This resulted in new interpretations of the feedback given. The reason and implication for lecturer tone remained undeveloped in this paper. Formative stage of draft feedback is better for essay development. Students and lecturers value annotation feedback. Due to tone neutrality, students have to interpret the lecturer's comments.
Ball and Regan (2013)	General action research (Regan, 2010) comparison with participatory action research methods (Ball et al., 2009)	The research methods of both studies were compared in relation to annotation practice. My study (Regan, 2010) was an individual study benefitting my own practice and understanding. In contrast, Ball et al (2009) was a group of experienced nursing lecturers reviewing annotation: identifying reflective themes applied to their own practice. Findings: Tone and frustration when annotation feedback was read back at a later date. This reinforced the need for reflective practice not too dissimilar to from other forms of nursing reflection considered essential for best practice.

3.5 Retrieved literature

As stated in section 3.4.2, the two categories of retrieved literature were first, literary sources (see figure 5) and second, research papers (see figure 6). The latter included two reviews of annotation, Ball (2009) on handwritten annotation in the context of nursing research and Wolfe and Nuewirth (2001) on sharing annotation to support digital reading through new technologies. I found a good starting point to develop an understanding of annotation was a combination of reading the literature reviews and literary sources. The literature suggests that annotation is distributed across multiple boundaries, and its influence found in many subjects (Jackson, 2001). For example, biblical, literary criticism, literary editors and publishers, translation, re-translation and educational annotation studies.

Early retrieved literature demonstrate that annotation is historical and a hierarchical genre used for multiple purposes with institutional power and authority prevailing over the rights of the individual as discussed in chapter two. What the literature identifies is that annotation lacks any instructions, or codicils, to support or uphold rigour (Jackson, 2001).

Annotation's presence in the earliest historical scientific and literary manuscripts identifies it as an important social force. Given this, in order to locate gaps in the literature for the purpose of annotation feedback, the wide variety of literature on annotation helps to recognise it as a historical activity closely linked to social policy (Jackson, 2001). Many of the findings from literary sources presented in figure 5 are contentious and contrast sharply with the findings from the research studies presented in figure 6, which generally concern education, rather than literary criticism. The latter were generally less lively than the literary sources, perhaps, in part, due to their wider use and purpose. However, the scholastic literature was the most illuminating aspect of the review because they brought history back to life to allow contemporary parallels. Annotation for literary purposes, such as editorial annotation to an author, is combative, challenging, baiting, undermining, and has an unsettling agenda (Benstock, 1983).

An author's footnotes aim to be an addition to the text and perhaps may refer to alternate literature. As a result, Benstock (1983) suggests "anything" added to the text brings to annotation an undefined and wildcard dimension because it often depicts overtly biased views, and is seen as a lively social phenomenon and force for change (Henige, 2002; Jackson, 2001). However, Lyons (1967) identifies the baiting attitude and negative tone found in some scholastic annotation in the form of footnotes, "poke fun" at an author's message in an attempt to undermine it with extra-textual dynamics. Literary sources therefore, continue the long tradition

of annotation aiming to share a reader's interpretation, often at the author's expense (Novak, Razzouk & Johnson, 2012).

3.5.1 Annotation in education

The literary annotation (see figure 5) were not characteristic of the retrieved research studies presented in figure 6. This led me to question its difference in educational studies. Figure 5 demonstrates annotation that was stark, lively, baiting in the use of sarcastic comments. In contrast, the annotation styles presented in figure 6 were conservative and constructive. In comparison, educational authors promote an awareness of annotation and rather than research, they discuss practical classroom initiatives. For example, Porter-O'Donnell's (2004) non-research study describes teaching ninth grade US high school students' annotation and a writing-to-learn strategy to facilitate an active dialogue with the text. After training students, the researchers identified the student had a more active dialogue with the text and making notes, which enabled a visible record of thinking to help with further reading.

Reading aloud portions of material promoted thinking about the text and making links to form new associations and Porter-O'Donnell (2004) conclude that annotation improves student comprehension, memory recall, understanding, attention to the text and moving from superficial to deep meaning. However, Porter-O'Donnell's (2004) discussion lacks evidence to support the identified six or seven ways of students responding in annotation such as making predictions, asking questions, slowing down the reading process, stating an opinion, analysing the author's skill, making connections and reflecting on the content or reading process.

3.5.2 A lack of evidence

Knoblauch and Brannon (2006a) suggest teachers' claims of quantifying learning and behaviour change is a myth, with annotation being informed instead by subjective socio-political baggage. Annotation papers often "talk up" the positive nature of teaching annotation without reference to demonstrable evidence (Knoblauch & Brannon, 2006a; 2006b; Straub, 2006). I therefore, sought evidence as to the significance such insights could bring to nurse education because of the health and social centric experiences of nurse annotators. Annotation aims to offer reference points to aid students learning but it is also important to understand what the content of annotation decorating a student's page says about the annotator's own attitude and understanding as professional educators.

3.5.3 Annotation research studies

I now discuss the 13 research studies from figure 6 (Ball et al., 2008; Ball et al., 2009; Ball, 2009; Ball, 2010; Feito & Donahue, 2008; Regan, 2010; Ball & Regan, 2013; Liu, 2006; Marshall, 1997; 1998a; Marshall & Brush, 2004; McColly, 1965; Wolfe, 2002). As stated earlier, Ball (2009) and Wolfe and Neuwirth's (2001) papers were literature reviews. The discussion on the different kinds of educational annotation will identify its variations, such as reader annotation on the page of an essay or book made by a student in order to engage with the text, or an annotator's formative comments on a draft essay or summative marking of an essay for assessment purposes.

3.5.4 Four dimensions: Competence, tone, appearance and fitness

McColly's (1965) research aimed to test the hypothesis that more writing alone means better writing and found it to be untrue. Therefore, 32 English teachers' annotations were studied to identify what happens during the corrective stage of grade nine students in a US college to their essay composition. The students were asked to write four compositions, which were then annotated by the teacher. This generated 313 compositions which were assessed using a 32 bipolar adjective scale and an analysis of the inter-correlated variables made. The findings indicated that the writing process alone is worthless, but gains are made functionally in the composition of essays through four activities of re-writing, revision, correction and discussion of revisions made. Therefore, McColly (1965) concluded that annotation possess four general dimensions; first, a content style factor which he termed competence and adequacy which reflected scope, depth and purposefulness. Second, there was a tone factor which he termed helpfulness and positiveness, reflecting the attributes of feelings and attitude. Third, a physical factor called appearance reflected the attributes of format and visual impact. Lastly, the fitness factor reflected the appropriateness of annotation in relation to the concept of annotation and how it was expected to be seen.

3.5.5 Print based and digital annotation: Capture and share

Although not directly linked to students in education, Marshall's (1997) research helped to bridge the gap between McColly's (1965) research for a digital audience. Marshall asks the question why analyse textbooks when the research related to hypertext? Marshall's (1997) original research study was disseminated in other publications (Marshall, 1998a; 1998b) to focus on digital library users within computer science advocating the use of annotation to aid reading.

Marshall suggests annotations on books are hyper-textual because they exist in non-linear relationships to the linear text and interrupt linear reading. This makes it relevant to choosing student digital annotation research extracts as a research method (see chapter 5.5).

Marshall's (1997) study was based on annotation found in university textbooks which allowed a buy back scheme and students re-selling their used textbooks. This allowed Marshall access to a community of annotators and the mechanics of making annotation marks in the books. Marshall (1997) then observed students discussing the books, flicking through them and interviewing them about their thoughts which are discussed in Marshall (1998a; 1998b) as well. In total, Marshall (1997) examined 15 sets of books totalling 150 in all and in the later study (1998a) extended this to 410 books, with 39 titles from 21 areas. The findings are much the same in both Marshall papers and are divided into form and function. Both studies collected data by examining the used text books one by one, much as I did with chapter seven's research theme of "the hermeneutic self" when reading the 100 essays to identify visual impact themes. Form refers to any permanent or transient marks made and any published or private annotation. Function refers to annotation characteristics, being telegraphic, and highlighting text to aid reading and writing. In particular, the functions of annotation are useful as a record of reading, a visual trace of the reader's attention, making links, interpretation, memory and building a path to understanding.

The non-linear annotation involves making associations, relations, anchors, and types of marks on paper. Associations are made at a word to word and collective level and referring to many sub sections of the book through the use of arrows, circles or a mark consistent with spatial hypertext. Anchors (such as a bracket) are annotations that set off a series of text. The findings concluded annotation emphasised parts of a text, either within the text or in the margins to organise the host text for future reference. Segments of text were highlighted with different colours as a code only known to the annotator. Often key works of an author were chosen with annotations present to ensure that the private annotations of another become public and helpful in the interpretation process. Marshall (1998a) then ends the study with an application of the findings to expressiveness of digital text with the prospect of having unadulterated text with digital annotation that can be clicked and opened up to read.

Marshall and Brush's (2004) US research study developed the capture and sharing of personal to public annotation on published digital work, which is a common practice for the workplace and classroom. Marshall and Brush (2004) make the point the study develops on from their past research related to paper and digital annotation, because the former allows for the immediacy of comments, a malleable surface to make marks, and the latter with ongoing prototypes being developed to promote and support online reading. Marshall and Brush (2004) aimed to anticipate and support how personal annotations can contribute to collaborative activities for the digital reader and the changes that occur when shared. In order to support the sharing of personal annotation, the study involved reading online assigned material with others.

Eleven graduate students were recruited from a human computer interaction seminar and the research methods involved semi structured interviews, personal annotations made and contributions to online discussions and summaries. Over 1700 shared annotations were tracked, compared and analysed with three hypotheses confirmed. First, only a small percentage of

annotations made when reading directly related to the annotations shared in discussion. Second, some annotations, such as anchors in the text, brackets, a margin bar, highlighting or circles indicating start and end points, were better for sharing. Third, personal annotations changed dramatically in how they were anchored in the text when shared. Anchor changes of 80% were almost as common as content changes when shared.

Not unexpectedly, the authors found that the assigned activities, discussion and shared annotations affected the kind of annotations they used. For example, reading text more carefully, being more aware of their annotation and even changing their styles in anticipation of being better understood by others. Another student identified topics they knew would be provocative to others when in discussion, therefore the findings indicate an anticipatory awareness of its effect on others, which has relevance for nurse education too. However, the students' personal annotations still reflected an "...un-self-conscious engagement..." (p. 352) with text and not understanding the meaning behind some annotations they had written. Marshall and Brush (2004) concluded the assigned reading and specific aspects of the system changed annotation when the user was required to identify an anchor for every note they made and the signification of such an anchor may change in time. This is of relevance to a finding in Ball's (2009) research study when lecturers reading their past annotation given on an essay identified a degree of tone but had forgotten what they had thought at the time. When they had re-read the annotation they could not see why they had been irritated. The signification of the anchor may have changed in time.

Wolfe and Neuwirth's (2001) review is concerned with the promotion of software that aims to distribute, display and share annotation to support digital reading. At the time this was of course cutting edge. The review originated from Wolfe's 2001 PhD thesis (Wolfe, 2001) and further dissemination of findings in Wolfe (2002) which I will discuss after this section. Despite developing the future of digital annotation, the authors review is organised around print based annotation and covers most of the issues discussed so far, such as biblical and medieval annotation, sharing folklore, defacement of text, scholastic copying of text and restrictions in the form and function. The effect of such sharing was noted to have both an impact on the sharer and receiver, with the former changing their views when the annotations were discussed. However, the purpose of the review was not to develop print based annotation but to learn from it in relation to how reading, writing and how annotation could be developed for new and emerging digital technologies.

Wolfe and Neuwirth (2001) viewed print based annotation as having many of the qualities of digital annotation which is of concern to this thesis research method when choosing both written and digital annotation to analyse (see chapter 5.5). The authors' technical discussion about software will not be discussed here due to the thesis exclusion criteria (see figure 2). However, their study suggested the printing press had enabled the availability of books to a wider audience, but this was suggested to have brought a minor restriction. Unless a book has editorial footnotes, new books are newly experienced and without annotation, unless there are footnotes from the author. In contrast to the hand copied books medieval readers, such as Galileo's letters on sunspots in 1613 (see chapter 1.2, p. 24) which offers a contemporary analysis, modern readers have limited opportunities to observe how others engage with the text.

Wolfe and Neuwirth (2001) considered topics already discussed in this thesis so far, such as private versus public annotation, the form of annotations such as circling and underlining and its function as an *aid memoire*, facilitate re-reading and re-interpretation, acting as an anchor to chart new understanding, resonance or meaning. What Wolfe and Neuwirth (2001) highlighted, that annotation helps students' reading, comprehension and that annotation evolves over time to have new meaning. Anything highlighted helps with the storage of memories and their recall through different levels of processing. Annotation therefore helps the reader's understanding, and re-reading helps bridge reading, writing practises and memory recall. In short, in relation to different stages of reading and comprehension, the review discussed the difference between novice and expert readers and its effect on annotation which I discuss in Liu (2006) and Feito and O'Donnell's (2008) work (see figure 4, and appendix 1). Wolfe and Neuwirth (2001) discuss the impact of comprehension research and study skills, which generally refers to studying for exams through annotation. The authors suggest that annotation could be taught to professionals, who they suggest rarely annotate in order to improve communication, and in academic settings tend to read to then write. The authors suggest annotation might produce useful information for the writer of technical and business text about how audiences shape their writing in relation to readers' processes and values, in order to avoid irrelevant material. The context therefore shapes the type and quantity of annotation.

The authors go on to compare paper to digital annotation; paper annotation is likely to be more legible than digital, more portable, easily annotated and allows the reader to go back and forth along the text. However, the authors concede that new technologies allow the digital reader to do all of these things and share more. Wolfe and Neuwirth (2001) paper sketched a broad picture

about annotation, but recognised its own limitations and concluded that more research was required on the effects of annotation as an interface to making, giving and receiving annotation, on memory recall and comprehension of annotation, with novice-expert comparisons of particular use. Ultimately, Wolfe's work advances a comparison between print based annotation and digital which helped me to appreciate the similarities for data collection (see chapter 5.5) and analysis later on in chapter six (6.6 and 6.7).

3.5.6 Imitation

Re-written from her 2001 PhD methodology, Wolfe (2002) had a total of 122 students enrolled on a lower level composition course in university agreeing to participate in the study. Of the 122, 52 students were enrolled in the first year introductory course and the remaining students, 35 second years and 31 juniors or seniors, enrolled in other lower level composition courses. The prompts for reading and writing were a series of letters to the editor of the New York Times and 26 instructors were asked to annotate them for the students to read and discuss. Certain annotation characteristics were noted; no annotation, underlining only and positive or negative evaluative comments. There were two control groups, one reading annotated text, the other reading without. The results were categorised to include memory recall, local, global attitude, process, essay quality and imitation of strategies perceived to be in annotation.

What was clearly noted was that receiving negative feedback affected students in contrast to receiving positive feedback, leading to the conclusion, negative comments influenced attitudes more than positive annotations. An interesting finding was the content of annotation influenced how the student perceived the persuasiveness and claims made in text. Students paid more

attention and recalled more accurately annotation they perceived to be strategies they should be imitating. With two thirds of students making no annotations, receiving annotation appeared to encourage them to imitate the style and frequency of annotation. In order to test this hypothesis twenty students were asked to evaluate the personality, position and influence of the annotator who had written the annotation. The conclusion was that the annotation was more rhetorical and influential, if an image of the annotator was evoked, and helped to shape the students view about the host text. Therefore, students imitating annotation they perceived to demonstrate critical strategies for reading and writing, affected their interpretation of the host text. Wolfe's (2002) research findings influenced my awareness of the annotator's use of rhetoric, imitation of critical strategies and memory recall which I discuss in chapter's eight and ten's research themes. The finding by Wolfe, that students develop an impression of the annotator's personality and attitude fits with earlier eras of shared annotation before the printing press and mass printing. The continued exposure therefore to multiple readers, means annotation is suggested to improve critical reading, writing and recall of the student, which brings me onto Liu's (2006) study.

3.5.7 Forgetting thinking and feeling when annotating

Liu's (2006) qualitative research study examines the difference between individual learning strategies among US students with varied writing ability and annotation helping to improve students' critical thinking and writing. Liu's study considered annotation as a meta-cognitive skill and essential for critical writing skills and exegesis, which is the tradition of explaining and understanding (Liu, 2006). Therefore, Liu suggests annotation should be taught to students in order to improve their critical engagement with the text. Liu's research methods include questionnaires, samples of annotation and an in-depth literature review and discussion.

Analysing the annotations of students over two assignments over a year, Liu hypothesises that deep learning occurs when students exhibit extensive and insightful annotation comments on the content of the text to eventually produce a better essay. As a proactive concept, annotation is found to maximise critical thinking and learning to write rather than leaving students to learn through trial and error. This hypothesis concludes that students who produce a lack of annotation demonstrate surface learning and in contrast, students who are annotators that are more skilful produce more analytical writing samples. These findings support Porter-O'Donnell's (2004) own assertions in relation to Iser's (2006) high and low self-efficacy readers.

The teacher-student annotation involves sharing of information from one more knowledgeable person to another. This is a simple concept but in reality, annotation is difficult for students to understand fully due to interpreting the annotation comments themselves (Liu, 2006). Liu however, suggests any time lapse means only the annotator themselves could know what they were thinking about at the time, and they may later struggle to remember what they had thought and felt. This was a finding also in Ball et al (2009) from experienced nurse lecturers, which identifies parallels between students annotating to make notes and those marking essays. In addition, in Liu's study students understood the text fully or not at all and made changes to their essay content by knowing what to change and why. However, a criticism of Liu's study is that all forty of the students were all taught and enrolled on Liu's mandatory writing course, who was also the researcher. The students therefore, may have felt a degree of coercion to participate in the study by being asked about the strategies helping revision (Grbich, 2010). Lastly, the author lapses uncharacteristically, by means of emotive words such as "made tremendous progress"

indicating a degree of subjectivity. Subjective consciousness and being more aware of one's actions is a theme found in the next research study.

3.5.8 Self efficacy and ownership

The issue of self efficacy or "ownership" is considered in Feito and Donahue's (2008) US study on the impact of classroom discussion for freshman students (see appendix 1). Analysis of this study using the hermeneutic circle is presented in figure 4, the resulting discussion here and notes made in appendix 1. The research design aims to obtain rich descriptions of the reading process of undergraduate art students' literacy improvements after lecturer annotation. The authors' target multiple undergraduate seminars to capture students' experience on first reading of Shakespeare's *King Lear*. Before reading great books, students were asked to self-survey their reading processes before and afterwards to identify their pre-conceptions about a pre-packaged photocopy of text requiring annotation. This allowed the researchers to state that each student had read the same text. Feito and Donahue (2008) then collected data from reading surveys, students reading annotations and observation of seminar discussion.

Perhaps when stating "...annotations are never neutral or arbitrary but represent interpretive decisions...." (p. 298, see appendix 1), the authors refer to what remains hidden in even the most sensitive and innocent annotation comment, which is the focus for this thesis. This perceptual dynamic identifies the scope of annotation as an individual interpretation (McColly, 1965). Like Liu (2006), Feito and Donahue suggest teaching students to annotate reinforces a sense of ownership over the text and annotation's make it personal, which in turn allows them to develop knowledge and language vocabulary. Feito and Donahue apply Iser's (2006) hermeneutic theory

to categorise annotation findings into four major groups; trackings, identifying gaps, individual and literary repertoire: "Trackings" describe the reader's attempts to follow a text with the use of underlining, highlighting, questioning or paraphrasing (Iser, 2006). A statement made in a sentence can be analysed to ask what it said and question what is not said. Therefore, reading requires the ability to identify the gaps in the individual's own understanding and every response is an attempt to overcome the gap found in the text, which may not be obvious to the reader at that time (Feito & Donahue, 2008; Iser, 2006). This confused state is therefore a search for new understanding and a more sophisticated reader turns these "textual difficulties" into a valuable resource by overcoming misunderstanding (Iser, 2006).

The individual "literary repertoire" refers to the personal and experiential factors such as gender, background, life style, levels of education and intellectual abilities, culture, values and reflective consciousness (Iser, 2006). Hence, a less experienced student needs to read more and accept the notion of ownership of the text in order to engage and anticipate the meaning of annotation. The relevance of this point in nurse education is that annotators too are likely to be studying at the same time and the same issues of literary repertoire will apply. Yet, as more effective readers, they still have the capacity to misunderstand the text, whether reading an essay or literary sources, due to misrecognition and the clarity of discourse. Hence, when annotating for marking, the clash between low self and high self-efficacy readers overcoming textual difficulties, may not be fully understood.

The literature suggests effective readers are more adaptable, flexible in their application of learning strategies and cognitive engagement and move from surface to deep learning (Prat-Sala

& Redford, 2010). The self-efficacy theory Feito and Donahue (2008) discuss illustrates that self-belief performs a large part in a student's own expectations and ability to write, and if learning is found to be difficult then the effect may be to stop and choose something more rewarding to read. Both low and high self-efficacy can result in meta-cognitive deficit such as low efficacy readers being more likely to be surface learners, less likely to be aware of learning strategies, consider comprehension issues and are more likely to react affectively than effectively (Bandura, 1977; 1997). This affective issue is examined in the research theme of the "reflective consciousness and slippage" in chapter ten (see section 10.8.2) and the perception of failure phenomenon.

The issue of reading ability is evident in research on self-efficacy and undergraduate students where a deep or strategic reading in students with high self-efficacy contrast with superficial, surface learning strategies of students with low self-efficacy (Prat-Sala & Redford, 2010). These characteristics are relevant to "ownership" relating to a confident self-belief and high degree of intrinsic motivation linked to teaching and learning strategies reinforcing or reducing motivation to learn (Prat-Sala & Redford, 2010). That is where the need for assessment of students' self-efficacy can develop new approaches to challenge students with low self-efficacy. Hence, annotation perceived as unconstructive to a student with low self-esteem may lead them to avoid negatively construed feedback and appreciate its significance to reading and writing confidence.

3.6 Consistency building

Feito and Donahue's (2008) reference to scales of significance that text has on each reader suggests every new reading reinforces a refined version of what is known. This is a hermeneutic

approach to reading and their study was the only one found to make this distinction. The expectations the reader has about the text is an attempt to identify a unified "meaning" that may lead to ignoring parts of the text that has little perceived resonance with the reader (Iser, 2006). This may not be the case in the lead up to first readings; unless the student has prior experience of the text based on cultural influences, and this is not likely for young students entering the nursing profession, unlike older students and post-graduate nurses. Consistency building involves a search for clues to help the reader gain access to the text based on what is familiar to them as a point of entry that enables constructive meaning. Therefore, annotation indicates to the student the degree of effort required in order to create a consistent narrative. The direction of the text therefore depends on reader sophistication and making links with other literary works (Prat-Sala & Redford, 2010).

3.6.1 Criticism- not hermeneutic and knowing colleagues practice

Feito & Donahue (2008) however, fail to develop the hermeneutic aspect of their study. First, Iser's well known notion of ownership and hermeneutics: memory, understanding and the power of text to transform are not discussed. Second, the naming phenomenon and putting a name to a thought, feeling or experience places restrictions on language is of relevance to ownership and self-efficacy (Gadamer, 2004a). This relates to inexperience and a lack of familiarity shaping the freshman students understanding of experience. Gadamer (2004a) argues that interpretation derived from understanding results from the familiar to the foreign in what he called a fusion of horizons. This occurs through a basic level of understanding or pre-judgment and searching for meaning by accepting the inner world of subjectivity (Gadamer, 2004a). This hermeneutic open-endedness is limited by the reader's own self-censorship as to what they want to understand and

that is a key gap in Feito and Donahue's (2008) research study. Hence, the reader has to be open "before" they can have their views transformed because transformation requires curiosity and the ability to exercise critical capabilities that recognise literary conventions, codicils and critical technique (Gadamer, 2004a). This means that the reader should get out of it what they put into it in the first place (p. 69-70) and relates to the hermeneutic self, I discuss in chapter seven. The students in Feito and Donahue (2008), like Liu's (2006) study, may have felt a degree of coercion because it fell within a compulsory module led by the researcher (Grbich, 2010). Therefore, student participation may have depended on their motivation to read, learn and progress in the course. Unlike Iser's (2006) assertion that literature has the power to transform thinking, more pragmatic baby steps of attitude adjustment for freshman may have been beneficial.

3.6.2 Experienced lecturers and reflection

The above issues of youth and naivety contrast with Ball et al's (2009) research study by experienced nursing lecturers reflecting back on their own essay annotation. The purpose of students annotating to engage with the text is in contrast to the nurse educator's reason for annotating. Nurse educators may annotate as a student themselves, or as a reader of literary sources but for assessment purposes they may use Microsoft Word documents track changes formatively or digital annotation for summative marking. The issues I discussed in sections 3.5.8 and 3.6 about trackings, consistency building and reading ability (Feito & Donahue, 2008; Iser, 2006; Prat-Sala & Redford, 2010) appear to be relevant here despite the changed context of assessment. In parallel, despite the temporal dimensions, they both relate to the annotator's experience of reading.

Due to the overlapping themes, a critique of Ball's (2009) research findings and literature review (Ball, 2010) on annotation are synthesised here. Ball et al (2009) focus on nursing annotation make a strong case for an increasing use of annotation within higher education due to distance learning, and multi-site correspondence which has the effect of reducing face to face discussions with a student. This indicated a growing movement of digital annotation and a face to face meeting afterwards being increasingly unlikely. Whilst acknowledging this shift, change means more annotation through digital sources, yet the study design focussed only on handwritten annotation at the summative stage. The case for digital annotation software allowing for the same kinds of activities found in handwritten annotation such as highlighting, circling, scribbles, adding comments to the digital page, saving annotation and immediacy of digital annotation, which are all factors in capturing the annotator's immediate thoughts.

3.6.3 Presence and tone

Ball et al's (2009) study refers to summative annotation feedback using semi-structured interviews for 249 students and 74 lecturers, the latter changed to a focus group of ten lecturers due to poor responses. The reasons for choosing summative assessment were not made explicit and this is a gap because whether formative or summative, annotation is different because of the timing (temporal understanding) before submission versus after. I discuss temporal understanding in chapter 4.8.4, 5.6.1 and at length in mimesis 1-3 in chapter 8.5 to 8.8. This relates to purpose, such as timely advice on draft essay work versus a marked and completed essay, and the stage of assessment such as reading to learn then writing to learn and the relief at passing an assignment.

Ball et al's (2009) findings identify that underlining words do not develop a student's understanding of the feedback because of the lack of explanation, and using marks on the page can be more useful when ticks are used instead to indicate where an essay gains marks. Students report that annotation helps to inform the next essay, provides specific feedback, identifies strengths and weakness and helps learning. Lastly, identifying spelling, syntactical and referencing errors can be underlined or circled is of limited value. Discussion however, reinforces McColly's (1965) work that the annotator's presence and tone was found to significantly impact on student understanding of feedback. As with McColly (1965) and Wolfe (2002) the tone of annotation is felt affectively to undermine student confidence, de-motivate and evoke perceptions of power dynamics when focussed on the negative aspects of an essay.

Ball et al's (2009) findings suggest the first and second markers are off putting to lecturers and students alike conceding an adverse effect on feedback and signposting the student to what and why change needs to improve the essay. The above finding is not analysed or developed in the section on the ten lecturers in the focus group who identify that feedback needs to be worded sensitively, constructive, aware of the implications of tone, be transparent, motivational and promote confidence. This finding is however, undeveloped in the discussion. The study concludes that in order to be more confident and comments are received well, annotators should be mindful of ten key points: making sensitive and helpful comments, being respectful, providing balanced comments to identify strengths and weaknesses, phrasing comments in the form of questions, explanation and justifying marks awarded. There is a need to clearly indicate where and how to improve, write minimally in the margins in pencil because lengthy comments can be saved for the feedback sheet and lastly, avoid the use of a red pen. However, lecturers

suggest some annotation when viewed retrospectively were found to be difficult to read due to a different interpretation at the time. The reason for the unhelpful tone, apart from the distance of time is not made explicit or developed analytically in the study to identify why.

Ball's (2010) review of the literature between 1965 and 2009 uses the search terms annotation, assessment, education and feedback separately or combined through Ovid online, IngentaConnect, SwetsWise, AMED, Medline, Psychinfo and Cinahl databases. Blackwell-synergy.com and Cambridge journal online were found to be helpful as well as Sage online, METAPRESS and Science Direct. 1500 publications were found reducing to 209 then 53 and with the addition of the search term handwritten reduced to eight. With few papers in the past forty years on annotation and now considered at the cutting edge of innovation through digital and hypertext technology, it is clear that annotation remains a largely inductive and misunderstood practice.

Ball (2010) reinforces three positive themes of annotation feedback: annotation needs to be transparent, nurturing and be given in the context of both summative and formative work. Annotation feedback should be directly linked to essay composition and text should not be underlined or a telegraphic comment made because they fail to make an impact on the students understanding. Lastly, annotation feedback given as an evaluative method is more helpful than the same feedback given without annotation. This brings me conveniently to Ball's (2009) participatory action research study which parallel Porter-O'Donnell in the promotion of critical abilities when learning to annotate.

97

3.7 Tone and evoked emotion

Wolfe and Neuwirth (2001) did not identify the issue of tone, originally mentioned by McColly (1965), but it is found in all nursing research studies on annotation. Ball's (2009) action research study promotes greater reflective awareness for nursing lecturers about annotation, but with an emphasis on the annotation tone previously under developed as to cause and effect. From 40 essays, five nursing lecturers shared their realities of giving annotated feedback in a focus group. Findings identify that essay content appears to trigger a reaction in the lecturer which is then projected and annotated back to the student. Annotations are open to interpretation and lecturers' felt retrospective reflections on their own annotations were beneficial, which I first discussed in chapter 1.2.5. Due to the strong impact of the lecturers' disclosures, their direct quotes are used extensively in discussion here. Again tone is identified reflexively with lecturers stating when the annotations are read back the issue of tone is the "...*startling revelation*..." to the annotator who had written it in the first place with irritation noted (p. 120). Lecturers' realised when marking, they did not feel that their comments were unfriendly or unconstructive, but in isolation there is a difference. The collaborative effect of the focus group drew authentic conclusions such as:

"...we felt we had woken up to something because the students themselves also read their essays in 'isolation'...(p. 118)...(and the author concluded)...we reserved the harshest judgements for ourself and knew that our own annotations were very similar to the sampled scripts, possibly signifying anger, critical overtones or undertone, an absence of reinforcement or support, and little facility to explain the telegraphic and incomplete marginalia..." (p. 119)

The last quote relates to assessing essays outside the context of the marking process and a realisation that students' too may read essay feedback in this isolated manner. When the context of reading is removed temporally from the marking process, annotators acknowledge that students' essay work triggers emotion. The emotion is in the form of anger leading to less motivation to explain more and support the student's learning needs. The question following this finding should have been why, but this line of enquiry is undeveloped in the study. Ball et al (2009) suggest Derrida's (1982) observation that annotation is more "annunciation" or a proclamation of something, than announcement to be significant, as I also do in this thesis. However, the word count limit for research publication appears to minimise the opportunity for authors to develop such points more fully. Therefore, I develop this issue further in the research theme in chapter ten.

3.7.1 Tone neutrality

My own research (Regan, 2010) focused on formative annotation on students' draft essays completed as an action research study. When reading previous research and literature reviews, it became clear to me that the same parameters for the literature search dates should be adopted, but it did not occur to me at the time to widen the literature search and identify the important contribution to annotation of critical theorists, editors, translation and second language scholars (see figure 5). The study identifies the temporal nature of annotation feedback at the formative stage being different to summative feedback. Related to the QAA (2006) at the time, I identified a policy gap in QAA guidelines because both formative and summative annotations were considered one and the same. This means commenting on the essay as an end product is the same as formative, when in reality commenting on the writing process as it develops makes its

temporal actions quite different. The research methodology was as such: I first sent an email questionnaire by global email to 600 students with 13 responding. Second, using the same method 112 were emailed and 22 responded.

As a methodology the use of questionnaires is criticised in Ball and Regan (2013) for annotation studies but the rationale was that questionnaires are useful when there is little known about a subject matter, or so I thought at the time (McNiff, 2013). The findings identify the same positive characteristics postulated by the then QAA (2006) which still relate to the QAA (2012a; 2012b) and explains why lecturers recognise formative assessment being as important, and involves the same skills and practises as summative annotation. The findings were that lecturers' feel generally untrained and unprepared to give annotation feedback and were informed instead, by receiving feedback as a student, as a personal tutor, observing colleagues own work and adhering to higher education standards. I found the notion of tone from lecturer to student obvious when received on summative assignment work, but not so with formative annotation, because of its timely, collegial and professional engagement.

Paralleling the unhelpful feedback tone found in other studies, students' felt frustrated by passive lecturer tone, which led to de-motivation. Students stated they had to "read between the lines" of vague and overly sensitive language from lecturers who were adhering to the positive tone of QAA (2006) professionalism. Criticism of formative and summative feedback is that that they both fail to engage adequately with the student and the promotion of sensitive language in feedback fails to stir up the emotions of a student's meta-cognitive processes (Juwah et al., 2004). Therefore, students prefer peer feedback which is generally more transparent, clear,

honest and perceived as having equal and collegial power (Handley, Szwelnik, Ujma, Lawrence, Millar & Price, 2007). If lecturers fail to make a student think in a different way, and a less than candid feedback tone results in many students having to read between the lines of the annotative feedback, lecturers risk losing credibility and to a large extent, authenticity. This means that a lecturer's use of constructive annotative feedback at times appears to be inconsistent.

3.8 Discussion

The literature review found the issue of tone reflecting feelings and attitude found by McColly (1965) remains prevalent in nurse education and higher education. However, despite admissions of anger by lecturer's reflecting on their own feedback, the reasons for annotator's tone remain undeveloped. The literature does not analyse whether lecturers' projections of emotion and attitude considered uncomfortable are ethical. If the same findings were found in other clinical nursing practice, such as following the *The Mid Staffordshire Inquiry* (2013), there would be systems developed to promote the quality of practice. This point is particularly pertinent for a branch of nursing practice involving teaching and learning to students who may perceive lecturers as role models, and should personify professional occupational characteristics of care, compassion, competence and reflective practice (NMC, 2011). However, for annotation to carry with it at times a degree of emotion and a lack of explanation requires further analysis due to the lack of reasons why, which remain undeveloped in the research.

Annotation is made more complicated by feedback to a student from an experienced lecturer "who should know" annotation opens them up to certain vulnerabilities. The student should be able to "borrow" the perceptive skills of the lecturer as a "...surrogate perception..." and a

parasitical "...*mouth to feed*..." depending on the student's academic independence (Ball, 2010, p. 139). The nurturing component of annotation should therefore link to relational aspects of the student-lecturer dyad. Hence, the projection of lecturer tone, most of it good, but some plainly negative could be more consistent and positive if lecturers' have the time and motivation to reflect on their own academic development and empathically understand the students' stage of development and choice of language-in-use. If they do not, then what triggers any negative tone remains defiantly under developed.

3.8.1 Hidden from the annotator

The literature suggests the meaning of the annotation may be hidden from all but the annotator themselves (Marshall, 1998b). This phenomenon was perhaps evident in my own annotations I made when attempting to engage with the reading and hermeneutic process in order to write critically about each retrieved study (see section 3.4, p. 64; figure 4, p. 66 and appendix 1). In short, what each annotation means changes over time because perception changes too. Conversely, Ball's (2009) retrospective analysis by lecturers on their own annotation comments suggests the meaning of annotation when read sometime afterwards, is hidden from the annotator too. In addition, because annotation is supplementary to the main text, it affects how the student reader understands (Bandura, 1977; Feito and Donahue, 2008: Prat-Sala & Redford, 2010) understands it. A lack of analysis on how annotation evokes emotion suggests the meaning of annotation more to annotation evokes emotion suggests the meaning of annotation as a device for analysing the inner thoughts of interpretation than is currently known.

3.9 Conclusion

This chapter is informed by the work of Barney (1991) and Straub (2006). Their work acted like an oasis from which to identify the key issues about annotation from many different genres. In this chapter I report on the search strategy (figures 1 and 2), method (figure 3), the hermeneutic circle and reading (figure 4), and findings (figure 5 and 6) of a hermeneutic literature review. A hermeneutic approach suggests that researcher decisions and the reading experience should remain open to wherever the text leads them because it is impossible to predict what will be found for a doctoral thesis (Boell & Cecez-Kecmanovic, 2010). This process is called over saturation (van Manen, 1997). The reading process initiates the hermeneutic circle (see figure 3) which I present in a transparent way by selecting Feito and Donahue's (2008) research study as one representative example of my own annotation and engagement with the text over time, when new reading informed analysis (see section 3.4; figure 4 and more fully in appendix 1).

In order to understand the full scope and context of annotation the review includes scholastic and literary disciplines (see figure 5) from a range of editorial, translation, education and scholastic genres before finding 13 research studies for higher education (see figure 6). In scholastic and literary writings, annotation is more than a comment in the margins or a word underlined to include anything added to the page to influence and inform the reader. Annotation that includes a heading, title, or a footnote explaining the text to the reader is a powerful social tool engaging with the author and reader, sometimes by an annotator with an agenda. Emerging themes are presented and annotation being considered one step in a long line of interpretation depending on each person's belief systems. Each addition to the text is considered a corpus, an evolving body of the text and its additions, which can see-saw either way and lead the reader to be unsure what

the annotator really thought. The annotator's attitude can come through with the tone of language-in-use however, their comments may hide their true thoughts about the text, even if admired.

Annotation in translation literature is criticised as being sometimes downright wrong and culturally ignorant (Mandel, 1965). Therefore, there is a lack of evidence about annotation, which may lead to educationalists talking up the effectiveness of annotation, with characteristic annotation including the use of imperatives, problem solving, reflective questions, Socratic questioning and paraphrasing (Knoblauch & Lansford, 2006a). I report on the findings of 13 research studies relating to higher education presented in figure 6. A decision was made to use the hermeneutic circle of encirclement or saturation, and ensure that any new papers offer something new to knowledge. The various themes I identified are: students can improve their reading and attention span to become deeper learners when taught how to annotate. Annotations are not equal and have different purposes because they: interrupt, are playful and reflect the reader's temporal engagement with the text (see figure 4 and appendix 1). What this last point indicates is that unclear annotation can lead to mis-interpretation and encouraging a student to use their imagination can help their understanding of the text. I found that understanding annotation comments depends on a student's literary repertoire, self-belief, consistency building, their ability to identify gaps, their reading ability and dealing with difficulty and tracking thoughts to make the necessary essay changes.

In relation to identifying gaps, I report on numerous gaps in annotation practice for higher education, which I will briefly summarise. There is a lack of research into annotation, which is a gap in itself. There is more openness to annotation in the wider scholastic fields of literary circles (see figure 5), and what is significant in literary disciplines, unlike nursing practice and nurse education, is their knowledge of the processes involved. I conclude that the meaning of annotation can be hidden from all but the annotator and, at times, even the annotator themselves. This is of concern perhaps because of the position of nurse education within higher education and its wider responsibility for professional education and ensuring public safety. A lack of interpretive exploration in the research findings appears to restrict analysis of research findings and no hermeneutic research study found to develop the issue in more depth. This was under-developed in Feito and Donahue's (2008) otherwise comprehensive study (see appendix 1).

In the higher education research studies, no research came from any other health discipline other than nursing (see figure 6). Why this is the case is not clear, but the nursing authors interest in linguistic interpretation and understanding annotation in nurse education are apparent (Ball, 2009; Ball et al., 2009; Ball, 2010; Regan, 2010; Ball & Regan, 2013). That is why the interpretive aspects of annotation appear to offer rich commentary on hidden aspects of nurse education feedback. Findings from the nursing research studies include: annotation when removed temporally from the marking process triggered emotion, anger and irritation in the annotating lecturer reduced the motivation to explain more and support the students learning needs. Tone and interpretation are factors in nursing research studies with lecturers affected by the essay text, the presence of first markers and conforming for continuity, and identifying poor practice and attitudes (Ball, 2009). Tone neutrality requires reading between the lines of annotation, which may lead to possible misunderstanding, and benign feedback (Regan, 2010). The literature identifies that attitude and tone come through annotation, but why is not developed

adequately in the nursing research. These issues transcend normal feedback processes and are associated for some reason with professional nurse education. However, when the practice is located contextually in nurse education, the gaps are evident in the feedback comments back to students. The gaps in the research literature directs enquiry into an exploration of the meaning of annotation in three ways. First, what is the imagined or real intention of the annotator's comments, and what is communicated to the student? Second, what is the impact of annotation in the context of nurse education and what infuences how annotation is received? Third, what is the effect of temporal processes on annotation and the writing task?

Chapter Four

Theoretical and philosophical positioning

4.1 Introduction

In this chapter, I discuss the reasons for choosing Ricoeur's textual hermeneutics as a research methodology with the aim of exploring the meaning of annotation in nurse education. I define hermeneutics, phenomenology and key concepts of meaning, understanding, recognition and the hermeneutic circle, which are relevant to the researcher's grasp of the analytical process. I then discuss Ricoeur's ideas in relation to writing and discourse and the role of rhetoric in asserting the rigour of an argument. Lastly, I discuss the notion of Ricoeurean temporality and making sense of the text through the phenomenon of time before discussing Ricoeur's theory of transference as a hermeneutic of the self, which are relevant to the identified research themes and the research process itself.

4.1.1 Aim of the research study

The benefit of using Ricoeur's textual hermeneutics for nursing research is an enhanced awareness of the plurality of language (Clark, 1990). I first presented this approach in table 1 (chapter 1.6). A qualitative approach to the research method is appropriate because of the literature findings reported in chapter three. In particular, annotation in nurse education requiring a lecturer's interpretation of essay content back to the student in the margins and spaces of the text is fraught with difficulty. The success of annotation feedback depends on reaching an understanding for both parties, which in the absence of a face to face meeting will remain largely unknown. Understanding "meaning" is therefore, the realm of hermeneutics, and my preferred method over quantitative methods in this research task (Ricoeur, 1999). This thesis parallels the hermeneutic challenge to understand other people's writing and our own as annotators, whilst overcoming any pre-conceived ideas about the text (Derrida, 1988). These are the ideas that are necessary to have when grasping the meaning of discourse and engaging with it over time. The parallel in reading and interpreting the theory and the research data was acutely felt. My engagement with Feito and Donahue's (2008) research paper is evident in appendix 1 and relates to my interpretation of the literature in chapter three and research data in chapters six to ten (see table 1, chapter 1.6). Annells (1996; 1999) suggests a researcher needs to understand the philosophical tradition their research drew inspiration from. Ricoeur's textual hermeneutics provides me with an opportunity for a deeper understanding of annotation by examining the effects of others on annotation language. Therefore, I identify key themes of Ricoeur's philosophy of language in relation to annotation and my chosen research methodology for this thesis (see table1).

4.2 The theoretical approach

The influences on Ricoeur's philosophy of language link the conceptual structures inherent in the phenomenon of annotation. Therefore, I will restrict reference to Ricoeur's textual hermeneutics to only those concerning the application of his work into an exploration of annotation. Paul Ricoeur (1913 to 2005) was born in Valence, France, beginning his philosophical training at a time when European philosophy was dominated amongst others by existential phenomenology
(Ricoeur, 2003). His significant body of work and encyclopaedic knowledge of philosophy developed a general theory of interpretation, called textual hermeneutics (Kemp, 1996a; 1996b). Ricoeur took well known concepts in order to critically examine life and his own work mapped out other philosopher's ideas, before a de-constructed analysis of their work was presented (Kemp, 1996a). Then Ricoeur would identify the points of convergence and departure. Ricoeur's reflective style of philosophy therefore, focuses on textual interpretations as the primary aim of hermeneutics to interpret language, reflection and understanding rather than the universality of being human (Kemp, 1996b).

4.3 The phenomenological influences on Ricoeur

Ricoeur's work encapsulates a broad-based linguistic philosophy and research methodology in which to analyse text. Hermeneutic phenomenology is the method of Ricoeur's philosophy, which he calls a textual hermeneutics to distinguish his ideas from other philosophers such as Husserl, Heidegger and Gadamer. Both the words "hermeneutic" and "phenomenology" are borrowed terms, after Heidegger (2003) joined the two together to signify a "way" and a method of interpretation and studying ontology (Gadamer, 1994). Both terms need defining before their method can be combined through Ricoeur's textual hermeneutics.

4.4 Hermeneutics and phenomenology

The word hermeneutic is derived from the Greek verb "hermeneuein" with reference to Hermes the divine interpreter messenger listening to the content of the message (Palmer, 1969). Hermeneutics is a "way" that sought to "lay bare" language and expand on the possibilities of human thought (Palmer, 1969). This playful kind of thinking enables phenomenological analysis to be as rigorous and objective as the natural sciences because it focuses on the symbolism of language (Heidegger, 1982). This is evident in empirical studies at the hypothesis stage and working through the significance of data.

Phenomenology is the philosophical approach that underpins Ricoeur's textual hermeneutics (Gadamer, 2004a; Palmer, 1969). Kant (1993) suggested a general phenomenology to analyse the principles of validity and limitations of principles relating to empirical sensibility. This general phenomenology aimed to act as a preliminary introduction to further study knowledge gained by actual experience and observation versus those ideas only known through theoretical deduction (Redding, 2013). Hegel (1998) later developed this notion to analyse consciousness, unconsciousness, self-consciousness and perception, understanding and the struggles of recognition in *Phenomenology of the spirit* (1807/1998). For Hegel (1998; 2008) the phenomenology of consciousness was through a phenomenology of concepts and challenging what is known about a subject in order to fully understand it.

For Husserl (1859-1938), the theory of knowledge began with a theory of theories, in the form of science (Husserl, 2008; Ricoeur, 1996b). In *Logical investigations* (first published 1900) Husserl promoted phenomenology to study the phenomenon of consciousness, which he called transcendental (or descriptive) phenomenology (Husserl, 2008). In other words, knowledge about objects begins with human sensory experience and processing information by drawing inferences or judgements from experience. This sensory experience led to a systematic body of phenomenological theory reinforced by evidence shaped through observation and experience (Husserl, 2008). This method remains a popular choice of phenomenology for the human and social sciences with the development of interpretive phenomenological analysis in psychology

over the past four decades (Hein & Austin, 2001). However, the method has less focus on a philosophy of language, which Ricoeur's textual hermeneutics provides for my analysis.

4.5 Phenomenology versus empirical science

Phenomenology contrasts with empirical science because the latter's scientific methods attempt to research an object, whilst removing all possible variables that would somehow influence the observer (Husserl, 2008). Husserl (2008) suggests empirical sciences are flawed for this reason because objects cannot be studied generally in a sterile environment and such a method would dilute empirical claims to human knowledge. Interpretation inevitably changes what and how something is experienced and in contrast to science, phenomenology aims to "know" the object intimately and know it in a different and consciously meaningful way through experience (Husserl, 2008). This "knowing" develops through "working out" any pre-conceived ideas about a lived experience in order to objectify it (Gadamer, 2004a). The "working out" process relates directly with the hermeneutic circle and "knowing" about preconceived ideas that impact on the process of interpretation. This reflective process, as can be seen, relates to the hermeneutic circle I discuss in chapter 3.3.3 (figure 3), section 4.8, and its application in chapters 5.7 to 5.7.3, chapters 6.3 to 6.8, and the thematic analysis of chapters 7 to 10. "Working out" the hermeneutic circle relates to an awareness of ontology, a study of Being which I discuss briefly here in relation to language-in-use.

Working through this "knowing" however, requires the use of language to think about the object and then to write about it, which in a way tests the limits of languages ability to articulate all life experience (Gadamer, 2004a). This is known as the problem of language in early existential philosophy (Regan, 2015) and Ricoeur realised this in his ability to fully articulate a direct reflection of oneself by way of the signification of language (Ricoeur, 2003). In other words, we know more than we can express about life and this realisation led Heidegger to identify the Husserlian phenomenological method to be fundamentally flawed, because of the limitations of language when analysing life more fully (Heidegger, 2003; Regan, 2015). Heidegger's radical departure from Husserlian phenomenology was made explicit in his magnus opus *Being and time* (2003) where Heidegger asks the question what is the meaning of Being, meaning "to be." Although an understanding is always close, the concept remains vague. For example, Being is perceived as the "…*amniotic fluid our thought naturally moves in…*" (Eco, 2000, p. 20) and human life is known and experienced before we can place the restrictions of language upon it, to constrain it.

In attempting to define Being and existential life, Heidegger suggested we should stop and take notice of "...*this Being which we ourselves in each case are*..." (Ricoeur, 2006, p. 354). Hence, by combining hermeneutics as a linguistic tool to analyse the hidden meaning of language, Heidegger considered phenomenology nonetheless useful and developed a new phenomenology called hermeneutic (interpretive) phenomenology (Gadamer, 1994). The process suggests that people experience the same things differently, whether listening to music or reading text and then interpret the meaning of that experience through its meaning for them (Ricoeur, 2008). This awareness gives hermeneutic phenomenology an authenticity through a philosophy of language over the methodological constraints of empirical science and descriptive phenomenology (Ricoeur, 2008).

The above point of view parallels an understanding of epistemology and ontology, yet epistemology relates to my own sense of reality, my understanding of life and its events on others and me. An individual's ontology is the study of their Being, their potentiality for being-real and authentic but Heidegger aimed to answer the question about Being human in its entirety (Regan, 2015). Yet to the individual, as it must have started with Heidegger too, a sense of oneself occurs when identifying the question and language in the first place (Regan, 2007; 2008). Hence, my understanding of life means, when I ask the question of life for others; I inevitably try to make sense of life for myself and with others; the question therefore, starts and ends with the meaning of life for oneself (Ricoeur, 2003). How this is articulated to oneself is through the language-in-use.

4.6 Ricoeur's textual hermeneutics: A framework for analytical exploration

In this section I introduce key concepts of Ricouer's textual hermeneutics. Each part of the remaining chapter relates to Ricoeur's interpretive framework and develops in more detail for exploring the meaning of annotation in later chapters. For Ricoeur (1994) the recognition of language enables making sense of words and signs captured and conceptualised in discourse. Communicating in social situations such as nursing, teaching and learning are achieved through the symbolic use of language which can be seen to directly impact on how people view the world, ourselves and others in a mutual cycle of interpretation (Ricoeur, 1974; 1994).

When this lived experience is understood, the meaning of language inevitably changes over time and temporally because we experience and think more (Ricoeur, 1990; 1996a). As a result of temporal action, Ricoeur's textual hermeneutics identifies why there are difficulties recalling thoughts which occur when writing annotation (see chapter 3.8.1). This temporal action is pertinent to reading and writing and is discussed further in chapter five's research study design because a researcher needs to be aware of their initial assumptions and interpretations when reading. The signification of language in the present often reminds humans of something in the past and so the reader has to be aware how temporal action informs their present interpretation. These assumptions and initial interpretation occur in a circular and dynamic manner when taking note of entry and exit points in the hermeneutic circle (see section 4.8). In relation to a hermeneutic approach to reading, Ricoeur's textual hermeneutics, as a theory, parallels the reality of the reading experience, for the reader attempts to understand the meaning of text by making connections with what is familiar and unfamiliar, which I found to be relevant to the analysis of research data (see chapter 5.5.4 and 5.6.1). Lastly, I discuss Ricoeur's advancement of textual hermeneutics through the phenomenon of conscious or unconscious projection of emotion onto another person through something called the transference hypothesis (Regan, 2012b), which is relevant to nurse education and annotation I explore later in chapter ten's research theme, the reflective consciousness and slippage.

4.7 Meaning, understanding and recognition

Ricoeur's deviation from Husserl, Heidegger and Gadamer led him to a much more linguistic and symbolic approach to the study of phenomenology. The hermeneutic process of reading relates to all interpretations of text and Ricoeur (1999) suggests Husserl's acts of knowing is restricted by the signification of words because mental imagery is "more than" what is represented (Ricoeur, 1999). This Husserlian critique of the image is important to Ricoeur's (1998a) analysis which I explore in the research theme of rhetoric in chapter eight, because the image and what is imagined always includes assimilation informed by schema towards a conceptual order. This conceptual order is a reflexive relationship with what is known and unknown and how this occurs through language (Ricoeur, 2003). For Ricoeur (1998a) the central problem of hermeneutic phenomenology is interpretation, with two issues determining its actions. First the independence of the text becomes "other" than what the author intended due to the subjectivity of the reader. Second, the interpretation of the text contrasts with the concept of explanation (Ricoeur, 1998a). Hence, what the text means to the reader relates to self-interpretation, what Ricoeur calls the hermeneutic self which involves negotiating the meaning of text versus the spoken word and interpretation versus explanation (Ricoeur, 1998a). The theory therefore, very much resonated with my own research journey as I immersed myself within the research process to try and "see" the meta-cognitive dimensions of reading theory about interpretation (see chapter 5.5.4, 5.6.1, 6.3.1 and appendix 3 to 6 for examples of the theory applied to my research notes).

In order to understand these problems, Ricoeur describes a general theory of symbolic language and perception connecting indirect and hidden meaning of text (Ricoeur, 1998a; 1998b). Ricoeur identifies hermeneutic rules of interpretation applied to everything that is a text, or like a text, in the sharing of cultural information. This cultural information in itself, limits objective interpretation of reading unless learning is made socially explicit to the reader (Ricoeur, 2008). This interpretive limit is in the form of bias, which the hermeneutic circle aims to overcome and which in the end, only time and more reading can develop deeper insight into (see appendix 3 to 6). I sensed this conundrum as I read Ricoeur's philosophy. For example, appraising the literature in chapter three (chapter 3.3.3, figure 3, and appendix 1), introducing the theory (in section 4.8), applying theory (in chapters 5.7 to 5.7.3 and 6.3 to 6.8) and thematic analysis in chapters seven to ten, with accompanied research notes in appendix 3 to 6 demonstrating transparency. Ricoeur (2003) therefore, develops Dilthey's notions of understanding to "recognise" and "to grasp" the meaning of what another person says or writes, means the act of recognition involves the capacity to change from ignorance to being knowledgeable. Ricoeur (2003) suggests this change or learning occurs in three ways. First, the acquisition of knowledge is gained through the recognition of signs and symbols as building blocks to the connection of ideas or in other words the meaning an image, word and phrase has for the reader (Ricoeur, 2003). This was evident for myself as a researcher in figure 4 (and demonstrated in appendix 1). Second, knowledge is revealed in the text's identity, such as the unravelling of a story's plot, its structure, organisation, coherence, use of grammar and punctuation. The structure of text resonates with the act of reading and re-reading text and when identifying any mistakes made. Third, learning involves the use of memory, which I thematically develop in chapter ten (chapters sections 10.7 to 10.8), where the above two processes, the signs and coherence of the text triggers recollection when seeing an object (Ricoeur, 2003). I will briefly define "meaning" and "understanding" due to their importance to recognition and the meaning of annotation.

4.7.1 Meaning and understanding

According to Ricoeur (2008), the word "meaning" refers to words pointing beyond the self to something valuable, significant and relevant. "Meaning" is contextually bound by the wider experiences of society, politics, culture, conventions and laws with experience considered the unit of meaning. With every new experience the capacity to modify meaning grows (Ricoeur, 2008). Hence, the Dilthean phrase the "lived experience" is often used in relation to

phenomenology and the external world experienced as a continuum where the future becomes the present and the present becomes the past. Meaning is therefore, making sense leading to understanding (Ricoeur, 2008). This mimetic experience can be seen in appendix 1 and the use of numbers 1 to 3 indicating new meaning occuring over time (see mimesis 1-3 in chapter 8.5 to 8.8).

The word "understanding" refers to an ability to grasp what other people say, gesture or write (Ricoeur, 2008). The term is often left undefined because its meaning is thought to be clear enough (Rickman, 1979). Humans need to understand each other's needs and without this capacity for understanding, the ability to communicate, co-operate and learn would mean that life would be very difficult indeed (Dilthey, 1988). Understanding, is therefore grasping, appreciating and sharing thoughts and feelings through language interpreted in different ways depending on cultural norms (Ricoeur, 2007). The process of grasping the meaning of words, phrases or objects resonates with the reader's understanding of what it means to them (Ricoeur, 2007). In short, we cannot understand what words, signs or symbols mean to others and then predict consensual meaning for all, without first acknowledging their meaning to ourselves (Ricoeur, 2007). However, it is possible to understand what is in a speaker's mind, for example, if they are in pain, without fully understanding the words they are using because of the act of recognition which I will now discuss in more detail.

4.7.2 Recognition

Ricoeur (2007) suggests understanding refers to the recognition of an object or sign in the course of its communication. He could have been writing about annotation, because, the act of the

annotator expressing themselves to the student aims to promote new insights through linking textual content and conceptual recognition (DiYanni, 2002). This was evident in chapter six (section 6.8) data collection stages and table 3. Through the process of recognition, three interrelated issues enact to generate the understanding process (Ricoeur, 2007). First, when visualising an object we can grasp it in our mind then join the visualisation with other images, perceptions or assumptions in order to identify the difference it has with other objects, helped by memory (Ricoeur, 2007). Second, recognition accepts to be true what is visualised and is taken as found. Third, the meaning of words are bridged by transitional processes; from the visualised image, to the idea to recognising the object and thoughts or action (Ricoeur, 2007).

Therefore, the process of understanding is inevitably interlinked with other processes, for example: social learning, environment, perception and experience and familiarity and context (Ricoeur, 2007). This interlinked process combines to influence thinking processes and what the text says to the reader is influenced by their previous experience of language (Ricoeur, 2007). Ricoeur (2007) suggests the word recognition and its impact on meaning and understanding had been overlooked. When reading and turning over a page, one word or phrase could mean exactly or nearly the same thing as another word or phrase in the same language, called a synonym, indicating that judgements of meaning are often intuitive. For example, a bucket and pail. This indicates understanding and meaning are both structured and intuitive and ensures a degree of prediction. If the sign is the basic unit of language, the sentence is the basic unit of discourse and supports the theory of speech and discourse (Ricoeur, 2008). Language requires a noun for meaning and a verb in addition to its meaning which has an indication of time (Ricoeur, 2007).

Once they both become interlinked as fundamental building blocks of language, then thought can begin to be made sense of (Ricoeur, 2007).

The integrated unit of verb and noun therefore, carry with it the action of assertion and denial, which can then be challenged by another assertion as to whether it is true or false (Ricoeur, 2003). This relates to the Aristotelian theory of argumentation, rhetoric and "saying it well" with proof which I examine as a research theme in chapter eight. Even at this stage of "saying it well" with proof, the unit of meaning makes a claim to truth, yet the question is decided in each instance. There remains, however, the phenomenon of guessing meaning, and because of the many factors serving to influence what is meant by discourse, meaning is both different and individual.

Ricoeur identifies the reader's negotiation of many possible meanings for a word or phrase existing in the same time and place called polysemy, to result not in chaos (Ricoeur, 2007). Even when confronted with two or more words written in the same way and pronounced differently, called a homonymy, Ricoeur suggests the gaps or lacuna randomly determine the use of language through an organising principle that orders the recognition of words (Ricoeur, 2007). The underlying assumption or principle is that the passage from one meaning to another occurs through imperceptible skips (Ricoeur, 2003). These skips are bridged and the many possible meanings of words negotiated by the reader's intuition, hence, the possibility of misinterpretation. Therefore, it is what is left unsaid rather than said that gives clues to what is meant, which governs the different meanings of words (Ricoeur, 2003). This is an issue for annotation because of the perception of what is said or left unsaid. There is of course the space between what is said and left unsaid, and that is what is concealed by a lack of openness which can be better understood through the hermeneutic circle (Ricoeur, 2003). An example of this is found in one handwritten annotation research extract (see figure 7 below):

Figure 7: Handwritten annotation extract

Gallagher and Hodge (2012i) believe that all health care professionals should have an understanding of what is required from them lawfully, professionally and ethically. Therefore this assignment is going to examine professional and legal regulations in place and how they impact on the way midwives are able to practice. Ethical theories and principles will be explored and applied to securics observed in clinical practice, with Trust confidentiality maintained at all times (in accord on ' the Mussing' the Code - Full title (unce) The law in the United Kingdom (UK) is divided into three main areas, European, Statute and

The annotator had added text to what they considered to be incomplete or inaccurately written sentences. In doing so, the added text aims to re-shape, mould and model an improvement in the text. In reality, the effect may result in a degree of misunderstanding because it is the annotator's interpretation of what can improve the text, rather than the student's. Any misunderstanding of course could be overcome with a face to face meeting.

4.8 Ricoeur, interpretation and the hermeneutic circle

Ricoeur suggests "proper" interpretation relates to the reference, intention and power of the text to trigger reciprocity in the reader and the hermeneutic circle (Ricoeur, 2003). This relates to the study design section in chapter five when discussing the hermeneutic circle (see chapter 5.6.1 and appendix 1) and the research data grids of first to third order interpretations in chapter six.

The objective meaning of the text is not hidden but addressed to the reader and the hermeneutic circle requires the reader to be aware of their bias, assumptions and pre-conceived ideas they may have before reading which may otherwise restrict interpretation (Ricoeur, 2003). The hermeneutic circle therefore, links the subjectivity of the reader with the subjectivity of the author (Ricoeur, 2003). When the reader is aware of their pre-conceived ideas then they can begin to be open to the text in a playful way towards more objectivity and being open to new ideas (Ricoeur, 1998a).

The inherent difficulty in the hermeneutic circle relates to the reader's authentic engagement with the text, which only the reader themselves can truly know (Gadamer, 2004a). Rather than pre-conceived ideas triggering an expectation of the text's meaning, the hermeneutic circle can identify bias, only, if the reader is genuinely open to theirs. Therefore, inauthentic engagement provides the conditions for conscious or unconscious mis-interpretation and the illusion of intellectualised rationality (Ricoeur, 2003). In other words, the reader pretends to have been open to the text and may even think they are, when in reality their bias is masked by intellectual analysis (Ricoeur, 2003). This issue is addressed in the research process by a triangulation of methods reinforcing the transparency of interpretation, decision making and judgements made (see chapter 5.7 to 5.7.3). Language is therefore, not a closed system, because words as a referent "say something" about the world and the author who chose to use those words in the first place (Ricoeur, 2008).

Ricoeur (2008) stresses that any relationship with the text remains one sided when there is no possibility of directly questioning the writer, no relationship with the author or if the writer does

not reply to the reader. As Ricoeur declares: "...*the reader is absent from the act of writing, the writer absent from the act of reading*..." replaces the voice of one to the hearing of the other (2008, p. 103). However, an authentic interpretation is more probable when there is the least interference from others in working through pre-conceptions and bias. Chladenius in Gadamer (2004a) suggests what the text actually says or means is unimportant, because it is the reader's own interpretation of the text that brings to life the static words on the page. This is fundamental to the Ricoeurean hermeneutic circle and the qualitative research process because it is the reader's or the researcher's own understanding of the text that is important as an entry point to develop new understanding.

When reading Ricoeur, the reader is helped by their mental capacity to deal with more than one idea at a time, before the most likely idea is grasped and made clear. This repetitive cycle of new projections enhances interpretation of the text because, when reading, the anticipation of meaning runs along the text like a rhythm, open to the reader's pre-conceptions, prejudices and judgement requiring them to challenge any hasty assumptions made (Chladenius in Gadamer, 2004a). Therefore, the language-in-use within discourse acts as the foundation for negotiating agreement between the text and the reader (Ricoeur, 2008) which I discuss later as a medium for meaning and understanding text.

In general, Ricoeur (2008) discourages meeting or discussing the written work of an author because the conversation may lead to a "...*profound disruption*..." in the reader's relationship with the text (p. 103). The author, too, is encouraged to maintain their pre-conceived ideas about the subject discussed because human behaviour in the writing process is as social as any other

human encounter, and lessened by a lack of it (Ricoeur, 2008). Conversely, discussion on the meaning of the text may increase the author, or annotator's own awareness about what they thought they were writing, as is so often the case when temporal processes make the author forget (Ricoeur, 2008). This is of relevance to exploring the meaning of annotation for two reasons which are discussed in chapter eight's research theme of rhetoric. First, as chapter three reports, because the annotator may forget what they thought at the time of giving annotation feedback (Ball, 2009). Second, because a natural conclusion is that interpretation for teaching and learning purposes would not be complete unless communicating with the student who is the author of the essay, and asking them what they mean in their writing and vice versa.

The opportunity to ask what is meant by the choice of language does occur in the research process when asking an interview participant during the interview what they meant by their choice of language and also in academic supervision on written work. Interpretation of the text is more likely to remain in a prejudiced state unless there is motivation to be open to what is written and that requires reflective curiosity, awareness of reader bias (Ricoeur, 2008); and for annotation, checking the meaning of the text with the author of the essay. The following section develops the Ricoeurean notion that discourse on metaphor, rhetoric, space and time, words are never static, but rather reflect an act of persuasion that has the power to be understood by playful engagement and curiosity (Ricoeur, 2003).

4.8.1 Writing and discourse

Ricoeur develops understanding and meaning through "discourse." Discourse means to bring something "out into the open" in order for it to become observable (Ricoeur, 1976) and this is the

crux about annotation and the intention of this doctoral thesis, to bring "out into the open" a hermeneutic of annotation. Ricoeur (1976) suggests the study of discourse aims to overcome the difficulties of language through exegesis, explanation or interpretation of text in order to get nearer to the truth of discourse. Truth in this thesis refers to the authentic narrative of the interview participants' lived experience. Therefore, Ricoeur suggests writing is "...*discourse as intention to-say*..." meaning a direct visual and graphic illustration of the signs of speech (2008, p. 103). Discourse "opens up" and "gives voice to" experience and the phenomenon of reading, visualising words and their emotional impact leading to thinking, interpretation and understanding (Ricoeur, 1974). Discourse is therefore, a way of reversing the in-authenticity of reading and reducing the risk of a biased interpretation masked by analysis. The need to be persuaded by an author, to convince the reader that their argument falls under the theory of temporal action, I suggest relates to the art of persuading another through discourse in annotation. Ricoeurean theory of rhetoric and the art of persuading a reader are examined in the research theme of rhetoric in chapter eight, which I briefly discuss next.

4.8.2 Rhetoric

Aristotle first conceptualised the theory of rhetoric as a counterpart to logic, politics and discourse through understanding figures of speech. Rhetoric refers to the act of persuasion and being convinced when something has been demonstrated. Rhetoric can be clearly seen in Aristotle's *Poetics* (1997) and is divided into a taxonomy of "muthos" (emplotment, shortened to plot), "mimesis" (imitation or resemblance, timely understanding) and "metaphor" (the trope of substitution/ borrowing) with a hermeneutic purpose. The art, developed by Cicero and Quintilian had been popular in Ancient Greece. The mind, considered a product of a conscious,

reasoned judgement about experience was generally considered to be complete and a mirror of how things are in reality (Knoblauch & Brannon, 1984). In its wider sense, all modes of persuasion constitute the art of rhetoric and include style, argumentation, proof and composition. According to Ricoeur (2003), rhetoric had become a forgotten discipline in the mid nineteenth century when science challenged the notion of truth and logic through experimentation. Therefore, Ricoeur (2003, p. 8), concerned about the decline of rhetoric as a discipline, viewed its importance as a social phenomenon beyond the text and re-conceptualised rhetoric for a contemporary audience. What is notable about Ricoeur's *Time and narrative 1* (1990), is that he develops rhetoric to understand narrative as a temporal action through mimesis 1-3, which I explore in the research theme of rhetoric and mimesis in more detail through research extracts in chapter eight.

First, I examine a story's plot as a melodic partner to mimesis and metaphor and its relevance to essays. Plot in relation to Ricoeur's "new" rhetoric incorporates metaphorical references to ensure that action is represented "as it appears" in a literary work through the sequence of events (Ricoeur, 2003). The imitation of plot, according to Ricoeur, occurs in all aspects of literary work, and rhetoric gives the text its structural coherence and underlying purposive human action (Ricoeur, 2003). Plot is the synthesis of the predicate, or everything in a sentence excluding names, in that it grasps together the events of a story into one whole to schematise the signification of language. Ricoeur suggests plot is therefore, comparable to the assimilation of predicates that structures an argument and its use in persuasion (Ricoeur, 2003). These factors are what an annotator is assessing in an essay and the cadence and clarity of language within it demonstrates a student understands through the plot's combination of storyline incidents. In

order to demonstrate the theory's application to annotation research data in chapter eight research theme, rhetoric was found to be relevant when annotators focused on corrections, on composition and plot. This is demonstrated in the "what if" interview research extract below (see chapter 8.7 for analysis):

"I like it when the student brings in their essay draft. I will annotate and make key points and if I suggest moving a paragraph to another page I will put a star on it or put "move A to point B." I find annotation is the context and I like the person with me and will say things like "what if you write this in a different way" and using metaphor to get a point across I will say something like "such and such is like... what if..." because they perhaps knew what they were writing about and I try to stir the student's imagination to imitate the nursing context. I am sharing my knowledge as if they are learning and developing themselves from me – we share....for example if the student has written something derogatory about old age then I may ask what if they were your parent, loved one, or perhaps the loss of a long term partner when aged and feeling left alone to live out the rest of their days....the issue is about them thinking about themselves in the future, not now as young fit people whose relatives are also healthy, it is about instilling empathy for future use..." (Lecturer interview 8)

The annotator is talking about temporal action and the ability for students to think about themselves in the future, in order to understand the phenomenon and death of a spouse. "Stirring the student's imagination" is rhetorical and depends on communicating the need for shared understanding. In *The rule of metaphor* (2003), Ricoeur revises Aristotle's theory of rhetoric

which he defined as the "...*technique founded on knowledge of the factors that help to effect persuasion*..." (p. 10). Rhetoric attempts to negotiate an equilibrium between the powers of eloquence and saying it well and probability or logic and when they both meet they become philosophical (Ricoeur, 2003). Let me first define the three phenomenon of rhetoric in more detail. Rhetoric aims to demonstrate the link between argument, logic and probabilities (Ricoeur, 2003), so inherent in teaching and learning strategies (Knoblauch & Brannon, 2006a; 2006b). This was demonstrated in the "what if" research extract (p. 126) as an example of a persuasive and collaborative encounter between annotator and student.

The term imitation or mimesis is "...the imitation of human action..." (Ricoeur, 2003, p. 40) and ensures that action is represented as it "appears" in literature and in the actions and speech of the actors within. Ricoeur aimed to "show" how metaphor can be categorised under rhetoric to clear the way for a "new rhetoric" (2003, p. 50). "Metaphor" is giving something an unaccustomed name usually to change its visualisation by reference to another object (Ricoeur, 2003). However, for Ricoeur, in order to generate new life from a dead metaphor, or something that is meaningless to the reader, new metaphor and meaning occurs with the collision of two semantic worlds (Ricoeur, 2003). This means the signification and meaning of words can be individual and experiential to enliven language (Ricoeur, 2003). Therefore, the use of words other than their ordinary meaning is transformed into something metaphorical, which is relevant when shared professionally in annotation. The similarity with annotation is that interpretation within the sentence allows the reader or listener a new referent because the word has a lexical code that remains within the semantics of the sentence. The meaning, therefore, is shaped by the context.

4.8.3 Productive imagination and metaphor

Ricoeur's re-conceptualisation of Aristotelian rhetoric suggests a new rhetoric includes the signification of language gained through a productive imagination, narrative, plot, metaphor, imitation and temporal action. For Aristotle, the basis of semantics takes the word or the name as the basic unit of understanding and so language and thinking hinge on this preliminary definition (Clark, 1990, p. 122). Part of this classical rhetoric focussed on the metaphor of the individual word, typically the noun carrying with it an original meaning. Metaphor, for Ricoeur (2003, p. 52-53) then, deviates from a words literal sense to extend its meaning, and this deviation is by something called resemblance. The difference between the metaphor and a simile comparison is the difference between two predicates, "to be" and "to be like" which then refers to something else (p. 53). Hence, the single-trope word connecting metaphor, mimesis1-3 and rhetoric allows Ricoeur to develop his concept of reference and theory of rhetoric, of tropes. The productive imagination is discussed further in the thesis (see chapter 3.7, 4.8.2, 4.7.2, 8.4 and 8.7).

The productive imagination is different from memory and imitation, which Ricoeur suggests is merely reproductive. First, thinking first brings objects and experience to the "mind's eye" as a visualised object and second, the imagery is eventually "represented" creatively. Therefore, for learning and understanding, what is re-created in the event of a productive imagination is hermeneutic and experienced through the dynamism of language (Ricoeur, 2003). The productive imagination is examined further in chapter 8.4 in the activation of an enlivened metaphor. For metaphor, the fresh metaphors resulting from a productive imagination engages the reader to make a mental leap from the trope of language. A contemporary definition of "trope" is: "...rhetoric. A figure of speech...consists in the use of a word or phrase in a sense other than that which is proper to it. Hence (more generally): a figure of speech; (an instance of) figurative or metaphorical language...." (OED, n.d, 1a).

A figure of speech leading to a new experience of the referent word. In Greek philosophy trope is "...an argument in support of scepticism..." (OED, 5, n.d), meaning metaphor is a function of language and generally follows the principles of the organising text, and recognition as discussed in section 4.7. Finally, with regards to the organisation of the text, its composition, structure, flow, story and coherence, plot is the "...combination of the incidents of the story..." (Ricoeur, 2003, p. 40). Ricoeur (2003) suggests plot is universal to all literary composition due to its structure, character and temporality. This means tragedy, acting, poetry, dancing, music and art are all affected by the phenomenon of initation and plot and are equally applicable (Ricoeur, 2003). Tragedy is just one example of literary art with similar characteristics to an essay (Ricoeur, 2003).

The tragedic plot is relevant to the nursing context when one remembers the tragic effect of illness, or death on the person and their family. Hence, in the imitation of professional action and text, Ricoeur relates to the student as a developing reader of literature and essay writing and the act of imitation is being watchful and open to the writing style of all authors, including the lecturer. This process is temporal and related to the meaning language has for us at any one time, with Ricoeur (2003) suggesting that the meaning of language changes too with time, because thinking changes with experience.

4.8.4 Temporality, reciprocity and distanciation

Unlike Aristotle's rhetoric, Ricoeur developed an understanding of rhetoric to include discourse and time. In short, Ricoeur's suggests the symbolic representations of words contradict any known meaning because meaning changes over time (Ricoeur, 2003). Ricoeur's discourse addresses the dissociation of meaning, the intention of the speaker and the verbal meaning of the text that needs to be clarified when spoken about by distancing the written word to the physical presence of the reader (Ricoeur, 2003). This is the phenomenon that explains Ball's (2009) research study in chapter 3.8.1 when annotator's had forgotten what had been in their mind when providing annotation feedback to students and when reading their comments, they could identify tone. However, on re-reading the essay they could not remember what they had thought at the time and why tone was evident. The issue of time and a story's plot is also examined in the research theme of rhetoric (see chapter 8.6). For Ricoeur, temporality structures language and narrative in structured stories which attain their full meaning over time (Ricoeur, 2003). Examples can be seen in appendix 1, my research notes in chapter six and appendix 3 to 6. Temporal experience is grounded in the recognition of a word in language linked to distanciation which involves an individual's living sense of history (Ricoeur, 2003). Temporal experience is also relevant to meaning and the act of consciously reflecting on the lived experience to think what is signified by language itself (Ricoeur, 1990; 1985; 1988). An example is found in the next research extract from a lecturer reflecting on this issue when:

"Words on the page versus verbal are a direct confrontation with yourself as an individual. Words on the page can be seen as disembodied for example "what's the message here, did I know that, could I have done something about it?" Invariably the answer is yes, I could have done. You see it as the next stage of "I'm not quite there yet in terms of managing that..." (Lecture interview 7)

The issues identified are: words being disembodied and temporal to ensure a direct confrontation with the reader's own analysis when reflecting back on what was interpreted and why. Ricoeur (1990), therefore, suggests the notion of distanciation is relevant to understanding the meaning of text. Distanciation is the distancing of the individual from their life events. When distancing oneself from the conscious meaning of the word, or experience articulated in language, the creation of space allows for reflection and making sense of textual experience, whether remembered or re-read. This happens when an individual reflects and returns to think retrospectively about an experience, which can then be objectified more clearly due to the distance of time and absence of emotion. In other words, "...phenomenology begins when, not content to live" or "...relive...we interrupt lived experience in order to signify it..." (Ricoeur, 1998a, p. 116). The process of making sense of experience therefore, requires the time and space to reflect and assimilate what signs signify, denote and mean. This notion is further examined in the research theme of rhetoric and mimetic analysis in chapter 8.6 to 8.8. The need for timely interruption reinforces why lecturers need time to think and reflect on the effect essay content has on them and their effect on the student in their annotation feedback. This is demonstrated in an extract from the research data:

"Annotation is one of the most powerful teaching aides we've got, but it has to be used at the right time, before and during provision – it should be continuous...." (Lecturer interview 8, appendix 2) The quote is quite clear about acknowledging the power and need for continuity for the benefit of the student's learning. However, as discussed in chapter three's literature findings, annotators forget what they might have thought at the time when making their comments and so the notion of distanciation changes what has been experienced: because the act, the gesture, and what are operative changes into a thematic from which to make sense of its meaning. This process also relates to the temporal process and students understanding the meaning of annotation over time and when making changes to the essay based on the comments received. Distanciation, is also an event which mixes up the perception of subjective and objective lived time which affects the consciousness of the lived experience (Ricoeur, 1985). The awareness of time and how experience is understood is suggested to advance the lived experience through the notion of historicity, and changing a person's perception of the present from their historical past (Ricoeur, 1998b). This leads me to discuss in the last section, the power words have to confront self-consciousness, perception and memory recall (Ricoeur, 1994).

4.8.5 Ricoeur and psycho-analysis

Ricoeur develops the Freudian theory of the transference hypothesis which I suggest is useful for understanding nursing attitudes (Regan, 2012b). Nursing has been suggested to be an ideal springboard for analysing the effects of transference hypothesis due to feelings and empathic understanding often going unacknowledged (Evans, Pereira & Parker, 2009). Ricoeur's reading and interpretation of Freud's psycho-analysis published in *Freud and philosophy* (1970) and *On psychoanalysis* (2012) had the intention of testing textual hermeneutics on the unconscious and what falsifies phenomenology (Ricoeur, 1998a). The transference hypothesis is related to

translation and hermeneutics, but its difference is due to past experiences being replayed and triggered by events in the present (Ricoeur, 2006). Transference hypothesis is a revelation and way of accessing the unconscious thoughts and emotion seen to be enacted through annotation actions (Ricoeur, 1970; 2012). Discourse, according to Ricoeur (2006), is the intention to bring something out into the open and when it relates to memories that are preferably forgotten, this exposes the minds' capacity for selectivity. This is of direct relevance to annotation in nurse education because the recall of memories based on experience can influence both the annotator and addressee. In this sense, transference is useful for understanding the meaning of annotation comments and any unexpressed views about nursing. This was a finding in the research theme of reflective consciousness and slippage in chapter ten.

Ricoeur, however, develops Freud's theory of transference, which has three types of memory recall: "practical" memory, "ethico-moral" memory and the "wounded" or disturbed memory (Ricoeur, 2006). This is examined in much more detail in chapter ten because in the case of the "wounded" memory, what may be hidden in the meaning of language may not only be hidden to the reader- the student, but also to the annotator themselves. These conditions are suggested to incite the affect-laden past experiences based on the principle that what is not remembered is likely to be repeated and acted out again (Ricoeur, 2006). Freud's transference hypothesis is, therefore, a reciprocal tool for reflective practice and annotation because the projection of emotion is two ways: from the author to the annotator and vice versa with meaning likely to be repeated (Regan, 2012b). This relates to discussion in chapters 1.2.3 to 1.2.5 on reflection in the NHS and HEIs and conditions which promote the replication of emotion triggered by the essay content. Transference hypothesis allows for a phenomenological glimmer of what is hidden

behind discourse and secretive thinking (Regan, 2012b). What maybe unconsciously revealed however is the underlying attitudes and inner beliefs behind the interpretation (Ricoeur, 1990). Therefore, transference hypothesis is a form of reciprocity which requires a need to clarify, question, and highlight to the reader a different interpretation than the one perhaps meant by the author, but its meaning may be hidden by symbolic and metaphorical language (Ricoeur, 1970; 2006).

4.9 Conclusion

In this chapter I outline the philosophical methodology of this research study through Ricoeur's textual hermeneutics. His textual hermeneutics act as an interpretive framework to enable an exploration of annotation in such a way that it shapes my interpretation of the research data. The key terms have been defined and Ricoeur's major influences briefly discussed. The relevance of Ricoeur to the analysis of annotation, and is made explicit through the hermeneutic circle as a visible form of communication identified through key concepts of meaning, understanding, recognition, writing and discourse, the hermeneutic circle, rhetoric, temporality and the transference hypothesis. Lastly, transference hypothesis is introduced because Ricoeur (2012) suggests discourse can be evocative and signify something hidden to the reader and writer that may be repeated or recalled differently. What is recalled depends on the new context and this relates to the phenomenon of annotation and why it is experienced in many different ways by different people.

Chapter Five

Study design and methods

5.1 Introduction

In this chapter I outline the research design, the ethics process, consent, data collection, storage and analysis using van Manen's (1997) three stages of interpretation and the hermeneutic circle (see table 1, chapter 1.6). The literature review identified the fundamental interpretive difficulties of linguistic meaning and Ricoeur provides a way to translate annotation effectively within the higher education arena. I had considered Heidegger or Gadamer's phenomenology as the research methodology and a number of publications were inspired by their work (Regan, 2007; 2008; 2012a; 2012b; 2012c; Regan & Ball, 2013; Regan, 2015). Feito and Donahue's (2008) research paper (see appendix 1) also suggested through the work of Iser (2006) that hermeneutic phenomenology was a good method for studying annotation due to the scope a doctoral thesis could bring to a subject. However, Ricoeur's work was chosen instead because he sets out a systematic discourse analysis, which I considered to be ideal for a hermeneutic analysis of annotation. As chapter four examined, Ricoeur's textual hermeneutics is a phenomenological approach concerned with consciousness which allows me to bridge the gap between thinking, writing, action and annotation, which I apply to a variety of themes and research extracts in chapters seven to ten. The theory, as I discussed in chapter four, and after writing for publication on the subject, appeared to be a fitting methodology from which to explore the meaning of

annotation in nurse education. This will become apparent in the research process and thematic analysis because the strength of this approach is that the researcher is central to the research due to a "working through" of the research data (see chapter six and appendix 1) in three steps, from first order interpretation, naïve guesswork to a deeper and informed third order interpretation (Ricoeur, 2008).

5.2 Research design

For clarity, I briefly outline the research design and discuss it in more detail in sections 5.5 and 5.5.2. The research study is a split site research study between two universities, both located in the North West of England. The first university has a decade long history, policy and research promoting handwritten annotation practices. Therefore, as a registered doctoral student at the first university, essays were selected for convenience from the archives (see section 5.5). In contrast, the second university, where I currently work as a senior lecturer in pre and post registration nursing, did not have an annotation policy or any archived essays with handwritten annotation to choose from. However, in 2012 both universities commenced the use of digital annotation and utilised Turnitin© for feedback to students on their electronically submitted essays on e-learn.

I considered that textual analysis of annotation research extracts was not enough to explore the human aspect of annotation and so I included digital annotation and the archived handwritten annotation. I also included individual interviews with lecturers and students, and a focus group with students in order to explore the lived experience of giving and receiving annotation within nurse education. All of the interviews were taped and transcribed for analysis.

5.2.1 Ethics approval

Approval to commence the research study was gained after I submitted separate ethics forms to both universities but written as one research study. Protecting the research participant's identity, maintaining confidentiality and maintaining professional integrity was of paramount importance due to my position as a lecturer. The ethical considerations are discussed in the forthcoming sections.

5.3 Data storage and confidentiality

An ethical researcher has to be aware of strategies for ensuring the safety and confidentiality of the data (Grbich, 2010; Kimmel, 1988). Ethical and moral issues in research acknowledge that any information collected during the course of a research study should be kept strictly confidential and all identifying details removed (Cresswell, 2007). However, the caveat to that principle is that if any information is disclosed that is considered a fitness to practice issue, or for example, threatens the life of another then the researcher as a nurse, has a duty of care to safeguard life, to discontinue the interview and report any such disclosures (Hawkins & Shohet, 2012; NMC, 2015b). Despite the availability of free internet service providers, known as cloud computing, facilitating uploads and easy access through virtualisation, I decided to use more traditional methods of data storage of the taped interviews to protect against digital piracy (Kshetri, 2013). Mobile data was stored on encrypted pen drives and kept in a locked cupboard within a locked room within a secure building. In particular, the right to privacy, confidentiality and protecting the data and identity of the research participants is considered essential in the research contract (Cresswell, 2007). The research data were anonymised and any identifying information replaced with a code prior to analysis and kept for up to five years. Transcribed

interviews of Microsoft[©] Word document files were stored on a password protected pen drive. I was responsible solely for photocopying and anonymising the fifty essays archived with handwritten annotation from the first university and I am the only person to know which material relates to each interviewee. The following discussion outlines the research methods, collection, and analysis through the three stages, hermeneutic circle and evaluative rigour.

5.4 Data collection

Morse (1991; 2003) suggests there is a risk of "muddling" of terms used in hermeneutic phenomenology because of a failure to acknowledge its distinctiveness. This criticism was also acknowledged by de Witt and Ploeg (2006) and Paley (2005). The term sampling generically refers to a selection of events, experiences, objects, phenomena and participants who are representative, called theoretical sampling from grounded theory and aims for maximum variation (Corben, 1999). In contrast to other qualitative research methods, hermeneutic phenomenology seeks research participants who want to share their lived experience and offers methods that capture in-depth insight not patterns of experience (van Manen, 1997). Research data captured in a phenomenological way suggests therefore, instead of sampling, the term extracts is more appropriate for use in hermeneutic phenomenology research (van Manen, 1997). Therefore, for clarity and simplicity I have used the generic term extracts in this thesis to indicate examples of all data, whether interview quotes or annotation extracts, accepting that phenomenological research does not aim for generalisability or replicability, but in-depth insight. Figure 8 presents an overview of the chosen methods which I will discuss in the forthcoming sections.

Figure 8: Overview of the research design



5.5 Handwritten and digital annotation research extracts

The lack of prescribed methods for phenomenological research should not indicate a lack of rigour (Giles, 2005; van Manen, 1997). Instead it is a "turning to" a phenomenon, rather than an abiding concern about research techniques. van Manen (1997) reinforces that a scholastic immersion into a subject matter enables the researcher to become a sensitive conduit for working with the many layers of meaning found in phenomenological research. This means that the researcher eventually becomes attuned to the methods for hermeneutic phenomenology which involve identifying discourse through texts and interviews (Paterson & Higgs, 2005). The hermeneutic process means the researcher is "caught up" in a cycle of reading, writing, dialogue, re-writing and a committed engagement through the act of playfulness (Smythe et al., 2008). This means suspending disbelief, working through pre-conceived ideas and being open to new possibilities of the textual meaning (Gadamer, 2004a; Smythe et al., 2008). I decided to collect annotation extracts from handwritten and digital annotation on student essays (see figure 9 below) and interview lecturers and students (see figure 10, section 5.5.1). The annotation extracts

were collected in two stages. First, once ethical approval had been obtained; 50 handwritten annotated essays were selected from the first university due to the availability of archived essays. The essays came from a mixture of pre-registration and post-registration essays and I purposefully chose research extracts that could facilitate further analysis. Following ethics approval, 50 essays were also selected from the second university's e-learning website with digital annotation on each essay from two consecutive pre-registration student nurse cohorts of 360 in 2012 and 370 students in 2013. The digital annotation research extracts came from essays from both years. The essays were purposefully selected and I chose digital annotation research extracts using the search criteria described in figure 9. I then used Microsoft© Word's snipping tool to copy and paste the annotation samples and then stored. The archived material (essays and digital annotation) from the second, university remained online on e-learn.

Annotated essays	Pre-registration nursing (levels 4, 5)	Post- registration/ post nursing (levels 6, 7)	Number
Students' marked hand written annotated essay feedback (see chapter 6)	yes	yes	n= 50 in total
Students' marked digital annotated essay feedback (see chapter 6)	yes	yes	n= 50 in total
Criteria for choosing annotation extracts	Imperative, problem solving questions, reflective and Socratic questioning, paraphrasing and corrections (Knoblauch & Lansford, 2006a). Visual impact, key words and tone (Barthes, 1964; 1973, Ricoeur, 2003).		

Figure 9: Overview of essay research extracts of annotation feedback from both universities for analysis

The criteria for choosing the annotation research extracts was informed by Knoblauch and Lansford's (2006a) review of the literature concluding that characteristic annotation includes the use of: imperative, problem solving questions, reflective questions, Socratic questioning, paraphrasing and corrections, which I used to guide a purposeful choice of research extracts from each essay (see figure 9). The selection criteria also included choosing annotation research extracts of convenience, were readily available and appeared to say something that could be further analysed using Ricoeur's textual hermeneutics. This started the intuitive decision making process of the hermeneutic circle.

As I examined each essay I chose extracts with key words which resonated with my understanding of the literature review findings. I noted the tone of the annotation and any clues to the thoughts of the annotator such as exclamation marks or well thought out comments. In particular, I chose annotation which had a visual and semiotic impact ranging from key words with little visual impact to a surprising impact on first viewing the essays and noting their visual effect (Barthes, 1964; 1973; Ricoeur, 2003). This visual criterion aimed to duplicate the student's first impressions of annotation when reading the feedback. van Manen (1997) suggests a raised awareness of a phenomenon affects all stages of the phenomenological research process and I was therefore aware that I needed to identify a balance of positive and negative annotation research extracts. This realisation moved full circle from expecting to find a large choice of research extracts with lecturer irritation, which I found were less, in comparison to the majority of constructive and balanced annotation comments. I realised the issue of tone reflecting emotion and attitude, which the literature suggested may be negative (Ball, 2009; Feito & Donahue, 2008; McColly, 1965), could be construed as positive, caring and professional. This was a surprise to

me and overcame certain assumptions that I had held for some time, in particular that tone was negative and neutral comments frustrated students (Regan, 2010). In order to choose from the many annotation research extracts, I used entry and exit points of saturation which meant collecting as much data as possible and stopping when nothing new emerged (Smythe et al., 2008). Some of the chosen essays had little, and some essays had a lot of annotation so it was difficult to state how many research extracts I collected from each essay. Sometimes one or two, sometimes all, depending on what they said and how they said it. Next, I discuss my methodology for inviting research participants to be interviewed.

5.5.1 Interviews with students and lecturers

I sent out invitation letters, information sheets and consent forms to prospective student participants from the two cohorts mentioned in section 5.5 by email. The working title at the time was "Analysis of annotation feedback from learners' assignments: How does annotation feedback facilitate student's learning development in health and social care?" This title however, changed in due course as the thesis progressed and evolved. I requested as an attachment to the email, a completed invitation letter sheet indicating an interest to be interviewed individually or in a focus group. The same method was used to request lecturer participation. However, no students replied to my request to be interviewed in 2012 which caused me concern, with all the respondents coming a year later from the 2013 cohort. I made it clear to students I had supervised that they could not be included in the study to lessen the risk of coercion, which had been a requirement of the second university's ethics committee. Once the student and lecturer invitation forms had been returned by email I then arranged to meet with prospective participants. When meeting the respondents, I introduced myself, the purpose of the

research study and then asked them to read and then sign a consent form. The students and lecturers, when interviewed individually and the focus group of students, had the opportunity to ask any questions, and I checked with them whether they were happy to progress. I reinforced that the participants could withdraw from the interview at any stage of the interview without prejudice (Grbich, 2010; Kimmell, 1988). Figure 10 below presents a summary of the interview schedule with students and lecturers. All interviewees were advised to bring examples of annotation on essays if that would help them to discuss their experiences, however, none did.

Figure 10: Summary of interviews with students and lecturers at the second university

Interviews at the 2nd university	0	Post- registration nursing (level 6-7)	Post graduate level 8	Lecturers
On experiences of receiving digital annotation	n=25 (total) n=20 focus group interview n=5 individual interviews	n/a	n/a	n=8 individual interviews
Interview questions (see section 5.5.5) General prompt questions such as "can you tell me more about that" (Creswell, 2007).				

5.5.2 Interview participants: Student focus group and individual interviews

For the study at the second university I recruited a focus group of 20 pre-registration students for interview and five individual interviews with degree students (see figure 11 for demographic details). In the focus group none were graduates, all were studying for a diploma in nursing and all were female. I organised a room for the 20 research participants, and when we met I discussed the study, obtained informed consent and then proceeded with the focus group. The interview was taped and later transcribed. In the individual student interviews, all were female, three out of the five had a degree in another subject already and all were studying for a degree in

nursing. None of the five individual student interviews had been in the focus group interview. For both the focus group and individual student interviews the age ranges were from 19 to 55 (see figure 9 for an overview of interviews). The demographic details of the research participants are given below in figure 11.

T			
Focus group	Aged 19 to 55. All female. Currently in their 3 rd year of studying for		
	a diploma in adult nursing.		
Individual interviews			
Student 1	Female, white, aged 19. Studying for a degree in adult nursing. First		
	year with two previous experiences of receiving annotation within		
	higher education. Previous experience of annotation in college.		
Student 2	Female, white. Aged 36, Studying for a degree in mental health		
	nursing. First year with two previous experiences of receiving		
	annotation. Has a degree and has experience of giving and receiving		
	annotation feedback.		
Student 3	Previously studied psychology. Female, aged 25, white, aged 19.		
	First year studying for a degree in adult nursing. Experience of		
	receiving annotation.		
Student 4	Female, white, aged 20. Studying for a degree in adult nursing. First		
	year with two previous experiences of receiving annotation within		
	higher education. Previous experience of annotation in college.		
Student 5	Previously worked in the manufacturing industry. Female, aged 35,		
	white. First year studying for a degree in adult nursing. Experience of		
	receiving annotation.		

Figure 11: Demographic details of interviewed students

5.5.3 Interview participants: Lecturers

My concerns during the interviews for both student and lecturer interviews were recorded in figure 13. Eight lecturers agreed to be interviewed (see figure 12 for demographic details) from the second university's school of health. All lecturers practised within pre and post registration nursing programmes and the range of disciplines included professions recorded on the NMC register (adult, mental health, child, health visiting, and learning disability) with some lecturers having multiple registrations. Due to the possibility of identifying participants by way of
ethnicity and uniqueness (for example, a male midwife, a male health visitor, a foreign national) ethnic details have been omitted.

Lecturer 1	Aged 35, male. Mental	Senior lecturer in mental health nursing. Five
	health nurse BSc (Hons),	years' experience in higher education HE.
	MA, PGC higher	
	education.	
Lecturer 2	Aged 38, male	Senior lecturer in adult nursing. Five years
	Adult nursing, public	experience in higher education.
	health nursing, MSc, PGC	
	higher education.	
Lecturer 3	Aged 40, female	Senior lecturer in adult nursing. Four years
	Adult nursing, health	experience in higher education.
	visiting, MSc, PGC higher	
	education	
Lecturer 4	Aged 44, female	Senior lecturer in adult nursing. Ten years
	Adult nursing, MSc, PhD,	experience in higher education.
	PGC higher education.	
Lecturer 5	Aged 50, white female	Senior lecturer in adult nursing. Eight years
	Adult nursing, MSc, PGC	experience in higher education.
	higher education.	
Lecturer 6	Aged 28, female, mental	Lecturer in mental health nursing. Two years
	health nurse, MSc, PGC	experience in higher education.
	higher education.	
Lecturer 7	Aged 53, male, adult	Senior lecturer in adult nursing. Twelve years
	nursing, MSc, PGC higher	experience in higher education.
	education.	
Lecturer 8	Aged 42, male, adult	Senior lecturer in adult nursing. Eight years
	nurse, MEd, PhD, PGC	experience in higher education.
	higher education.	

Figure 12: Demographic details of interviewed lecturers

5.5.4 Interview questions: Follow the lead

I will now discuss interview questions, the interviews and related issues (see chapter 4.6 to 4.8 for a background to the theory). The purpose of interviews in hermeneutic phenomenological research is to capture thick description of every day lived experiences (Smythe et al., 2008). The phenomenological conversation in an interview is suggested to have no clear plan, questions or

awareness of direction in order to encourage an openness that allows the researcher to engage instinctively (Smythe et al., 2008). However, being new to phenomenological research I had to ask some questions as indicated by the approving ethics committees. Smythe et al (2008) suggest the interview should be neither too tight nor too loose, but should always be about encouraging the participant to be relaxed and find their voice. A relaxed interviewee means more potential for openness (Smythe et al., 2008). Despite asking some questions as prompts, conducting an interview naturally affects the flow of conversation and phenomenological spirit (Smythe et al., 2008), each question had an underlying rationale which aimed to identify the thought processes about annotation experience. However, I realised they had more than a hint of presumption because the questions aimed to identify issues between lecturer and student communication, the choice of words, phrases used and how they were perceived, any affective (emotional) impacts, of development and previous experience of annotation. Interviewing stages for phenomenological research therefore needs to be not only conversational but relatively unstructured in order to allow relevant questions to be asked through probing and checking (van Manen, 1997), so the questions I had as prompts soon changed to be unstructured and literally going with the flow of the interviewee's thoughts.

As I prepared for the interview process, I realised that reading around the subject had led me to be more open to the hidden meaning of the language-in-use than I had anticipated because I was absorbed in the research process. The approach therefore aims to identify phenomenon of the lived experience through the collection of rich, descriptive data and so the emphasis is on being "open" to what may be overlooked or unnoticed. This is what Smythe et al (2008) refers to as the phenomenological conversation and being aware of the hermeneutic circle. I asked each question in a conversational manner to ensure each participant was relaxed, and when it seemed the right moment after conversational flow sometimes slowed, I used another question prompt (van Manen, 1997). What was important to convey was valuing each interviewee's unique contribution to the study (Smythe et al., 2008). This was not hard to do as I appreciated all who took the time to be interviewed, especially after being concerned the year before when no respondents came forward to be interviewed in 2012.

With regards to the student individual and focus group interview questions, I asked them to tell me about the comments they received by the marker on their essay (see chapter 6.4, student 2 for an edited transcript and research notes). Their replies tended to start with the annotation comments that were most memorable and sometimes participants struggled to remember what they were. This recall issue was made more difficult by the fact that no interviewee brought with them any examples of essays or annotations and instead relied on their memories. Students were asked what the impact of annotation was on their learning, understanding, motivation, insight, interpretation of annotation. I also asked what did they think and feel after reading the annotated comments. I was concerned there may be some confusion about what annotation was in relation to other forms of feedback such as email feedback on an essay or face to face meetings with the supervisor.

For individual lecturer interviews, a selection of questions aimed to identify their experiences and thoughts about annotation; coherency, rationale, logic, and objectivity by asking them to tell me about the comments they made and a rationale for giving them. A partial transcript of an interview with lecturer 8 can be found in appendix 2 demonstrating the effect of my questions on the interviewee's answers. Like the students, no lecturer came with essays and annotation examples. I aimed to identify a lecturer's working philosophy about giving annotated feedback on the page of a student's essay. Lastly, if there were any issues of interpretation, misinterpretation, assessment, rationale for comments made, relevance to knowledge and writing, I asked each lecturer to tell me about their thoughts and feelings they had when they had read the essay.

Initially, with the standard questions in front of me I had naively thought they would be enough, but during the interviews it quickly became obvious, from a phenomenological perspective, that they stifled the conversational dynamic (Smythe et al., 2008). This was a moment of realising that the phenomenological research theory actually did apply to the research experience, and it became more than theoretical to be real. Therefore, as the interview was unfolding my interview questions soon became unscripted and I began to appreciate what was meant by the phrase the "phenomenological conversation." This means keeping the phenomenon being talked about in central focus, and with an awareness of theory, keeping questions flowing. The more interviews I did the more confident I became and I felt abler to make mental notes, later documented in my reflective diary, about emerging themes (see chapter six). This was in contrast to the first one or two interviews where I felt overly self-conscious and "lost" with the interview "washing" over me. Figure 13 identifies some initial concerns I had before, during and after interviewing.

Anxiety	"That I might dry up." "My anxiety may be obvious." "I need to be mindful of the interviewee's anxiety and be supportive."	"At times I felt lost and didn't know what to say when the flow of discussion was poor."	"I feared the tape wouldn't capture all of the interviews" (2 interviews were irretrievable).
Emotion	"Excited when a comment appeared to 'fit' with the assumptions I had about annotation."	"There were flat responses at times during the interviews." "I had to deal with heightened states of emotion on one occasion."	"Excited that I was finally at the interviewing stage and a sense of gratitude that colleagues and students took the time to be interviewed."
Interpretation	"I feared I may not have understood or interpreted the answer incorrectly. I did leap ahead at times and my questions may have then led discussion."	"I asked questions at times which were biased and leading." "What does Ricoeur say about this or that issue?"	"Did the interviewee understand my question and the underlying reasons for the question?" "Do I understand the underlying assumptions of my responses to the interviewee's replies?" "What hermeneutic theory can be used to develop the data for analysis?"
Genuineness	"I felt the professional use of language restrict discourse from being authentic- both mine and the interviewees at times."	"Intellectualising on the possible meaning of the words at the time of the interview may have obscured the phenomenon that was attempting to be revealed."	"It was clear the interviewees cared deeply about the teaching and learning experiences." "Lecturers clearly understood their role in the assessment of theoretical knowledge and students were keen to discuss teaching and learning experiences."

Figuro	12. M		during (tha int	orviowa	with l	acturara	and students
riguic	13. IVI J	133003	uuring i	ine mi			cciui ci s	and students

The interviews with lecturers were quite different for a number of reasons: our collegiality, insights, motivation, knowledge and working educational philosophy, annotation practice, disciplinary difference (adult, child, health visiting, mental health), teaching experience and experience of annotation. However, the same interview processes were followed for lecturer interviews. I found it a little embarrassing at times to interview colleagues I knew very well because I had to remain professional, keep focussed and lessen the spontaneous use of humour which I use at times as a defence mechanism when things get too serious (Regan, 2007). I was also aware that they may have other pressing things to do and I was taking up their time. I was concerned in case I came across as being obsessed with annotation and them thinking "…surely there are more important things to research..." I worried they may think that I should know about the real time pressures of working life (see figure 13). These concerns, however, lessened dramatically as I became more confident in the interviewing process.

5.5.5 Post interview analysis

Methodological rigour required me to identify any bias and assumptions and work through them in a cyclical manner to generate analysis (Smythe et al., 2008). I write this in the present tense now, because the methodology is ongoing even in revision. Memory recall is helped by the use of note taking during interviews and afterwards. To help me in the analytical process I remind myself how I felt, and what I thought in the interviews at the time, which I have mentioned already in section 5.5.4 (see figure 13). In particular, I need to remember how I experience what was being said which leads me to follow the conversational lead being presented. What did I feel? What did I think? What impact did I have on the proceedings? Lastly, how can a heightened awareness of reciprocity help the phenomenological process, are all appropriate questions to have (Smythe et al., 2008).

The affective awareness of the interpretive process is therefore important to acknowledge because I need to be open and transparent to all that is said by an interviewee and its impact on my understanding (de Witt & Ploeg, 2006). A degree of self-interpretation occurs during the interpretation process because of the subjective connotation of language and experience (Ricoeur, 1998a). This is what Ricoeur calls the "hermeneutic self" (see chapter seven) and the meaning discourse has for one-self needs to be made explicit before a more objective reading can occur (Ricoeur, 1998a). The hermeneutic process is, therefore, a process of reciprocity and during the interviews I began to appreciate my own experience linked to interpretation of the data and being understood (Smythe et al., 2008). Smythe et al (2008) suggests that the phenomenological researcher becomes exposed in the interview and they need to take notice of what they were thinking about during the interview. This process is important because the thoughts and feelings recalled during an interview can illuminate any assumptions that had been made for interpretive purposes as figure 13 demonstrates. More about analysis in sections 5.7.1 and 5.7.3.

5.5.6 Transcriptions

Normal procedures for dealing with taped research material were followed. That meant as soon as the tapes were finished I uploaded each interview through a USB portal onto my computer. I removed all identifying features and used a simple code to identify each interviewee. I wrote in my reflective journal how I thought the interview had gone and issues that had been raised (van Manen, 1997). The interviews were transcribed by a colleague experienced in using dictation equipment which saved a lot of time and effort, with over fifteen hours of transcribed material (see chapter 6.4, student 2, p. 184 and appendix 2, for examples of lecturer interview 8 transcript). When I received the transcriptions back I started the analysis by making annotations on each document and making copious amounts of notes (see chapter 6 research notes, figures 14 to 18 data grids and appendix 3 to 6). When reading the transcribed notes, I checked for any information that could identify anyone for reasons of confidentiality (Kimmel, 1988).

5.6 Method of data analysis- to use software or not

Ricoeur's textual hermeneutics is the method for analysis and yet I had to make the decision to use software for data analysis or not as I believed it was expected for rigour in qualitative research. Data management computer software packages are good at processing large amounts of data and identifying key words, codification and thematic analysis (Cresswell, 2007; Smythe et al., 2008). Coding themes involves the grouping and labelling of data in order to make them to be manageable for presentation, and obtain clarity and breaking into the data especially if there is a large amount of data collected (Grbich, 2010). A criticism of software use for data management in hermeneutic analysis however suggests the software process can de-contextualise the meaning of data which would not have been useful to my naive yet developing understanding of the data and Ricoeur's theory (Grbich, 2010). Heidegger (1977) suggested the only difficulty with technology is how it is applied and in the end I decided intuitively not to use a software coding system because I wanted to be fully immersed in the analytical process.

A more reasoned exposition for my choice not to use software came very late in the research process and was reinforced after reading Giddings and Wood's (2001) suggestion that a coding software programme is not desirable when using a hermeneutic phenomenological approach. This is because the hermeneutic process is as much about the "meanings being revealed" when the data leaps out at the researcher through an immersion with the data. The choice of Ricoeur's textual hermeneutic phenomenology is the priority methodology, not convenience and thematic analysis and I reasoned that I had waited a long time to get to the data collection stage (September 2013) and felt it would be counter intuitive to distance myself from the interpretive process.

The chosen research methodology ensured my own interactions with discourse are considered central to phenomenological analysis (Smythe et al., 2008). This is because the phenomenological researcher through the process of immersion makes the research itself a lived experience because they are re-living it (van Manen, 1997). My experience of others experience is the key to phenomenological research, made even more important when analysis is combined with writing within the semiotic-hermeneutic circle. Hence, the phenomenological researcher "becomes" the research in many ways and embodies the circuitous notion of the hermeneutic circle. This approach was repeated in all stages of the data collection process and allowed for my own reactions to the interview experience (see figure 13), annotation extracts and transcribed text to be articulated before analysis could move through the levels of understanding.

After attending a second seminar with Professor Liz Smythe from Auckland University of Technology (AUT), a pioneer in hermeneutic phenomenology for nursing and midwifery, I became aware of the scholarly commons library at AUT which allowed me to select and read ten doctoral theses (with the majority supervised by Professor Smythe) using hermeneutic phenomenology. Not one had used a coding software programme (Bernay, 2012; Crowther, 2014; Giles, 2008; Jhagroo, 2011; Paddy, 2010; Reed, 2008; Rossouw, 2009; Sutton, 2008, Walker, 2008; Young, 2011). Instead the researchers made good use of the hermeneutic circle which I now discuss.

5.6.1 The hermeneutic circle

The theoretical aspects of the hermeneutic circle have been introduced in the reading process when appraising the retrieved literature and the theory is developed more in chapter four (see section 4.8). However, the hermeneutic circle also relates to all stages of the interpretive research process I discuss in this section (and appendix 1), which I outline here as a method for organising research data using van Manen's (1997) three stage model before analysis of the themes in chapters seven to ten using Ricoeur's textual hermeneutics (see table 1, p. 36). I discuss why van Manen's (1997) three stage model was chosen below. A number of the research notes have the addition of numbers one to three on them to indicate the temporal process of interpretation through the hermeneutic circle (see chapter 6.3, figure 14). The continuity of the hermeneutic circle to make explicit every stage of interpretation is important because in a phenomenological research study design, analytical processes need to be consistently transparent (de Witt & Ploeg, 2006) which I have attempted to do throughout the thesis.

Authentic engagement between the chosen philosophical theory and research methodology requires me to develop hermeneutic approaches to interviews, data capture and analysis (Smythe

et al., 2008). However, the hermeneutic phenomenological research process has generally undefined steps and the research journey is, at times uncertain, overwhelming and confusing for researchers new to the methodology (Smythe et al., 2008). Analysis, however, moves from description to interpretation and the uniqueness of hermeneutic phenomenology can be found in the detailed presentation and analysis of emerging deep and rich information (van Manen, 1997). The hermeneutic circle is based on the rationale that phenomenology of the lived experience is best served when the researcher's own experiences of the data is lived too (van Manen, 1997). This interpretive process involves the researcher and the different meanings of the lived experience through a process of "guessing" the meaning of language for both the research participant and the researcher (van Manen, 1997). Understanding, therefore, develops through trusting, thinking and "letting go" to allow new meaning to be revealed (Smythe et al., 2008).

Three issues are relevant here: probability, self-interpretation and falsification: First, the hermeneutic circle enables the use of personal resonance, identifying assumptions and bias, working through and using a "guess" and validation circle of interpretation (Smythe et al., 2008). In working through the meaning of data, its accuracy is made more certain when "...*an interpretation must not only be probable but more probable than another*..." (Ricoeur, 1998a, p. 213). Hence, any "guesses" are understood to be the starting point to working out the meaning of research data once transcribed. In this manner, indirect language requires negotiating the double meaning of language and appreciating a fuller understanding of views that inform communication.

Second, an initial significance a word has, involves a degree of self-interpretation based on its influence on myself as a researcher (Ricoeur, 1998a). Third, Ricoeur suggests the role of falsification is possible because of competing researcher interpretations and exposure to all previously discussed dominant discourse: for example, power, gender politics, family, and societal and other technological discourse informing a particular worldview (Ricoeur, 1998a). I realised the hermeneutic circle only made sense when I began to experience the research data at the collection point when the interviewee spoke about their experiences, and that is the nature of phenomenology (Smythe et al., 2008).

Discourse when articulated by the research participants' conversations allow for thoughts to be aired, which means if they are not articulated then the thoughts may remain unchallenged or unprocessed (Ricoeur, 1998a). Therefore, when the internal world of the person is externalised, new insights can be gained (Ricoeur, 1998a). The research participant is thinking-it-through and speaking-it-through and the researcher being central to this process, when first listening and later writing about the phenomenon, also parallels the same process in making the past tense the present. This is because they are thinking-it-through the hermeneutic process of writing in its constant revisions. In a way the participants lived experience finds its voice again in the researcher's own developing understanding of the shared phenomenological conversation (van Manen, 1997). This means the research participants lived experience becomes a relived experience which is relived in each revision. Therefore, discourse once articulated by the research participant, becomes "something else" and I initially made sense of it for myself using van Manen's (1997) three stages of interpretation.

5.7 Three analytical processes within the hermeneutic circle

The above reflections involve three interpretive processes evident in the hermeneutic circle (Ricoeur, 1998a). When working with the data I was guided by van Manen's (1997) three stages of analysis and the ten health care doctoral theses retrieved from AUT (previously cited in section 5.6). In the ten doctoral theses most had adapted van Manen's (1997) description of identifying themes divided into three cyclical categories: level one "explanation," level two "naive interpretations" and level three "indepth understanding." However, I found the terms first to third order interpretations were clearer to me and so I discuss the model in those terms. The third order interpretation is the generation of likely themes. I now discuss van Manen's (1997) stages in more detail.

5.7.1 First order interpretation

The first order interpretation which van Manen (1997) calls the "explanation" stage, very much relates to an initial experience of the data when in an interview or when first reading the research extracts (see chapter 6.2). I made initial notes in my reflective diary after the interviews and when transcribed. I made marginal notes, annotations, key words, and phrases on the transcribed page and did the same when reading the annotation research extracts. The explanation stage relates to the internal parts of the text. This is a level of holistic reading where words, sentences and phrases capture meaning of the text as a whole, yet the contents remain disjointed because they have only been experienced fleetingly in the interview or when reading the research extracts (van Manen, 1997). This stage was at the level of making notes about the main details and what the text actually said.

5.7.2 Second order interpretation

The second order interpretation, called "naive interpretation," offered me a more considered interpretation of the research data whilst still remaining at a temporally naive stage of interpretation (van Manen, 1997). Again, I am using the present tense because of its temporal action in revisions where the past becomes the present. The process is inherently temporal and the meaning of the research data gains in time (Smythe et al., 2008) and the process involves trusting the meaning the research data has for me and viewing the text as a whole. This is a stage of living-with the data, and being immersed and open to what the data may mean analytically. This stage, therefore, involves a high degree of commitment, motivation and a playful openness to the meaning of the data, without which analysis would remain at a pre-conceived and undeveloped stage (Smythe et al., 2008).

Key words and phrases were written down to help me to think and write and this process became more significant due to new meaning developing over time. This second order interpretation offered me the opportunity for more selective reading when key phrases revealed phenomenon I had not previously considered. This is what Smythe et al (2008) refers to as a phenomenological approach to identifying themes. A theme in qualitative research refers to the replication of words and phrases grouped together, however, thematising research data in hermeneutic phenomenology risks de-contextualising an experience from its unique temporal situation (Smythe et al., 2008; Thorne, 2000). From a phenomenological perspective, when identifying themes, the researcher needs to think they are in the process of conscious un-concealment, which is hermeneutic (Smythe et al., 2008). Understanding, therefore, is the process of recognising something which the researcher may not have otherwise found significant (Gadamer, 2004a; Smythe et al., 2008).

5.8 Third order interpretation

Finally, the third order of interpretation is called the "in-depth understanding." This stage involves a more detailed reading of each research extract, sentences and paragraphs, themes and allowing further revelations to emerge (van Manen, 1997). This is presented in more detail in chapter 6.8, table 3 and figure 18. Therefore, it is not good enough to simply state analysis is deeper at this level, because, reading and the significance of words change in time, with immersion, more life experience and further reading around a subject. The third order interpretations are developed later in each thematic chapter.

Thematic analysis occurs through "playing" with the many possible inferences of a word, sign or phrase has and acknowledging its relevance (Ricoeur, 2003). This reference to a hermeneutic of language is by way of an example, what Heidegger refers to in the last line of Stefan George's poem entitled *The word* and "...*where the word breaks off no-thing can be*..." (1982, p. 60). Heidegger took each word and analysed their inference and suggested a re-configuration of the line to be "...*no-thing is where the word breaks off...*" Therefore, the phrase "breaks off" is a diminution, meaning to take away, to lessen, and "no-thing" is where the word is lacking because it names a given thing (Heidegger, 1982).

When "playing" with words or a phrase, I took the above quote from Heidegger literally to refer to "something beyond" what the word infers because where the word breaks off it is important to follow the direction of what it alludes to, what it signifies to me uniquely and what is given to me in the phenomenological context. Hence, understanding the meaning of words can "go beyond" the normal consensus of opinion to be individual and reveal to me in the phenomenological process, its significance for analysis. This revelatory experience can be called a concealmentuntil-it-is-experienced-in research and where naive interpretation become more developed through immersion (van Manen, 1997). van Manen (1997) suggests phenomenological research can develop insightful analysis through writing, reading and more writing, and at the third order of interpretation, some of the lesser sub themes became less important because Ricoeur's textual hermeneutics opens up new directions in the textual themes.

The theory, therefore, gave me an entry point into the research data and to view the data through the phenomenological lens of Ricoeur's work. I can write this in the past tense and I can relate it to the present because the process is never closed to interpretation especially when reading and writing further revisions (Ricoeur, 1991). This realisation became more resonant as I interpreted the research data informed by Ricoeur's theory. This third order of interpretation (from a temporal perspective) is distinct from the previous stages of interpretation because understanding moves from naive knowledge of what is known in the present to deeper knowledge of the research data gaining resonance over time (Ricoeur, 1991).

The third order of interpreting the research data is in unison with understanding the meaning of Ricoeur's work. In other words, I realised when I wrote about applying his theory to the research data that it was initially tentative and it could not have been anything else but at the time. The tentative application of theory to research data is naive at first and follows the temporal process

of misunderstanding to understanding, which I write about in chapter eight in the research theme of rhetoric (see chapter 8.6 to 8.8). van Manen's (1997) model literally was iterative and dynamic, yet de Witt & Ploeg (2006) suggest qualitative researchers use of hermeneutic phenomenology need to ensure analysis meets three characteristics. First, a description of the philosophical ideas; whether the theory resonates with the researcher and its relevance to the research studies aims. Second, an in-depth integration of philosophical concepts with the research methodology and findings throughout the research study are crucial as can be seen in chapters' seven to ten. Third, a balance needs to be maintained between the study participants' voice and the philosophical concepts used. In addition, my own thoughts and decisions should be comprehensively transparent as an interpreter of the research data and as an interpreter of the hermeneutic phenomenological theory. This last point means that my interpretation of Ricoeur's thoughts should be representative of all of his work.

5.9 Conclusion

In this chapter I discuss the research design and methods which allow me to capture the research phenomenon of annotation. Hermeneutic phenomenology requires the researcher to adapt qualitative methods to collect phenomenon and analysis (van Manen, 1997). The research method for this thesis includes interviews with lecturers and students to enable data collection on the lived experience of giving and receiving of annotation. Essays were collected and annotation extracts from handwriting and digital sources help to identify the different kinds of annotation style, the use of language of essay and annotation text and its effect on the reader. I outline the research process, the split site study, the ethics application, consent, and storage and data collection. I also summarise the hermeneutic circle and three stages of interpretation using van

Manen's (1997) model. In particular, the chapter changes from past to present tense at times to ensure the temporal dynamism of the method, with interpretation changing with every revision (Ricoeur, 1991; 1998a). Finally, the importance of maintaining a degree of openness and transparency for me and the reader was also discussed and is a key to the process of interpretation and understanding because there is a need for researcher bias, pre-conceptions, and presuppositions to be negotiated openly (de Witt & Ploeg, 2006).

Chapter Six

Initial findings and four identified themes

6.1 Introduction

In order to organise initial findings, this chapter presents the research data using van Manen's (1997) three step model of first, second and third order interpretation (see chapters 5.7 to 5.8). As I discussed in chapter 5.7 van Manen's (1997) model was developed because hermeneutic phenomenology has generally undefined steps for anaysis which the research process requires. van Manen's (1997) model is popular amongst researcher's using hermeneutic phenomenology because it facilitates the detailed presentation of data, enables analysis of emerging deep and rich information through the work of the chosen hermeneutic philosopher (van Manen, 1997; Smythe et al., 2008). Therefore, findings will be further explored using Ricoeur's textual hermeneutics in chapters seven to ten. Table 2 presents an index of the chapter.

6.2 Emergent themes leading from first and second order interpretation

The chapter is organised by presenting research notes and research grids with coded key words followed by summaries of the findings in each section (see figures 14 to 18). Some of the research notes number one to three (see appendix 1 as an example of the same method I used to review the literature in chapter three) to indicate the temporal nature of the hermeneutic circle. The research notes were more significant to me than the research grids because they were a combination of the working out process and my temporal understanding (see appendix 3 to 6 for

research notes on the four themes). Lastly, at the end of this chapter, I collate the key words and sub themes in table 3 and identify four themes: the hermeneutic self, rhetoric, individualism and the reflective consciousness and slippage (see figure 19), which I explore further in chapters seven to ten. The first order interpretations, according to van Manen (1997), are my initial interpretations, which are considered superficial and immediate in relation to the meaning a word or phrase has for me. These are based on my pre-conceived ideas which I present in the research notes (see appendix 3 to 6), grids and two columns to identify the key issues. In the first order interpretations because they are the interviewee's own words, and the emphasis placed on the quotes by interviewees themselves make them significant enough to be a starting point of interpretations aimed at identifying key words and phrases (see table 3, section 6.8) to identify the third order interpretations and develop research themes.

Sections	and	Interviews	Research notes and grid		
figures					
Section	6.3,	Focus group	4 pages of annotated research notes, numbers 1-3 indicating		
figure 14			temporal process, grid, summary notes		
Section	6.4,	Student 1	2 pages annotated research notes, numbers 1-3 indicating		
figure 15			temporal process, grid, summary notes		
		Student 2	Full 10 page interview transcript with annotated research		
			notes, numbers 1-3 indicating temporal process, no grid due to		
			the amount of notes presented, grid, summary notes		
		Student 3	2 pages of research notes, grid, summary notes		
		Student 4	2 pages of research notes annotated research notes, numbers 1-		
			3 indicating temporal process, grid, summary notes		
		Student 5	1 page research notes, annotated research notes, numbers 1-3		
			indicating temporal process, grid, summary notes		

 Table 2: Index of figures and notes

Section figure 16	6.5,	Lecturer 1	2 pages of research notes, grid, summary notes	
		Lecturer 2	2 pages of research notes, grid, summary notes	
		Lecturers 3 & 4	1 ¹ / ₂ pages of research notes, grid, summary notes	
		Lecturer 5	3 pages of research notes, grid, summary notes	
		Lecturer 6	3 pages of research notes, grid, summary notes	
		Lecturer 7	2 pages of research notes, grid, summary notes	
		Lecturer 8	2 pages of research notes. Selection of the transcript in appendix 2	
Section figure 17	6.6,	Digital annotation extracts	Grid, 3 extracts, summary notes	
Section figure 18	6.7,	Handwritten annotation	Grid, 3 extracts, summary notes	
	2	extracts		
Section 6.8	8	Second order interpretation and sub-themes		
Figure 19		Third order interpretation: Four identified research themes		

6.3 Student focus group interviews (n=20)

Figure 14 is entitled *Extracts from researcher notes of student focus group interviews* of narrative quotes I collected during the focus group with pre-registration nurse students. As previously stated the research notes are presented followed by research grids and two columns of first and second order interpretations. At the end of each grid, a key of the abbreviations identifies the key words/ phrases found, followed by a summary of the notes and grids such as those circled numbers 1 to 3 and triangles/ pyramid to organise the research data.

Figure 14: Extracts from researcher notes of student focus group interviews (n=20)

Notes for there in psychotro Focus group interviews p.1) DE spant to keep poorsed on the frank as logs work. . Student () - Made Student want to cry -2 Londran 3) disheatered. confidence () The color red Scondarts evoke a reaction felf estean () for some Students - Referred back to hand and then feedback as a child and its inspaced on child chitd like 2 feelerps endred monory - child - diserponennait 0, when Sense of enpotture independing recented and promoted for teaching and leaving puppes (-De Perception of tone in two annotation (+/ transferei Modelino (MB iscuident yet when meeting the annotator a thitude there is little - so the witten would appear, to the. to ender trigger, Refect tone perception of The about indicates Missulequetection tore projected or of the withen word (eccary) can would in reverse of tap Missitephetethen of anuclation received/+ ktood a U' Save annotation lacted detail - lactor THOOD ~el &-3/14 relevence (coherence) "Good" says title . 'Yord' or one wood lacks flow Cherence Specificterionic of ideas & text / lactors narrative / detal lades engagement with the text. Good annotations happy (St/4) Bad " = unhappy (St/4) undoor 1000 etonic /studiet drence

Annotation made sense when face to face. before then clid not GROWN (2) individual AJCUSED cartning of annotation quality, cartinuity Expand of essay expediations, interpretation of reasons ()essay guiddines and discause all related Suason discarte to ensure insight, expertise, entrusion, - interprot knowledge, interprotochian & nientepretection Selt If lecturers do all of the narradine aspects interprotation of supervision preparection (module metetings, MZ1" moderation of essays, session prep, session () presentation, mass marting) there is an undertal EXPECTEDAT Oxpectation of Cartmunity. Hangues discourse Fairs day to the connotation VS mis (2 of words, phraces, meaning and dindestending to frally of everyone moduled in the epistemological (\mathcal{N}) leco - Unternetation <> mismitproster piocess (2) undestadupt misundestadup +0 knowledge From the author's an rescord, the studends better undertandy Epsteme reading of the another's wet the and interpretation of the observer i the markers 179) interpretation of cinnelection. The annotation tempcral trocked then starts another cycle of interprotection replacing the "betty de at the authors own TIME poseard Stage (hermoneutic circle).to rheton (minders Start again. Then when ne need, due to the temporal native of understanding, not ally the Student but later the another engage temperally to somewhat improve their undertrendig.

focus group P.3

So the above P.2 discussion brings the varying Factors influencing interpretation together, but they are any namifest in the nonent of new realization tothe individual Denburn. 2) the quality process (guidelines, marting criteria, learney - underlading Moderation) the individuals, involued and their interpretation Estudent, marker then annotating, student reading Superision, annetator re-reading, student re-reading, No-untie cutt ve-Subrussie). > poducture insognation ? (Ricourd (student 19 Said (unsaid) and vica versa - interesting suggestion or mudication the impact of the spoten / unspoten would and the nexter mayery that every pror (3 hourd 2) - some understanding both articled at. I'm cheer primet develoption harding the process many inting thence the unsaid inducates the process many inting magery because what is not said, left insaid the near 38 3) then allows the reader to leap a head corde to group some mading of the gaps they 3 Fillion. That is their issue not the vagancies GEASPING of the annotators unsaid projections Individual mounty It I also likely that what I said also ahead to has to "negotrate" the inflections, the shadows/ maning of what it refers to, even fleetingly, to then Fillinpin had tothe said (unsaid or said fue gas voual integral and regolierty with unsaid SAID/ Visual intageny) (Imperceptote UNSAD Casaditi . · gaps being ind magnation meaning chescome. visual unconsailer reflection of wood many C Mathere Forsions Prod. inconcine

focus gove Pitt

Essay · Essauy parallels practice - allaws for accessment Fractic (.) Of the enderth's traileage, attride, bis, competence, andence based practice. The annotation received 3 also parallels attitudes for musing eg modellig nusary modelly feedback, modellip priciples Annotative of practice and writing conventions -BM essen The ecsay is the safe space to experimentontent (2) upon (a theme seen elsewhere). Hence annotation applicably is affects the essary & It cantent which shaper 40 ideas, knowledge, critical thinks for direct pract-q because fip aphatha into the clinical setting (3) assmof F.T.P Aressto Pract producinge & a Hitrody ave also applicable. studies modellig behavit if essay is safe space to develop does r knowledge for practico. MB Mmes 75 1-3 Suasany discuse FB (annotation) Time= photonic Shapes & applies to persuade Caninta clinical practice & Graspirf Student believery visual imagery (productie urdestandy Tone -mgreetc reapup middlip about Initatia (ansarans U= uncarsciens of pufession (said / unsaid) belowers/ Initedia O.F ? Shippage WATTPETYLe

Focus group: 1 st order interpretation- initial stage	2 nd order interpretation- naïve stage
"Reading written feedback on my essay made me want to cry. I'd put so much work into the essay it's as if I'm on trialit cuts deep" (Student 1)	Tone, emotion and projection
"Lowered my self-esteem- felt incapable, felt down, disappointed- anxious, unconstructive feedback-unsure about the expectations, previously mostly positive" (Student 3)	Lowered self- esteem, feedback was unconstructive Disempowerment Need guidance, challenged thinking
"I needed help to steer me in the right direction lecturers' are very accessible and caringand I can seeinfeedback One profound comment asked how this make you feel really did? Can you dig deep and talk about the real emotion you felt? Why is it important to reflect as nurses?" (Student 11)	Attitude of the annotator Touched a nerve
" Some lecturers' essay comments ooze attitude yet when you meet them it's as if they are a different personsometimes the essay comments are nice and constructive and when I meet them it's as if I've got them on a bad day" (Student 11)	Feel like a child again. Painful comments
"I received feedback that touched a nerve. The feedback was sensitive saying something like 'I can see that you have been touched by nursing this patient, well done because patients' and their relatives need to see that you really care'I can actually imagine their face as I'm reading it, approving and praising my passion" (Student 17)	Negative comments, critical and condescending
"Some of the feedback comments can be painful I have to brace myself first before I read them as I know they may hurtit makes me feel like a kid againMy essay comments were very patronising in parts, there was a flavour to the comments, they made me wince and tears welled up in my eyes I'm usually able to deal with feedback but this felt personal" (Student 5)	Felt patronised, tearful, wincing, it felt personal
"Written in RED! Negative comments- critical-condescending, exclamations throughout! How did I feel? I lacked confidence, criticised, lowered my confidence and self-worth. I felt like a kid again, chastised by the teacher. Even the positive comments in red looked negative!" (Student 16)	Poor clarity of discourse in some annotation, needed specific detail, unhelpful comments were not convincing or coherent
"My essay writing got worse because of the hoops I jump through worse because now I have to write in a formulaic way, a way that is less genuineI feel pressured into writing and thinking a certain way because if I write and think differently my feedback and marks will reflect their expectations not my own" (Student 12)	Writing and thinking became inauthentic, formulaic
"I felt very frustrated with the lack of relevant comments - there were a lot of track comments like "good" which told me nothing, then there would be lengthy track changes which made the page look busy. I felt	Felt frustrated, some comments tell you nothing

"The supervisor said my essay wasn't comprehensive enough, and. I thought, what do you mean, where can I be more comprehensive, which sections? When it's not very specific it's meaningless" (Student 14)	gless
с с .	ing grammar, ation, can see the n

6.3.1 Explanation of focus group research notes and grid

The four pages of research notes in figure 14 demonstrate the student focus group (n=20) had a mixture of annotation experiences. As mentioned in section 6.3 the number 1 circled in the research notes indicates my initial grasp of the interview content. The number 2 circled indicates me making links and naïve judgements about the interview content. The number 3 circled in the research notes indicates a fuller temporal understanding of the data and the process follows mimesis 1-3, which I discuss in chapter 8.5. For example, mirroring the same hermeneutic process of appendix 1, the circled number 1 in the research notes was my initial interpretation and 3 my final thoughts. This process was repeated for all sections in this chapter. In the research notes and grid's key words, followed by a brief discussion of the meaning the research notes and grid had for me.

6.3.2 Pyramid and triangle significance

Page four of the focus group research notes (see figure 14) has an example of an intuitive pyramid diagram, which I used as an *aid memoire* to identify the emerging key words. Each triangle making up the pyramid had an intuitive purpose from a semiotic perspective. The term's

signification include words such as professionalism, learnt behaviour, NMC guidelines, safe and competent practice, showing how things should be done, learning by observation, childhood and adulthood experiences, to name but a few. This was repeated for each triangle in the research notes to add a level of signification that was hard to articulate and yet easy to forget, unless notes were made about the process. The pyramid therefore stopped me from forgetting what I perceived to be an accurate interpretation of the research data at the time. Therefore, the research notes in themselves are an example of rich description of my temporal interpretation of the research interview content and saturation. The columns of first and second order interpretations therefore, put in context statements referring to my interpretation at the time. The subsequent addition of comments and numbers 1, 2 and 3 indicate the complexity and emerging nature of my understanding changing through the hermeneutic circle. There were a variety of issues identified in the research notes to inform the interpretation grids, such as annotation comments modelling professional behaviour in the tone, knowledge and content. The issue of handwritten annotation in red ink, along with the negative comments and use of exclamation marks (!!!) had an impact. This led to a lack of confidence, frustration, and some students feeling their essays worsened as they addressed each feedback comment. Another key point is annotation correcting grammar, punctuation and referencing led to students' feeling like children again because the feedback had tone and projected authority. Now onto the individual student (n=5) interview data.

6.4 Individual student interviews (n=5)

Figure 15: Extracts from researcher notes of individual student interview (n=5). I present a selection of notes I made on the interview transcript followed by the research grid I referred to in section 6.3.1 regarding the circled numbers placed on the research notes.

Student 1

In relation to regain to the So your moral compass says that when you see, hear and read it, you know it's not right. And what would you N.Q think would be going through their minds? That they are better than you. 1= Initial thoughts KEY 2= Thoughts on repear readings So when its written down from a professional, it feels good, its constructive and it values you. If you ever get something the opposite then it would have the opposite effect wouldn't it? A Yes, it would knock my confidence if I came across it early on. But I try not to take it personally, that's the way lam. Vin glad you've had a good experience so far and long may it continue. People who teach are obviously nurses Q first and on the register. In the past they've been hands on and clinical but now it's talking about it and so 1 when you meet someone or read what they say or hear how they come across, it gives you a sense of the nursing context. That's what we are here for. It is a different type of nursing practice, only teaching nursing Role model practice. So when its sensitive and professional it's like modelling. 2) How to become Putessian Definitely. We are looking to be taught how to become professional nurses and if we are looking at people first who are teaching us and getting negativity from them, that's not teaching us the right way. You can't teach 11 someone my age, (I'm an adult) I know manners and I treat people how I would like to be treated and not everyone is like that but in that respect I think people need to be more-professional. Q Why do you think they're not? А I haven't come across anyone who isn't yet. 0 Good. It's your first module, isn't it? If there are parallels with the nursing practice, why is someone positive and constructive in the nursing (that's an inappropriate question if you've not been on the wards yet)...... it's basically stress that would be underlying reason why someone has a bit of an attitude, or they may have difficulty in a culture of doing lots of work with a lack of staff so they feel students should suffer. I'm just putting words into your mouth as you've not experienced that. The opposite is what you've experienced so that reinforces how you should be, but human to human guard drops sometime. If dearly see the parallel between (3) the content of an Essay developed a nuse for practice So the positive constructive feedback - thinking about the way it's written - could you use that language or the flavour of it to talk to patients? (5) ferriforcing (MB) Shee for doubled 12 read Space to develop. A Definitely. The way mine was. The fairness and straightness and the manner in which it was spoken was how it should be. Its honest and its truthful without degrading , like when you're in a situation where you're telling somebody something, it's got to honest but done with respect, Q When you thought of the person who wrote it, what perception did you get of them? Fair. Did you think they were writing it in a nice warm office? tone to (1) А Q Probably. A (3) <u>space</u> to develop in a nessering. / a safe spap where no actual housen Can occus. Annotation is the engagement (support of such 9 proces 0

А Yes, it stands out.

Q What if it was written in red?

- Α No, because red is like one those signs, an angry sign. Colour inclucates
- Did your annotation have a section in yellow? Q
- Α No.
- Q I was just wondering. That's why it didn't look busy.
 - Is there anything that you want to ask about annotation?
- А No, it was all covered in all the feedback.
- Q You passed didn't you?
- А Yes, I got 56.
- Q Thanks very much for your time.

Spredent Interest (1 Purpessioned "they know what they were telling Rove modelling peodback erbant" Teachop us haw it Should be done" - Colour of annothed Sudant the essays FB parallels other brids of feedback kneulage, altitude, projessanding FTAZ - Annahathan is no diff- to other former _____ Of FB in the need to be clear, ROA of implinup text, be transparsed, fair & valuing the andert's efforts (no depredation) footback often stated " cohonent' Moving be caliqued, in the structure, flow, content, plot. · Focussed on enous, punchaten specting, grampion discouse? Finished with praise - haped with my sense of

fecture good - (ave)

That is imposed because it is generation) Essay being the Safe professional (MR Space to experiment 100

FB= feedback (2

mood langer.

MB = Modeling belavour. FTP = Fitness to practice assessment

Care

MB

Student 1: 1 st order interpretation- initial stage	2 nd order interpretation- naïve stage		
"Annotations were clear and easy to understand. That made me feel good because I could understand it and put it into action"	Clarity of discourse, positive feedback		
"We are looking to be taught how to be professional nurses"	Annotators model nursing behaviours (MB) and principles.		
"Manners and morals are important in nursing, not degrading but valuing people is essential to be communicated. What comes through is the lecturers caring attitude"	Manners, morals and valuing people are important. This comes through a caring attitude to the student in annotation		
"Cohesive comments made me think differently and made my arguments clearer. It also made me think more clearly and critically"	Annotation helps to clarify thinking		
"Correcting punctuation errors is part of life to receive feedback on how to improve something is the lecturer's job"	Corrections are expected because it's the lecturers job		
"Annotation is no different to other forms of feedbackbecause there is generally a sense of dread and deflation when receiving negative feedback"	Sense of dread when receiving feedback		
"The feedback finished with praise which helped raise my mood"	Praise in feedback helps mood and confidence		
"If feedback is sensitive, fair, straight and truthful, it promotes caring and professional attitude"	Constructive feedback promotes professionalism, morals, manners		
"I can see that the lecturer is checking to see whether I know my stuff, and on one essay they challenged what I had written and encouraged me to get up to date with current evidence based practice. I realise what I had written was potentially dangerous to a patient"	Modelling behaviour by challenging the available evidence Key of abbreviations : MB =modelling behaviour, FB=feedback and FTP=fitness to practice.		

6.4.1 Student 1: Explanation of the research notes and grid

The research notes and grid for student 1 identify role-modelling and feedback styles, parallel professional attitudes and good writing styles. For example, annotation being coherent, clear and concise reinforces a sense of professionalism. Assessment is an opportunity to identify the student meets the intended learning outcomes of the module, and demonstrate the students safe and competent practice. I now present student interview 2's research notes and data.

Student 2

* Destructure Student 2 Student 1/V 0 What do you think annotation is? I think it's the feedback that's given onto our assignments. A Q And you've experienced the digital annotation haven't you? So, the typed feedback comments directly onto your assignment? All regatue Yes. A Q How was that experience for you? A Not good. I failed the assignment and the feedback I got I found very destructive. There weren't any positives; it was all negative, negative negative. Even the bit I firstly submitted to look at for supervision suddenly wasn't ok by the time it got submitted whereas it was for supervision. So I found it very destructive arcused c and it had a very negative effect. Do you want to know about the recorded feedback? Oplasar. Jourealy cheating Yes, anything that is relevant. Q On the recorded feedback, the marker had said "these are not your words", so he had accused me of A cheating, or that's how I had interpreted it. Obviously they weren't my words but I had referenced badly, but the point of that exercise was to know how to reference, so I took that very negatively. When I contacted Vou dut him about it he thought it was my fault that I took it negatively basically, so he told me I had a duty to tell people how to give me feedback I was under the impression that my ILP (Individual Learning Path) was given to tutors. He said it was (ny duty to let people know that Lused to be a college twtor and I would not have assumed that feedback would be so negative without any positives. I nearly didn't come back. in education (profestioned Really, was it that bad? (quite shocked 0 А Yes, I wasn't pleased. Ox learning Q Was that your motivation to do this interview then? (7) No, to be honest. If anyone could learn or benefit from having that kind of feedback then that would be NOT A togive FB. good. Because it was really unpleasant. Q Well, that sounds like a bad experience and if you thought of not coming back, it's obviously important to get on the record. A Yes. The other thing from a module perspective is to feed that back to the module leader. DQ matre I did. It's been put in my hands to go and have supervision with the person in question, which I felt a bit А intimidated by, because of the comments I didn't really want to. I have now raised a meeting but I feel very nervous about it. An wiety (2) It sounds like remembering about the comments might be painful? Q A Yes. With the supervision, if the bit I sent by email (he didn't offer to go face to face) to get an e mail back saying it was fine, (whatever fine means) for it not to be fine when I resubmit was a little of an anomaly. I 2) Misriteprotohan / Rerception / Consident AB

1-V (2

Q

0

А

aware of emen-If you remember the digital feedback comments, do you mind remembering them and giving some examples?

Some were "No reference to the target audience" but I didn't understand what he meant by that and there was no explanation and he said I should have given two examples of evidence based practice but when I looked through the assignment given to us, I couldn't find where it said that and there was no indication from him where I could find it. The way the assignment was written was unclear as well and I have misunderstood now. I have an any degree and language is very important to me and I really can't see where I went wrong with it. So to actually have not understood where I went wrong and for the feedback not to explain what was expected, I don't know how I can move on and make it better.

- I can see the frustration, with your interpretation of it and with someone else reading it and reinterpreting it Q and saying you didn't interpret it right.
- But if I didn't interpret it right and got it wrong, then I hold my hands up. But what would have been good 🐔 А he had said what / should have done. Clarity of the feedback. He should have given me something to go on. Mismtenstate. 2

So the referencing and the comments, what comments do you actually remember? Anxiety? (an one over teo Q

"Attention to detail". Well, I'm really anal and I read everything a million and one times. It doesn't just get submitted, it gets looked at and looked at. The reference list had been put half way down the page and that wasn't how it was when submitted, but nobody would accept that the format had been changed. And yet I've spoken to two other people and they say it happened to them. Now if I print out what I had submitted it doesn't print out with the reference half way down the page, it prints out at the top. So something had itely happened to my formatting. figorous but does that headness with the finte prototo , definitely happened to my formatting.

That happened to me with a student I marked, in fact the conclusion was before another session ?????? Ana

said it doesn't happen but if you talk to students, it really does. That was something out of my control. And the difference between people marking as well, I spoke to someone who was concerned about this and they were told by their supervisor that if it was only formatting then they didn't need to worry about it. But I lost marks because of it. There needs to be some sort of rule of thumb or some acknowledgement of technical error, but it definitely did lose some for my formatting.

Q So comments you remember vividly and you've mentioned referencing, is there anything that stands out as being destructive?

Well, the accusation of cheating really, I was mortified and crushed. The whole thing was to summarise (as you know) which I did, but I referenced it badly so I had a too big a thing with only one bit at the top. He put "these are not your words". They're not but I put Barkley 2012. What I should have done was done a joiner, and he further goes on to say ... ddddddd. That would have been helpful if he'd put that in instead of saying I'd cheated basically but I don't see how I could have interpreted it any other way. He e mailed me back and said he was sorry I'd seen the feedback as so negatively but there wasn't anything positive there to read into it and also I'm the sort of person, if you tell me 99 things that I've done right, I will hear the 1 thing /ve done wrong and that's what I'll focus on because I want to get it right.

Q Any other comments that were vivid, that hurt, other than the allegation?

А That was the main one really, that I'd plagiarised, in effect. That's a bad thing to do and not a thing to be accused of when I'd just referenced badly.

~ / integrity / M. B / Professionan MB



- I III LINIKING, ILS ODVIOUS LIAL THE ANNOLATION COMMENTS ON THE PAGE NAV AN EJJECT ON YOU affect. А Yes, I went into meltdown, which might seem like an overreaction but it's important. Q If those comments were on a separate sheet instead of directly on your assignment, would you have perceived them differently? It was actually good/that it was on the relevant bit, the bad part was not having any positive. When I took Α the teaching qualification, it was sandwiched (the feedback) and that really does work because you think "oh yeah, ok that was good" or "you need to do that better". It leaves you with a positive feeling, so I think if it had been constructive on the actual work, then that would be good. It's relevant and where it should be and you know exactly what you're looking at and it's clear. Whereas if it was on a separate piece of paper you'd be reading it and trying to find it on the actual text. Q. What about these very negative words, when they were actually on your work? They are there like q parasite А It was kind of like that. Q It makes it more personal, doesn't it? The comments on the feedback sheet said "access wise, learn how to write academically". I've been out of А education for a while but I do know how to write academically. It was a little bit insulting. Q It felt patronising?
- A Yes it did really.
- Q You can see the pros and cons can't you? Those comments directly on the page show you directly what they're referring to and also I asked the question because I know it is a factor that because they were quite negative, then being directly on the page makes it more personal to you.
- A If there had been something positive or if I could have read the feedback on the page and it said, "you didn't use two pieces as required by...." or given me something to go on then it would have made it less negative and I would have known what I was doing. I don't know whether to screw it up and start again or jiggle what I did. There was no indicator as to what I've got to do. With the result in that when I came in this morning and looked at it, I shut it down again.
- *Q* So you sound like you're overwhelmed with what you think you need to do.
- A To be honest I'm a bit worried about failing it again and then I'd be off the course.
- Q Worst case scenario is that you'd have a third or fourth attempt.
- A I've only got two, because I've been stressing about it.
- Q You'd just make your case. Because they value your experience, its highly likely they'd give you a third or even more because they want people to stay on. It's formative development in the first year particularly, which should be a nice experience, feedback wise, for the first assignment.
- A Obviously the person marking it doesn't know me, and there are 400 of us but I would hate to think that other people had got that negativity. I know they are marking a lot but go away and have a drink and a break. (3) I was acutely awave of high state of emotion.

178

wes good

On page it You acknowledge that the anhotation directly on the page is fine if its constructive but for the reasons you Q mentioned, because it was so negative, it brings a different dimension to it. What is that? regaine effectacordent Destructive. That's the only word I can think for it - it completely crushes you. It's just an essay, but it's so A important and we want to get it right. So, if something is so destructive it demoralises you and you really don't want/to go back into it because you just look at it and say "oh no, I don't want to look at it again" but I've got to 🗸 ve got to do it. The best word is destructive. I know everyone isn't like me and I do have issues, I have a mental health condition that makes me more hypersensitive but I don't think somebody has a right to anxiety make people feel so bad about something that's taken so much time and effort. It wasn't easy to do and it was the first piece of work and I took a lot of time on it for him to say "attention to detail". How many times do you want me to read it through? Hypevigilance? As well as the time, and only you know how much time you put into it, whereas the marker often doesn't Q know and might presume something but the writer knows the reality and how much they give up, like missing their favourite tv programme or playing with the kids and you think "I've got to do this assignmen rentes I don't watch much tv these days and I've given up a lot to do this course. А Its high risk assessment, aren't they. That's another point, having to pass because some people give up a lot Q and some others are straight from college and they don't know much about life and haven't experience other things. There's paying the mortgage and all that that, some students don't have that. I have seven horses which I'll probably have to down size to come here so it's been a huge decision. А One other issue is, nursing lecturers are nurses first and constructive criticism has the positive strengths and Q weaknesses, When it is perceived as being negative, what impression does it give you of the lecturer? I a does Where was the caring nursing side there? Did they miss the lecture where they were told how to give A feedback. I just think it's an essential....I used to teach special needs and you do have to be very clear - you can't molly coddle, you have to tell them what's right and what's wrong. You need to leave them with a feeling that they've done something right and that motivates them to go on/and think ''''ll do it right next time". Without that, what have you got? You have no motivation and even though I know I've got to do it I've no motivation to do it. I have a meeting next week and I'm really scared. Q It's about the care isn't it? Your view of them as nurses and were they like that in practice? Exactly, and ethics. I'm into the ethics side of it all and you think, where was that? Because there's a certain А ethical obligation to not crush people. Why do you think it's so destructive and why are the words so negative? Q Because it just told me everything I'd done wrong and nothing I'd done right and gave me nothing to build 11 А on. What does it say about the lecturer themselves for choosing that attitude? Q He'd either had a really bad day or marked too many assignments or maybe he thought he was actually A being constructive because I e-mailed him and said "I find your comments destructive and not constructive" and he put the onus back on me by saying that it was up to me to tell him how to feedback to me, which 1didn't think was right. He said he was sorry that I'd taken it negatively and he didn't accuse me of cheating. How else can you take "these are not your words", I can't see any other way of taking that. You mentioned the nurse lecturer (well I did actually), but trying to get as much as I can out of lecturers and Q nurses. The nurse first, who then (for whatever reason) the comments are consistently negative and you the 4

lecturer?

А He didn't give a damn. It didn't matter how he made us feel. It was like, "I'm the lecturer and you will take In other words you don't take that attitude and why should you? (was I colludy?) what I give you". Q To be honest, coming here today is a big thing for me because I've put hyself in front of the firing line and I А would rather be invisible and rather get on with it and just do it, but I felt quite strongly. I appreciate you taking the time to come and you want to continue? COVERT Q А Yes. In practice then, what was your view of them in clinical practice? 0 А I think they would be very critical and if you had to work for those people in clinical practice I think it would be quite demoralising and probably recognition would be very low on their agenda. 7 Recognition grospip / meaning Q What about patients? 2 I think if you see somebody treating staff in an off-hand manner or critical manner then I don't think it engenders trust from patients point of view, it might look quite bullying and it may not engender trust in that person to look after them in a caring and compassionate way. I know sometimes you have to take RYC somebody to task but there is a way to do it and if someone is very efficient for example, and that's why practi they are very prescriptive of what they say to people and it can come across as very cold. People don't want that in hospital as they can be very vulnerable and they have to rely on these people to look after them. They should be equal in their dealings with everybody. Q In whatever environment a nurse finds themself in. Whether that's in clinical practice and then going into university. 1/c freacting - Education + Chilical Numa nustin () Α You don't switch off your humanity switch when you walk in as a lecturer and then switch it back on when you walk into a ward. No I don't think you do, some people do. Do you think then that they've got something they need to get off their chest? Is it an attitude: Q thoughtes! Α It just came across as thoughtless. I don't think they'd got anything to get off their chest. They are trying to do a job and they've got a lot to mark but when you get to the point when you think "oh, not another _???????? bibliography" you should walk away for a bit and come back when you are..... the first one should be the same as the last one and in the same frame of mind. It hard to do. I know because I've done it myself. but at that point when you get to the point when you are thoughtlessly putting comments down then you should take a break. 19A SOLN Q What if they took a lot of breaks and they got back to the feedback and the comments were still negative? А Well, they need to go on a how to give feedback course or retraining. Q Wouldn't that give the impression that they've got a certain mind-set about nursing or students, or patients? А I don't see how, if it's negative, its negative. Everybody brings something with them about what they read, so I would bring my experience to something and you wouldn't interpret something the same as I would. Bu

if something is negative, its negative – there was no ambiguity. XMY is the provacate cer aff chast

() interpret lotter no ambiguity
- Q It sounds like in your e mail to them, didn't you say that they thought they were constructive?
- A

I can't remember now. He just said that he was sorry that I'd taken the feedback so negatively and I had an obligation to let people know how to feedback but I was under the impression that my ILP had been sent round to tutors. I checked with my tutor and she said it was just for placement, I thought I'd better check for here as its fifty fifty. I don't think it was a stupid assumption that it was passed around because it is something that's important. But it's not up to me to go round to every lecturer. I will if you want me to but not everyone wants to hear it.

- Q If there's anything identified as an individual learning need then it's always best to e-mail each new () Identified learning disability added to serve of negotivity supervisor.
- I do get extra time in exams, I get 25% extra time. My problem is also supervision, you can only do it up until А a week before and I work best under pressure. In fact I wonder if it's because I had so much time on this one and I didn't do it under pressure, maybe that's the reason why it wasn't as good as it could have been. I do work better when the clock is ticking. Supervision isn't always that easy to access when you work at the last minute. We all work in different ways, and I work better under pressure.
- Q That will be handy in nursing.
- А Reading feedback comments:

"It's a shame that you've interpreted the feedback in a negative way, I haven't accused you of plagiarism, however, to the reader, poor referencing style can look identical on the page".

"My feedback stated that when you use other peoples work in an assignment you must reference appropriately".

"The ILP is used to inform your personal tutor of any reasonable adjustments that need to be made in relation to both the theoretical and practical aspects of the course. It must also be clearly stated that you have a responsibility to inform those who you deem appropriate to know of anything in the ILP".

"For future assignments I would recommend you access face to face module supervision and inform the marker of any issues that you feel would be necessary to know in marking your work".

But he never offered the face to face, it was e-mail.

- Q How about the audio?
- 3) Sounded reaconable and had geod tone А The only bit I can remember from that was the impression that I cheated.
- Q You mean it reinforced what was written?

Yes, it made it more so because he said it. I couldn't misinterpret that, "these are not your words". If they're not my words then they are somebody else's words, which means I've plagiarised - that means I've cheated. Rather than "this is a big paragraph with only one reference" (for example) "it would have been better if you had used a joiner". But I've come to that assumption myself.

Q So you are actually modelling how best to feedback in that situation. You are giving an examples of how you would have liked feedback. That's your tutor's hat on. You are saying it constructively and based on what he 3 (MB) modeltie by the Shoeen should have said differently and may do in the future. It is a reflective-process .

191

1

models and there is evidence based practice for giving good feedback which he has pet followed. 2 Q Some people will look at their own feedback and think "oh, where is positive in that?". There was a study a while back where lecturers looked at their own feedback to students where they identified that tone and attitude came through. They were mortified, so they need feedback and you are actually modelling QRACTURL constructive feedback on the lecturer who you experienced as not giving constructive feedback. DECALL' You have given a lot of information about the emotional aspect of receiving feedback. We need people in nursing to have emotion because it demonstrates caring, compassion and enthusiasm to actually want to do the job and pass each module to become a registered nurse. (1) Valuing Students experience. А I would never personally make anybody feel bad. If I did make anybody feel bad then I would have the nerve to change it. It's not unusual to expect people to not make you feel bad. Q To do no harm, as in maleficence, (can be good, moral stuff which is on the other module) do you think the lecturer purposely or unconsciously actually did harm. Unconsciously. I don't think they would do it personally. I might be a bit naïve but I don't think it's done on ſ purpose. Q So a degree of emotional effect which might take that to do harm. A I cried all day. I was mortified. So you need to pass this module and get it out of the way and get onto a different type of assignment Q writing because this one is a bit characteristic in it's a certain way of doing something. A I've got a mentor for an hour a week from RAMSTAT. She is going over it with me on Wednesda Q Have we come to the end of what you want to say? Have you come to the need of what you want to ask me? Good point! А Q Well, I think a good question to end on would be - I've asked some questions that are my interpretation of what you want to say. Has anything I've mentioned been in the wrong direction or close to what you, were saying and words like "unconscious message" that came out? 3 Aware OF my bras A I don't think he sat down and thought "I must be really horrible to this person". Shouldn't be in this sett or in this job. I just think he didn't think about what they were putting. (3) Dents advocante) Q Empathise perhaps. The empathy may have been removed for a short time. Do you think is it anything about what you wrote and how you wrote it that triggered in them a certain reaction? А I don't know because I honestly didn't know what I was supposed to do after reading it... I thought I'd done ok. I didn't think I'd done blindingly well, but it's been a long time since I've done a thousand words on anything. Somebody sent me theirs to look at and when I read it, theirs didn't make sense to me. Now, I'm not from up here and some of how this one that was sent to me was written was very much in the vernacular of the north west. Well, if that's how I've go to write then that to me is not academically writing then I can't do it - I don't speak north west. That may sound strange but the language and accent does throw me sometimes and I don't always understand and have to ask what is meant by that. It is very different and I have a combined arts degree and majored in history with an English component and I got a

2:1 so I can academically write.

- Q Clearly.
- A And that was with a child and five horses and travelling to Nottingham and Leicester every day. There is nobody harder on me than I am so 40% is a low pass mark so to get a low pass mark was really quite a blow.
- *Q* It would be, because you are not used to that after getting a 2:1.
- A I don't like getting things wrong because I don't like getting in trouble. I will move heaven and earth to not get in trouble. It seemed ridiculous that I couldn't get 40%. It knocked me down a bit. I can take constructive criticism but there wasn't any.
- Q That's a good observation because how are you supposed to take negative feedback?
- A If you've nothing to build on how are you going to build and improve?
- Q And the natural defence to that would be to feel horror, shock, perhaps anger and feeling an emotional effect of those words. I was just thinking about Eric Burns, Parent to Child you mentioned authority at the beginning and I might have chipped in "power", is that a factor as well, feeling like a kid again?

It is really. The minute you get into that role, (without going into too much detail, I had an abusive upbringing) and if someone takes that tone with me then I either react aggressively or I revert, and it's very much parent and child. It invokes the same response in me as it did when I was a child and I'm very aware of this but if you're constructive with me then I can build and move on with it.

- Q (That's a very profound reflection, isn't it?
- A I've been doing a lot of that recently.
- Q The feedback that negative re-enacts feelings, thoughts and emotions from the past revisited in the present.
- A You're under someone's control and power 2
- Q Feeling childlike again, as an adult.
- A Well, I'm nearly 50 so I shouldn't be feeling like that.

Q (1) That's the past come back to hurt again?

- A Yes.
- Q I know you spent the day in bits and cried all day as a result, which is important to acknowledge. Did you actually feel as raw as you did in the past with those comments?
- A Yes. At the moment I'm in the period of recovery. So I am not where I should be which is why it probably hit harder. I guess that's why it had a big impact as well.
- Q That's an important thing to note. Ok. I've got to be aware of the boundaries and the purpose of this interview. Just to summarise (again I use my own word "raw" to emphasize) you said you felt similar after receiving the feedback?
- A It's like a physical pain, like your skin is made of nettles and prickles. And what was needed is what I could have done to build on it to make it better. But there's no way of doing that and alleviating the prickling.

8

Desychosonoutre pain Shin /proches.

of parallel pivers

power

last

ansterence

Lisemparenet

- There's no way of using your defences to ward off..... Pain / un carsonaus / Parallel to actual Pain? Q
- Short of going to see the person straight away, which would have been impossible to do at that time so I've А left it as late as I can to try and get some perspective.
- But because, as you see it as so negative, you get the physical sensation which if there had been some Q constructive feedback then your defences would have warded them off.
- А You can get something from it "ah well, it wasn't completely useless" then you accept more the negative comments if there's positive there because I've come away thinking the whole thing was completely rubbish. I don't know how something you formally said was fine (which is a bland word) suddenly isn't fine on actual submission. That's seems like I've been misled a little bit. Because if wasn't fine when I sent it in then he shouldn't have said it was fine as I would have changed it. It was a bit of a side swipe when it came back and it suddenly wasn't fine.
- Q It sounds like it knocked you for six.
- А It threw you now.
- That's another word it threw you, mentally and physically probably. So it became a physical phenomenon. Q This might sound a bit odd but you obviously know the lecturer.
- А I only met them once.

Bact to EMIC BURNS

- Just going back to reliving and the effects of the negative languages of parent/child perception and like a Q relived memory in the present, did they remind you of anyone in the past? Does it matter -head (Frithan - the great is real?
- А No.
- Q You're not physically remembering them as the lecturer?
- A No.
- That would obviously be another physical transference. (Cleasly leaping a heard to hypothesse) Q
- A That would be really unlucky. I can't really picture them to be honest because I've only seen them once and I'm very short sighted anyway.
- Q Sometimes people....
- А Yes, it triggers something, you see something that you are expecting to see because you've seen it before but I probably wouldn't recognise them if I walked passed them. Iranster

Q So it's just the words that had that effect?

- truted to tole of annotation (Ball et al) And the tone. A
- Yes, the tone. And re-enacted, childlike responses, is that fair to say? Q
- A Yes.
- We all feel like that sometimes, if you go into a trot and slip all of a sudden you're a child again. Q

- par when you we brown objion pend anowed to do anything wrong and nothing is minor then it does have a really big impact later on in life.
- Q Well it would be for self-esteem in particular. Is there anything that you've disclosed that you want to keep out?
- А No, not really, I'm an open book. You can use anything useful.
- Q Do you want me to send you a copy because it will be transcribed by a colleague of mine. You could read through it and think "oh no, I don't want that"?
- А No thanks. I trust you.
- Q

i Learn hew not bgive FB = modelling behavior? (i)Missinteprotation - not understandig what was meant not being able to move an = stephaticn? / // · Suspicion of cheating = M.B./integrity · lationisty = Power/ine/child toadult · Parastical fann Aatan. stud Not knowng · Negativity/ thoughtless encything done wrong (nothing done night. · Parallel 1/c dividal practice = Mrs if negative - devalue (Dv) (2) RECOGNITION / being valued (K) · First Same as last - Misconcepte Modelly by the Student, (2) Stephection, Bart hurs again in presents · Emotional reaction 10" J crued all day" () Thiggers Someting (Physical / enchand

6.4.2 Student 2: Explanation of the research notes and grid

I present another tier of research notes made on the full interview transcript prior to writing more notes and filling in the columns of the research grid. These research notes demonstrate a comprehensive process I followed for all of the typed interview transcripts in this chapter but, due to space, I present one example here due to word restrictions. This process was unplanned, otherwise I would have mentioned it in chapter five's research design, yet it demonstrates how I instinctively identified the first and second order interpretations, annotated when making notes, and used a key to codify emerging themes. Unlike the other sections in this chapter, due to the format of the research notes, I present no research grid here because the steps for first order interpretation on the transcript are self-evident in the selected key text and use of the numbers indicating my later thoughts. However, the initial key words of emotional over-reaction, feeling patronised, destructive comments, anxiety, perceptual issues of the student, repressed memories from child-hood, the annotator demonstrating un-professionalism in the student's eyes, were all relevant. The issue of perceiving the intent of annotation was significant and the emotional impact on the student was quite acute. I demonstrate my own part in the interview process by my interview questioning, which I realise clearly affects the flow and direction of the interview, unless there were no questions at all. The circled 1 (see section 6.3) indicates an initial, first order interpretation because for a time they were the only comments I made on the transcript. Faced with a mass of interview data, it can take time to make sense of the emerging themes, especially if reading the interview content after the event. My own second order interpretations were transference, anxiety, mis-interpretation, power, modelling behaviours and recognition. The pyramid at the end of the research notes (see section 6.3) identifies the emerging second order interpretations in the research notes. I now present student interview 3's research notes and data.

Student 3

- Going back to childhood, if your teacher wrote "well done" then the writing was always different than if they A were disappointed with you, I think.
- Q Have you experienced any hand written annotation here?
- Α No not on this course. I tried to do a midwifery degree but failed my final module which I did here, and there was one tutor that when she did supervision she had a digital copy and you had to upload it to her via e mail. She sat with the digital copy and she would highlight whilst she sat there and that was more beneficial. Even though it was face to face you could tell it was sincere and she was trying to help.

"Social

Q Engaging with the text and not, therefore, not making it a personal attack, as it were.

Student 1/V (3 А Yes.

- Did she start making changes and asking "what do you think of this, if this moved here Q
- А Yes.
- Q How did you find that approach? In other words, you are both together and your work is on the pc, highlighted, and perhaps working through the changes, how did you find that?
- Because I could see what she was thinking, I understood a lot better what she meant because I was reading А all the communication then - the voice, the body language and she was showing me where things could go.
- And <u>she wa</u>s experiencing the text as you <u>were reading</u> it with her so you were shown and it <u>b</u>ecomes more Q obvious. And then did she ask "where should that move to?" and "where should that go?". Donno new or
- A I really struggled with the construction. I could get all the information down but it was making the assignment flow, so she would say "this sentence needs to go next to this sentence". So even though it was my work, she was helping me.
- That's a great example of face to face use of annotation and a more collegial approach to showing how an Q assignment can develop. Which I then presume, when you are on your own and you've got that same piece of work, you can then more or less follow how it got to that stage and the changes that occurred. Then start to Anxiety / Pos. repairs make your own.
- Yes. You make it a blueprint then and you understand how things will read better, because I've got mild А

dyslexia and sometimes lose everything even though I know what I want to say. My train of thought isn't straight down. Whether are any more questions. Thanks very much for your time. Q Cleas AB · Passt Associations find Together theread) · Demansfrote

Student 11.3

Modellup behavar (MB) through Superingan Annotator worked through the annotations after student came for feedbeeds-Then modelled, demanshaded here to impore the text, make charges, ask quotions like a what should go here ... how can this / Section le improved ... Coungousee?" D = Demanstratile - Showing by doing - door F-B / he's happy / recognition / rewood - face to fact feedback and wated they ph annetation - Benefit OF PB an sum notice which to reinforce modelling. - Notes outre transcript) - fiz = feedbact, HW = hardwitig AD = demanstrative A = Annotatia factitation undertand. Accodence SV > Superisien. - Questienp /prouterp/ reinforcing -- Proscriptive advice - Stick to learsy automer - Part associations M HIW - Love B' charges tonted to the poor - telus

Student 3: 1 st order interpretation- initial stage	2 nd order interpretation- naïve stage
"Annotation did help me visually to make sense of what was being read"	Clarity of discourse
"Feedback to identify strengths and weaknesses"	Promote confidence and self-esteem (psychological health)
<i>"It is important to be constructive, fellow student received all negative affecting confidence"</i>	Zoom in on the negative
"Encourage the effort, focus on what to work on Previous experience of receiving handwritten annotation"	Need positive strokes, modelling positive and negative behaviours (MB) in supervision (Sv).
"Got used to their handwriting (HW) style"	Handwritten (HW) feedback (FB) linked to past associations and different to
"Reminded of child hood experiences of angry handwritingThis is hard to do with digital annotation, so less communicative"	digital for that reason. Felt like a child again, anxiety/ memory/
<i>"Like to see if they took their time or rushed it. Changes</i>	transference
depending on mood, motivation, energy, perception of what is written"	Looked for cues as to the annotator's mood
<i>"If normally curvy, look for when it gets slanted, sharp, spidery, aggressive"</i>	Word association/ affecting emotion, thoughts, confidence as a child, remembered as an adult
<i>"If unconfident as a child, when an adult you associate the</i>	
same feelings when reading annotation again"	<i>Key of abbreviations: MB</i> =modelling behaviour, Sv=supervision,
"Moved around a lot as a child so had the need to fit inFor	<i>H/W=handwriting, FB=feedback and</i>
example when in the Middle east the teacher drew a smiley face and that made me laugh"	<i>FTP=fitness to practice.</i>

6.4.3 Student 3: Explanation of the research notes and grid

My research notes from student 3 identify a mixture of annotation comments on the interview transcripts and research notes. The modelling of behaviour and elucidating both negative and positive emotions from the student was a key finding, in particular, observing, mirroring and imitating good communication. The student's childhood experiences were also key to their perception of feedback, stating they felt child-like, disempowered and patronised by the annotation experience. Therefore, my second order interpretation identified transference, thoughts and emotions experienced in the past evoked by present triggers. Student 4 is next.

Student 4

А

Yes. I think I'd also accessed supervision prior to that and from that I'd had the digital annotation so I'd had the same thing from the supervisor from when I'd met him previously. He put the comments in the same format as I'd had for the final feedback so I think that helped because it wasn't something completely new.

Q And consistent? Student 1.1/4 () Yes. And the way he structured it and laid it out was consistent as well. That helped, especially for the A 202 section where I'd had the supervision because I could refer back to the section where I'd done wrong or right (to improve it for the final) and compare the end result to show that this is what he'd said when I went for my supervisor meeting. And now I can see that he is happy that I've taken his advice on board and made that change. He's happy So you're even monitoring your changes which were perhaps fed back previously? You see your draft before, then you get the feedback, you then make the changes, then you get feedback. sometries (counciliation) of each word Then I can use that feedback for my next piece of work, which I'm hoping I've done correctly That's often an issue about feedback and annotation – measuring what has changed as a result of the $\int e^{-\sqrt{\alpha} d}$ feedback (3) I have an isone with measurement in the - Miked To Management in the - Miked To Science and the NHS It's not something I'd done before but I think it's because I didn't have the ability to do that. Usually you get feedback and its verbal or just a few comments, but I don't find that very useful because sometimes it's very difficult, if someone's giving you some feedback (on say a whole section of work) and they say "this is quite good" or "this isn't right" - you can't always relate it specifically to what they're saying. Whereas in the way the feedback was structured, it was always highlighted to one specific sentence or one specific paragraph. I could then compare, to make sure I had understood what he'd said originally, and done something about it, O Super / accout A/prin because it's no good if it's not helping me. It's good that you can actually see the effect of feedback and the changes you make, and using that for μ_{Λ} future writing. Syrensiz That's the whole point of getting feedback, there's no point if I'm not going to take it on board. Whether good or bad. Did you get a good mark? The grads? Sup atoke industrial 2 facil Yes, I got 93%. I didn't have that much wrong. accent GRIDS INDIAND Wow, well done. I've never come across that before. Der But I don't think I'll get that again in the next three years that I'm here. You must have impressed him very much, is he a very hard man to impress? 0 I did work very hard for that grade, I don't think I'll see it again, but I did work very hard for it. Preuse p ina I just wonder how you've written so well so early on in your career? M I don't know. There's one thing that I stuck to and it's a thing the factor of a regular basis - "always stick to your learning outcomes, don't drift away" and I did not do the work without the learning outcomes next to me so that I was sure and always focusing on those, because that's what is asked for , whereas, with the second assignment that I've done, it's a mistake I've made. I've not seen my feedback yet and I don't know what I've got but I know it's a mistake I've made - I didn't continue what I did on the first one. Stick to fearning actences / fours

A Yes, the feedback is there for a reason. Like yesterday, there was a technical error for some people on the next module when they got their results - I've read that a couple of people said "I know I've got a bad result" but they've had no feedback because of the technical error, they just got a grade but no idea why they got that grade. Its like having that grade and not knowing the mistakes they've made and if I was in that position I would be quite...... getting that grade and not knowing why you've not done that well.

Q Have you anything else to add?

A No. I find it definitely works, I found it all very helpful. Like the collaboration of different types of feedback, it wasn't just one, it definitely helps. My tutor was able to put one comment on one section and expand on what he'd said, to make sure that I'd personally understood that and if it was just one of them, then maybe I would have phoned him up said "actually, I don't understand this, can you explain this to me" (and if you've got 25 students doing that....). I think because I've got a record of both, I can go back to it at any stage and read back and listen back and see what I've done and what I could have changed, so I don't forget.

Thanks for your time. "Take it an back - good or bad" Q Constant annotation Style Accent - non monorable another Student I.V No accent - not monorable another (4) · fronte / comment / explain "What she dave & how she dere it " Specifie dotail ." What she dave & how she dere it " Specifie dotail . unhappy wingerence provise - wart dotail. . Measuring drage has occurred / charge in pospective . Loving for errors/prentised the errors, . Nad made of . Fear of know p mistake then admarkedged by annotedor . Distanciedia - luxury of time · Miltakes = life expensive Mildakes = life expension
Mildakes = life expension
Mildakes = life expension
Porstheredge can by matters
Borallel with output more in domo nutrie
Mild helps throthy about = fractice
Whip helps throthy about = fractice
Konte + apply to practice / rescord =
Repetition / Preparation / rescord =
Monary
Matter No one warts a nuse
Monary
Matter (Mohrothan) - why? Integrity - Social
Theorem of porallel individualism - sett.

Student 4: 1 st order interpretation-initial stage	2 nd order interpretation- naïve stage
"The annotation stated simply "awkward sentence" leading the student to think what does that mean?"	Clarity of discourse
"Annotation e.g. good introduction, told me that I had done right, not which parts were right"	Positive strokes yet unclear what to change. Praise stands out/ Constructive feedback Negative comments are disliked
"I remember praise; it was well structured"	
"Don't like negative commentsmistakes occur despite proof reading"	Mistakes highlighted by a professional nurse/ Accountability (Ac).
"The feedback is from a professional nurse, so its nurse related"	Writing about nursing parallels clinical practice/
"If I make a silly error in clinical practice I could kill them"	Accountability
<i>"Writing helps in many ways; preparation, knowledge, insight, repetition, theory applied into practice"</i>	Patients deserve better
"Patients don't want someone who has scraped through their training, they deserve better"	Modelling behaviour (MB) in annotation
"Annotation modelled feedback for practice; what I've done, how I've done it, why I've done it. Specific detail and analysis"	Helpful because the lecturer wants me to learn, supervision (Sv) is consistent
"I find it all helpful, because the lecturer wants my essay to improve and want me to learn"	Key of abbreviations: MB =modelling behaviour, F=focus, U=understanding, Sv=supervision,A=annotation,
	Ac=accountability and FB=feedback

6.4.4 Student 4: Explanation of the research notes and grid

The research notes from student 4 indicate a few key words circled with 1, 2 and 3 (see section 6.3) to strengthen the three step research approach and understanding of the data. The annotator was viewed as a positive role model for writing and professional nursing. The student valued the annotation as an opportunity to learn and improve their writing and is a good example of a high self-efficacy reader (see chapter 3.5.8). As a result, the student lowered their guard because they viewed annotation to be worthwhile. I now discuss student 5's research notes and data.

Student 5

Student 5: 1 st order interpretation- initial stage	2 nd order interpretation- naïve stage
"I think good rapport in face to face meetings with the supervisor after I have received annotation feedback	Feedback as a continual process
definitely helps me to understand the feedback"	Supervisor relationship and learning through positive attitude and modelling
"When discussing the annotation face to face I can see the lecturer's point but also can see their passion for nursing	behaviour (MB)
which comes through by the way it's communicated"	Praise promotes sense of professionalism
"I love praise such as this is good and that's exactly what we discussed in our meeting"	Good rapport in supervision is collegial and promotes good attitude
"I see supervision as being a time for nurturing, a time when I can be comfortable in their presence, be open to what is said and discussed"	Lowers defence, respond to the content not the person
"Lowers defences, promotes engagementResponding to the writing not the person"	Lowered guard when there is good supervision rapport
"My guard is down not up during supervision with good rapport"	Flow of the essay content relates to interpretation by the student and annotator
"Cadence of the written word depends on the reader's expectations, fore-thinking"	Flow is improved in an essay if better understood
"I think an essay will flow better if I understood the article but often the essay reads badly if I can't make sense of it"	Use annotation in the form of flags, post its Annotation is engagement, sense making, feedback and feed forward
"I highlight my own work. Flags, post its"	recubick and recurrent wind
"Annotation for myself needs to make sense only to my senseI annotated because I want to make sense of the article, I want to engage with it, to see where it's coming fromto see where it is"	Key of abbreviations: MB =modelling behaviour, A+=good attitude, FB=feedback and R=rapport

6.4.5 Student 5: Explanation of the research notes and grid

The research notes from student 5 promotes face-to-face meetings after receiving annotation and the student's "guard" being lowered due to the establishment of good rapport. This rapport was an example of the annotator modelling positive behaviours, good attitude, warmth, and nurturing student learning. These conditions were key to establishing acceptance and behaviour change. I will now discusss the individual lecturer interviews (n=8).

6.5 Lecturer interviews (n=8)

Figure 16: Researcher extracts from notes of individual lecturer interviews (n=8)

-----· Comments at bottom aread related directly to carteit . Comments need to be clear + concite · Annotation - Shows you have need it . .V1 · Prouse - companye / contract. . Has to sand posttike · Deuelop shalents critical abilities - Dep tig Prep · The above is imperfect for drin ad practice bo . D. consterarg uneuts · FB prop for clinical predice - Digital ear ann from gevenit AB but track changer addatail Xt. Carteur aleAsto F.T.P - attitude, bre of Gutent . Academic PTP - Clinically FTP ET Gutent · hove and real featurer meneral it as. · l'agranism - laginess & descript Equates to divi. practice. · L= feels disappointed fruction by direction although L = mining ~ modelling (M-B) ABT modelling "Rde : 5 to Incubato a richitop attrude ANOT to allow poor practice -('Cane into lecturing Roots Annotation Rootes poor prostice out · L = only human - D + F if cheaded Eugeano · Need more training in annot : + making · Not trained in ANY way to anotate MARK . We are modelled on senior people. e.g. moderate AB . Hence, annotation & not standardized excellent. or systematic - FLAWED because inconsistent · Ann-on drafts - 1:1 V. useful re. publicity querieir PTO -> 1'.V. (1) · Stempest

Lecturer 1: 1 st order interpretation- initial stage	2 nd order interpretation- naïve stage
" many lecturers haven't worked on a ward or in the community for a long time so they are thinking back to their pastuninformed by new clinical experiences. Many leave the ward with differing experiences, positive and negative so	Lecturers who haven not practised for a while will refer to their past experiences and memory recall may be problematic. Modelling behaviours (MB)
they bring with them a certain perspective. Their experiences	Modelling behaviours (MB)
may have led to a biased view about what is relevant"	Products of our past, modelled on
<i>"We are products of our past. I seem to have been modelled</i>	positive or negative senior nurses worked with which may come through
on the most senior nurses I worked with, good or bad and my	in the annotation. An improved essay is
ten years' practice has been difficult- so that may come	judged by the student coming for less
through in annotation. I identify specific parts of an essay and how to improve. The less coming back for subsequent feedback	supervision
the more they have learnt"	Annotation prepares the student to
	develop their critical thinking informed
"Annotation feedback prepares them for professional practice;	by evidence. Frustrated and
it can identify the rationale, focus, evidence, develop the argument, consider the research, what is the argument,	disappointed by plagiarism
counter argument and conclusionI am disappointed by	
plagiarism as it is cheating and deceitful"	Annotation makes the essay individual
<i>"Essays tend to blur into one and annotation makes them</i>	for the lecturer, it can alert them to poor attitudes and practises
individual. The content can alert you to poor attitudes. Student	Key of abbreviations: MB=modelling
observe and imitate practice one had written when	behaviour, FB=feedback,, id=identity,
restraining a patient "we had to deck thempinned them to	PP=poor practice and FTP=fitness to
the floor had to jab them"	practice.

6.5.1 Lecturer 1: Explanation of the research notes and grid

The research notes from lecturer 1 identifies after assessing the student has met the learning outcomes, a significant concern for them is to identify their fitness to practice (FTP), and protect clients and patients against poor practice. Lecturer 1 identifies one of the ways a student can demonstrate FTP is to inform their essay discussion with evidence, which is considered a moral priority for nursing. The essay is by all intents and purposes, considered a safe space to practice and test ideas and parallel professionalism, honesty and integrity for clinical practice. Hence, the lecturer was disappointed and frustrated by any student plagiarism. The research grids parallel some of the findings from student 5 about modelling professional behaviours, attitudes and

annotator's view of nursing practice when out of date they have to refer back to past clinical experiences. I now present research notes and grids following the interview with lecturer 2.

Lecturer 2

19

Lecturer 2: 1 st order interpretation- initial stage	2 nd order interpretation- naïve stage
"I found receiving annotation both constructive and supportive for my own studies but can only assume what others think of it when receiving them. My knowledge and understanding communicated to the student"	Modelling behaviour (MB)/ empowerment
"I try to empower not destroy, because that is our role. As nursing lecturers, we have to be aware that what and how we write something will influence the student somewhat, it also gives the student an impression about our nursing attitudes"	Empowering students is art of the lecturer's role. What an annotator writes will influence the student and gives an impression of nursing principles
"Annotation on the page helps engagement with understanding the text. What is unknown is what the student does with the feedback. How do we know what works?"	Annotation on the essay helps student's and lecturer to recognise issues, understand
"Once I've annotated I realise nothing will change unless the student is motivated to make the necessary changes. I try to facilitate thinking not	and engage with the student
correcting thinking"	Changes based on the annotation depends on a
"The lengthier annotation indicates more problems than are commented upon because there is a possibility of too much feedback. I don't want to destroy their confidence by pointing out all of the problems"	motivated and informed student
"Unfortunately the writing process is solitary. From the student writing the essay, when I am reading the essay and making the annotations and then when they are read. A better use of supervision and annotation would be developing a working relationship during and after the essay	Maintaining balance in annotation comments ensures students are informed yet remain confident
is written and marked"	Reading and writing are solitary acts. Annotation and
Key of abbreviations: MB=modelling behaviour, FB=feedback, R=recognition	supervision can foster alliances that are supportive and consistent

6.5.2 Lecturer 2: Explanation of the research notes and grid

The research notes identify the issues of balance, advice, not being too directive and feedback influencing the student's essay. In particular, annotation should inform the student fully, yet, it may not be fully understood, and not knowing what the student will do after receiving feedback identifies the paradox of annotation. The notion of positive strokes, and short comments indicate good essay content, and lengthier annotation identifies areas of textual concern. The annotation

comments aim to be de-personal because they relate solely to the essay content. I now present the research notes and grid following the interview with lecturer 3 and 4.

Lecturer 3 and 4

A2 I'm thinking about a positive comment "the best bit of this piece of work is your literature review". I can still for me it is - one they going to tail a PL? remember it now. I can't think of anything negative. Arethey (- Get annoyed with another alleopt to submit Frustration with the system that appears K87 uder to supplet t Serp Ret whispereivellas RGYERP never Col her goves und do who true is the perception lora Behavers in (lass (txtip) can be Seen in essay = lazy A viva could make the difference. Face to face of gettipthogist/but poception less comp (.') Annetation much passer to expain 1.') face to face (less essay Notters Have to Face / less so nct 1:1 30-40% house not st only dane pre-veg here

RUING It a teasine to show them the script + charges . Maide Changes to demonstrate where imperiences could. be made ' Students may not undertaid the language of EB grans Print 0.ft - Good practice TiF comes naturally Davidord & print of them annotate study annatate and menuisie · Don't give à to them - Q' What dos & mean? ' I don't undertand that ' - to Is easing to undertend than other anchation Sametime Repair 16 Andert helps overcome mitundertady Huwan Connection - Know you & thous theor - take advice Woolly - lact of misudoseder - Formatrie is development - WHATI Shocking Section - Anoid Manisik Could be 3rd Sub & Same as 181 -"Pivillaged Job- to me if Janys student does not can " - Often poer musip chidents are failing in practice too - Ann reinforces apoint inswelly wouldn't the level of an ... be Judged depending on the Content of the essay? its Studie, Coherence, Content, context? - mediciser m(ment " concern ve your approach, rationable - FT * Open Connect an practice from the "withing" * FTP - Need toget bases rout before they go auto Specialite open dant see what you mean, after 1:1 they do see - Need to demonstrate being safe practitions Many don't read MB - don't knew how the initale PTP oad and Sefety Scheenis imposend MB attogues at UCLan - Essays haup Carl > hay not meet the shalet. studelt Peer pB / collegral and and 1.1.34

Lecturer 3 & 4: 1 st order interpretation- initial stage	2 nd order interpretation- naïve stage
"We both share an office and the same method for annotation practice; we print off essays and annotate, collect the comments for the generic feedback sheet which the student then receives" (Lecturer 3)	Modelling behaviour (MB)/ collegiality
"We peer review each other's annotation informally. We both annotate to formulate our thoughts prior to final analysis, collation of annotated feedback comments and mark" (Lecturer 3)	Peer review annotation, annotate to formulate feedback
"I print off all essays as I can't read or mark them online. I think it helps with my concentration, vision and attention because otherwise I may miss things. Making online makes the depersonalised online essay even more depersonalised" (Lecturer 4)	Formative feedback (FB) considered useful
"I only show annotation if requested in supervision after mark has been received- hence can show the thinking and detail behind the comments" (Lecturer 3)	Annotation helps the clarity of discourse
"Annotation can overcome resistance and defensive reactions as the student can see the issues" (Lecturer 4)	Annotation helps overcome resistance and misunderstanding Spontaneity of annotation
"Annotating and formulating thoughts as they read- I consider this to be spontaneous thinking and helps analysis" (Lecturer 3)	Visual impact of the word
"Reinforce the visual impact of the word, thinking and action by stating something like WHAT!" (Lecturer 4)	"What" indicates and mirrors poor practice, WHAT gets attention, and so it should
"I recently wrote "WHAT! What is this?" In the margins to refer to the shocking errors and lack of references. I wanted to make my irritation plain and unequivocal" (Lecturer 4)	An aid memoire, diagnostic
"Used to remind me on one occasion it was their 3^{rd} submission and was literally the same as the 1^{st} and 2^{nd} . If the first submission I wouldn't have put it. Students re-submitting their 3^{rd} essay tend to be failing in practice too. Hence, they need a short cut to wake them up to the identified issues." (Lecturer 4)	in relation to a lack of changes made to the essay generally means a lack of understanding
"I promote the fundamentals of nursing, which is very important in pre- registration nursing because it's the bedrock of the profession. The 6cs, care, compassion, competence, communication, courage, commitment are	Fitness to practice (FTP) - is it safe?
now considered basic attitudinal conditions for good practice borne out of the Mid Staffs enquiry into systematic organisational and nursing neglect" (Lecturer 4)	Critical thinking in an essay parallels practice- FTP
"For us it's about, are they going to kill patients, are they going to be safe and can we put our professional name to passing them. It's about professional integritywhat a nurse ought to do in a given situation" (Lecturer interview 3)	Key of abbreviations: FB=feedback, FTP (fitness to practice), 6cs (DH, 2012), MB (modelling behaviour).

6.5.3 Lecturer 3 and 4: Explanation of the research notes and grid

Both lecturers wanted to be interviewed together and their collegiality is evident. The office colleagues disclosed "bouncing off" each other by reading each other's annotation. As my research notes suggest, the issue of fitness to practice were paramount to the lecturers' philosophy of nurse education and assessment feedback. The research notes identify the lecturers promoting the fundamentals of nursing because it is the "bedrock" of the profession. They mention the 6cs (DH, 2012), assessing attitudes that indicate safety and competence and reports of NHS failings as a driver to promote these principles. The lecturers had a keen sense of what it meant to demonstrate safe practice in an essay. The research grids identify lecturer 4 writing "WHAT!" to indicate poor practice on an essay in order to get the student's attention and for their irritation to be made clear. Lecturer 3 printed out essays as an *aid memoire*, which they found helpful when a student came to discuss their essay, the mark and annotation. This was useful especially when the student was resistant to the rationale given for any marks or annotation given. The issue of formative annotation was considered timely and a key point, because it was preferred over summative annotation feedback due to an increased motivation to change the essay content. I now present the research notes and grid following the interview with lecturer 5.

Lecturer 5

difficulties. It could be family background - hinder them, pre reg – they haven't got a family but they've got other problems. So all this, annotation – if a nursing tutor has thought about doing this, it is very helpful to a student. But I must say that it takes a lot of time, longer than a general comment.

- Q But in your view, is it a better type of feedback than a generic feedback sheet?
- A It is better, its quality, but I haven't compared the time I've saved though. Because if a student just reads your general comment, they will e mail you back. You still have to respond and read back the assignment to respond. So I've not compared the time saved if I don't do a proper one.
- Q Can it be improved to include annotation? Or is time the main factor?
- A Time is a main factor because in a mass production time is crucial and even if you tried to sell people the idea they would say that they know about it but whether they want to practice, they have a choice.
- Q Anything you want to ask or add before we finish?
- A Is there a way forward to promote good practice in this after your study is finished?
- Q What is interesting is that despite this university not having a formal annotation policy, they discourage it, but it is widespread. It's almost instinctive to add a comment on the page because that's what we do academically anyway when you're studying. You get an article or book and make a note of what's relevant and if it makes sense to you. The dialogue the word on the page, it seems to be natural but here it's under resourced and unacknowledged and that may need to be bought to the attention of the executive.

track chapes - use of Socratic Q' - what next THE is a big factor
 THE is a big factor
 Mass of FB in pre-reg. musch - FTP
 Mogether but happens anyway
 Nopotry but happens anyway
 Note helpful in larger escars 4k and above To Charled myrane - pour deque level - the quality OF - pour deque level - the quality OF - fue pulled of practice imprive too. 2 Wa - Newses First, lectures second suces a A vinant grage lecture content when Steidort Public view nuits as a practical puters acad. hid staffi hou jun - Good nuse (Poor essay (don't See Poor essay (Good nuse (G contracticts - Matechapes to harder the becknorer 242 Show then the charges pelate 5 FLENT FR

Model good structure for writup . 187 essay need nothing - description Studie (noteners / Syndax, Structule, assofil granmer, puedent - New to the So tellip on board advice from Reary - As a studat) focused in - not t - policy-conflict no - not allowedy to annotate Then alland to a gradenal latthetine Get open/black word doc/read essay/ nation notes as read but not a escary/ an the FB Shoot Notes as new on reced/andate / unte/feedback meddly set · (1 do too!) Novise 18+ / lecturer and unto as I know best acertu riga are doese modelly schere My issue trigger, Syntax ACTISAT = PNE. TSar. RMN = Qaahtarb Connierts VS &GN - B-O/ BM/ VBP! Backgrund affects annotat ptp outred .es Syntax stude R.W WATE as FB Read as a form of 1.1.5 annotation asgoalar reading /

206

Lecturer 5: 1 st order interpretation- initial stage	2 nd order interpretation- naïve stage
"I think it is important my annotation, whether formative or summative, mirrors good principles of constructive feedback"	Modelling behaviour, nurturing, attitude, care, rapport, read to write (RW) and inform practice.
"When I'm reading an essay I am assessing whether they are ethical and safe"	Structure, syntax (S) argument, understanding is evident in an essay. It can also relate to clinical practice.
"I can see from a poor essay, with a poor argument that they do not know the evidence to back up their actionsthere is a gap between what they can do, how they can write and joining the dots"	Evidence based practice should be seemless from essay to clinical practice
"I am looking for attitudes that would concern me, like writing about a patient as an object, or being dismissive, uncaring"	Fitness to practice (FTP). Good attitudes (A+), promoting individual care (pc), modelling professional principles (MB).
"I worry that we are reinforcing the wrong things now. It seems ok to do things online, communicate by email, even mark online and give feedbackthis reinforces that it is ok to be impersonal and making decisions when not face to face"	Reinforcing the use of technology may collude with person centredness (pc) and impersonal practises.
Key of abbreviations: Abbreviations in the pyramid, made up of semiotic triangles are: MB=modelling behaviour/ modify behaviour, A+= positive attitude, FTP=fitness to practice, R=rapport, S=syntax, RW=read to write, pc=person centred.	

6.5.4 Lecturer 5: Explanation of the research notes and grid

The research notes identify lecturer 5's use of Socratic questioning through annotation, such as when asking "what informs your next point?" The lecturer is trying to get the student to think critically and the annotator being a nurse first and a lecturer second is significant because of the fitness to practice assessment component of an essay content. In reality, this shows that their unease for patient care remains a priority. There is apprehension by the lecturer about the quality of writing, the student's capacity for critical thinking and its impact in clinical practice. The lecturer is concerned that annotation, whether given formatively or summatively, should reflect good principles of constructive feedback and person centred care. Therefore, the lecturer's

positive, consistent behaviour as an annotator is important because they reinforced the "right" nursing principles to the student. Reinforcing good feedback principles was necessary, whether given face to face or online. The lecturer reported an expectation to model professional behaviours to students, in relation to showing how to give good feedback. For example, good communication, developing a good rapport, having a positive attitude and being person centred are all relevant for the practice setting. I now present the research notes and grid following the interview with lecturer 6.

Lecturer 6

New to teaching analys Annotation can model fied Structure back to the Student " Give more feedback that was modelly analysis · Meaningful annatstorte / meaningful to anotato: and student · Keers in firence my annotation. · As a student I focussed as the negative fg - Cannor remainder most of student appenieros . Uni. regulations did State that we could not annotate but then charged their missely, · 80 even when NOT being allowed lectures Still did become it was natural, Nawing Can quie the annotation feedback to the Student. annotadia, . when not allowed) saw Jechness annotate but did not pass the FB to the Student ! - Andatia alla for intque FB / pandup iniquers ·) lask for person autral news in an easily Person Cathed vess + Positive altitude, negoticitien, not Judging, collegred and conficuence - Lectures have to model for students e-gattinde, bis, calabaatan . If ready badly worded as mensive createst then I may - F.P. * Parallel between chilical Practice and the essay - but "forght" is the education Setting

Leoner didit know if Mental health training Came out in annotation back to the student. But whey / how could it not? The feedback on M.H Y.2 Mussip back to students is discipline specific-So the techner Questicing this indicates a Autor of clarity in what they annotate , Autor is draw ald of reall * 'I draw all to of feedback from the way I keep records * for chrical practice and it informs have I mark." " youget abit cross with some students" you may think "No, whey have you written this?" Then She thought " NO , I've not understood this atall" An example I can think about 15 in an ethics essay and the patients mental capacity which I'm big on " "I'd thinks, nos wouldn't put Freet. Q. Josk would you put "Nol," anthe essay? No - as digital anotation / so save discurrect between timing write and foodback to suder? " feading on q P.C Scheen is a new Shirt - a feats formatting of essing smotries, lacks spontanents because you have to Click on so vary thisps and thinkip charges in the black of on ege . " · Lecturer wants Standordized feedback to students to anois grang mixed nessager - My thangut were what about the jouraltels of standardized care athe with it not theather people as being unique - Stenderdited = and not possible clave - Lecture, wants Carstering -(artext A 2r Misintepretation to interretation concisive. Find · However when one thinks of a consistency is Noi/ 22010 an ideal that waybe unrealized. e Jf a Suday has failed - aligital annitertain is ideal because the student can set west toggan and see the annelester + here it relates to the text. As it is explained and forme re-utited, I see the essence content as cifrednester Shudowi class - They Figet what fliery whole and so day so reading the annotation is important in higher of how if now reacted sense, has I now read and intervet.

(3) 1. 1.6 a trave the student now reads the assay charges in As me) and even the annotestion changes of the The words stay the same but their reschance and meaning Thayes in the - when ? They we read, thought, hved More Since waiting the essay. The & anuchetter worthing wood of the energy essentially comes aline Egain.) It may not be important what A said but what does it ween now? To ether of us. As time changes meaning, 80 the too does space where reading occurs matters, where with draws does too. This student with at have and headup their essay at have charges when A is need in a duffevent context, context is king. · Timie duryes thinkip, If J've annotated and 15 days later I masked to remember what I thought lid say) had little idea-· 36 J'm modelling has to deal with the <u>charging mouning</u> of bands and life. There is no Static mealing, I because meaning always dranges. " inflammade les statements rile me " to: examplé they dait have capacity be cause they have demotion". that thes me. I'd get quite wate; Assignments mucht · facts at figures, a retue attributes your things then about allows we to ases their (FTP) It's the need for pusar contrad care - This is an expressing Canise So altitude is all important." "Students need to be ever mind ful of Mainlayange " It gives you a sense of what they TIME charges practice thuir, haw they write and here they practice ! - This has an effect Essay - what how on neuse lectures. they with and practice

211

Lecturer 6: 1 st order interpretation- initial stage	2 nd order interpretation- naïve stage
"I print essays and then annotate. If students come after receiving their mark it is a useful aid memoire"	Promote good nursing standards but fought within the confines of HE and modelling professionalism
"As a student I focussed on the negatives not positives"	Frustrated by the tone of an essay content
"Riled by inflammatory statements, I get cross with the	
student. I think No! But wouldn't write it"	Promoting good attitude (A+), communication by demonstrating the same
"The language used to describe a dementia patient indicates a certain mind set which may be repeated in	Identify poor practice through the essay
clinical practice"	Modelling academic writing style
"I can also see cultural differences in writing sweeping statements"	Observed by imitating peers and referring back to own experience as a student
"To deal with the many themes we find in essays, for example short paragraphs and errors, writing annotation is important to model analysis, good structure, syntax, coherence and grammar"	
"Modelling occurs by engaging, listening, reading and watching peers work. Especially when new to higher education"	
Key of abbreviations: 6cs= (Compassion in practice,	
DH, 2012), FTP=fitness to practice, PC=person centeredness, A+=good attitude, FB↑=increases	
uniqueness of feedback, A=annotation, T=time lapsed.	

6.5.5 Lecturer 6: Explanation of the research notes and grid

The research notes identify a number of characteristics such as modelling behaviour to the student in the form of linguistic style, structure, meaningful feedback modelling analysis, and avoiding negative feedback. Lecturers also have to model other professional qualities such as the use of language, attitudes, care and the 6cs. The issue of technology affects essay feedback and brings depersonalisation into the equation. The annotator identifies the issue of time (see research note p.3) and how the student perceives the annotation and essay. In a spatial sense, the words stay the same but their resonance and meaning do not. This begs the question why and it may simply be that the annotator has lived, read, written and thought more since. The effect on

reducing individuality is an issue because technology inevitably devalues the quality of supervision due to the lack of human contact and communication. The annotator searches for fitness to practice issues and person centred care in an essay. The qualities they sought were things like having a positive attitude, being non-judgemental and valuing the individual. I now present the research notes and grid I made following the interview with lecturer 7.

. Prevenery told NOT to annotate as it could affect the Students Self Obern. . It is a chore to be told to do it but natural for reading "There are about of digital connecte we can use · Lower score commente? Only telling what is wrong and not saying what is night. · fligher Scove comments? "whent do I need to change?" FB Sandwitch "Inritation when have lots of annotation" Because if well dave - need to challenge throwing ... back of annotation comments are used loss, " TOO Surple" - the better the mark the horder 7 is to feedback as discussive and try to improve it sanchow. . HIW annotethen which more like speech - in spartareity. . If typey and you epit as you go along you have to funk and topled More-sout changes, . "Adding why? not sive if student under pood the use of the wood and why two withen". . "It level 7 writig stendard when I used the word why we met with defensioners? " Instead very "Please expand arthis point May be More positive I trandy ne comments that are generic and in a back of comments is depergenalised ". They say less because they are typed, other peoples wards and B tau DA gereinie "- " utrat are me rentorcomp have with rechnology of feedback?"

· Digital comments are in addition to the 1.0.7 technological developments in musiq f.2 Previously I'd privile of the occay and add hardwitten feedback auto it. That was fersond the Naw I'm replaced by a back of Connects! what does that havinfarce, that its ok to be hedravitic? She the past when priviting off the escay Sid paruse it touch it, take more true with it, and adding the gent to the payer. How it is TURDON R, login, find the black back module, login, find the cissignments, refer to the tet. identifying municer for anoninaly, chickarthe essay, lockarthe originality", add commants digitally of select from the bank of connents. Then old use the antilo feedback Section, add a number to the marking grid. It is dependentised, no cartout with anything other thoma keybacood and my momony of nuscip. ! Then when they read it they tog into the R, ete - read the comments and many come to see Ne to falk about the comments of mark -. It is at this point when Scan fuely come not my own and reinforce, model, coach, inspire and be personal and person contried y. Several times addrey Solo this and it is aform OFNILISM · Moral heresy - IF Juinte badly thostident May mittede My Style and thank of SOK V Sougard . (mproved writig = 4 thinking & 4 practice satured Handwith FB /Pc Depersond

Lecturer 7: 1 st order interpretation- initial stage	2 nd order interpretation- naïve stage
"Showing students how to write, whether a medical doctor, a paramedic or a nurse"	Modelling behaviour (MB)
"I think feedback is about nurturing, caring, positive role modelling. The developmental process starts from the acceptance of literature and growing a healthy scepticism"	Promoting critical thinking, professionalism, care and empathy through feedback (FB)
"Previously told not to annotate as purgative to the student- risked lowering self esteem"	Mixed messages from the institution regarding annotation
"Annotation comments are often anodyne and ambivalent"	What is the purpose of neutral comments if no action occurs?
"The whole process of online essays is onerous, switching on etc; ironically the student goes through the same process to receive the annotated feedback. What is obvious to me is the student may be overly defensive when reading those comments"	Technology changes the supervisory relationship and feedback online is depersonalised
<i>"I print all essays off and annotate and from them I produce the feedback sheet"</i>	Technology issues, reading digital comments online
"Essays online are different, difficult at times to read, sometimes distorted formatting. I'm unsure at times so come back to it at a later date"	Reading from a screen is an issue, sometimes distorted
"Depersonalise d experience of reading from the screen, but real when in one's hand"	Depersonalisation and technology
"Like to see good structure, direction, meeting learning outcomes, sign posts, safe practice"	Plot, flow, academic rigour and safe practice
"There's a moral heresy of sloppy writing generally means it won't be implemented in practice and patients will suffer"	Parallel between critical thinking in practice and an essay
Key of abbreviations: H/W=handwritten, FB=feedback, Di=digital annotation, D=depersonalised, Pc=person centred. The pyramid shapes made up of triangles indicates key themes.	
6.5.6 Lecturer 7: Explanation of the research notes and grid

The research notes identify the annotator's reference to the past when not being allowed to use handwritten annotation on an essay and send it back to a student for fear of causing upset. The reason for this was one of the HEI's at the time thought that annotation written on an essay would cause student distress, yet the lecturer continued to annotate on their printed off copy to aid the formulation of ideas. However, now in a digital form, annotation is considered acceptable. The research notes presented before the grid (p. 1, 2) identify the annotator using digital annotation and feeling like they were being "replaced" by a digital bank of comments which made feedback somehow sanitised, impersonal and generic. The annotator therefore asks the question "what are we reinforcing" with such a practice? The annotator concluded that a degree of depersonalisation is inherent in the contemporary assessment process, from online formative feedback, online submission, digital annotation and online summative feedback. From an ontological perspective, the lecturer believes technology lessened the amount of contact with a student, and the full potential for teaching and learning is inevitably reduced by the lack of human contact. This is of concern for nurse education because of the modelling of professional behaviour that is critical to the development of a student's own understanding of professional nursing behaviours. The parallel between the essay and clinical practice, in relation to critical thinking on the page, shapes student learning for clinical practice. I now present the last of the research notes and grid made following the interview with lecturer 8.

Lecturer 8

· See the wat and Say " what do you see here ? - Taylog to generate their under Dendup · May Storkethragh (May edit / May worke connent · Hence, Supportie partnership . The time of teckes to show, all question, water out the text is withely twe concurring but the student will learn to us note by modelhy 6 Eg keys A to B etc. Collegral approach to writing nonice to Ann. is about Moving pape along. · Ann i. is the context - It is then not seen in isological. . or by telephere - I mas numbered etc. · the essay becomes a tring thing , poesn't lite summadrie ann. tostate / non dynamic - Sharing Knowledge - hearing & dailograp from me. e lised as a teaching and shows nos (RAPPORT) - not destudio FB o Eg My superisor - She knows nos (RAPPORT) - not destudio FB Mutual understandig promotos leanop from become regisert - Mutual unaversion (worknosces / pisces / betreves in) (underfields Strengths / worknosces / pisces / ketreves in) - l'essenable + Professionel - No one explains the nythead barmer Supp. fa Anerthy of decc. to analysis - 4-5 · NO Mystery- Show theirs lesson Rapport centied Ax2 Showing (1:1 many 667 people days / procide water Mutud is a two way aundretter process under - tous to move from one with level to - If not show how to analyse - this PROACTIVE (2 way) proces can show Wenhue.ed fiestident & Shar fue (1) asthey donother by Sharp the best here to afficulate from of contract with 1.1.2

, Interal monacher to come into nuise educate of stendards of care through up a consup nuses / improve · Charlenge fixed prevention put · we can inspire minds, get to them first · Teach them to get their ninds" Intunquestion : Challege the lecture. " Say what you think" - you're sand this but meant that a what say a what worke o Ann. Poursfel tools me have but I have to bo before & driving to be panelfull - not after. It should be continuous • Hitand run - open PC, open up, write acrime connects'. "and you are frippall sons off in there"- IF marked 15, Sea 13 same gener pland suddenly your fore & grammer Changer" " you finerate anes at a this poor unknost idents' feedback and rentral T PB need to be ablentic & depensandized · Danger can be come hustful ; depersondised + out of coulert" . Faiting Andents - Lotsof 1:115 ? 312 heurs · Even if bodly with are they safe is the pierty Fur · Never personalize - ment neutral + del. NOT writing it for them, they are writing, I surplut. PROBE Q - what's within - understanding . Writig is like a darkart because no -one feadues yer 1.V.8

6.5.7 Lecturer 8: Explanation of the research notes and grid

Appendix 2 has a partial transcript of this interview and like student interview 2, I present research notes only in order to reduce the word count. The first order interpretation are the direct quotes and the second order interpretation the identification of key themes, which I present in the form of a codified key. For example, R=rapport, MB=modelling behaviour, SP=supportive partnership, PC=person centred, A2=enlivened by discussing its meaning and MU=mutual understanding.

Lecturer 8 viewed their role as a guiding hand in a student's writing development. The annotator talked about how they supervised using annotation to improve the student's essay if they had failed at the first attempt. Their style included sitting with a student, with the essay open on the computer, and referring to their annotation comments. The annotator used Socratic questioning such as "what do you see here?" in order to get the student to think differently and then write differently. This process, the annotator agreed, was labour intensive but a kind of practice that develops the student's self-sufficiency and confidence as a writer to develop new skills to write independently. The annotator acknowledged from a modelling perspective, it was well worth the effort, because the student then went away with a quality experience they were likely to learn from and use again. Therefore, annotation was a process of using one to one experience to its fullest potential, to ensure a supportive partnership was in place for imitation and modelling. The annotator had been motivated to come into nurse education to make a difference, improve standards and "...*teach* (students)... *to open their minds*..." to model behaviours considered self-perpetuating and professional.

Lecturer 8 identified receiving timely formative annotation to be an issue because it is a developing stage of the writing where the student is encouraged to think critically and in a dynamic way. This timeliness contrasts with summative annotation feedback and meeting afterwards to correct a failed essay and to correct errors. Instead, annotation is considered most powerful when continuous, timely and received before an essay is marked. In this manner, the "writing and thinking process" can be analysed rather than forgotten. This issue reminded me of the theme of rhetoric for chapter seven (mimesis1-3) and reflective unconsciousness in chapter ten.

6.6 First and second order interpretation from digital annotation extracts (n=50)

In this section I present a selection of annotation extracts (reduced from 130 pages of research extracts) from 50 chosen essays found to have annotation on them with figure 17 entitled *Extracts from researcher notes of digital annotation*.

Digital annotation extracts(n=50)		
1 st order interpretation- initial stage	2 nd order interpretation- naïve stage	
Extract 1: The annotation suggests improving clarity in the essay content. Evidence based writing was well used and impacted on care. The word "it" starting a sentence was of concern to the annotator because of not being specific	The student is attempting to interpret the concepts they have read about and identifying their dialectical position	
Extract 2: Pink highlighted text related to evidence based practice. The highlighted "they" in green when opened up said "who are they?" The use of prescriptive and objective language is de-personalised	Evidence based practice, promoting person centred care, and discouraging the use of depersonalised language in an essay	
Extract 3: The extract indicates the marker was irritated by the lack of referencing which they linked directly to poor clinical practice	Reinforcing the need for evidence based practice, and its link with poor clinical practice	

Evidence based practice incorporates the best researched evidence with clinical expertise and patients' needs and values (Burns & Grove, 2011). It is seen to be the gold standard for the delivery of safe and compassionate healthcare (Brown, Wickline, Ecoff & Glaser, 2009). Best evidence based practice will be used to implement this by utilising a literature search. Burn & Grove (2011) state research in nursing is vital to develop and enhance the knowledge nurses use to improve clinical practice. It is used to promote positive outcomes for patients and families, lowering health care costs, regardless of this it is not being constantly used (Wallis,

2012). Different 💭 practice, new ev detriments includ needs and prefe



lementation will be successful as best implement themselves, these ent and staff attitude and behaviour, rol ,Wensing, Eccles & Davis, 2013). vas found that lack of time and

Despite a number of awareness campaigns and the acknowledgement of alcohol and its effect on the body and the individual's health, research has shown that the number of adults being cared for in the health care provisions is increasing. This piece of work is going to be looking at the affect that alcohol has on the body. The main target audience that my work is going to be based on is adults and the way in which alcohol affects them. I will be using a range of sources to help support my arguments.

Evidence-based

Evidence required to sup

Evidence based practice is extremely important within nursing and the health care provisions. When working with patients, it is part of your job role to help make decisions for them if you feel that will benefit their health and there state of mind. Nurses are trained to Evidence based practice is used to improve the quality of care for the patient by using individual experience and external clinical evidence. They use the evidence and experience combined together to help make appropriate decisions for the patient, taking into concept their beliefs and values (Larrabee, 2009). Theory, research and practice are bound together to produce evidence used to support practice within the health care sectors. Research will be completed by professionals against a certain topic, theory or activity, and once the research is complete it can then be brought forward and used as evidence to support cases relating to the topic, theory or activity (George, 2010). Reflection plays a key part in evidence based practice because your experience is used to help strengthen the use of evidence based practice. In order to teach an individual how to practice in a safe manor, you must also have had the experience (Bulman&Schutz, 2008).

The nurse can only try and explain the outcome, side effects and somehow persuade but not force the patient to reconsider their decision. In relation to the moral dilemma chosen, the issue was that the patient refused treatment for the UTI but was deemed to not have the mental capacity to make an informed decision, so therefore the MDT then had to consider the patients best interests and had to go against the patient's wishes, thus the patients right to autonomy was overruled by beneficence and in regards to consequentialism, the consequence of the action of giving the antibiotics benefited the patient by treating his UTI.

Lastly, the NMC (2009) imp ensure that good record keeping a improve accountability, to make su to the patients care, the delivery of clinical decisions. Good and clear i and complaints or legal processes.



6.6.1 Explanation of digital annotation research notes and grid

The digital annotation research extracts 1 to 3 identify three key findings. Research extract 1 identifies the annotator taking exception to the word "it" starting a sentence. This was an entry point in the three step model allowing me to later identify the hermeneutic self as a theme using Ricoeur's textual hermeneutics. The initial interpretation of this extract indicates that the student, when attempting to make sense of discourse does so by making a statement about their dialectical posture. In essence, I felt I was viewing the student's own engagement in relation to making sense of an-other's work which involves weighing up the relevance of ideas for the writing task ahead. The research extract 2 identifies the annotator's concern for the student using language, which depersonalised patients as "they" (highlighted in green). The annotator therefore

was reinforcing the ethico-moral nursing principle of person centred care, which links directly to making judgements with known benefits and risks. The research extract 3 identifies the issue noted in the lecturer interviews, and that is the need for evidence-based practice in an essay and its absence may indicate a less than rigorous approach to their nursing studies. In short, the judgements that the student makes unsupported by the literature, may adversely affect patients in clinical practice if there are interventions unsupported by efficacy.

6.7 First and second order interpretation from handwritten annotation extracts

The handwritten annotation research extracts totalled 136 pages and I reduced it to 41 pages before identifying three handwritten annotation extracts below in figure 18 below entitled *Extracts from researcher notes of handwritten annotation*.

Handwritten annotation extracts (n=50)		
1 st order interpretation- initial stage	2 nd order interpretation- naïve stage	
Extract 1: 10-month-old "James." Use of ticks, underlining, deleted words, corrections, additional words, questions in the margins and in the spaces of the text	Professional values are promoted, such as person centred care, a need for objectivity, evidence based care, questioning the clarity of discourse	
Extract 2: "What?" question refers consistently to the word "it" starting a sentence. The "what" question indicated the annotator was irritated by the lack of clarity about what "it" relates to? Plot, structure, clarity of discourse indicated by the question "by whom?"	The research extract refers to the student's understanding of the literature they cite	
Extract 3: Arrows drawn to suggest linking the paragraphs together, grammar changes, spelling errors, lengthening the text for more detail, challenging statements made e.g. "meaning what?" Identifying assumptions and asking what is their meaning? Starting a sentence with the word "it" is interesting and I wondered what it could mean, in "all" of its possibilities?	As above, the extract identifies the student at a naïve stage of interpretation indicated by the lack of clarity of discourse. The student in the extract is writing about an experience in a personal way	

Figure 18: Extracts from researcher notes of handwritten annotation

James is a 10 month old baby who was admitted with a fractured left femur. The information that was given to the doctors was that there was no history of any trauma. James was placed in his baby walker and he refused to place his JAMES INSTITE left leg on the floor. This was 2 days previous but mum felt that it was not serious so did not seek medical advice. Mum then reported James continued to refuse to use his leg and it began to swell. Mum took James' sister to their local GP to ask advice about head lice treatment. The GP was checking to see if James had head lice when he noticed the swollen leg. The GP advised Take thus mum to attend accident and emergency which mum did. Concerns were raised when mum was concerned regarding the head lice but not that James' leg was swelling for no apparent reason. - OK - NOW appy what ye have mum to Attend accident reason. - OK - NOW appy what ye have mum to mum way by the way of the mum in the strend accident is the swollen is but not that James' leg was swelling for no apparent reason. - OK - NOW appy what ye have mum to mum to attend accident reason. - OK - NOW appy what ye have mum to mum the apparent reason. - OK - NOW appy what ye have

Extract 2

This assignment will define and describe what assessment is and now important it is ouring the nursing process. This assignment will highlight the different sources of information gathered during the assessment process. How it helps nurses during the nursing process. The assignment will identify and the describe Early Warning Scoring system and how important it is at identifying patient needs during the nursing process. It will describe how nurses work with their patients to develop a positive relationship during their time at hospital. Assessment is important in nursing care as it allows the nurse to gather information about a patient's health, identifies health needs, strengths and weakness of the patient. It allows the nurse to identify problem's and treat them to the best of her or his ability (medical dictionary 2012).

It is during the implementation of the surgical safety checklist that the incident occurred. \sim

A child presented in the anaesthetic room with their mother for corrective orthopaedic surgery for a chronic disability. The child was therefore familiar to the surgeon and to the anaesthetist as this was not their first bout of surgery and the anaesthetist did this surgical list on a weekly basis. be more speare about surgery and distant

During the preoperative team brief the surgeon did in fact mention that the child had missed a couple of outpatient appointments and that the parents had questioned whether the operation was completely necessary even after being informed of the high success rate for this procedure It was also mentioned the mother did not appear to be very educated and the staff in the anaesthetic room should take this into account when conversing with her. Meany Mat ? IS fluts a Sudgemental was fully statement?

Whilst in the anaesthetic room the child appeared to be distressed as is often the case. The f factors anaesthetic room is considered a strange, unusual even scary setting to a place where the child finds him/herself under unusual circumstances. They are surrounded by at least three strangers, dressed in strange clothes. The parent or caregiver although present does not what appear to the child to be in control of the situation invoking fear in the child. If you persend the child to be in control of the situation invoking fear in the child. If you persend the child to be in control of the situation invoking fear in the child.

It is at this point that it was noted that both child and mother appeared detached from each other. The mother did not make any attempt to comfort the child either verbally or physically. The the the Neither did the child reach out to or plead to the mother for some form of comfort. To What are yes may have the to the mother for some form of comfort. To Mangetrin When checking the surgical site the child's body is partially exposed to reveal that the surgical site is marked and that is the correct surgical site. The initial check revealed that the

child appeared to be malnourished and unkempt. During a more detailed examination when the child was eventually anaesthetised and the mother had been informed of the concerns raised.

6.7.1 Explanation of handwritten annotation research notes and grid

The research extracts 1 to 3 identifies key words of concern to the annotator. Research extract 1 aims to improve the clarity of discourse and the gaps of the text. The colloquial use of terms such as "mum" ensure the extract is personal and subjective, and a lack of evidence is noted by the annotation "to whom, by who, who with?" The aim of the annotation therefore challenges the student to identify what they have left out, with the possibility they thought there was enough information in the essay section. In effect, the annotator is triggering memory recall, interpretation of events and any evidence considered relevant.

Research extracts 2 and 3 first order interpretation identified the "it" word starting a sentence to be a concern for the annotator. From a mass of research data, I collected prior to choosing the three research extracts, it was clear to me that some annotators did not mention this phenomenon yet others were irritated enough to comment on its frequency and lack of clarity. From my perspective, I could understand the student's use of the word "it" when starting a sentence and in effect this directed me to wonder about the personal nature of interpretation, and reflect on the visualisation of the student "holding" concepts, ideas and judgements about the text in their mind, which they then externalised onto the essay page.

6.8 Second order interpretation and sub-themes

In the next section I present the key words from all of the second order interpretation columns and research notes and summarise them in table 3 entitled *Figures 14 to 18 collated in three steps*.

Table 3: Figures 14 to 18 collated in three steps

Figures 14 to 18	Sub-themes	Identified themes (see figure 19)
Self-awareness, reflection, mis/understanding, mis/interpretation,	Terms, concepts of "an-other"	The hermeneutic self
insight, instinct, reasoning, knowledge, pre-conceived ideas (bias)	(text) and the object of thought,	
when reading, the word "it" starting a sentence, interpret through	represented by the word "it"	
own embodied experience of nursing, naivity, leaping ahead to		
grasp meaning, intent, making decisions, judgements	Hermeneutic circle at play	
	Embodiment, depersonalisation	
Annotation/ supervision models professional behaviours such as	Plot, imitation of style, cadence,	=Rhetoric
clarity, Socratic questioning, praise, style, structure, coherence	proof, convincing, character,	
(good or bad), critical thinking promoted, correcting essay	action, clarity, persuasive	
content, grammar, spelling- all model, parasitical annotation	discourse	
affects the host text, handwritten annotation more communicative		
(mood, colour, look for changes in style), promoting evidence	Theory of argumentation	
based practice, clarifying meaning, evidence		

Grasping meaning, recognition, identification, misunderstanding	Use of metaphor, visual imagery	
leads to understanding, mis-interpretation leads to clarity,		
modelling good feedback, observing professionalism, action,	Observation and imitation of	
imitation of professionalism, demonstrative, rapport, writing style,	professional behaviours	
how annotation looks is significant (e.g. circles, arrows,		
underlining of text, use of colour), imitation of normal	Productive imagination	
conventions of essay/literary composition, visual imagery,		
character and action (mimesis), annotation is learnt through		
imitation/experience, nurse first-lecturer second		
Understanding changes over time, spatiality, from initial to the	Temporality, distanciation	
latest realisation in the moment, hermeneutic circle, glass of water		
analogy, formative annotation is better placed temporally than		
summative, action on annotation feedback is unknown		
Diagnostic assessment, facilitatory, person centred care, the self,	Individuality, ethico-moral	=Individualism and
nurturing, fitness to practice assessed in an essay, poor attitude,	practice	being "just"
identifying needs, knowledge and competencies, e.g. 6cs, Francis		(fairness, equity,
report, political policy versus reality, reports of NHS failings,	Mutuality	justice)
maintaining professional quality and standards, the essay is a safe		
space for experimentation, face to face supervision is a supportive	Justice, social contract	
partnership, facilitating a positive use of annotation promotes care,		
empathy, morals, social norms, use of praise, constructive	Empowerment	
feedback, promoting individual and collective identity, integrity,		

accountability, rapport, warmth, valuing the student,		
empowerment, promoting confidence, critical thinking, revisions,		
societal ethic-moral principles, individual participation, manners,		
trust, self-interpretation, nurse first-lecturer second, power,		
authorised permission of nursing, use of technology risks		
depersonalisation, objectified language also depersonalises, less		
face to face contact with students, annotation aids learning and		
memory, amount of annotation indicates concern or not,		
mutuality, fitness to practice, competence		
Tone, projection, parasitic annotation, emotion, patronising,	Transference, abuse, use of	=Reflective
expresses mood, anxiety, frustration, disappointment,	memory has a number of	consciousness
psychosomatic impact, evoked feelings, attitude, collusion, and	contingencies e.g. what and why	
projection. Disempowerment, self-esteem, feel like a child again.	something is remembered	Slippage and
Neutral comments do not inspire change. The past can return to		transference
bring pain, authentic, inauthenticity, truth of recall, what is left	Perception	
unsaid projects something, conscious and unconscious reflection,		
memory recall changes in time, perception and purpose,	Anxiety	
intentionality, negativity (meant or perceived) is unconstructive,		
devalues and affects self-esteem, need positive strokes		





6.9 Conclusion

This chapter presents a selection of research extracts from each research method: a focus group interview, individual interviews with students and lecturers (see table 2). From 100 essays, half with digital annotation extracts and the other half with handwritten annotation were selected. The chapter organises the emerging research findings using my research notes, grids with first and second order columns and coded key words followed by explanatory comments. These grids followed van Manen's (1997) three stage model of interpretation of data from the initial experience of the research data, to a more nuanced yet still naïve interpretation of the research data and lastly, to the stage of interpretation where sub themes (see table 3) could be identified before grouping them into four themes (see figure 19). A surprising finding was identifying that annotation has organising principles which are temporally bound to the repetitive act of reading, writing and the whole process is contingent on the meaning text has for the student and annotator at any given time. Both read and interpreted text depending on the professional context, self knowledge, understanding of the lived experience and experiential plot lines that surface when writing. Finally, four themes were identified in figure 19, and they are the "hermeneutic self", "rhetoric," "individualism" and finally, the "reflective unconscious and slippage." The next four chapters present each of the four identified research themes.

Chapter Seven

The hermeneutic self



7.1 Introduction

In this chapter I explore the theme of the hermeneutic self. In particular, I suggest the theme is revealed by a specific writing style found within the research data, namely; a student starting a sentence (or paragraph) with the word "it." If discourse is the intention to bring something "out into the open" (Ricoeur, 1976), what is revealed, is a student's continued subjectivity (see chapter 4.8 and 4.8.1). Due to the process of temporal understanding, subjectivity remains an issue until the literature is better understood and as the various research extracts demonstrate, if there is a lack of clarity noted in the essay by the annotator, this likely mirrors the student's own misunderstanding (see chapter 4.7.1). In order to explore the hermeneutic self and the word "it" starting a sentence, I define both terms, discuss subject-object relations, student embodiment, ontological discourse, recognition (see chapter 4.7.2) and the self within the hermeneutic circle (see chapter 4.8). The first research extract below identifies a frustrated annotator suggesting this writing style is "sh(it):"

Evidence based practice incorporates the best researched evidence with clinical expertise and patients' needs and values (Burns & Grove, 2011). It is seen to be the gold standard for the delivery of safe and compassionate healthcare (Brown, Wickline, Ecoff & Glaser, 2009). Best evidence based practice will be used to implement this by utilising a literature search. Burn & Grove (2011) state research in nursing is vital to develop and enhance the knowledge nurses use to improve clinical practice. It is used to promote positive outcomes for patients and families, lowering health care costs, regardless of this it is not being constantly used (Wallis,

2012). Different practice, new ev detriments incluc needs and prefe Wallis (2012), sta

of th of th Som	I content and use of evidence however the use e word "it" starting a sentence reduces the flow e essay and should be revised to write better. sone once wrote when starting a sentence "it es with?!! for essay writing!
<u>Save</u>	Associate a criterion

lementation will be successful as best
 implement themselves, these
 ent and staff attitude and behaviour,
 rol ,Wensing, Eccles & Davis, 2013).
 was found that lack of time and

Good content and use of evidence however the use of the word "it" starting a sentence reduces the flow of the essay and should be revised to write better. Someone once wrote "it rhymes with?!! and for academic writing should be avoided when starting a sentence!

What is noticeable about the "it rhymes with?" research extract, is the frequency of starting a sentence with the word "it" and the annotator's apparent scatological reference. I say apparent, because the assumption by the reader is arrived at quickly to avoid rhyming "it" with other words like "fit," or "bit," and due to linguistic cues given, the reader reaches a timely conclusion. The annotator therefore provides the student with a word substitution, or to put it another way, one word that can be replaced by another. What is certain is the recognition of words aims to negotiate the existence of many possible meanings for a reader, without resulting in textual chaos. The research extract reveals something about the annotator and the student, which I discuss later.

偂

I was surprised that an annotator had been so candid, because the comments had the potential to cause upset to the student, and if a complaint had resulted, upset for the annotator too. When stating "it rhymes with," their humour is indicated when the annotator stated "...someone once wrote..." to support its apparent humorous intent, and their use of rhyming is a literary tradition that allows for representations of themes, memorising and sharing stories, songs and poems (Rubin, 1995). In rhyming, whether knowingly or not, the annotator is using plot and the cadence of the rhyme to draw a mental image for the student to remember. So the next time, when precipitously writing "it" at the start of the sentence, the student may think twice. Whatever one makes of the "it rhymes with?" research extract, not

many annotators would have dared to cross this professional boundary and so I will discuss it later in the chapter. I now describe how the theme was identified.

7.2 Identifying the theme

My research notes (see appendix three) identify a number of theoretical and experiential issues related to the word "it" starting a sentence, which was an entry point to the theme of the hermeneutic self. I thought the positioning of the word at the beginning of the sentence was the theme for a long time and started by identifying my assumptions such as: "grasping" the meaning of literature, the subject (reader) and the object referring to the literature. I also wrote about embodiment (see chapter 8.4.1), pre or unconscious processes informing thinking, and the word "it" starting a sentence somehow referring to the student remaining present on the page of their essay.

I identified the hermeneutic self as a theme in the handwritten and digital research extracts in chapter 6.6 and 6.7, leading me to collate key words in table 3 (chapter 6.8). The theme emerged when identifying the possibility that students use of the word "it" starting a sentence was a manifestation of their hermeneutic self in action within *their own* hermeneutic circle. The hermeneutic self refers to an individual's own interpretive understanding (explanation, moral conscience, action, temporal understanding, ontology, identity, recognition) within the hermeneutic circle through recognition, which I discussed in chapter 4.7 to 4.7.2 (Ricoeur, 1994). I realised I had not mentioned the hermeneutic self in chapter four, and its emergence as a theme indicated a gap in my understanding of the hermeneutic circle's central characteristic, the interpreting reader. Perhaps, part of the reason for missing this point was that I too was experiencing the same and had not identified fully with other students'

experiences until the theme emerged. I now define the word "it" before exploring the significance of the hermeneutic self theme further.

7.3 Defining "it"

A definition of the word "it" was illuminating because it reinforces the reader "grasping" meaning, whilst not wholly understanding or even being able to define its meaning. The Oxford English Dictionary (OED), under the outline entry, pronoun, adjective and noun, give a comprehensive lexical definition of the word "it" and the following definitions here and in section 7.7 are relevant to the theme. First, the word "it" according to the OED is a pronoun and relates to:

"The subjective and objective case of the third person... (the) I. Subjective uses. 1. The thing previously mentioned, implied, or easily identified...as subject or subject complement..." (1a).

The "thing" or object "appears" through the senses before it can become objective, is a reminder of the purpose of phenomenology and its referent terminology (see my research notes, appendix 3, p.2). Ricoeur (1967; 1999) suggests phenomenology is the science of "appearance" of things, and perception and in this sense the word "it" is a reference to an "...*abstract thing, or a matter expressed or implied in a statement, or occupying the attention of the speaker*..." (OED, 1b). The word "it" can also be used as a humorous reference to a person (OED, 1c) which we saw (or not) in the "it rhymes with?" research extract. The various definitions so far identify a number of key words of relevance to the self-referential theme of reading, such as: the self, reflection, an abstract thing, attention and subject-object relations. The definitions are useful because the word "it" in a

phenomenological sense, underlines its use as a reference point to self-understanding which is the starting point in identifying pre-conceived ideas in the hermeneutic circle. Once identified as an entry point to the hermeneutic self, my research notes (see appendix 3) directed to a large extent, the remainder of the chapter discussion. For example, the reaction to the word "it" by the annotator was perplexing and the entry point to the hermeneutic self as a theme was the notion of the "thing" being read and an object the reader thinks about. When reading, the ideas of an-other, they are in reality, not the reader's own initiated thoughts and whilst figuring out their relevance, the ideas metaphorically speaking float in and out of conscious awareness to ensure when being cited in an essay, the student continues to have a preconscious presence on the page. Ricoeur's (1976) notion of the interpreting self involves grasping the meaning of discourse, as it moves beyond the structure of the sign, symbol and language to refer back to the speaker. This is central to the mimetic stages of temporal understanding which I discuss in chapter 8.5 to 8.8 and the crux of this theme corresponds to "how" something is perceived, signifies what is conceptualised (Ricoeur, 2006). The next research extract entitled "x3 its" should therefore be viewed as a diagnostic opportunity, rather than be considered a problematic writing style.

7.4 The annotator

The research extract entitled "x3 its" identifies annotator irritation and the negative visual impact of annotation on essay content. The annotator's comments in "x3 its" when starting a sentence indicates a degree of frustration when used three times in such a short paragraph. The apparent lack of clarity found in the essay is repeated back to the student who, judging by the flow of the essay content and annotation comments such as "blunted" and "stop/start," may not have understood what they were writing about at the time:

may then comply with the request of the nurse and work with the decision to improve their health and avoid serious disease Anorexia is also a serious illness and can lead to fragile bones, stopping of periods and constipation. It also can have an effect on the mind ad can cause depression. This is why it is very important to screen all patients with MUST. It indicates that patients are at risk and are in need of health care professional intervention. The NICE guidelines indicate that healthcare professionals should screen all patients on admission to hospital, be fully trained in how to deal with malnutrition, offer patients who are malnourished sips of water and snacks at regular intervals. It also states that patients need to be aware of their treatment and offer support at all times including support to their relatives. (Ref.) Too Many Sentences Stoph with the usod it e.g. With above. You Second the work the patient of the patient of the patient we above. You Second the work the patient of the patient we also states have a sector work you need to enpatient we also sector needs as "bluetted" and war for all times including the above. You Second the offer and the patient of the patient of the patient we also sector needs as "bluetted" and war for all the patient.

Reading the annotation comments is instructive. The annotator, when stating "...too many sentences starting with the word it..." and the student "...wrestling with your need..." sounds frustrated by the student's apparent inability to empathise with people suffering from depression or anorexia. The annotator suggests the word "it" in this position, lacks clarity, which is a fair point to make because the student appears to be subjective, or empathise with the patient's situation. The research extract identifies issues of objectified language, such as "comply," "co-operate," and perhaps the student is struggling to make sense of NICE guidelines and judging the relevance of academic terms and concepts to the writing task. The research extract also highlights that a sentence beginning with "it" is problematic for the annotator and their way of dealing with this is to highlight the frequency of the "it" word and questioning "...comply...what happens if they don't... how so...?" The annotator is frustrated by the essay tone, which is clipped and unemotional, and there are glimpses of lay perspectives of the yet untrained nurse. For example, when stating the essay content has an "...effect on the mind..." and a lay reference to "...stopping of periods..." rather than a medical term such as amenorrhoea, the naïve stage of the student's writing is illuminated.

Add to this naivety, the absence of any supporting evidence in the research extract, and it indicates the student's incomplete understanding of the literature, and early stage of development. When the annotator suggests the student is "wrestling" with the subjective word "it" coupled with a lack of specificity in the essay content, it is perhaps understandable that a degree of irritation from the annotator creeps in, because the student appears to know very little. Therefore, for me, this research extract reinforces the hermeneutic self is present when the student writes "it" when starting a sentence and a naive stage of temporal understanding (see chapter 8.6 to 8.8).

7.5 Temporal understanding

The research extracts "it rhymes with?" and "x3 its" ensure the annotator's purpose is to fill in the gaps of the essay for the student through a number of annotation styles such as circling, slashes, numerals, curly brackets, underlining, text. There is first, an attempt to improve the clarity of the writing process for the reader and second, lead the student to re-think their writing style for the next time. The clarity of annotation needs to be as clear as possible and this depends on reflective insight and realising that the dialectical posture of the student is somehow evident (Ricoeur, 1976). The annotator, may assume the student would understand their comments, which means when writing about a subjective experience, the words and phrases used subconsciously reveal a viewpoint that may not have been fully articulated before, until the moment of sharing it with others. From the work of Raivaisson's traits of character, Ricoeur (1994) refers to the temporal and overlapping dimension of character thematised into two distinct selves. These are a "…*set of lasting dispositions by which a person is recognised*…" (Ricoeur, p. 121) and what Ricoeur calls the problematic of the "ipse" or "who" and "idem" or "what" formed out of habit, and in a constant state of flux. This may refer to the student spontaneously processing ideas in their mind and trying to get inside the mind of the author. Discourse as a phenomenon, may also ensure the reader accepts new meaning without acknowledging what has led to the change (Ricoeur, 1976) because a priori thinking based on theory and knowledge involves a kind of inductive reasoning. Ricoeur suggests the possibility of not thinking deeply enough before writing is a temporal phenomenon, which I discuss further in chapter eight (see sections 8.6 to 8.8) and eleven. In discourse, this process is cyclical, in the sense that having a lack of time to think is linked to a lack of self-understanding up to that temporal point of interpretation (Ricoeur, 1976). For example, when writing down a latest realisation and returning to read it later, the written word is judged by interpretation, which will often lead to more revisions and clarity. This appears to be happening in the hermeneutic process I am referring to. Ricoeur (1976) suggests this subject-object tension refers to the notion of pre-conceived ideas that inform all experiences of language when listening, reading or speaking to others. Therefore, in order to avoid making mistakes, the reader requires the "time" to reflect on their interpretation, before communicating those ideas to an-other reader. Time is therefore an aspect of the hermeneutic self because it takes time for the reader to identify pre-conceived ideas that may otherwise affect their objectivity (Ricoeur, 1976). In contrast, what is learnt through the empirical senses determines how discourse is interpreted through feelings (Marcel, 1965; Ricoeur, 1976). In my view, the "it rhymes with?" research extract is an example of a posteriori learning, and the corporeal aspects of subjective experience (Ricoeur, 1976) which I discuss next.

7.6 The student and embodiment

The word "it," I found, relates to the verb "to be" (OED, 2) under the OED outline entry, entitled pronoun, adjective and noun, and defined as attention to "...the person or thing in question...statements or questions regarding identity..." (OED, 2a). The definition "to be"

has an ontological place in understanding the hermeneutic self and the work of an-other. Notably, for academic purposes and the hermeneutic self when "...quoting from books and other written sources; as it says, it tells...usually expressed by the passive "it is" said, it is written..." (OED, 3f). This process is a reference to the third person and the "...reflexive use of it..." (OED, 6) literally speaking, occurs when the student reader assimilates the ideas of an-other with their own (Marcel, 1965). A definition of the word "self" is useful here because it indicates unequivocally that the reference to the person or thing mentioned is not merely, to an-other (OED, A, n. d) but also to the interpreting self. The next research extract entitled "neglect" identifies the hermeneutic self, embodiment and caring for a child suspected of physical neglect by its mother:

רבחבית הסתפת ית Which Were? It was concluded that the child was suffering with signs of physical neglect. Neglect is defined by the absence of provision for childs basic needs (Gough 2005) And also And X in relation to this other forms of neglect possibly emotional neglect based upon the child's interaction in the anaesthetic room prior to being anaesthetised. - he Specific peed to give nove detail " they's It is a worrying that during the experience I found the child's physical appearance very distressing and found myself slightly disturbed and disgusted by the behaviour of the mother to her own child/It is therefore evident that at this point I was finding it very difficult to remain objective and I think it showed. bands - need to be ver Judgemental and Denoticual Overates track - has ? Model? A more in-depth assessment of the child was needed than given to possibly determine whether there was a deliberate omission of factors beyond the parents control. Because of this there could be inconsistencies made when assessing

a good parent (Howarth 2005) of differences of opinion and judgement of what makes 50 what will five more less of different grania and Judgeweit? The staff nurse in the research extract "neglect" is writing about the possibility of emotional and physical neglect of a child by the mother and appears to find it difficult to remain objective. The staff nurse focuses on embodiment which recognises the mutuality of sensory perception. Indeed, the staff nurse refers to themselves in the first person and uses emotive words such as "disgusted" and "disturbed." In response, the annotator advises the student to refrain from using emotive language when writing and so leaves the student little room for being-authentic. In contrast to the student's concern, the annotator's responses are abrupt, superficial and filled with parallel emotion, noticeable by the visual impact of the extract itself and the annotator when writing "…*need to be non-judgemental*…" ironically does not take their own advice. They do not however acknowledge the issue of ontological discourse evident in the essay extract.

7.7 The student self in an essay

The staff nurse's comments about feeling "disgusted" and "disturbed" in the "neglect" extract is an example of ontological discourse which becomes realised not in thought or emotion but in the consuming ethics of care, of existential guilt and dread (Ricoeur, 1969). Care, in relation to doing what is right and mutually beneficial, which I write about in chapter nine (see chapter 9.4). The staff nurse's reflective awareness is important because they challenge the care they gave and acknowledge their profound feelings about the experience. However, whilst empathy is evident in "neglect," the use of supporting literature and understanding its relevance is not, and the staff nurse appears to doubt themselves which may explain their difficulty in remaining objective. The research extract below entitled "Ben" also demonstrates ontological discourse in the repetition of the word "it":

6+03 fleese use more recent Supportup perceture Very gastric emptying (McDonald et al 1991). Ben's naso-gastric feed was stopped four hours nil by mouth prior to going top theatre, children have to be starved prior to theatre to reduce the risk of So teret isk of vomiting in theatre to minimise the risk of vomiting in theatre when ventilated. Ben was ventilated in theatre when he was under general anaesthetic to protect his airway-and wash his burns s and apply a dressing. It was easier and less painful to dress his burns in theatre and kinder to him as Children can be very traumatised and upset during a dressing change. (Reference?) potential complication of an anaestuctic is as informed concert deterned? By whom? what Principles are noche Ben was nursed in a cubicle for isolation purposes, to minimise the spread of infection to himself (Wilson, 1992). It also reduces risk of infection MRSA, Streptococcus and Pseudomonas. Members of staff are expected to use the seven stage hand washing technique National patient safety agency 2004.) It is a recognised national guidelines which reduces the risk of cross-infection. A review up date is due in September 2011. Link nurses are designated on the ward to attend to dates and circulate important information to the rest of the staff (Infection control nurses association 1997). Descriptive and an ecdotal text above. Endence in prectice Should be reinforced in an ecsay to wippose Ben's outcome.

As the research extracts "neglect" and "Ben" suggest humans *visualise* an object, before we can grasp "it" in our mind, and what is grasped visually by both staff nurses was the "look" of the neglected child in "neglect" and "Ben's" third degree burns. What may have been visualised in a pre-conceived way was the dread of experiencing such an event in the first place because of the emotion it triggers. This relates to other memorised images, such as in "neglect," the notion of motherhood, the wounded child or dressing a traumatised Ben's wounds when awake. Ricoeur's (1994) theory of recognition accepts that what is visualised is considered to be true, because the phenomenon comes first, then its relationship to the signifier, the subject, comes second (Ricoeur, 1977). In this sense, the staff nurse's

experience of Ben's pain came first, and the writing of the essay came much later. Despite coming second, the emotional effect of writing about the experience was still clearly evident, and so the student's internalised emotion present on the page of the essay. Therefore, temporal processes linking thoughts to action (Marcel, 1949; Ricoeur, 1999), such as what happened when "Ben's" pain was lessened when anaesthetised, means the staff nurse's subjective feelings will eventually subside. Until that point is reached, the staff nurse is left judging what literature is relevant to include in the writing task, whilst dealing with their unprocessed, subjective emotions. In relation to both research extracts, the hermeneutic self, which I suggest is present in the use of the word "it" when starting a sentence, is a reference point to understanding sensory experience preceding the writing act. In Being and having (1949), Marcel suggests the two poles of sensory experience oscillate between "being" and "having" meaning being conscious of oneself as a thinking person and having a conscious awareness of one's own embodiment. In this sense, Marcel was referring to the primordial predisposition a person has for an-other which is evident in the staff nurses' motivation to care for others. Both the "neglect" and "Ben" research extracts identify the possibility of writing to improve the existential outcomes of care and when writing an essay, they are attempting to move beyond the subjective to write objectively. However, when the staff nurse (as a post registration student) is still processing their thoughts about a traumatic experience, the question may turn to the meaning life has for them.

7.8 Moving beyond itself

Ricoeur suggests a phenomenological pre-supposition relates to any kind of ontological question, and for the staff nurses in "neglect" and "Ben" this may relate to being-in-the-essay itself. Perhaps, being present in the essay in the precipitous use of the word "it" when starting a sentence? Hence, the forgotten question of being, of life and its meaning which becomes

manifest in the individual's identity is an ontological question (see chapter 4.5) addressed through discourse, reformulated perception, memory and language (Ricoeur, 1998a; 1999). Add to this phenomenon the possibility that both staff nurses were reliving their experience through virtual language, which has no physical form other than what it signifies to them, then the word "it" when starting a sentence almost seems a logical indicator to the hermeneutic self. The research extracts "neglect," "Ben" and the reference to the body needing to be healed and restored back to health, suggests the essay in these two extracts was an opportunity for both staff nurses to work out their emotions and knowledge, which they embraced in the writing process. Therefore, writing about nursing experience is more than likely to involve the notions of embodiment, pain and suffering communicated in the essay with some emotion. However, the research extracts demonstrate that emotion is likely to be discouraged by the annotator, who instead attempts to objectify any, and all meaning informing student discourse, and in doing so, they may negate the ontological significance of writing for the student reader.

7.9 Conclusion

In conclusion, this chapter has explored the first theme of the hermeneutic self. I demonstrated in the research extracts that many essays used the word "it" to start a sentence and some lecturers failed to comment on the word "it" starting a sentence, whilst others were not so forgiving. The word "it" used in this manner was an entry point to the theme of the hermeneutic self theme, and explored the phenomenon of reading, writing, and making sense of other people's discourse. The words of an-other, I suggest, are assimilated and externalised in the student's dialectical posture and precipitous use of the word "it" starting a sentence. I examined the research extracts and the notion of embodiment was found in two research extracts entitled "neglect" and "Ben." The notion of embodiment appeared to relate to the

emotional recollection of the students' lived experience which they attempted to write objectively about. This was clearly difficult for them due to the traumatic nature of their recollections. However, in their search for ontological discourse and making sense of embodiment, again the recognition of the meaning of words was restricted and externalised.

Chapter Eight

Rhetoric and "saying it well" with proof



8.1 Introduction

This chapter introduces the second research theme of rhetoric. Rhetoric, as discussed in chapter 4.8.2 in more detail, includes: plot, metaphor and three kinds of imitation called mimesis1-3 (the numeral 1 is meant to be close to the word) and refer to understanding the meaning of language and time. The research findings suggest to me the characteristics of annotation and student essay writing can be organised, explored, analysed and better understood using Ricoeur's model of rhetoric. Plot, metaphor and mimesis1-3 are discussed through a number of research extracts (see table 4) which explores annotations rhetorical traits that include "saying it well" (Aristotle, 1991) with proof, the use of the productive imagination to enliven the meaning of discourse for students and temporal understanding.

8.2 From level two to level three analysis: Naming through tropes

My research notes (see appendix 4) identify the process of negotiating initial assumptions and naive hypothesising (my hermeneutic self) are necessary stages of the hermeneutic circle. The process of revealing a phenomenon first to identify its hidden dynamics relates to mimesis3 which is the state of knowing at the very latest moment when things finally become clearer to the reader. This cyclical process is evolving and changing because the reader is exposed to new literary ideas (see chapter 7.7). Rhetoric relates to a reader's pre-conceived attitude to discourse which starts right away when reading to restrict an open interpretation and due to the dynamic of time, understanding increases when the dots begin to get joined up which I discuss later (see sections 8.6 to 8.8). In order to clarify my research narrative, figure 20 entitled *Ricoeur's new rhetoric*, identifies the five aspects of Ricoeur's taxonomy: plot, metaphor and mimesis1-3 which the research extracts are organised into. I will use figure 20 before each of the chapter's sections with the addition of colour to indicate visually its content.

Plot

Metaphor

Mimesis 1-3

Annotators aim to reinforce the strength of an argument in a student's essay and I suggest this relates to "saying it well" with proof, improving the essay's plot, using metaphor to compare, and modelling through imitation and temporal understanding (mimesis1-3). These taxonomies are considered necessary to promote critical essay writing and have a direct impact on clinical practice. In developing a Ricoeurean analysis of annotation, a number of research extracts are used to organise the chapter by using tropes to name interview extracts which I present in table 4 below, with the rhetorical taxonomies listed for convenience. The research extracts promote, imitate literary styles and enliven understanding of annotation through the use of metaphor and temporal action.

 Table 4: Tropes elicited as research data extracts

Type of research extract	Titles elicited from research data	Rhetorical taxonomy
Lecturer interview quote	"What do you see" in the art of nursing reinforced through the imitation of writing styles	Emplotment Metaphor Mimesis1 Imitation
Digital annotation extract	"Practice or practise" focuses on the principles for essay writing and "saying it well" with good composition	Emplotment "Saying it well" with proof
Handwritten annotation extract	"Phlebitis" reinforces the need to give a rationale and evidence for care in an essay which also parallels critical thinking for clinical practice	Emplotment "Saying it well" with proof
Lecturer interview extract	"Nebulous" relevance of reading to the embodied experience	Rhetoric Mimesis3
Student interview extract	"A dead thing to do" refers to the relevance of language. The antonym of dark is light, and metaphorically speaking, light illuminating ideas	Metaphor Imitation Mimesis1
Lecturer interview extract	"Incubate and nurture" Suasory language, the value of supervision, developmental stages	Imitation/ Mimesis1 Metaphor
Lecturer interview extract	"Derogatory terms" imitating integrity and character through social and symbolic meaning of language.	Rhetoric Imitation

Lecturer interview extract	"What if' we follow reading by thinking "then and then" followed by "and then what?" to refer to an external relation	Mimesis2
Lecturer	"I wonder" learning from the annotator	Mimesis2
interview extract		Imitation
Lecturer	It's like "a dark art"	Metaphor
interview extract	Bringing something to light, the opposite of dark is to illuminate	Mimesis3

8.3 Plot

Plot

Metaphor

Mimesis 1-3

The following three research extracts, two here and one in section 8.3.1 demonstrate "plot" which I suggest encompasses the key words identified in table 3, chapter 6.8. The first research extract entitled "what do you see" demonstrates the annotator's rhetorical philosophy of annotation when meeting a student to discuss the annotation they received:

"I may strike through sentences and whole paragraphs and ask the student to tell me in lay terms what they think they meant and then ask them to re write it or make suggestions. When I ask the student "what do you see here?" I'm trying to generate their understanding of the knowledge. If it appears to be different to what they've written that's where we start with the annotation because then I'm showing them how to develop their work so annotation is part of a tutorial but is essentially student led. I'll pick out grammatical mistakes, show them examples and say "such and such is like..." to make a point they can understand. If I point something out I will write something down for them but they do the work..." (Lecturer interview 8)

The "what do you see" research extract identifies a number of assessment issues annotators consider when giving feedback to the student on the essay content. The issues include: promoting student understanding through editing essay content, questioning their
interpretation, identifying the symbolism of words used and using metaphor as a bridge to understanding. I came to realise that this extract appeared to connect Ricoeur's rhetoric to the annotator reinforcing the art of nursing to the student. When the annotator picks out any grammatical errors, they relate it to plot and composition and asking the question "what do you see here?" attempts to get the student to recognise something they had not previously taken notice of, such as an error or lack of evidence. The same question also concerns plot, the clarity of discourse and the strength of an argument. When making "...a point they can understand..." the annotator is using rhetoric, persuasion and an action that can be imitated which is reinforced when the annotator says they are "...showing...how..." to improve the student's writing style. Therefore, Ricoeur's rhetoric and imitation of writing styles are bound temporally with the symbolism of language before a student can act and understand more fully the annotation they receive. Plot, in its fuller literary sense, whether poetry, music, or dance, refers to a well-constructed story with characters, style, flow, structure and the imitation of human action (Ricoeur, 2003). Through the story line, assertions are made and what informs the impact of any statement is the rhetorical power of eloquence, and what Aristotle (1991, p. 255) called "saying it well." However, when asking the student "what do you see here?" and identifying grammatical errors, striking through sentences and whole paragraphs, the annotator is in fact promoting an awareness of plot. This is because the composition of an essay invariably starts with a structure that develops as the "story" unfolds on the page. However, this process requires the reader (the student) to understand the annotation before attempting to re-shape the meaning of the text.

The annotator in "what do you see" attempts to convince the student about the logic of the argument presented and "...trying to generate their understanding of the knowledge..." by "showing" the student another way of writing. When stating "...such and such is like..." to

get the message across, the annotator's comments demonstrate the rhetorical traits can all be seen to work in confluence with one common purpose, for the student in their essay, to "say it well" with proof (Aristotle, 1991, p. 255, Ricoeur, 2003). Hence, all of these processes are learnt through the act of imitation in plot and learning about different meaning, and meaning developed from the referent word to action (Ricoeur, 1990). In the context of annotation, action occurs when the student makes changes to the essay content. This action is promoted in the next research extract entitled "practice or practise:"



behavioural models to smoking and cessation. One point to consider is that despite a

knowledge of the negative effects including health problems, in 2011 20% of the UK

The "practice or practise" research extract identifies a number of issues: the word "practice," a noun or thing and "practise" an action is stressed to inform the student. The annotation asks for more critical discussion, corrects grammar and punctuation. The corrective nature of the annotation when stating "*no need for a comma*" is grammatical, yet restrictive when not stating why. However, a need for "...*balance between pros and cons*..." is reinforced when asking the student "...*is this the most important criticism*?" to focus on developing structure and a balanced argument. Ricoeur (2003) suggests an awareness of plot fostering the mimetic conditions of integrity and critical thinking to ensure the successful "imitation" of moral action, which is relevant to the next research theme of individualism in chapter nine. In the nursing context an essay is a "safe" and experimental arena to act out ideas and interpret evidence with the expectation the student will apply what they have learnt from writing the essay into clinical practice. If the clinical experience had occurred already, then the essay is a safe space to think retrospectively about the evidence based care given. This assumption is reinforced by the notion that nurse education is focussed on the practicalities of professional education.

8.3.1 Plot and proof

The third research extract in this plot section is called "phlebitis." I chose this extract because it demonstrates well, the annotators' need to frequently reinforce academic conventions of an essay with regards to plot, argument and evidence, even in such a short section. The research extract underscores a need for evidence in a nursing essay has a direct impact on clinical practice: and 5% glucose, however if fluid restricted may be given undiluted, preferably via a central winy? Explain further Critique the guidelines were not very clear. As Sarah was fluid restricted the guidelines, labetolol was undiluted and was being administered via peripheral cannula, however Sarah's What phlebitis score was monitored hourly and the cannula site easily observable. Phlebitis is the Guiden General Sector for the function of the tunica intima of the vein (Macklin, 2003). The cannula site was slightly the Guiden General Sector would indicate early stages of phlebitis, the rationale used by both Sarah's nurse and the pharmacist for stopping the infusion. At 11:00 hrs Sarah was reviewed by the Mebits score of the tunica intima of Labetolol and that systolic had remained stable for the tunica infusion running, infusion completely stopped.

Not knowing the consequences of action is considered unprofessional and immoral (see "Ben" and "neglect" chapter 7.8 and 7.8.1) and the lack of supporting evidence is reinforced when the annotator asked "...*why? Explain further*..." and "...*what is the purpose of a phlebitis score*?" I wrote about this briefly in my research notes (see appendix 4, p. 1). Therefore, when the annotator asks the student to explain the rationale behind the essay and "prove" their rationale for care, if unable to the annotator may use more productive metaphorical language in order to relate to the student's current level of understanding.

When asking "...what evidence supports the guidelines..." and "...why, explain further?" the requirement of proof is a key focus for the annotator because of its direct relevance to clinical practice. When stating "...phlebitis score?" and "...revise sentence structure..." then "...sentence breaks the flow..." the annotator is modelling a sense of plot and "saying it well" combined with reference to proof (Ricoeur, 2003). Therefore, Aristotle's suggestion that "saying it well" and "...proofs must be demonstrated..." (1991, p. 255), became a key realisation for me, and so, I combined "saying it well" with proof, as a metaphor for the rhetorical union of Ricoeur's five taxonomies. When there is a lack of evidence, the proof is

simply not there on the page, yet the student may have thought there had been enough to convince the marker.

The student may not have realised that a lack of evidence demonstrated on the page relates directly to knowledge applied into clinical practice, and the annotator in the "phlebitis" research extract needed to convince the student of its importance. The annotation comments in "phlebitis" were critical, corrective and, ironically, when reinforcing a need for proof, they offered none themselves. Annotation, as a model for imitation therefore needs to be mindful of the idiom, to practice what you preach, whether student or annotator, because the word "proof" means producing enough evidence to establish whether an assertion is likely to be true or not (Ricoeur, 2008).

An essay, I suggest, should also function to communicate a sense of character of the self and others, such as self-awareness, bias and changing opinion based on persuasive argument. In other words, a need for proof is the need to convince the reader and the writer who happen to both weigh up the strengths and weaknesses of the argument to determine its worth (Ricoeur, 2008). When that reader is the nursing student, their interpretation of reading is important because of the dualistic notion of evidence and the clarity of discourse. If the evidence is not used to support an assertion being made, it may mean that the student was uncertain of its significance at the time. In parallel, the annotator also was unconvinced and a lack of clarity in an essay suggests the same in other aspects of clinical practice, which would require remedial attention. Conversely, the annotator is attempting to convince the student that corrections are a normative aspect of annotation because after an essay has been submitted, errors are invariably identifiable. This need to convince and persuade the student about the

importance of "saying it well" with proof, as I mentioned briefly overleaf, may be in the form of metaphor and productive imagination which I discuss now.



8.4 Metaphor and the productive imagination

In the following section I examine metaphor, productive imagination (see chapter 4.8.3) and the embodied experience in promoting the rhetorical clarity of discourse. As my research notes indicate (see appendix 4, p. 1-2), an alternate analysis is possible with a "play" on words and testing whether an argument makes sense. A lack of clarity in the student essay is an assessment issue indicating a need for remedial action to be taken, in order to address the student's misunderstanding. This is demonstrated in the research extract entitled "nebulous" below:

"An essay is a nebulous piece of work in progress and the student may or not understand that what they present is only a small section from the original source, and they have chosen to use it. So I have to think it was meaningful to them when they read it and then chose to use it. So I might ask them to describe why it was meaningful and on the rare occasion they don't know why then I may say something like "when I read it I felt lost by the point you were trying to make, it seemed to go off in all directions, and sending me down different paths..." So I'm going to have to trust them that their interpretation is accurate enough and by being honest in how I felt when reading it perhaps I can give them something to think about as I can't read all they have read, that would be silly. By accepting their interpretation in good faith I can then see if they are making sense of knowledge as a part of the essay towards a fuller understanding of the subject matter. I'm going to help them to make me understand by writing clearly. So, I see it as a partnership, and conversely using supportive annotation I'm showing them how to develop their essay by sharing meaning with the student..." (Lecturer interview 4)

The "nebulous" extract like the "what do you see" research extract, appears to reinforce the annotator's role in "showing," to use a phenomenological term for appearance, the student how to write and think critically. I thought this term was relevant because understanding "appears" with realisation to bridge the gap between what Ricoeur (2003) calls the negotiation of misunderstanding to understanding. I discuss this point further in chapter 4.4 and 4.7.1. The noticeable quote for me from the "nebulous" extract, is the annotator's use of themselves to encourage the student to think differently. For example: "…*when I read it I felt lost by the point you were trying to make, it seemed to go off in all directions, and sending me down different paths*…" is an example of the tactical use of self and metaphor to activate the student's imagination and enliven the meaning language has for them. The annotator could simply state "*please explain*" or something similar but this has little potential to activate the student's interpretation of discourse from the reading act to the writing act. Therefore, the use of imagination may ensure that both the annotator and student will be inspired by the reformulation of discourse from the reading act to the writing act.

The annotator's use of metaphor indicates the acknowledgement of stasis and a state of unknowing (Ricoeur, 2003). As an advanced reader, however, the annotator's thoughts are unlikely to be in stasis, but they may use metaphor to stimulate the student's misunderstanding (Iser, 2006). When the annotator states going "...off in all directions, and sending me down different paths ..." and "...I'm going to help them to make me understand

by writing clearly..." the use of metaphor is proffered because when "...using supportive annotation I'm showing them how to develop their essay by sharing meaning with the student..." In order to make sense of this phenomenon, I examine Ricoeur's use of metaphor further.

8.4.1 Metaphor and the embodied experience

Ricoeur (1990) calls for the use of metaphor to have a new "semantic pertinence" and move beyond the literal displacement of words such as "like for like" as found in the "what do you see" research extract to offer a new predication. This is a key point. This "living" kind of metaphor aims to bring new meaning to a sentence through semantic innovation, synthesis of plot and temporal action, or the effect of time on the clarity of discourse (see chapter 4.8.2). This means the annotator, when attempting to relate the student to the parts and the whole of the essay activates the imagination of the student to clash with any assumptions made. This clash of semantic meaning, which the annotator attempts to evoke, uses metaphor and the productive imagination of the reader to be productive, in the sense that new understanding is generated, whatever that may be (Ricoeur 1990, p. ix). This production enlivens language, perhaps pictorially, as a "...new thing... the as yet unsaid, the unwritten..." (p. ix) and its relevance to the reader, the student and the annotator is activated. Ricoeur (1990) states this living metaphor "...springs up in language..." (p. ix) as a semantic innovation and Ricoeur's metaphorical activation therefore is made "alive," only as long as the reader perceives it in the first place to mean something new and act on its relevance. This also depends on whether the student can remember what they assimilated as new knowledge. Ricoeur identifies this process being very difficult to achieve because of the automatic act of assimilation and new meaning merging with old in an almost indistinguishable way, unless taken notice of at the

time. I discuss this phenomenon more when I examine mimesis1-3 and the embodied experience of understanding.

In addition to the semiotic concepts of a sign or symbol and what it signifies, Ricoeur (1990) suggests metaphor helps develop the coherence of discourse through the productive imagination. All linguistic signs are followed by non-linguistic imagery which can be viewed pictorially, such as meaning, concepts or things, but the depth of meaning is ultimately based on a person's embodied and corporeal experience. This means, the more physical experience a person has about a subject read about, such as cardiac nursing or public health, then a fuller understanding may occur (see chapter 7.6 and 7.7 on embodiment). Conversely, a lack of embodied experience, such as a student nurse reading about something they may not have experienced, means their understanding is lessened by a lack of application (see chapter 2.6 and 7.7). Understanding, therefore is based on embodied experience mediating in terms of another or what Ricoeur calls resemblance, perhaps pictorially, to contemplate similarities and likeness (Ricoeur, 1978; 1990). Therefore, language is the trigger of the rhetorical structure of plot, imitation and metaphor and seeing-in-terms of the language being used.

Far from metaphor at once having meaning as an event in plot, the effect is suggested by Ricoeur (1990) to be bipolar and temporally meaningful, because misunderstanding at the time reduces its meaning. Therefore, metaphor has the potential to be dead or alive, asleep or awake, literal and personal with meaning ready to be triggered by the reader. The annotator's challenge therefore, is to enliven feedback to make sense to the student, which in the absence of the student's embodied experience, is typically intuitive. An example of this is found in the following research extract entitled "a dead thing to do:"

"I find it frustrating when writing an essay because I read all of this material but instead of saying what I think, I have to use other people's ideas. Even annotation reinforces that my voice is pointless and many times I've been told to "write in the third person...and find the essence of nursing." I know I can choose what to write about and how to phrase things but I want to make a statement too about what affects nursing, so it feels like a dead thing to do, to write about something someone else wrote means that your own voice isn't heard..." (Student group interviewee 16)

The research extract entitled "a dead thing to do" is temporal and the annotator in supervision needs to "bring along" a student to a certain point of view in order to ensure the student grasps the relevance of interconnecting ideas. Meaning and understanding, which I discussed in chapter 4.7 to 4.7.1, is relevant here due to semantic differences acted out by the student and annotator on the page of an essay. The supervisory relationship usually culminates in annotation feedback, unless annotation is used later to develop the essay content with the student for re-assessment. However, the extract "a dead thing to do" identifies the imitation of nursing by using metaphor. When stating "...feels like a dead thing to do..." the student's difficulty in grasping the relevance of nursing literature is due, as I mentioned overleaf, to inexperience. I wrote about this in chapter 2.6 in relation to discursive versus more scientific essays and when reading other people's work, the "grasping" process of meaning is clearly deactivated. This deactivation relates to the student perceiving reading and writing about academic work and due to the absence of an embodied experience it "...feels like a dead thing to do..." If not considered relevant to the student, they may not appreciate the relevance of what they are reading until they are more experienced. Hence, the often repeated student comment "I wish we'd been taught this earlier" when in fact they had, but it had gone unnoticed at the time.

The metaphorical declaration by the student interviewee stating reading "...feels like a dead thing to do..." indicates a number of embodiment possibilities: that the nursing discourse does not make sense to the student or discourse has partial meaning or the power to elicit meaning is redundant. The first part of the quote and writing about nursing "...feels like a dead thing to do..." has both literal and analytical potential. The "dead thing" is what the student refers to in nursing literature because the words of an-other are redundant and not of their making (see chapter seven theme). Literally, a "dead thing" to do is a contradiction in terms because when the student refers to a "dead," "thing" and "to do" it is an antidote to inaction, which is the awakening of life. Again, awakening is metaphorically speaking, being open to the light and being illuminated. The student may find reading and writing "...feels like a dead thing to do..." because the canons of nursing literature lean towards formality and academic objectivity, where even the first person narrative is considered a poor substitute for evidence. This evokes Ricoeur's (2003) pre-understanding of symbols because only when de-valuing the importance of the nursing literature and subsequent annotation, could there be any sense of it being "dead" and this identifies a responsibility to read in order to gain new understanding from a position of misunderstanding.

8.5 Mimesis and temporal action

Plot

Metaphor

Mimesis 1-3

My research notes (see appendix 4, p. 3) identifies the starting point for the question of temporality, and its relevance to annotation and rhetoric, discourse, or teaching and learning in general. Clearly, the process was cyclical and temporal because I started with the research extracts first, then found that I had no idea what or how Ricoeurean theory could be applied to them. In a continual process of back and forth motion, I identified theory to the relevant

research extracts. Hence, the term mimesis being "...the imitation of human action..." (Ricoeur, 2003, p. 40) is relevant (see table 3, chapter 6.8, imitation and modelling behaviour).

In classical Aristotelian rhetoric, mimesis and the imitation of action included language, dancing, rhythm, and harmony are engaged either singularly or in combination. Therefore, in dancing and harmony the human actions of character and emotion are imitated which according to Ricoeur's (1976) reference to Aristotle's work, is an instinctive kind of likeness and finding delight in reproducing something. I could relate to this notion because I attempted to view the research extracts consistently through the medium of Ricoeur's work. Imitation of art then depends on three Aristotelian differences: medium, objects and manner. The "medium" refers to things such as acting, dancing, and singing, playing an instrument, narration or discourse. Imitating human action as an "object" (see chapter 7.3) is suggested to fall into moral differences which I discuss in chapter nine in the research theme of individualism and moral character. Lastly, Ricoeur (1976) suggests the "manner" of artistic imitation is divided into poetry, comedy, tragedy, storytelling where the author takes on the character of another or writes in relation to their own experience. This is all relevant to annotation, as will be shown.

Ricoeur (1976; 2003) suggests when the medium of narration is written down and imitated it takes on a different dynamic to extend its meaning. In the context of the research theme, mimesis can be seen in the imitation of nursing action and nursing discourse. In other words, behaviour is observed, read and written about as an act of imitation. This temporal process occurs when a student observes, listens and are taught by the lecturer who then goes on to annotate in the course of essay feedback. What is annotated back may be imitated or not and

indicates that time and the perception of relevance are factors in choosing what to be influenced by. Temporal action and distanciation were discussed in chapter 4.8.4 and inform an understanding of mimetic action. I clarify this issue through Ricoeur's (2003) mimesis1-3 which I present next in figure 21 entitled *Ricoeur's model of temporal action through mimesis1-3* corresponding to more research extracts. As the remainder of the chapter progresses, each stage of mimesis will be discussed in more depth. Although new to me at the time, I found mimetic theory applied directly to the hermeneutic circle and all aspects of the research process, whether in the preparatory chapters one to six or thematic analysis of each chapter.

Figure 21: Ricoeur's model of temporal action through mimesis1-3

Imitative process requires a semantic action between structure, symbolism and time.	Mimesis1
Structure includes plot, metaphor and rhetoric, symbolism and signification of words	See 8.6
through time and playful (and productive) imagination (see chapter 4.8.3).	
Allows the reader to follow the story because plot organises narrative events to	Mimesis2
"grasp them together." The possibility of the "as if" (Ricoeur, 1990, p. 64).	See 8.7
Connection of ideas, reading better understood in the fullness of time. An endless	Mimesis3
spiral going past each point with understanding occurring at different altitudes and at	See 8.8
different times. Ricoeur likened this spiral process to an ancient intersection of a	
road where understanding meets and changes.	

8.5.1 Ricoeur's model of temporal action through mimesis1-3

According to Ricoeur, humans experience time in two ways. First, time is experienced as a chronology, in linear time, with time passing in hours, days and years. Second, phenomenological time is the orientation to discourse, in the past, present and future (see chapter 11.5). In mimesis1 (see section 8.6), the imitation of action is expressed in the ability to ask questions such as who, what, why, how which are structural, symbolic and temporal (Ricoeur, 1990, p. 54). This relates to humans prefigured ability to understand the symbolic system which allows the meaning of gestures such as raising an eyebrow, a nod of the head or

following discourse to be interpreted (Ricoeur, 1990). Hence, symbols provide the rules or structure of meaning and allow for interpretation.

Mimesis2 is a configuration stage which concerns the imagination, which I discuss more later in the chapter and what Ricoeur (1990, p. 64) calls the "kingdom of the as if" (see section 8.7) which I discuss in the research extract of that name. The "kingdom of the as if," allows the reader to follow the story through the organised narrative of events and plot in order for them to grasp meaning and this involves recognition, or recognising something from which to grasp some meaning. Lastly, mimesis3 (see section 8.8) is an integration stage where the real world of the reader is informed by the hypothetical world of the text. This last stage ensures the reader gains a fuller understanding and connects ideas at the last instance of time. The process is open ended because when a fuller understanding is achieved it adds new insight. So, until new insight is gained, perhaps in an "aha" moment, a state of flux occurs between symbolic meaning (mimesis1), the grasping of meaning (mimesis2) and the fullness of understanding reached in time (mimesis3). I will examine mimesis in more detail because mimesis1-3 concept develops classical rhetoric into Ricoeur's new rhetoric and I thought a relevant analogy may make it more meaningful.

8.5.2 An analogy of mimesis: The glass of water

Mimesis1-3 becomes active in time and the analogy of a glass demonstrates the benefit of metaphor in order to visualise an idea. The pictorial image of a glass as the symbolic structure holding water relates to knowledge and the glass to indicate language (mimesis1). When the glass is filled, the tap is turned off (mimesis2). This relates to ideas gained through reading to reach a level of understanding and new learning. After a period of time, the water in the glass may need filling up and ideas and understanding may lessen, leading to more

reading and hypothetical ideas testing reality to inform the reader in the last instance of time (mimesis3). The remainder of this chapter will explore the annotation research extracts in relation to Ricoeur's mimesis1-3.

8.6 Mimesis1

Plot

Metaphor

Mimesis 1-3

The figure above emphasises plot in the colour purple, because it is relevant to analysis of the research extracts and mimesis1-3 which I highlight with the colour red. Ricoeur (1990) identifies that the composition of plot is grounded in a series of inter-related preunderstanding of the world formed through individual and social symbolism. First, in structural terms actions imply goals. Second, every narrative presupposes the familiarity of terms that relate to the language of "doing-something," being able to "do something" and "knowing how to do something." This is especially relevant to developing an argument, writing an essay and annotation discourse. This mimetic action depends on the cultural traditions which organise the plot to involve the ordering of events and inter-connecting action of sentences into the total syntactical action of the story (Ricoeur, 1990, p. 56). In other words, does it make sense? Therefore, word signification moves from the virtual to actual signification through the sequence of linguistic events.

The mimesis1 theory and plot are evident when an annotator asks the question "what are the relevant issues?" This approach is instilled in people from childhood and finding pleasure in imitation during play (Aristotle, 1997). Hence, in Ricoeur's work, the reader can identify a playful engagement with the text through the notions of narrativity (telling stories-fables,

allegory), plot, imitation of action (mimesis1-3) and the signification of simple to complex language through playful imagination (Ricoeur, 2003). Chapter 4.7 identified that image and what is imagined always includes assimilation informed by schema towards a new conceptual order. This conceptual order is a reflexive relationship with what is known and unknown and developed through rhetoric and discourse from one person to another engage in argumentation. Ricoeur's (2003) rhetoric therefore ensures the meaning of metaphor is more than mere utterance and should embrace context, thought and semiotic discourse at the level of the sentence. These are the imitative conditions that enable a rhetorical approach in annotation and the research extract called "incubate and nurture" demonstrates this pragmatism below:

"I think our role effectively is to incubate and nurture these individuals in positive ideologies and positive role modelling. That means shaping a student's thinking through annotation amongst other feedback activities. I honestly think the reason I came into lecturing is to encourage people to self-actualise and be the best they can be, and to provide the environment, facilities and tutelage. Sadly, what we have realised is that not everybody works that way and the system can be manipulated, it is also our role to root out bad practice, poor attitude, poor belief systems and offences that are bordering on fraudulent, cheating and so we have dual roles..." (Lecturer interview 1)

Words like "incubate" and "nurture" relate to the annotator's role in modelling nursing, clinical skills, critical thinking, knowledge and in writing. The stages of development go from embryonic dependence to independence first as a writing student and second, as a nursing student. The research extract indicates the dual purpose of annotation as a social function of

control and its regulatory function because the annotator identifies themselves to have a "dual" role as a marker and giving feedback but this does not address the complexity of nurturing the student's development.

8.6.1 Dual role

The "incubate and nurture" research extract also identifies the issues of temporal action in an essay content which the annotator, for assessment purposes, is searching for. These actions are the predictions of safe practice demonstrated in the essay to relate to past and future practice. This was first discussed in chapter 2.7.2, 2.8 and here, reinforcing "...the presentation of alternatives and arguments..." (Ricoeur, 2000, p. 63) being so important to demonstrate and develop. This temporal action relates to mimesis1 suggesting that narrative composition and practical understanding lies in the symbolic resources of the tradition that it refers to, in this case nursing. This feature governs the aspects of "doing something" and the "meaning of something" mediated through rules, symbols and social norms (Ricoeur, 1990, p. 57). In other words, the student's interpretation is mediated through the symbolism language has for them and the cultural basis underlying its significance. The annotator too interprets through this cultural mediation and the duality of their roles confines the supervisory relationship to be transformative. This intuitive symbolism is important to acknowledge at a practical level because it is forged individually through life experience, before a social and autonomous symbolism influences the meaning of discourse (Ricoeur, 1990, p. 57). This symbolic system allows people to interpret the meaning of gestures such as raising an arm to hail a taxi, or raising an arm to indicate implied consent for their blood pressure to be taken, of attitudes and readability of actions. Hence, symbols are considered a quasi-text and providing the rules of meaning to allow for interpretation of the parts of the plot (p. 58). The next research extract entitled "derogatory terms" demonstrates the symbolism of language in nurse education by raising the issues of corrections, structure such

as an undeveloped introduction, old references and terminology:

This essay will explore whether old age and depression are related to socio-economic, demographic and geological factors. The risk factors and impact on <u>old people</u> will be developed. Further research will compare the United Kingdom to another international country and show if being depressed and being geriatric is a common problem not only linked to one geographical area. Year use of words / terus peods to be constrained (Madge / Regard) Old age is considered to be a major factor in the onset of depression in the United Kingdom affecting (Blazer, 2003). Called late life depression risk factors are suggested to be stressful life events, changes to daily living, the death of a spouse and self critical thinking (Blaze, 2003). Due to the age of aged sufferers, the consequences of later onset depression are suggested to be more significant. However, this is contentious due to increased suicidal risk of young people, who are the future of the United Kingdom (NICE, 2007). Does that wate young people work in portaut? How would the death of the genes young beople work in portaut? How would the death of the genes

The student makes an error of terminology with reference to demographic data and "geology," rather than geography. The implications for the student are when the annotator immediately identifies the ill-considered terms in use to indicate an attitude of concern and one that may require a follow up meeting to talk about their inappropriate use of language. The word "geriatric," once used as a professional term is now considered derogatory. What a word denotes is first a literal meaning then second, it signifies personal meaning to the student. Notably, it is the annotator who may view the word as derogatory, when in reality it is only derogatory as a colloquial term and the essay extract demonstrates that corrective annotations relate again to promoting positive attitudes in clinical practice. The annotator aims to root out the potential negative effect of ill-considered language "as if" the student would be negative too in clinical practice. The next section develops what Ricoeur called the

Kingdom of the "as if" (Ricoeur, 1990, p. 64) through mimesis2 and a research extract using those terms.

8.7 Mimesis2 and plot: As if

Plot

Metaphor

Mimesis 1-3

Ricoeur (1990, p. 64) suggests mimesis2 is the "kingdom" of fiction, whether real or imaginary, and the notion of the "as if" in the plot of narrative. This is the stage of grasping the meaning of discourse. The ability to follow a story draws narrative in the direction of a linear representation of time and Ricoeur suggests we follow reading by thinking "then and then" followed by "and then what?" to refer to an external relation (Ricoeur, 1990, p. 67). Next the sequence of reading events allows the reader to add something to the text "and so forth" (Ricoeur, 1990). This is clearly seen in the annotation research extract entitled "what if" below:

"I like it when the student brings in their essay draft. I will annotate and make key points and if I suggest moving a paragraph to another page I will put a star on it or put "move A to point B." I find annotation is the context and I like the person with me and will say things like 'what if you write this in a different way" and using metaphor to get a point across I will say something like "such and such is like... what if..." because they perhaps knew what they were writing about and I try to stir the student's imagination to imitate the nursing context. I am sharing my knowledge as if they are learning and developing themselves from me – we share....for example if the student has written something derogatory about old age then I may ask what if they were your parent, loved one, or perhaps the loss of a long term partner when aged and feeling left alone to live out the rest of their days....the issue is about them thinking about themselves in the future, not now as young fit people whose relatives are also healthy, it is about instilling empathy for future use..." (Lecturer interview 8)

There are a few rhetorical taxonomies in the "what if" extract such as plot, persuasion, metaphor, productive imagination and temporality of understanding. Briefly, because I want to focus on the temporal aspects of the extract, when trying to "...*stir the student's imagination to imitate the nursing context*..." and "...*what if*..." the annotator imitates the use of metaphor and provokes the student's productive imagination (see chapter 4.8.3) in order to create new meaning. When "...*annotation is the context*..." as lecturer interview 8 suggests in "what if" the foundation for questioning in annotation is the springboard to effect change. The invention of the flexible terms "what if" and "as if" ensures that cognitive change occurs (Ricoeur, 1990). However, the annotator has forgotten to make explicit that the essay is also the context which allows imagination and imitation through annotation.

The underlying dynamic of these rhetorical processes is temporal action, and mimesis2 is considered a necessary stage in the development of understanding discourse. When "grasping" the significance of the plot and working towards a coherent whole, the annotator is using understanding and time to affect a student's changed perspective. The temporal action can be seen in the use of phrases like "...moving a paragraph..." in the "what if" research extract and "...what if you write this in a different way..." to indicate future essay changes and thoughts. The "what if" is a suggestion of thinking relating to how it changes in time. Learning and "...developing themselves from me..." therefore, indicates action and the annotator being instrumental in that action. When promoting students "...thinking about themselves in the future..." and "...instilling empathy for future use..." the extract

demonstrates the temporal action of mimesis2 can be applied to annotation to transform the almost automatic signification of language learnt over time to follow meaning, but not quite there yet. The research extract "what if" follows what Ricoeur (1990) calls the "irreversible order of time" common to physical and human events (p. 67). However, he also suggests that the story as a whole, the entire plot, can then be interpreted as a whole theme, or one thought.

The plot also allows for episodic reading of the parts and a sense of ending when the story can be viewed as a whole (p. 67). This can also be seen in the structural function of a plot when the story is re-told. Hence, a re-telling, or a re-reading can enhance the quality of understanding, of recognition, of its symbolism in time (Ricoeur, 1990). In reading the end at the beginning and the beginning at the end, Ricoeur suggests we learn "as if" to "…*read time itself backwards*…" (p. 68). This schematic application of "what if" explains a more thoughtful annotation practice when demonstrating the mimetic word "as if" in relation to structuring an essay.

8.7.1 My glass is full

The "what if" research extract is about the student using derogatory terms to describe old age which may be enacted as an attitude in clinical practice. The extract states "...*we share... learning... and developing from me...*" knowledge, experience of reading the essay, sharing feedback as annotators, and focus on derogatory terms the student may be innocently unaware of. The annotator in the extract acknowledges the essay is a transitory stage of the student's thinking towards schematic development and when asking the student to think about losing a loved one when elderly in annotation, they are reinforcing the need for empathy. The annotator is educating the student in a nuanced understanding of language, applied to the past experience of the lecturer, to the future experience of the student. For

example, and I quote from the "what if" extract "...sharing my knowledge as if they are *learning and developing themselves from me*..." is referring to the personalisation of the annotator as "me" and "...sharing knowledge as if they are learning... from me..." This is also demonstrated in the next research extract entitled "I wonder" below:

"I haven't had the chance to think or talk about annotation before, so what I am saying makes me think because it's a kind of practice that we haven't been taught about...if I am reading a weak piece of work, how I can balance my comments appropriately. The student needs to be clear about the point I'm making set against the likelihood they themselves may not have understood what they had written, or why. But I wonder... I need to make the student realise what's to be done to turn it around so I would use phrases like "what if..." and I will pose a question, or a point that leads them to a different way of thinking about the content. But again annotations would be utilised to steer the student to think of how they could improve things...." (Lecturer interview 7)

The research extract "I wonder" identifies a thoughtful view of annotation to weave a connected approach between the lecturer's philosophy of nurse education. When the annotator states in the extract "...*but I wonder*..." there is a clue to an alternate thinking about annotation and its temporal action. The annotator had not thought much about annotation until being interviewed and their thoughts indicate a pre-understanding of annotation which identifies schema and a history of annotation for the lecturer and their own practice. Again, the annotator in "I wonder" uses the "what if" question.

Ricoeur also had something to say about the word "if" which I found interesting because it allows the use of questioning in literature in the form of testing the adequacy of an assertion (Ricoeur, 1990). This hypothetical testing allows for the defence and challenge of any assumptions an assertion is based upon. Ricoeur suggests the word "if" can be used in two ways: first, as a synonym for the organisation of narrative and second, as an antonym to historical narratives claim to the truth of narrative (p. 64). However, as a trope the word can also relate to tone and "how" something is read and the spirit that forms it (Ricoeur, 2003, p. 58). The word "if" becomes a fiction in a reader's negotiation of meaning for fictional narrative, and whether lecturer or student, both attempt to make sense of what is being read in order to then write about it with more insight (Ricoeur, 2003). The word "if" therefore, refers to the aforementioned "imaginary" and "reality" or the "what ifs" in the revision of discourse and a reader's search to understand the context. The annotator however, needs to make clear what their annotation means to themselves first and this is not always the case when "steering" the student to think in a different way or when "...posing a question..." When being interviewed the lecturer began to reflect on their own practice and therefore, the intended audience of annotation may not only be the student but rhetorically, the annotator themselves. When questioning "as if" to elicit a response perhaps by re-phrasing a question in a different way, it gives the annotator a different perspective of the question too. Therefore, the posing of questions should not presume in annotation in the first place, that the questioner knows the answer.

8.8 Mimesis3

This final stage suggests discourse gains its fullest meaning in time (Ricoeur, 1990) and the

colour of the diagram above indicates the chosen research extract includes metaphor and mimesis1-3. The research extract below entitled "a dark art" from lecturer 8 (see appendix 2 for a partial transcript, indicated by one asterisk) demonstrates how the taxonomy of rhetoric occurs in annotation practice:

"It's like a dark art because no one teaches you what to do...maybe the annotation is just a veneer...not the real issue here but it involves capturing the thoughts I had about the essay at the time of reading and if we do meet the use of time when sitting with the student. Reading back the annotation with the student helps them to make changes and we can talk about what the annotation meant to them and me. We can use the time to discuss the temporal nature of reading and understanding the annotation within the narrative of the essay. We can come to some sort of agreement of its meaning...the dark art is in the issue of time when the student reads to write for the essay, then reads... what they have written, then reading the parts and the whole. Then you add the same process for the annotator, drawing from time and experience and then get them both together in a room to discuss what essentially is both their understanding of the narrative at that moment in time...clearly interpretation changes in time..." (Lecturer 8)

Using metaphor signifies the limits of language as explanation, instead using language as illustration; such as suggesting annotation is a "a dark art," so I asked for clarification (see appendix 2, marked with two asterisks). In particular, my interest was aroused by the addition of the word "veneer" which somehow made the "art" perhaps a dark one at times because of the possibility of conscious and unconscious attitudes. The lecturer replied:

"...I might make suggestions "what about this?" I might pose a question that triggers a response but they offer the answer. I offer them a way of writing it in a slightly different way and that takes time and that takes skill which is what we should be able to do.... not everybody knows the mysterious art of moving from the different levels of description to critical analysis...it's a craft, an art, to do what we do and we should use the skills. Yes, caring is part of it but the skills should be to communicate with the student...to improve their standard of life and their standard of living, health of the student. Life and wellbeing..." (Lecturer interview 8, last paragraph)

The lecturer's annotation identified an instinctively rhetorical practice. I felt that annotation as a dark art referred to not knowing theoretically what was involved but being instinctively persuasive. However, the annotator "shows" students a way of writing differently which is transformative and impacts on their "…*life and wellbeing*…" and so my task was to analyse the surplus of signification from the metaphor of "a dark art." I paraphrased this by stating (see appendix 2, indicated by three asterisks):

"When you are with a student, it's about finding out what's within them, perhaps understanding is another aspect. So it's finding their understanding and the meaning of the words they are using or how can they get the message more clearly or succinctly on the page. What is in them, but you are trying to get it out of them..."

First, the "dark art" with the addition of the word "veneer" was a reference to conscious or preconscious attitudes which I naturally linked to chapter ten's theme of the reflective consciousness. The lecturer realised the need to be aware of counter-productive views that would impact negatively on the student's learning. Second, being aware of the impact of annotation, the lecturer aimed to illuminate the student's writing with this in mind. Hence, I thought both possibilities were inferred by "a dark art" metaphor. This was reinforced soon after in the interview when the lecturer stated (see appendix 2, indicated by four asterisks):

"We are unbelievably powerful people and we have the power to inspire, enthuse, intimidate, frighten, de-motivate and destroy or not care. We can do that with our looks (because we are all experts in communication), what we say and certainly what we write and I have become aware that annotation can carry that attitude..." (Lecturer interview 8)

Perhaps "a dark art" refers to annotation and balancing the conscious and unconscious acts of imitation, coaching, persuasion, power and temporal understanding. Ricoeur (1990) suggests mimesis3 is the final stage which allows understanding to reach a sense of fulfilment as a narrative model. Ricoeur (1990; 1991; 2003) therefore presents mimesis1-3 as an endless spiral going past each mediation point of temporal understanding. These mediations occur at different altitudes and at different times and like any dynamic model reflecting the complexity of understanding does not remain at a fixed point, but is in a constant state of flux, ever evolving as new stimuli serve to inform a different perspective. The pragmatic nature of this understanding reaches a threshold of application and then it is tested and refined and the effect noted. This fulfilment therefore, identifies the fleeting nature of reaching a fuller understanding through temporal action because it will be revised later on as and when required.

In this temporal state of flux, understanding travels the semantic structure of symbolism, schema, and of time especially noticeable when a narrative plots ending appears to be linked

to the start and the start to the end (Ricoeur, 1990; 1991). This is according to Ricoeur, where understanding of the text converges at an intersection of temporal action, at varying times where paths or roads meet to be applied. An image of this process (using a productive metaphor) would be similar to looking at an intersection where several pilgrim and ancient routes are still evident but a new road has been built over it to indicate all of the past routes taken at different times. Then the temporal action continues in its spiral of interpretation and in a literal sense, this application is demonstrated by the annotator and the student's narrative.

The research extract "a dark art" demonstrates a number of rhetorical taxonomies which I needed to make sense of first, by defining the terms in use. For the interviewee, "a dark art" metaphor describes the art of annotation in the absence of a theory of annotation. For the interviewee "a dark art" may indicate a semantic collision of two worlds to motivate them to develop a spark of interest for themselves. The power of metaphor to promote understanding demonstrates the temporal nature of learning and Ricoeur's (1998) recognition (see chapter 4.7) suggests accepting something read to be as true as you find it, whilst being aware of any assumptions which may cloud its meaning. Therefore, the meanings of words are bridged by transitional processes, such as being indebted to the author for sharing their ideas, and from the symbolism of the idea to its recognition in the individual (Ricoeur, 1990, p. 12).

Consider the phrase the "dark art." The word "dark" suggests illuminating the meaning of something being obscured. The word "art" (OED, n. d) refers to skills in the practical application of principles and that is being "shown" through annotation on essay content. How could I progress to analyse this extract with the meaning inferred by the interviewee? I thought it logical to start with an antonym and the word "dark" literally refers to its opposite meaning to activate the word "light." There is something about annotation that aims to "bring

to light something, or bring something into the light" and that was my feeling when interviewing lecturer 8 (see appendix 2). I was linking several other possibilities with this analysis which appeared relevant which relates to the notion of "doing something" I discussed in section 8.6.

The "what if" section (see section 8.7) argued that what should be brought to light relates to the student and the annotator being confident in their ability to promote learning. However, the research extract "a dark art" whilst focussed on the student being "brought into the light," also demonstrates the annotator's insightful thinking. The underlying principle is that the student is enlightened through the rhetoric of annotation, however it is likely both student and annotator will be enlightened when reflecting back. The relational aspects of this process means the lecturer demonstrates to the student, good principles of writing, such as rhythm, syntax and grammar, content, character and elocution, style, and flexibility of ideas. This may lead to the possibility of "saying it" rather than "saying it well" with proof (Ricoeur, 1990, p. 12).

If the annotator has the opportunity to re-read annotation comments and talk with the student, then both are likely to better understand the rationale behind annotation. When attempting to "bring to light" new understanding, the annotator's ability to engage in suasory discourse and promoting change from ignorance to knowledge is significant because they are also demonstrating the dynamic nature of interpretation and time. The recognition of words, revealing one's identity (for example, reflecting on annotation), unravelling the plot and the use of memory are all enacted. When new understanding is gained, the "saying it" well with proof (p. 12) can then move from speech back to text. When considering temporal action, one particular section of the "dark art" quote in my view demonstrates mimesis3 well:

"...the issue of time relates to when the student reads to write for the essay, then their interprets what they have written, then reads the parts and the whole. Add the same process for the annotator...when they both get together in a room to discuss their understanding of the narrative, at that moment in time their interpretations may change..." (Lecturer interview 8)

The annotator is acknowledging the conclusion of mimesis1-3 in plot, metaphor, imitation and time, and ensuring the fullest understanding of narrative at the moment of analysis. The overshadowing of the past annotation is understandable, because it literally was a reaction at the time of reading, and one that shapes new thinking when re-reading the essay content. The imitation of action then is found in the imitation of writing and using language to understand the annotation process. So the student can "see and hear" the dynamic nature of annotation in the lecturer's own thinking evoked at different levels of cognition and at different times. However, it is the temporal process of enquiry that directs annotation towards new understanding because it is the first time the annotator has articulated their views on the page. A note of caution for mimesis3 stage is the fact that it appears to be the shortest and perhaps least important stage because meaning and understanding is constantly evolving.

8.9 Conclusion

This chapter has presented a number of research extracts in the form of digital and handwritten annotation and interview extracts. The former has been presented to identify what kind of annotation feedback is received by the student and the latter adds texture to the annotator's working philosophy of annotation termed "the dark art" by lecturer eight. In particular, in "a dark art" research extract the lecturer appears to use annotation as suasory discourse in order to forge links for the student and mapping concepts. The annotation data suggests making corrections of content; structure and attitude are literary phenomena having parallels with clinical practice. This is reinforced by the nursing context which the research extract suggests the annotator views as being central to their role. Therefore, the annotator has more than a dualistic notion of what informs annotation.

"Saying it well" with proof (Ricoeur, 1990, p. 12), I suggest, is a standard for essay writing and the students' observation of lecturers' spoken, behavioural actions and written work allows a form of imitation, not only of writing style, but for nursing action. The imitation of writing is also gained when reading the literature and annotation feedback on the essay. The process should be considered continuous and not a one off process and the culmination of the supervisory relationship is ongoing, temporal and its meaning unconfined by the moment (Ricoeur, 1984; 2006).

Chapter Nine

Individualism



9.1 Introduction

In this chapter I explore the research theme of individualism identified in table 3 (see chapter 6.8). Individualism relates to the moral predicate to do the right thing for individuals in society. I discuss how I worked out the research theme and the emerging key concepts of responsibility, the "just" and the political labyrinth. Using a variety of research extracts, I explore moral philosophy (Ricoeur, 2000) and discuss ontology, person centred care, organisational neglect, annotation promoting equality and what one annotator called "the defence of nursing." Lastly, the issues of prescriptive language, technology and disembodiment help to explore the meaning of annotation in nurse education.

9.2 Working out the theme: Setting the scene

The theme of individualism was identified after following a careful reading of the research data and research notes culminating in the key words identified in table 3, chapter 6.8. I then had to find a theme that captured the essence of the key words summarised in table 3. This was an emerging process and in order to check the accuracy of my interpretation I started by defining the terms. For example, the term "person centred" refers to being directed towards individuals (OED, P5b), and the emphatic pronoun the "self" refers to a person (OED, A) or the noun himself or herself. The word "empowerment" means to gain more power over one's life, and "ethics" refers to moral principles that include law and politics. Therefore, table 3's key words are clearly evident in this theme and Ricoeur's moral philosophy of the "just" of relevance to the theme, which I define in the next section.

My research notes (see appendix 5) represent my initial thoughts about the theme. I wondered about the reality versus rhetoric (policy, political rhetoric) and citizen's constant exposure to the political labyrinth, whether they liked it or not. By writing down my thoughts about the

theme and the research extracts, I could work through the temporal dimensions of the hermeneutic circle to realise individualism starts and ends with the perception of the self and an-other. The first research extract entitled "person first" demonstrates this well:

"What are we here for? Do we care for the person, the product, the system or do we care for the person within the system? What is the answer, I think of the person first and that drives my practice... Think about the Ombudsman and Francis report, they are just the tip of the iceberg and I want to instil in our nursing students the need to be advocates, be critical and have courage. That's what I've done for the majority of my career so that's probably what my philosophy on care is. It should be individually tailored care and not generic and you could almost draw those parallels with annotation feedback ... so I can draw a lot of parallels from my clinical practice...." (Lecturer interview 8, see appendix 2)

The "person first" interview extract identifies the underlying relevance for the annotator of various reports identifying neglect in the NHS. The annotator in "person first" states a key moral principle for informing action in the ontological question "what are we here for?" This question highlights a tension for the annotator within nurse education when promoting person centred care against injustice, in the various reports they mention. The ontological tension for the annotator occurs when reality and the rhetoric of social justice appear not to match. The annotator in the research extract "person first" refers to the ombudsman and *Francis report* and it may be useful to briefly summarise the recommendations of these key reports because of their significance to the annotator. First, the Parliamentary and Health Service Ombudsman (2011) report entitled *Care and compassion? Report of the health service on ten investigations into NHS care of older people* identified neglect such as: poor

communication, lack of empathy, loss of dignity, clinical induced malnutrition and dehydration, to name but a few. Second, known commonly as *The Francis report*, *The Mid Staffordshire NHS Foundation Trust Public Inquiry* (2013) final report found there was a lack of basic standards of care and dignity, callous treatment and a lack of staff, patients left unwashed, un-toileted and unfed to name but a few points. The recommendations suggested patients were let down by the system and there should be a return to a patient centred culture, transparency, care and compassion in nursing.

What these various reports indicate for the annotator is a real (in contrast to a rhetorical) concern for patient care within the nursing profession and society. I suggest the reports demonstrate what Ricoeur (1965; 2000) called, a form of violence, or social injustice to individuals, which I discuss in more detail later. Violence may sound an inappropriate word to use, but the definitions of violence make it relevant and relate to subjecting someone physically to violent behaviour or treatment and "to violate" through the use of "power or authority" (OED, 1). If violence is a "...violation or breach of something..." (OED, 6), it relates to the various reports cited in the "person first" research extract and to nursing in general. In contrast to the reports on NHS failings, the annotator in the "person first" appears to think it is morally right to care for the person first before all other organisational considerations are made. When asking "what are we here for?" the annotator, by using the word "we" reinforces the collective experience of nursing and shared purpose to care for a person on behalf of society. This consensus, indicated by the annotator's quote, appears to be a constant intuitive reminder to nurse educators how to care for patients with the right moral intentions, and why the theme of individualism in essence captures the meaning of the key words collated in table 3 (see chapter 6.8).

9.3 A working definition of individualism

The promotion of individualism is demonstrated in the next research extract entitled "selfactualise:"

"I think that our role effectively is to incubate and nurture these individuals in positive ideology's and positive role modelling. I honestly think the reason I came into lecturing is not to allow people poor practice...to encourage people to self-actualise and be the best they can be, to provide the right environment, facilities and tutelage so they can reach their optimum capability. Sadly, what we have realised is that not everybody works that way and the system can be manipulated, so it is also our role to root out bad practice, poor attitude, poor belief systems and offences that are bordering on fraudulent and cheating. So we have dual roles. When I first came into nursing I didn't think I'd be rooting out poor practice... and annotating essays gives me the power to make a difference and get a great insight into student attitudes...." (Lecturer interview 1)

The art of teaching nursing, and annotation, is an act designed to respond to the many social pressures exerted on an individual for the good of society. The notion of individualism attempts to juggle the rights of the individual within the system, organisation and society. The rights, according to Ricoeur (2000), are the issues of primary social goods, which need to be evenly distributed to underpin the "...*basic structure of society*..." (p. 62). Primary social goods are promoted through the notions of equality, self-esteem, autonomy and the fair distribution of rights, duties, advantages and burdens (Ricoeur, 2000) which I use as a working definition for the theme of individualism. When these conditions are met, the right for autonomy and freedom is determined. However, when the conditions are unmet, the

resulting distrust of the system disturbs the notion of the collective identity. When the annotator in "person first" instils in students "...the need to be advocates, be critical and courageous... individually tailored care" they are unwittingly advocating a Ricoeurean moral philosophy which promotes the notion of society organised to be mutually beneficial to all its citizens. Mutuality was also demonstrated in the "self-actualise" research extract when the annotator intends to incubate and nurture individuals and "...root... out poor practice..." whilst at the same time negotiating the tensions of obligation and duty (Ricoeur, 2000). The social dynamic of responsibility therefore, involves the public authorising and legitimising governance through political consensus (Ricoeur, 2000). This is the political consensus which authorises the lawful organisation of nursing as a registered qualification and in a sense, how the annotator wields that authorised power, depends on their understanding of the principles of responsibility and being "just," which I discuss next.

9.4 Responsibility and the "just"

The annotator in the "self-actualise" extract is aware of their professional responsibility and indicates their motivation to train as a nurse in the first place. This is instructive because it refers to Ricoeur's (1994) concept of "responsibility" and a sense of obligation and mutuality of the social contract to include justice, fairness, altruism, politics and the governing rules. Responsibility comes from an acceptance of the laws that govern society and punishment, if a responsibility is not adequately met. In nursing, as an authorised profession, this obligation is a response to the expectations placed on it to do no harm, or as little harm as possible. When examining the concept of responsibility, Ricoeur (1994) suggests the verb "…*to respond…*" (p. 12) is an appeal by society to act for the greater good and an appeal socially informed through the language of moral imperatives, which I discuss in section 9.7.1 and chapter ten on memory recall, obligation and duty.
Ricoeur (1994, p. 206) suggests responsibility starts from the intention to ask the moral question "what ought I to do?" in any given situation. This also refers to knowing what not to do, and the answer to what is right to do, relying on the will for a good life (Ricoeur, 1994). Therefore, the individual's sense of duty, when collectivised, becomes a social contract authorised, yet constrained by the public. Developed from John Rawls A theory of justice (1971), Ricoeur's (2000) theory of the "just" identifies the widespread acceptance of the utilitarian notion of the social contract that binds the collective of individuals together. In The just Ricoeur (2000) discusses the twofold principles of justice where any free and rational person concerned about their own position in life accepts the equality of others. This is demonstrated in the research extract "person first" when asking, "...do we care for the person, the product, the system or do we care for the person within the system? I think of the person first and that drives my practice..." This individualist notion of nursing philosophy relates to expecting a political system ensures that its citizen's right to fairness is a priority. Ricoeur's (2000) theory of the "just" helps place the two moral tensions of individualism and the system of governance and theoretical framework in which nursing dwells. The "just" concept refers to an ethical motivation to understand the self through others and a drive for autonomy (Ricoeur, 1994). This reinforces the point I made in my research notes (see appendix 5) when stating individualism starts and ends with the notion of the self. Such a concept is socio-cultural within a "political labyrinth" which brings me to the second tension of the system of governance, politics (Ricoeur, 2000).

For Ricoeur (1994), politics refers to a scheme of co-operation and social system designed to advance the good life for all those taking part. This "taking part," or participation is based on the ethical principle of mutuality and citizens being responsible to work for the common good. Ricoeur did not define his meaning of the word labyrinth but in my view it is a metaphor for a vast maze or a complicated, torturous arrangement (OED, 2a). This term refers to the political landscape in which citizens have to traverse, whether knowingly or not. Ricoeur (1994) therefore identifies a contrasting motivation for a sense of self and social responsibility within the political labyrinth, and this is the obligation to put things right or to suffer a penalty when things are not put right and deteriorate with inaction. The annotator as a citizen first and a nurse second, acts on behalf of the common good.

9.5 The annotator as a citizen: Promoting equality

Ricoeur (1994) refers to the moral autonomy to act on behalf of society as a principle, and a burden on the individual to think about their actions and its effect on others. This is their sense of agency, of action and ability to do something. Kant (1999) called this the moral autonomy, the legislative authority that can belong only to the united will of the people. The legislative authority means to promote the notion of the equitable citizen and perhaps for a nurse, to believe in the moral tenets of society. When stating in the "self-actualise" research extract to "...root out... poor attitude, poor belief systems... and annotating essays gives you a great insight into student attitude..." the annotator, as a citizen is exercising the notion of transferring individual autonomy to the authorising legitimacy of the State to rule over all citizens (Ricoeur, 2000). This ruling, given to nursing, as I said earlier, through the State made laws, gives it the authority to be self-governing and regulatory (Nursing and Midwifery Order, 2001). However, this legitimacy reinforces the perceived benefit of mutuality and the idea of individuals excluded from a distributionist perspective over the priority of social fairness, of advantages and disadvantages (Ricoeur, 1965; 2000). Ricoeur asks therefore, can we speak of fairness? In answer to that question, we need to find a way to minimise inequalities. One way to demonstrate the protection of equality is found in an argument of an essay, where alternate viewpoints and "saying it well" with proof (Ricoeur, 1990, p. 12)

occur and where the annotator can nurture a sense of equality. Hence, in the "self-actualise" extract to "...*root out... poor belief systems*..." the annotator is engaging in the rhetorical traditions of annotation when first, assessing the contents of the essay and second, linked to the rhetoric theme, when persuading the student to think and act professionally through an essay argument. The annotator when making their argument public to the student does so within an organising system that ensures transparency and commentary. This is demonstrated in the research extract entitled "moral heresy" below:

"What we did discuss is what post registration students are writing about change and improvements in patient care. There is a moral heresy that goes back onto the student which they must address. "If I'm sloppy about writing about nursing then this is likely not to be implemented and patients will suffer. If I get better at writing about nursing, there is a better chance what I'm proposing will benefit patients ... The annotation is one part but I would hope that in terms of students' dealings with me as an individual is that they know if I'm giving them information then it's because we must have integrity about and it's not just for the sake of it..." (Lecturer interview 7)

The "moral heresy" interview extract uses a strong theological word with "heresy" to refer to the annotator's responsibility when encouraging students to write well in an essay. The extract suggests there is a direct correlation between good writing and good thinking in clinical practice. Conversely, the annotator is also suggesting that bad writing may reflect poor thinking which could then affect patient care in clinical practice. They are attempting to identify poor practice or ideas before they can do harm in reality. The word "heresy" therefore refers to the annotator's almost spiritual belief in the responsibility to write well and view the student as a conduit of ideas to transform practice. The annotator in "moral heresy" indicates the constraining notion of the social contract in their annotation has an impact on the student to avoid being "sloppy." The annotator states a benefit to the student is promoting public safety and "...*if I get better at writing this there is a better chance that what I'm proposing will be implemented and patients will benefit*..." In this way, a fellow citizen annotates back to another citizen to ensure equality.

Rawls (1971) first principle of justice is relevant to the above position because it includes the principles of distribution and fairness that underlines the notion of individualism. Ricoeur (2000) suggests in the principle that: "...each person is to have equal right to the most extensive total systems of equal basic liberties compatible with a similar system of liberty for all..." (p. 63). A person in a position of authority who can then make a positive contribution to the life of another defends individualism. This can be seen in the "moral heresy" research extract, which identifies the annotator's direct impact on student thinking and "...*if I get better at writing this there is a better chance that what I'm proposing will be implemented and patients will benefit*..." Therefore, it could be suggested that annotation promotes a moral philosophy of the "just" against the possibility of colluding with what is unjust, such as poor or unimproved care.

Rawls (1971) first principle is determined by the conditions under which some inequalities are held as being preferable to even greater inequalities. This is called the "principle of difference," (Ricoeur, 2000, p. 64) which is a pragmatic interpretation of the social contract. This principle may be seen in nursing through decision making and practice which values equality and autonomy. When inequalities occur as a result of organisational failings referred to in the "person first" research extract (see section 9.2), I suggest a principle of indifference relates to the misrule of authorised priorities. Hence, the example I gave for the principle can be reviewed when there is a perception that authority has failed, to result in injustice. What is required therefore, is the development of reasons for action and agency. The power of the social contract, according to Ricoeur (2000) involves an acceptance of arguments developed from "...*considered convictions*..." (p. 67) and this directly applies to decision making and reasoning in essays. These convictions develop from rationalising doubt and prejudice and the adjustment of conviction through argument (p. 67) and in the context of an essay, the next research extract demonstrates the student being guided through annotation to identify their mistakes.

9.6 In defence of nursing: Self and others

The following research extract demonstrates the link between difference and perceived injustice entitled "defence of nursing:"

"In annotation the nursing lecturer defends the nursing profession through the motivation to aid students to learn, understand and be critical of the big picture.... that means being politically aware too. But it is getting harder due to the mass of numbers and varied forms of feedback we use. So I may write something like "your approach to this is somewhat concerning, you must consider the patient first and foremost...if you don't base your rationale upon evidence and as a qualified nurse you will be accountable for your actions." I am defending nursing autonomy and person centred care because students will be exposed to many challenges that directly and indirectly affect care.... for example, we live in a society where the NHS is used as a political football, with short term goals, where party politics appear to be the priority and where change may have unintended consequences..." (Lecturer interview 4)

The "defence of nursing" research extract is quite significant, because it identifies the annotator's perception that nursing needs to be defended from internal and external forces that may result in social instability. When stating to the student "...your approach to this is somewhat concerning, you don't base your rationale upon evidence and as a qualified nurse you will be accountable for your actions..." the annotator is identifying two issues they are concerned about. First, the need to "say it well" with proof which I discussed in chapter eight. Second, when stating "...you will be accountable for your will be accountable for your actions..." the annotator means "you" "must" be accountable, which is an imperative to act accordingly to promote the social contract and goods. Notably, the democratic process referred to in the "defence of nursing" being like a "political football," identifies the threat of short termism in society and the political labyrinth which I referred to in my research notes (see appendix 5, p. 4).

Ricoeur (1994) asked "...how are we to move from the individual at large to the individual that each of us is?" (p. 30) and I realised the notion of the individual self as a reference point underlines the Ricoeurean principle "... that nothing can be identified unless it refers to oneself or another..." (p. 31). The student as a reference point, was examined in chapter seven and the hermeneutic self and in this theme the State and the collective reference points are fundamentally threatened when individualism is devalued. To counteract this threat prescriptive language is used in nurse education and annotation to reinforce moral action, which I discuss next.

9.7 The use of imperative statements

In the research extracts "moral heresy" (line three and seven) and "defence of nursing" (line five) the imperative word "must" appears to be significant to responsibility and being "just." I was perplexed about the language of imperative and found amongst other resources, reading

Hare's (1952) *The language of morals* helped shape my understanding. An imperative relates to a practical rule of will and is an objective principle, such as saying something would be "good" to do or should be refrained from doing and "...*all imperatives expressed by an ought*..." (Kant, 1993, p. 24). An imperative refers to commanding a necessary action I identified in chapter 6.6 and 6.7 and the following short research extract demonstrates that some lecturers judge an essay content with imperative structures in mind:

"...when I am marking I am looking to see if the student has the right attitude, is caring and compassion and is not a danger to the public..." (Lecturer interview 4)

The research extract suggests a sense of responsibility is actively searched for in order to promote mutuality. Even a more serious concern underlies nursing assessment as the next research extract "are they going to kill?" demonstrates (see lecturer interview 3 and 4 grid, p. 202):

"For us it's about, are they going to kill patients, are they going to be safe and can we put our professional name to passing them. It's about professional integrity...what a nurse ought to do in a given situation..." (Lecturer interview 3)

Therefore, the research extract identifies the language-in-use in an essay has both scholarly and disciplinary implications for the nursing student, and this situation is reinforced in annotation. The use of the "ought" word in the research extract "are they going to kill?" above, is an example of an imperative. There are other similar words used in annotation which appear to have similar meaning, such as "…*where is the evidence*?" (see chapter 6.6, extract 2, p. 223) or "…*need a reference*…" (extract 3, p. 224), or in the sense that they are

stating "what" should be added to an essay in order for the student to improve their understanding. The words are value laden and interviewed students in the focus group stated that the choice of words used by an annotator often made little sense, for example, annotation comments such as the word "good" failed to identify "what" was good about the referent text (see chapter 6.3, figure 14, student 3). Ricoeur (2000) suggests a lack of definition is inevitable because all value laden words refer to an *a priori* deduction given to signify something in one's imagination before it is experienced. This is complicated by the annotator knowing only at the time of commenting, and as the literature review identified, they would probably forget what they had meant when reading it later. The following section explores the use of imperative or prescriptive language and individualism further.

Modern nursing regulation comes from a civilian body identified by law to uphold professional standards and ensure public safety. Ricoeur (2006) calls this authorisation, legitimisation. The imposition of caring for the good of society is modelled by a prime example of prescriptive language, the UK's nursing code of conduct entitled *The Code* (NMC, 2015a) which the annotator would be familiar with. From the perspective of the public authorising and legitimising governance through political consensus, the NMC attempts to shape attitudes, care and nursing practice through the medium of language. The social systems I discussed earlier in relation to responsibility and the "just" are monitored and proposed in their *Standards for pre-registration nursing education* (NMC, 2010). The standards suggest nurse lecturers and students have to abide by the agreed principles in order to maintain safe and competent practice. Both the standards (NMC, 2010) and *The code* (NMC, 2015a) have a high use of imperative language and a quote from the latter demonstrates the use of imperative well:

"Always practise in line with the best available evidence. To achieve this, you must: (6.1) make sure that any information or advice given is evidence-based, including information relating to using any healthcare products or services, and (6.2) maintain the knowledge and skills you need for safe and effective practice..." (NMC, 2015a, p. 7).

The language of ethics, power and control appear to run through the NMC code and the word "must" perhaps indicates a pervasive attitude of authority over collegiality. The tone of paternalism permeating the code is a throwback to past traditions and widespread expectations of female nurses thinking of themselves last and being obedient first (Holden & Littlewood, 1991; Urban, 2012) which raises concerns about perceptions of gender inequality. There is a clear intention in the code to be unequivocal because clarity is important to maintain standards for professional values (Griffin, 2006). We learn from the generalisation of instances (Hare, 1952). In other words, when instructed "...you ought to have used (then) and you ought to use (now)..." (p. 157), after being told a number of times what one "ought" to think, or "must" do, there is an expectation to do the same again (Griffin, 2006; Hare, 1952). This is what McGuire (1961) called the "normative instances" presented in the above NMC quote which can be challenged if we are to accept that value laden language guides conduct. If actions require imperative language, we must assume that language, or those wielding it for reasons of the greater good, exert an authority over the human will, or at least tries to.

Understanding the imperative of nursing becomes more important when problems are more complex and prescribing action can be done in more than one way by asking the question "...what shall I do when I'm in this situation?" The words "ought" and "must" may be used

before and after an event in the form of common sense and duty. The value laden words are there to instruct a person and by doing so act as a prescriptive imperative. This act is first instinctive and second, can be taught by the prescriptive use of language. However, the latter less so in predicting the right actions will be taken by an unconvinced person, which I found of relevance to the rhetoric research theme I explored in chapter eight.

Therefore, imperatives cannot be logically justified because we assent to do something based on the evidence supporting the reasons given (Hare, 1952). When humans identify themselves as having done something they should not and "ought" not to have done they realise the moral principles that may have been dismissed as non-guiding. However, it is important that learning occurs "without" being taught, which nurse education tends to do too much of in the prescriptive justification, persuasion and language-in-use of annotation, assessment and feedback. When aiding a student's understanding of nursing against societal pressure that may result in injustice, the nursing profession tries to defend itself and its patients, its moral charges.

The "defence of nursing" (see section 9.6, lecturer interview 4) research extract states the annotator's concerns when, and I quote "...we live in a society where the NHS is used as a political football, with short term goals, where party politics appear to be the priority and where change may have unintended consequences..." The annotator's "defence of nursing" against perceived injustices, seem to be justified. The issue of authority impinging on the right of citizens to self-determination and autonomy, and the abuse of trust by nurses in the various reports on NHS failings, is of concern to the annotator in the "defence of nursing" research extract. Perspective is therefore limited by the phenomenon of societal heritage being threatened when there are attempts to erase traces of its own historical traditions (p.

97), such as higher education institutions pursuit of modernity, or changed priorities for the NHS (Regan & Ball, 2016). Therefore, in the "defence of nursing" extract, when promoting individualism against any unintended consequences, one possible effect of modernity on traditional teaching and learning methods within higher education, is the rise and development of technology.

9.8 Technology

The "defence of nursing" research extract infers there are unintended consequences of technology and "...mass of numbers and varied forms of assessment..." in nursing to have an effect on individualism and a need to assess using secondary forms of information (QAA, 2012a; 2012b). I am referring to the rise of technology in a technological society and because technology is relevant to annotation, nursing and individualism, it requires further exploration. A technological society is one where the application of scientific knowledge for practical purposes, such as making and using instruments, are then used to improve life and increase productivity and efficiency (Kaplan, 2006). The question of technology is ontological because, as Heidegger (1977) suggested, it is not technology that is dangerous, but how it is applied that matters. Technology is a form of revealing knowledge in order to apply it and knowledge that had hitherto been unknown or concealed (Heidegger, 1977). The benefit for society and individuals of improved technology is in how it transforms life and more importantly how many lives are saved calculated by how it is applied and its intentions. This concern is amplified in Ricoeur's (1965) criticism of technology because, at the same time that its benefits are promoted, the negative aspects of technology are insidiously developing to change society. These insidious changes are: a tendency to homogenise, create a dependence on its products and lead to a form of subtle destruction (Ricoeur, 1965). This

tension is demonstrated in the next research extract called "standardised comments" on digital annotation criticising technological innovation:

"We have a bank of comments at the side and I use them, but sometimes I edit them and change the meaning of them. If I didn't then I wonder where it would stop, perhaps with software marking the essay. It worries me because it takes away the personal nature of communication which is vital to communicate the essence of nursing. The worry is that it is depersonalising and what message is it giving to the student and my colleagues, and where will it end? Again it may be that writing our own comments supports a more meaningful and a more sustained approach to feedback..." (Lecturer interview 7)

The promotion of technological innovation as a benefit to teaching and learning in the "standardised comments" research extract is concerned with what may happen next, to further reduce the value of individualism. This became more poignant to the research process because when I started writing this thesis, digital annotation was quite undeveloped and a recent addition for assessment purposes. I realise now that I found the annotators principled narrative quite reassuring. Whilst being promoted as innovative technology, feedback for nurse education, however, is more nuanced than the typed word on a page found on a personal computer. Unlike past practises, when the student submitted a paper copy which was then annotated, marked and returned, the student and annotator had a hard copy to touch, feel and sometimes smell. Often the marker could detect a smoker, or a student who had eaten a curry, or had placed a coffee on the paper leaving a ring. This made the essay feel more "real" and a connection could be made with the student. When this changed to online submission of an essay, online reading and marking, online digital annotation with a bank of

comments, then online access to a student anywhere in the world or online feedback by email to a question, the human dynamics changed dramatically. Gone are the smells, the touch, and the embodied experience of the essay replaced by a screen, keyboard, and effectively a sanitised experience. This suggestion of sanitation occurs because technology invariably leads to what Ricoeur calls a depersonalised attitude to the human condition and authenticity (Ricoeur, 1965). The depersonalised effect of technological systems is keenly felt by the following research extract entitled "conveyor belt" below:

"Meaningful annotations are considered as part of process and it's not a conveyer belt approach - it needs to remain individualised. What are we modelling otherwise? In the past student's picked up their essays and could come and get more feedback if they wanted their annotated feedback explained. Because it was handwritten I generally remembered what I had written and why... not now with digital annotation. It's looking at the essay and looking at what ideally we would like to see – has the student achieved, and if not, where have the shortfalls been and give the feedback comments appropriately? Again the student individually picked up their essays, not now it is online, so whilst the comments were generalised the context of the comments were not and were individualised..." (Lecturer interview 2)

The research extract "conveyor belt" system is concerned that annotation should not be a "conveyor belt" system and a bank of comments risk depersonalisation. The annotator in the extract is concerned that the moral theory behind their intentions to promote individualism, instead models technology as a credible assessment method. The annotator is suggesting that the easy access to the essay and feedback online ensures a disembodied experience for all involved and the context to promote meaningful feedback and persuasion is lessened

somewhat. Whilst there are benefits of accessibility, what is feared and modelled for nursing is that technology is preferred over individual and personal assessment methods of old. The authentic tension in promoting individualism is therefore felt not only by the student but also the annotator, which is demonstrated in the research extract entitled "meaningful" below:

"I am certainly looking for person centeredness and a patient is at the focus of the essay. For example, I recently failed somebody on the care planning assignment for writing "you are obese and overweight and need to go on a diet," and there was no evidence of negotiation and working with the patient. Certainly patient centeredness is something that I look for. In the essay there has to be a collaboration and consideration of the service user/carer perspective, rather than "I am the expert and this is what I think is right." That's one of the main things I look for..." (Lecturer interview 6)

The "meaningful" research extract identifies that individualism or person centred care is a priority the annotator looks for in an essay. The annotator identifies this through the student's use of language and any objectification of the patient being evident in the insensitive use of words in an essay. The "meaningful" extract suggests that the annotator wants the essay to demonstrate behaviour that is morally right and fair. The annotator is searching for person centredness and therefore, being "just" in annotation ensures the state of conformity of nursing principles that binds generations of nurses together, is communicated to the student. This is in part due to the shared familiarity of action nurses have for caring for those with physical, mental and emotional needs.

This ontology relates to Ricoeur's (2000) suggestion that every person fears a badly managed illness or death and it is society's sense of fairness to the population that aims to ensure procedures are in place to reduce the risk of incompetence. Hence, when changing from clinical practice to nurse education, lecturers inevitably bring with them the binding principles of their nursing experience into the new context. A sense of being "just" is demonstrated in the next research extract entitled "Chief Executive" below:

"No matter where she is at, whether she is a Chief Executive or working two or three days a week she will treat people with respect because we treated her well and supported her. It doesn't matter about the system, it's just me and her both sitting together to talk about the annotation, like a focus for a conversation. We annotated. I think there is a link there and it's what goes with the annotation, the investment in the student and it has to be done with the student. Right there face to face, or live on the end of the phone, I've done it with a webcam before and that does work or the other end or Skype and held up the annotation to the camera so they can see it. But it's really person centred student learning like the old way... is a learning journey, it's a process, and how can you tell if they've captured what's required..." (Lecturer interview 8)

The "Chief Executive" research extract is a timely antidote to the perceived negative effects of technology and using Skype©, Twitter and webcams to communicate with people individually. The emphasis on person centred care in the extract ensures a timely engagement within the supervisory process that does not stop with the awarding of a grade and receiving annotation. This is a process, as the research extract suggests, with no negotiable end because the start and end points of understanding are unknown. The annotator in the "Chief

Executive" research extract, with the student beside them, suggests the benefits of discussing annotation "...*right there face to face or live on the end of the phone...or held up to the camera...*" However, when practised in such a collegial manner "face to face" or on the telephone, the annotator is overcoming the negative effect of remembering what they had written because the student is there to question what was meant. This overcomes the difficulties of memory recall and perception, which I explore next in chapter 10.7 to 10.7.3 and instead annotation becomes individual.

9.9 Conclusion

This chapter used the research data to explore the theme of individualism and the "just" using Ricoeur's moral philosophy based on Rawls (1971) work. Ricoeur's theory of justice and the political labyrinth were identified as the landscape which nursing has to negotiate with, in order to be meet its social responsibilities. Throughout the chapter terms have been defined in order to examine their relevance such as the primary social goods which underpins the "...basic structure of society..." (Ricoeur, 2000, p. 63) and distributes equality, self-respect, autonomy, fair distribution of rights, duties, advantages and burdens, and a working definition of individualism. The research theme of individualism identified an awareness of the need for care and compassion which is reinforced in the Ricoerean (2000) notion of the social contract for citizens as nurses. This was evident in the research extract in "defence of nursing" where the annotator attempts to promote societal notions of individualism in order to defend nursing and the individual against pressures such as poor care, technology and procedural systems. The discussion also identified that some annotators need to be aware of the notion of individualism and the use of imperative language which inadvertently promotes the omniscient narrator and subjugates the authorial voice. Therefore, a system valuing the

notion of individualism, and one which underlines the principles of nurse education and clinical practice, reinforces to the student, that every contact matters.

Chapter Ten

Reflective consciousness and slippage



10.1 Introduction

In this chapter I define what the reflective consciousness is and how the research theme was identified. The research extracts identify the use of single word tropes, imperative, proof, slippage and evoked memory in relation to discourse, annotation and memory. In order to undertake a fuller exploration of Ricoeur's reflective consciousness I examine tone, the unconscious and the automatic processes in the mind waiting to be recalled in annotation. Therefore, Ricoeur's (2006) three uses and abuses of memory: ethico-moral, practical and the wounded memory are used to organise and explore the meaning of the research extracts. lastly, I explore the automatic replication of thoughts, actions and emotions from the past triggered and projected by something in the present, called the transference hypothesis (Ricoeur, 2012). First, let me define what the relevant terms mean.

10.2 Identifying the theme

The reflective consciousness was identified after analysing the data collated in table 3 (chapter 6.8). Reflecting and being consciously aware of the impact of the self in my view sum up the key words. For some time, I had a preconceived idea that the issue of tone would be significant and negative due to the results of the literature review. However, I found the reverse and if tone were evident, it was generally constructive. That led to the identification of organising concepts of annotation and promoting professionalism in nurse education. What was also identified was a degree of slippage from either the annotator or student which could be used to identify key professional principles and perceptual difficulties related to memory recall and interpretation. As I pondered the key words in table 3, I wrote down my initial thoughts in research notes (see appendix 6) which identify a series of assumptions and questioning the theme. Like the theme of individualism in chapter nine, some prior reading informed the research notes (page two) and relevance of Ricoeur's three abuses of memory. I

later questioned why abuses, rather than just uses was significant and realised I related it to interpretation, mis-interpretation, mimesis1-3, and memories likely to change depending on how they were perceived in the first place. I then organised the research extracts to develop an exploration of the theme to annotation.

10.3 The reflective consciousness

The research data collated in table 3 (chapter 6.8) demonstrated that words can trigger emotional reactions not only in the student reader but also the annotator. I needed to understand why this occurred and began to think that annotation for nurse education is grouped into a three-pronged act of interpretation. First, the author being read includes the annotator's interpretation of the author's message at some point in the past meets the student's more recent interpretation of the author's message. Second, the annotator reading the essay and giving feedback back to the student indicates their initial thoughts about the essay. Third, when annotation is finally read, the student has to process the annotator's interpretation. This process may have a fourth dynamic to it if the student and annotator meet afterwards to discuss, and then the process starts again. This interaction then ignites a reflective consciousness by way of thinking about something that needs to be revised and rewritten. This refers to perception which occurs through "...a process of becoming aware or conscious of a thing ... " (OED, 1). Reading the research data from initial to in depth understanding and noting the significance of words to the research participant and myself ensures a conscious awareness of words hermeneutic appeal. The research findings therefore led me to draw inferences about the conscious thoughts that informed perception, particularly in light of temporal understanding (see mimesis1-3, chapter 8.6 to 8.8). First, let me examine the use of single words as a trope in annotation to hint at something unsaid. The basic premise of Ricoeur's (1970; 2006) reflective consciousness is that a forgotten memory is likely to be communicated at some point in discourse, because it remains in memory, ready to be used. This relates to section 10.7 with regard to exploring annotation, evoked memory and emotion. This distinctly Freudian (1915; 1961; 1997) idea suggests the recall of memories is temporal and involves memory to shift an individual's consciousness. Ricoeur (2006) proposes that a memory forgotten is unlike a memory remembered, because the accuracy of the forgotten event has worn away to such an extent that it is hard to remember the truth of what actually happened. How the memory forgotten influences the memory recalled is difficult to examine because it relates to the childhood memories that are distant yet still influence to some extent emotion, thoughts and actions as an adult, which I discuss later on in the chapter. In contrast, the memory recalled has an influence that is more obvious and its historical influence easier to follow. This means recall remains reflectively conscious. Ricoeur (2006) suggests the translation of recalled past events from one person to another involves three uses and abuses of memory: the ethico-moral, practical and wounded memory (p.69) which I use to organise the analysis of research extracts in this chapter to explore the reflective consciousness. In particular, as the chapter draws to a close I synthesise Ricoeur's (2006) wounded memory with transference hypothesis to suggest the abuse of memory is more susceptible to the memory forgotten. I will now define the reflective consciousness further, and develop the notion of single word tropes before examining Ricoeur's three uses and abuses of memory from research extracts.

10.4 Single word tropes

The research extract 2 (see chapter 6.7) entitled "what?" identifies the annotator's reaction to the word "it" I examined in chapter six as a theme in relation to overcoming the selfreferential nature of reading. The "what?" word according to Ricoeur's recognition is visualised and its meaning grasped and understood when identifying the difference the word has from other visualised objects:

This assignment will define and describe what assessment is and now important it is ouring the nursing process. This assignment will highlight the different sources of information gathered during the assessment process. How it helps nurses during the nursing process. The assignment will identify and the describe Early Warning Scoring system and how important it is at identifying patient needs during the nursing process. It will describe how nurses work with their patients to develop a positive relationship during their time at hospital. Assessment is important in nursing care as it allows the nurse to gather information about a patient's health, identifies health needs, strengths and weakness of the patient. It allows the nurse to identify problem's and treat them to the best of her or his ability (medical dictionary 2012).

In the immediacy of the "what?" extract we can see that for each "it" (as I discussed in chapter seven), is the annotator writes "what?" to indicate the need for clarity and asks what "it" the student refers to? However, for the student when reading the annotation, the single trope may have a visual impact to hint at something unsaid. What that hint is perhaps, depends on the perceiver, yet the use of a single word carries with it a "what?" declaration devoid of social niceties. Why this is the case may relate to the annotator's perceived role, as chapter nine indicates by their dualistic role as a citizen and awareness and signification of language. In the "what?" research extract the annotator is unsure what "it" refers to (see chapter seven's research theme) and this misunderstanding played back to the student may start a chain reaction of differentiation with unknown consequences. When declaring "what?" the annotator had communicated clearly to the student then there would have been less misunderstanding. Hence, the "what?" research extract could refer to Ricoeur's (1976) clarity of discourse which suggests that misunderstanding is the starting point of new understanding

310

through the recognition of words (see chapter 4.7.2). Let me develop this point further in relation to the clarity of discourse and the next research extract called "no:"

Nursing models are usually used as a guide in patient assessment to monitor the success of care and if any problems should arise they would be detected. A nursing model may be distinguished No - a Nursung Model in a concept or philosophy to direct the assessment components rather than a method used to monitor care provided.

The annotator first corrects "nursing models" with the use of capital letters, underlines a segment and then gives an explanation underneath. The extract is more communicative than the "what?" extract but it also appears to be tinged with authority due to the strident corrections, which are not collegial and defining what a nursing model is. The authoritarian effect of all these factors combine to reinforce the value laden use of the "no" word.

10.5 Imperative and proof

The word "no" stated in declarative terms is an example of the imperative use of language (see chapter 9.7) to indicate authority and doing the right thing. However, this does not address the issue found in both the "what?" and "no" research extracts, that imperatives cannot be logically justified in the absence of proof (Ricoeur, 2003) and so excuses itself from "saying it well" with proof (Aristotle, 1991, p. 255, see chapter 8.3.1) as discussed in chapter eight. Therefore, annotation without using supporting evidence to convince the student of a need to revise the essay content reinforces an attitude of "do as I say, not do as I do" when added to the text. As a result, rather than promote clarity and critical thinking, annotation parallels the lack of it. Instead, the absence of any references, theory or evidence to corroborate the annotated point, colludes, rather than convinces the student. This is

demonstrated in the research extract below entitled "we are not trained" which attempts to deal with the clarity of discourse in annotation:

"We are not trained in any way to annotate...it's not necessarily part of the induction process that we are all trained to mark. Many lecturers haven't...worked on a ward...for a long time so they are thinking back to their past, and a past that is uninformed by new clinical experiences. Many leave the ward with differing experiences, positive and negative so they bring with them a certain perspective. Their experiences may have led to a biased view about what is relevant. We are modelled by our past experiences of nursing and the giving and receiving of feedback which generally is negative in the NHS, you tend to get feedback when you've done something wrong, because of the workload the priority focuses on what has to improve, not what you are doing right. Hence, constructive feedback is difficult because of the change in culture from clinical to educational...We then don't then have the lack of time and need to consider our own views more analytically..." (Lecturer interview 1, chapter 6.5)

The quote "...we are modelled by our past experiences of nursing... and the giving and receiving of feedback which generally is negative in..." is an important mimetic action because it underlines what shapes an annotator's point of view, their memory recall, feelings and perception. The annotator makes the point that in their experience of the NHS, rather than being told what they were doing right, there was a tendency to prioritise negative feedback which is problematic for the clarity of discourse. In addition, a lack of training, other than experience of writing essays, heightens the conditions for a lack of clarity in annotation discourse.

The above point refers to what I discussed earlier in chapter 2.2 and the quote the "...*Emperor's new clothes*..." (Knoblauch & Brannon, 2006, p. 2). The quote referred to when educators maintain an "illusion" for themselves and others that annotation on essay composition can lead to any measurable improvement, despite a lack of empirical studies to back up that assertion. However, in the "we are not trained" extract, the annotator indicates they might agree with Knoblauch & Brannon's quote that the "...*Emperor (still) has no clothes*..." (p. 2). Hence, the annotator's concern about their lack of training is heightened due to the reflective consciousness and lack of evidence about annotation. With a lack of evidence of annotation, the annotator instead reverts back to their knowledge-in-use based on experience which may be flawed because it might refer to the abuse of past memories recalled (Ricoeur, 2006). The next section examines the phenomenon of memory to annotation in more detail.

10.6 Memory and the absence of the thing remembered

The research extract "we are not trained" indicates a problem for nurse education. The problem starts first with the well-meaning annotator's lack of linguistic training and as the last paragraph demonstrated, there is a lack of evidence on the effectiveness of annotation for teaching and learning purposes. Second, when annotators are referring back to their clinical experiences, which may have been some time ago, then they are literally recalling events. Lecturer 1 mentions this as an issue and what is referred back to in the absence of training and proof, is a potential abuse of memory recall and the annotator's perception of nursing issues and past experience. This is potentially problematic because Ricoeur (2006) suggests the truth of memory is challenged by the absence of the physical thing remembered and memories being a flawed repository for learning. Therefore, misunderstanding is heightened

because knowledge of ideas informs experience of language when listening, reading or communicating something to others (Ricoeur, 1976).

We have seen in chapter three's literature review, that most annotation comments are done quickly and with little time to think, so it would seem the conditions are ripe for communicating something unwanted and intemperate to the student. In speech or text, this process is cyclical and closely linked to the restrictions of self-understanding based on life experience up to that temporal point (Ricoeur, 1976). A lack of time to think is problematic because Ricoeur (2006) suggests memory constantly informs perception. In reality, memory is practically overtaken by the events of the moment to blur the truth of a remembered experience. This blurring of reality is due to the absence of the "thing remembered" and the mode of its representation, it is vulnerable to the three acts of memory recall (p. 58) which I now discuss to promote their relevance to the research data on annotation.

First, Ricoeur (2012) suggests the ethico-moral memory occurs when "...commemoration rhymes with rememoration..." (p. 57). Commemorations are memories which are recalled, celebrated and respected and involve a sense of obligation to the past. The term rememoration refers to the truth about a memory, the intention to remember and what may change when remembering it after a period of time (Ricoeur, 2012, p. 57). Both terms have the potential for use and abuse in the act of rememoration. Second, on a practical level, a manipulated memory moulds memory to fit selectively to the present situation. This manipulated memory is pragmatic and therefore changes with each situation. Lastly, the disturbance of a blocked or wounded memory (therapeutic and pathological level) identifies the potential for something communicated, "that should not be" which I discuss later in

relation to transference hypothesis. I will now examine how the research data relate to Ricoeur's three acts of memory recall in more detail.

10.7 An ethico-moral level of memory

Ricoeur's (2006) ethico-moral level is the kind of memory that feels obligated to remember the past. So for a nursing lecturer this would include being informed by their own clinical experience, but also the historic and traditional aspects of nursing, perhaps views about the founding principles of the NHS in relation to recent reforms, changes in nurse education, practice or the key recommendations found in *The Mid Staffordshire Inquiry* (2013) about maleficent care. The obligation to the past would also refer to key people, incidents and experiences that forge a worldview about nursing that may be at odds with new ideas, practises or philosophies of care. Therefore, the use and abuse of memory relates to a nursing philosophy that drives ethic-moral practice, to do no harm, to identify competence in others and oneself in the act of rememoration. This is demonstrated in the research extract entitled "a bit cross" below:

"You get a bit cross with the student and a few times I've thought comments like "No, why have you written this?" and "No, I've not understood this at all" to show irritation would be ok. I am the guardian of nursing in identifying fitness to practice issues, and poor practice which...could lead to harm...I think "No!" that wouldn't have been tolerated when I was a ward sister but I wouldn't put that because I fear how it may come across. I would get quite irate and that the difference between us and something like engineering course where you only have to learn ratios and facts. But then I think in the past I may have thought the same, or written the same and time changes your thinking, and meaning changes with the context, now I fear I may make a slip of the tongue which would be problematic...well some students' use inflammatory statements sometimes and that riles me a little bit." (Lecturer interview 6, see figure 16)

The annotator in "a bit cross" identifies the ethico-moral drivers that ensures their assessment of students do think and act professionally within a beneficent nursing philosophy. The quote "*No, I've not understood this at all*" reinforces that a single declarative word, for example, "no" and "what," may also indicate the annotator has not understood and so parallels a lack of clarity. What "inflammatory statements" triggered the annotator's irritation are not stated but they clearly demonstrate the annotator's irritation and view of professional norms. The inflammatory statements perhaps are those that depersonalise the individual, a lack critical argument or grate against the perceived obligations and traditions of nursing. The annotator is stating they are the "guardian of nursing" (see also "defence of nursing" extract, chapter 9.6) and the reference to "flawed" statements indicates a semiotic impact on the annotator due to the perceived obligation to nursing and memory, which may lead to slippage in reply to the essay content. This is perhaps an example of a manipulated memory leading to a critical stance which was tempered in the research extract "a bit cross" by them remembering that in the past, they too may have written something ill considered. This is where the ethico-moral memory recall merges with the practical memory in promoting professionalism.

10.7.1 A practical level of memory

Coming from clinical nursing the lecturer typically brings with them clinical skills, competencies and experience and so at a practical level they will refer to these qualities first in order to negotiate a path in education. This condition is demonstrated in the next research interview extract entitled "ethos:"

"I genuinely think that when you move from clinical practice into lecturing practice you bring the same mentality and ethos across. Everyone in our school of health has worked within the NHS or private practice and is a nurse or healthcare professional so you cannot help bringing those core values and beliefs into how you practice. Despite not having had experience of clinical supervision I often think back to my nursing experience when annotating because that is the context of feedback. I also relate all I read to the nursing context whether a student essay or a research paper. I generally think you can have both sides of the coin, being proactive or very reactive. Either way it has an impact on student feedback...." (Lecturer interview 1)

The research extract above suggests the movement of core values and "ethos" from the NHS into nurse education occurs and annotators "...often think back to their nursing experience when annotating..." The "a bit cross" extract when stating "I am the guardian of nursing" also reinforces that what is transferred is of relevance to the examination of practical and manipulated memory. The interviewee in the research extract "ethos" also states they have not had experience of clinical supervision in practice but do refer back to their nursing experience which ensures the conditions for a manipulated memory are heightened due to the reflective consciousness (see chapter 1.2.3 to 1.2.5). The annotator when relating "...all I read to the nursing context whether a student essay or a research paper..." indicates their central position. Learning and teaching is a two-way process and the reflective consciousness of the annotator is an important process in assimilating new knowledge, skills or competencies into their professional practice. The perception of what is recalled in class, with a student or in annotation should be used to develop a new perspective about the experience. However, after a few years teaching and referring back to clinical experience, or not if new subjects are taught, what is perceived from memory may change to be untrustworthy. This is

because time tends to reduce the amount of detail that can be remembered, such as the trigger of emotion.

Ricoeur (2006) suggests untrustworthy memories occur because a reflective consciousness should always search for alternate ways of perceiving memory in order to learn from the bank of memories. Perception makes who the person is based on the signification of language, experience and memories manipulated to the practical situation, so long as the experience remembered leads to learning (Ricoeur, 2006). However, the regular effect of such a recall process means what may be re-fashioned for the practical use of memory may distort the truth of what is recalled (Ricoeur, 2006). This means annotation could consciously be used to communicate something useful to the student or the opposite, because a manipulated memory reveals a view that may not have been fully articulated before, until that moment of sharing it. This mimesis3 phenomenon, relating to the latest and fullest realisation was discussed in chapter 8.8. The message therefore, may come as a surprise, especially if little thought has been given beforehand to the immediacy of ideas expressed in annotation. This is a normal phenomenon of speech and when thoughts are written down they also indicate a world view open to challenge and interpretation (Ricoeur, 2003). The manipulated memory, Ricoeur (2003) suggests, may reveal the perceptions that make the person who they are as individuals (ipse) in coming to terms with painful professional memories (idem).

10.7.2 The wounded memory

The link between student and lecturer perceptions can be found in Ricoeur's (2006) third act of memory which relates to disturbance of a blocked or wounded memory on a therapeutic and pathological level as a hermeneutic tool (see chapter 4.8.4). In particular, the first two acts of memory, ethico-moral and practical were relevant to some of the research data extracts; also exist in unison with the wounded memory of the annotator and student, but at varying stages of being triggered. This was alluded to by the annotator's comments in the research data entitled "unbelievably powerful" below:

"We are unbelievably powerful people and we have the power to inspire, enthuse, intimidate, frighten, de-motivate and destroy or not care. We can do that with our looks (because we are all experts in communication), what we say and certainly what we write and I have become aware that annotation can carry that attitude..." (Lecturer interview 8, appendix 2, indicated by four asterisks)

The "unbelievably powerful" research extract identifies the power of language when used with the perception of authority. However, the quote "...we are all experts in *communication*..." is perhaps said with irony. The extract identifies the polar opposites of inspiration and destruction through the annotator's comments which may carry attitude and tone. These are the attitudinal conditions in which all three acts of memory dwell in triadic unison, and each competing to blur the reality of the past recollected by a trigger in the present. Ricoeur (2003) calls this a dissociation of meaning from the present phenomenon to the past and the following research extracts identify the reality of the third kind of memory abuse, the wounded, disturbed memory being relevant to the student experience. The remainder of the chapter develops the reflective unconscious, the wounded memory and transference hypothesis (Ricoeur, 2006; 2012).

10.8 The reflective unconscious

The research interview extract entitled "unbelievably powerful" rests on the perception of annotation by the giver and receiver. As previously discussed, words can evoke an emotive

effect on the student and this cycle starts when reading, writing and finally, when reading annotation. However, the last part of this cycle further reinforces the potential abuse of memory due to the delay from essay submission to receiving annotation. How the student perceives annotation will change due to different locations when reading, at home or on holiday and meaning of annotation changing in time. However, an examination of reflective consciousness and memory would not be complete without developing the antithesis to reflective consciousness and that is the reflective unconscious.

The reflective unconscious involves the annotator, student and any future reader of annotation related to temporal understanding and mimesis3 I discussed in chapter 8.8. This refers to a temporal state of flux and a fuller understanding of annotation when a fuller grasp of meaning is gained in time. In addition to this, perception also changes after reading more to change meaning and the clarity of discourse (Ricoeur, 2006). The process is cyclical and constantly evolving with the abuses of memory for Ricoeur to suggest that unconscious thoughts inform conscious thoughts. The following research interview extract "sometimes offend" demonstrates this point:

"The one only time I received very negative feedback has stayed in my memory but it was vital to me passing essays...I was appreciative of the help, but the negative comments made me rethink my essay. The annotation can sometimes offend you if you think you have done a good job because it seems to be irritated, with an odd tone to it, like being facetious..." (Student group interview 14)

The research extract "sometimes offend" states annotation to have mixed qualities such as being helpful at times, and at others having an "odd tone" and being offensive. However, words like "irritated" and "facetious" indicate something is projected from the annotation. Whether it is the annotator transferring something, perhaps emotion to the student or the student transferring something to the annotator is unclear because neither is made explicit. The research extract "a bit cross" however when stating "*No, I've not understood this at all...*" then attempted to avoid showing any irritation in their annotation leading to the conclusion that something unknown had finally been made conscious (Ricoeur, 2006).

Ricoeur considers the concepts of transcription, transposition and projection of a thought to be equivalent in language [homonymy and synonymy] (Ricoeur, 2003). This is of relevance to the perception of annotation because in the search for clarity, the same name is given to the same thing, yet the unconscious chain of expressions and substitution means the mind may not yet see the difference between thinking and saying (Ricoeur, 2003). Again, the abuse of memory is applied in the unconscious projection of discourse which is spontaneously communicated in what Freud (1915; 2005) called "slips of the tongue" and Ricoeur "slippage" due to repression. In the case of annotation with identified tone what is projected risks being a "slip of the pen," or less poetically, "a slip of the keyboard." Ricoeur suggests the unconscious expressions of language should be viewed with the mind as a virtual dynamism not a rigid structure because the mind attempts to make sense of uncertainty though socio-linguistic pattern formations (Ricoeur, 2006; 2012). Unconscious thoughts therefore link the symbolism of language through the manipulation of the meaning it has for a person. The next research extract called "you get annoyed" identifies the annotator's conscious and unconscious projections:

"...you get a bit annoyed with the student especially when the essay content does not reinforce the necessity of person centred care, so I wonder what they'd be like in practice. A few times I've thought "No, that's irrelevant to care, you've got it wrong!" An example I can think of now is the ethics assignment they are doing now and I am very big on the mental capacity act and when they get it wrong I get very cross and think "No!" I wouldn't put that so I have to write something else..." (Lecturer interview 6)

The annotator is referring to their inner thoughts when reading a text and the priorities for them when searching for essay content which relates to moral-ethical practice. This is another key extract, like "a bit cross" because the annotator is being very honest about what they were thinking when annotating back to the student. What is translated back to the student demonstrates the internal mental image can be transferred in annotation even when attempting to avoid slippage, for example, when thinking "no" then annotating. The "…you get a bit annoyed…" quote therefore indicates the risk of projecting attitudinal tone which I discussed in chapter 3.6.3 to 3.7.1. Perhaps, "finding the right pitch" is an apt metaphor for annotation and nurse education.

Ricoeur (2003) suggests textual tone is linked to the Aristotelian notions of mimesis, where imitation is produced by the rhythm or cadence of music and language therefore evokes a mood in the intonation of voice and the written word. Hence, the annotator when giving feedback on the page of the student's essay is not only aware of their role as a nursing lecturer (mimesis), but they are aware of themselves as an educationalist assessing essay composition (plot) and persuading the student about the need to make the necessary revisions (rhetoric). However, I would like to develop in the remainder of this chapter, feedback when it first applauds, reinforces or corrects and, second, when feedback is perceived as "negative" or "destructive" as one student (student interview one) states later in an interview extract. I

use the word "perceive" carefully because it appears something is projected and something is registered.

10.8.1 Perceived tone

The annotation tone of the "you get annoyed" research extract, to use another metaphor, is like "entering into uncharted territory" because the mental environment is unknown and unfamiliar and there is a possibility annotation may lead to a degree of surprise for all involved. This issue is evident in the contrasting moral nature of annotation feedback, to do no harm and the potential to do emotional harm. If the annotator is aware of feeling irritated, as the "a bit cross" research extract indicated, it is likely they would not intentionally be unprofessional, due to modelling a need to be constructive and risk of complaint. Therefore, when annotation has tone, what is projected by the annotator is more likely to be pre or unconscious and perceived very differently to that which was intended and, as as I discussed in chapter 3.7 and 3.8, it may remain hidden from the annotator as well. From a student's perspective of annotation, the conscious, sub (out of immediate awareness but accessible) or unconscious effect relates to their awareness of its cause and effect in time. This refers to the student's own reflective consciousness and the following research extract entitled "what is underlying" below demonstrates this issue:

"I mean some comments are so well thought out that I ask myself, what do they really mean? I'm sure when they read my essay they had spontaneous thoughts but you wouldn't read that into the feedback comments, so it's what is underlying that makes me wonder what they actually mean, and that makes me unsure...then when I read their comments I wonder why I experience them so differently..." (Student focus group 6) The research extract asks an important question because the student intuitively thinks something else is referred to yet currently fails to understand the nuances of meaning. I say currently because with time, experience and more reading around a subject their perceptions will change (see chapter 8.8). Ricoeur (2012) suggests the relevance of intentionality concerns the reflective consciousness of a human being to the extent that we are unconsciously empathic and concerned about the other person. The intention of annotation and student perceiving its relevance is therefore of concern.

Conversely, the notion of self-awareness, according to Ricoeur (2012), is secondary to the unconscious empathic communication for two reasons. First, because the unconscious is rarely reflected upon and second, the mass of unconscious stimuli and memories always outweigh what is consciously known. Ricoeur (2012) suggests we can see a glimpse of this unconscious phenomenon when taking the time to think about meta-cognition. What is remembered is changeable and what gets in the way of truthful reflection is consciously thinking about something only after avoiding the distraction of other activities (Ricoeur, 2012). Hence, Ricoeur suggests life being filled with distraction means when taking time to notice things, it suddenly heightens realisation. Thinking then can be worked through to clarity when thinking about meta-cognition itself (Ricoeur, 1994). This reference point gives a glimpse of the unconscious memory and why, as Gadamer (2004a) suggests, language is inadequate for the purpose of ontological analysis (Regan, 2015). An accurate use of memory is considered impossible to achieve, due to the status of the unconscious and the knowing subject (Ricoeur, 1970). As the next research extract entitled "no supervision" suggests a lack of reflection may be part of the reason why "slippage" and tone occurs in annotation:
"If lecturers don't get supervision, then isn't feedback a little one sided? How can they promote our reflective practice when there appears to be little chance they have to make sense of their comments to themselves and others...?" (Student focus group 8)

The research extract entitled "no supervision" identifies a key issue for annotation in that the un-reflected upon according to Ricoeur has primacy over what is reflected upon because it remains a prominent assumption under the guise of "knowing something to be true" (1970, p. 379). The student interviewee makes a valid point that nurse education may not be practising what it preaches when not promoting reflection on educational practice. I examined this issue in chapter 1.2.3 and 1.2.5 to be a significant issue for annotation because of the potential for "slippage" and viewing memory on past experience as being a valuable learning tool. As stated, an emotional link appears to still exist between past clinical practice and the present memory recall to remain active in a state of unconscious repression. As a result, this emotional link remains un-resolved and yet to be let go of (Freud, 2005; Ricoeur, 1970), to be then grasped by the student in annotation. This is an issue for annotators because as Feito and Donahue (2008) suggest annotation is never neutral or arbitrary because they are the result of an immediate interpretive decision. Let me develop this issue further which the student in the next research extract identified as being significant to teaching and learning.

10.8.2 The student as a child

Annotation because it is momentary hints at the inner thoughts of the lecturer and is incomplete because it is left up to the student to grasp the intentions behind it. Ricoeur suggests the transference hypothesis helps to understand the process of learning from the effects of discourse in the act of recognition. This process starts from identifying the trigger to the present which is a commemoration or rememoration of past events, and individuals find ways to negotiate a sense of professional dissatisfaction and repression when dealing with external and internal resistance (Ricoeur, 2012). This is what Ricoeur (2006) calls the wounded memory (p. 69). In order to reinforce this phenomenon, a variety of students in the research data had declared annotation led to some form anxiety and hurt. The following research extract entitled "it cuts deep" demonstrates this pain:

"Reading the written feedback on my essay made me want to cry. I'd put so much work into the essay, it's as if I'm on trial...it cuts deep...I felt disheartened ...the feedback was all negative so it made me feel unmotivated and disheartened and defensive..." (Student focus group 1)

The research extract "it cuts deep" indicates an emotional effect on many levels: physical, emotional and even spiritual in the form of being wounded and a remembered child-like situation of being made to feel vulnerable (Ricoeur, 2012). Then to find annotation comments which they do not understand fully, do not accept, or appear to be written with a degree of tone, may lead to the repetition of past associated emotions (Ricoeur, 2012). Ricoeur (2012) therefore suggests discourse is the link triggered by past experiences repeated in the present which still remain active in a state of conscious repression, un-resolved and "yet to be let go of." I make note of the effect and the cause because annotations are never neutral or arbitrary and represent an immediate interpretive decision (Feito & Donahue, 2008). Hence, the past informs how annotation is experienced in the present. The following research interview extract entitled "crushed" is more profound:

"My essay comments were very patronising in parts, there was a flavour to the comments, they made me wince and tears welled up in my eyes... I'm usually able to

deal with feedback but this felt personal... The last essay feedback crushed me...it drove me to tears...I couldn't get the lecturers' words out of my head for ages...and it took me a while before I got less anxious about essay writing..." (Student interview 2)

This kind of "crushed" disclosure indicates the emotional reality that some students have to endure when receiving annotation feedback. The emotive language indicates how annotation can be perceived as personal loss despite reason. The response for the student was again physical: "wincing," tears, anxiety and unpleasant memory of the language used. The annotation therefore indicates the temporal nature of transference hypothesis and words from the past triggering an emotional reaction in the present (Ricoeur, 2012). The annotation itself was past, as it had been written on the student's submitted essay, and that indicates the effect of memory recall on the student's perceived effort of writing the essay. The temporal nature of the wounded memory is a mental defence to an impending threat, which I discussed in the "we are not trained," "what" and "no" extracts and the possibility of negative feedback dominating nursing experience heightens this state of hypervigilance. The next research extract entitled "negative, negative, negative" reinforces this phenomenon and the reasons for evoked emotion:

"I had an abusive upbringing and if someone takes that (annotation) tone with me then I either react aggressively or I revert, and it is very much parent and child. It invokes the same response in me as it did when I was a child and I'm very aware of this but if you're constructive with me then I can build and move on with it.... I failed the assignment and the feedback I got I found very destructive. There weren't any positives; it was all negative, negative, negative. Even the bit I firstly submitted to look at for supervision suddenly wasn't ok by the time it got submitted whereas it was for *supervision. So I found it very destructive and it had a very negative effect...*" (Student interview 2)

The "negative, negative, negative" research extract identifies the emotional reality of the student's current experience of annotation affects their self-esteem to the extent that the student reported feeling emotionally affected for some time. The student discloses an abusive childhood, which they were attempting to work through. The ensuing emotional reaction from the "destructive" annotation relates to the "we are not trained" extracts which felt overwhelmingly personal and negative and which may be literally related. The student identified that annotation "…*triggers something and you see something you are expecting*…" Therefore, there is the possibility of transference or counter-transference and "something" being transferred in the "negative, negative, negative" extract.

If the student had passed the essay then perhaps the annotation comments would not have been perceived to be "negative, negative, negative" and would have been positive. This is hard to say as the phenomenon was linked to being negative and judgemental because failing an essay is a negative judgement in reality. However, the perception of negative tone "...triggers something and you see something you are expecting..." indicates a state or preparedness for negative comments for psychic defence (Ricoeur, 2012). Again, this is presumption and does indicate the transference of the past behaviours replicated in the present and the role of perception in understanding the present annotation. The following research extract entitled "prepare to fail" identifies this preparedness for failure phenomenon in another student: "As a child, I moved around and went to six different schools before I was eight and I had a lot of different teachers during that time. I remember two specific teachers, one was when I lived in the Middle East and if you did any good work she used to draw a smiley face on your hand and you were always special if you had the smiley face on your hand. Then when we came back to England I had this other teacher (who I still see now and I quake when I see her) who humiliated me. I had only been in the class a week, so I knew straight away that I had to impress the teacher in order not to be labelled stupid...after that with that teacher I was never liked just because I think it's the same. I think that's why I look at the markers hand writing style.... if you're not a very confident person from being a child and you read somebody's handwriting and associate it with the same feelings you had as a child then it's not going to do your confidence any good. So therefore it could even prevent somebody from putting the effort in or giving up half way through and saying "I'm only going to disappoint them again anyway. Whereas when annotation is done digitally you can't interpret it you have to just read the words and then see the comments..." (Student interview 3)

The "prepare to fail" extract and the student's experience as a child of disappointing a teacher is evoked in the present, relates to the abuse of authority which led to the student's anxiety in the first place as a child. The quote "*I'm only going to disappoint them again anyway*..." is sadly, not uncommon in the childhood experiences of adult students due to less than humanistic styles of past teaching (Freud, 1992). This student, when a child, had learnt to look for cues from the way the teacher wrote and the tone projected to them in annotation. They were in reality trying to write an essay that "pleased" the teacher and to avoid the label of being called "stupid" and "unliked" that could be interpreted from the handwriting style of the annotator. Therefore, the expectation of tone may indicate a necessary state of preparedness of negative comments for psychic defence (Ricoeur, 2012). The research extract suggests by not seeing the style of handwriting, digital annotation may present a sanitised form of discourse and in a format that lessens communication and apparent mood of the annotator.

The "prepare to fail" research extract clearly indicates the child-like reactions of an adult when presented with triggers that replicate emotion from the past (Freud, 1992). This is why Ricoeur (2012) used Freud's hypothesis to understand the emotion of interpreted language. The emotional pain is psycho-somatic and indicates an unconscious and unresolved issue from the student which could be further explored if the student went to the annotator to discuss their reactions as an adult. The research extract below demonstrates the emotive physicality of annotation entitled "physical pain" below:

"It's like a physical pain, like your skin is made of nettles and prickles. And what was needed is what I could have done to build on it to make it better. But there's no way of doing that and alleviating the prickling. It's destructive... That's the only word I can think for it - it completely crushes you. It's just an essay, but it's so important and we want to get it right. So, if something is so destructive it demoralises you and you really don't want to go back into it because you just look at it and say "oh no, I don't want to look at it again" but I've got to, I've got to do it..." (Student interview 2)

Reading the student's research extract is also painful because their emotion is raw and unresolved. The student reports a strong emotional and physical or psychosomatic pain, which reinforces the physical manifestation of the linguistic phenomenon caused by annotation. The effect is corporeal and existential (Freud, 1992; Ricoeur, 2006). The annotator however, would not have been aware of the student's history or the effect annotation had on them when reading it. There is the likelihood that if sent to another student, the same annotation may have been perceived as benign. Hence, on this occasion due to the student's disclosed past trauma, it is their emotion that appears to evoke a transference of emotion back onto the annotator when meeting them soon afterwards for supervision or in the future. The wounded memory therefore appears to relate to the reflective consciousness of both the annotator and student alike as they progress in nursing.

10.9 Conclusion

Ricoeur's (2006) three acts of memory underline the conceptual framework for this chapter's exploration of the meaning of annotation, the reflective consciousness and slippage. The premise of Ricoeur's (1970; 2006) underlying term, the reflective consciousness is a forgotten memory likely to be communicated at some point in discourse because it remains in memory, ready to be used, abused and finally translated. How the theme was identified was then explored and I presented a variety of research extracts which identified the use of single word tropes, imperative, proof and projected tone to the student. From discussion emerging from the research extracts the chapter later was organised using what Ricoeur (2006) referred to as the use and abuse of memory: ethico-moral, practical and wounded memory in commemoration and rememoration. This model allowed me to then explore discourse and memory and from the student interviews attitudinal tone perceived in annotation. Ricoeur's reflective consciousness was then examined and the automatic processes in the mind waiting to be recalled from consciousness. This last point brought me to discuss the idiom of slippage and the mass of memories which can help reflection once they become known. What is forgotten by the annotator indicates the abuse of memory as a premise is likely to be communicated consistently to the student through slippage. The student too, in their developing reflective consciousness, will be exposed to the same processes in their reactions to the annotation comments. The notion of transference hypothesis communicating something from one person to another, in conclusion, is that words are powerful and the impact of annotation discourse may be unknown and unintended.

Chapter Eleven

Discussion



11.1 Introduction

In this chapter I discuss the thesis' original contribution to current knowledge. Organised in two parts, one is written in relation to exploring the meaning of annotation in nurse education; temporal understanding and the hermeneutic self. Part two is written in relation to limitations of the study, recognition and misrecognition.

The research themes identify that annotation use in nurse education has a set of organising principles informing textual action. The intuitive use of rhetoric promotes an empowering philosophy of nurse education expressed on the page of an essay. As nurse educators we promote both the art and science of nursing. The rhetorical orientation promotes principles of empathy, care, autonomy, equity, individuality and the binding social contract that drives the annotator to view an essay as a safe space for students to experiment. The last organising principles relate to the hermeneutic self of the annotator, student and memory recall organising knowledge and experience. However, due to the temporal processing of stored memories, recall and the present context shapes the meaning of interpretation. This brings me to the difficulties of interpretation.

The difficulties I wrote about in chapter three's review of the literature are examples of the hermeneutic process of reading and writing which were prevalent throughout the thesis. I had written about Ricoeurean theory in chapter four but it occurred to me later that writing the thesis paralleled the research methodology itself. The process was emerging and temporal. A turning point for me was reading Boell & Cecez-Kecmanovic's (2010) paper to influence the literature review, and applying theoretical links to analyse first the literature and second, the research data. The writing process was fluid with many of the chapters written and re-written consecutively and in varying temporal and spatial locations. Therefore, when I collected the

research data in December 2013, the themes began to emerge six months later in June 2014. I had not realised that the hermeneutic writing process also related to the literature review (Boell & Cecez-Kecmanovic, 2010) and once remembered, my part in the research process became more dynamic. Every last revision led to the latest realisation, and when re-visited at a later date, they were replaced by the next (Ricoeur, 2003). I concluded at the end of chapter three that the gaps in the research literature did two things: they directed my enquiry to explore what annotation feedback stated, and what it did not state but indirectly inferred. Following on, I explored the meaning of annotation in nurse education through the lived experience of the annotator and the student, informed analytically through Ricoeur's textual hermeneutic. In chapter three I used a simple three step stage of reading the literature (see appendix 1), which unbeknown to me at the time imitated the praxis of plot, mimesis1-3 and the hermeneutic circle I wrote about in chapter 4.8, chapter 6.3 and my research notes (see appendix 3 to 6). I was beginning to make links between theory and its textual application.

Writing chapter four allowed me to organise Ricoeur's theory and gave me an anchor from which to analyse the research data. The opportunity to understand the hermeneutic circle as it was being experienced revealed my need to have some understanding in order to link ideas (mimesis2-3). I realised I had written about this process theoretically (chapter 5.7.1 to 5.7.3), in organising the research data, later identifying the four key themes (see chapter 6.8) and thematic analysis of each chapter. The next significant example of temporal understanding was writing transparently throughout the thesis to "work through" any interpretive bias I had in relation to the hermeneutic circle. As I wrote chapter five's research methodology I was acutely aware of the need to balance my own pre-conceived ideas and experiences (see Regan, 2010) and being open to glimpse new phenomenon.

The hermeneutic circle was a constant process throughout the thesis and I realised it could be identified in the research data, (students and annotators), supervisors and my engagement with discourse. I did not fully understand the hermeneutic circle until near the end of the research process yet working out pre-conceived ideas is crucial to the hermeneutic circle because it is a virtual, psychic phenomenon and in a constant state of flux. In other words, it is difficult to identify the moment ideas change. These levels of immersion are triadic, to use Freud's terminology, because preconscious, unconscious and conscious terms relate to what one thinks in time and how understanding changes over time. Self awareness, recognition, and understanding are temporally cyclical and in chapter six the research method came alive and the clarity of discourse, so significant to interpreting the research data and Ricoeur's theory, was worked through in the first two stages of van Manen's (1997) three step model to identify the emerging sub-themes. Hence, the hermeneutic circle was not applicable until I had enough research data to engage with. These issues relate to what Ricoeur (1988) refers to as doubtful aspects (an aporia) of temporality that appear to configure time and narrative when superimposed onto the lived experience. Thinking about discourse in the past, present and future fails to take into consideration the notion of self-constancy, temporal understanding and the hermeneutic self, which I discusss in the remainder of part one.

11.2 The hermeneutic self

Reading and "grasping" the meaning of the research data paralleled the textual difficulties of reading Ricoeur's work, and identifying the four research themes in a transparent way (see table 3 and chapter 6.8). When writing I was keenly aware that the "object" was the "thing" being read, and I was the interpreting (hermeneutic) self. This process parallels the experience of the student, the annotator and myself in writing about the clarity of discourse in annotation. The narrative identity, the hermeneutic self, is communicated in the temporal

dimension of emplotment because "all of the time" the reader's inner voice, their constant awareness of themselves, is at work to find resonance and meaning of the text. This obvious phenomenon is easy to forget about because it is constant and when I noticed this, the theme reinforced readings of a self-referential nature. When the student decides to write, another temporal process occurs, and it is no longer the temporality of the text but the temporality of the narrating self, the inner voice talking to the reader now writing (Ricoeur, 1985). This is the narrating voice of the reader when the choice of grammar speaks and a temporal revelation of the narrating self, and the hermeneutic self becomes resonant, as found in the sub themes of table 3 (see chapter 6.8).

Ricoeur (1985) suggested an individual's almost constant sense of themselves is an ontological concern to identify the relevance of the thing, the object, and in this sense, literature and the reader. The narrating voice of the annotator is suggested to be self evident when subjectively judging an essay content and "...looking at a textual mirror of themselves..." (Straub, 2006, p. 4). Again, this relates to self-understanding and interpreting language. Like the student starting a sentence with the word "it" the annotator who did not perceive the theme as an issue, and there were many, may have viewed the dialectical posture of the student as a developing linguistic style. However, this style of writing is developed when thinking and "holding" the terms, concepts of an-other in their mind's eye and writing whilst attempting to make sense of the literature. The student is trying to understand the text, or some of it, and the theme demonstrates to a certain extent, their narrative identity, because interpretation has to pass through the filter of the reader's self-awareness. The knowing student who identifies bias in his or her own writing will reduce its use in time, as I did when realising my own naïve interpretations paralleled van Manen's (1997) first or second order interpretation and explanatory stage of understanding. The theme addresses a gap in the

literature and Feito and Donahue (2008) (see chapter 3.6.1) failing to develop the hermeneutic aspect of their study. Their discussion, possibly for reasons of word count, could have included memory, understanding and the power of text to transform understanding for the reader were not discussed. Therefore, the theme explores the hermeneutics of the self, memory (in chapter ten), recognition and the representation of words to create meaning through rhetoric. The difficulty of understanding the meaning of language is made more explicit in the next theme of rhetoric.

11.3 Rhetoric

The second theme synthesises Ricoeur's new rhetoric to frame the meaning of annotation: through persuasion, "saying it well" with proof and the use of metaphor to teaching and learning. I mentioned previously that this organising system may be obvious to some, but for myself it was unknown at the time and from a nurse education perspective, it appeared to fit well with nursing praxis for the greater good. The question of temporal action was a new addition by Ricoeur to rhetoric which he developed to understand discourse and the temporal process of recognition. Ricoeur had re-conceptualised emplotment through mimesis1-3 and the hermeneutic (or the narrating) self, and rhetoric depends on the inner world of subjectivity and what the reader "wants" to understand. This is of significance because the ancient use of rhetoric had resulted in metaphor considered to be a redundant discipline from Aristotle's original triadic model of argumentation, with proof, eloquence and persuasion restricted to a theory of style, then a theory of tropes. The use of metaphor and the productive imagination enabled students to understand and the novelty of finding there was a pattern to understand the research data related to my sense of revelation in finding a conceptual model to organise the actions of annotation. If I had been an educationalist first, like Knoblauch and Brannon (2006), or a philosopher, then perhaps the theme may not have been identified, or organised in such a way due to being obvious. I had to accept the value of the research process because I rationalised if I did not know about the theory then it was likely my nursing colleagues and students were unaware of it too. This made the theme more relevant and probable. Therefore, in my case, for the first time I felt the development of meta-cognition because the clarity of writing, often lacking until reviewed, appeared to directly relate to the clarity of my thinking. The theme therefore offers a way of structuring and conceptualising annotation, addressing the gap in the literature to organise its eclectic characteristics and be applied to the research process and data.

The rhetoric theme addresses some gaps in the literature which Liu (2006) and Ball et al's (2009) research briefly mentioned. Ball et al's (2009) research findings identified that feedback needs to be worded sensitively, be constructive, aware of the implications of tone, be transparent, motivational and promote confidence. However, a finding was the annotator thinking back to what they had thought and not remembering what they had been thinking at the time (see chapter ten). Liu (2006) suggested that due to the issue of time lapse only the student would know if there had been any internalised thinking. However, both studies did not develop how annotation related to temporal action, perhaps due to the word count restrictions of published research, which is a gap in the research. Therefore, the application of rhetoric to annotation illuminated the issue of temporal action which I examined in chapter 8.5. The theme moved beyond an initial understanding of the imitation of human action to view discourse as changing in time due to the reading act. Rhetoric identifies the imitation of writing and thinking about nursing praxis occurs both passively and actively, whether textual or through observed action. What the rhetoric and mimetic theme identifies is unique in its application to annotation for the student and annotator, because the act of imitation takes time to develop before a closer approximation of action is obtained. Connors and Lunsford's

(2006) 1986 research study found that 77% of teachers had written rhetorical comments on 2,297 marked essays but appeared not to comment in personal or polemic ways and instead suppressed value laden comments. Connors and Lunsford's (2006) study did not develop any specific conceptualisation of rhetoric nor did any of the research studies cited. However, the rhetoric theme developed an insight into meta-cognitive processes by suggesting new rhetoric may be helpful to understand how deeper learning occurs when students are capable of producing insightful annotation themselves. The mimetic act of imitation, which the theme reinforces, is modelling nursing and imitating writing styles, one of which is annotation as discourse. The gap in the literature was lessened, I believe by the examination of mimesis1-3 to conceptualise human action, thinking and understanding discourse in all interpretive stages. Knowledge once acquired changes, is tested, refined and new ideas merge without trace of what it had replaced. The relevance of this process is to take notice of the strength of proof behind an assertion and to identify in a meta-cognitive manner, what new proof is required to change the readers view point. The issue therefore, is to be wary of the burden of proof, the rigour of argument and evidence influencing new perspectives, and be suspicious of the text (see chapter 2.7.1). The process is both hermeneutic and individual, which brings me onto the next theme of individualism.

11.4 Individualism

The third research theme, individualism, discusses the legislative authorisation given to nurse educators and as annotator's their responsibility to promote what is "just" in society. Section 9.4 identified the annotator as a citizen and the underlying principles that a social system should promote, such as equality, self-respect, and autonomy, fair distribution of rights, duties, advantages, burdens and mutuality. The "defence of nursing" was a key research extract because it suggests individualism appears to be a defence against any perceived violation by the state. Therefore, the promotion of individualism in nurse education has to overcome, to some extent, the difference between the individual personality (ipse, see chapter 7.5) and the sameness (idem) of professional nurse education. In promoting the notion of individualism, it is perhaps ironic that in the defence of nursing, the concept of individuality is transformed through an organising system of professional education, that despite its best efforts, tends to objectify people even more. This is because the assessment process helps to overcome any resistance to transformation from the individual in order to progress and conform to collective norms of the nursing profession. This appears to be a necessary stabilising process in "defence of nursing" because individuality is replaced by the collective "we" of nursing individuals to finally become the embodiment of nursing. That is the level at which the annotator appears to be working in order to negate the devaluing impact of systems of governance.

The theme of individualism explores what the research literature failed to do. For example, none of the retrieved research papers I examined in chapter three's literature review considered ethico-moral philosophy, the promotion of social justice, equality, fairness or the risks of technology. This research theme therefore identifies a concern for the nursing annotator motivated by their sense of obligation and responsibility to society, the profession, sense of mutuality, patient care and the next generation of nursing students. In particular, because nursing is an authorised profession, the theme identified the annotator protecting the integrity of individualism against the pervasive influence of technology and the political labyrinth. Technology was not an issue in the nursing research papers but it was in the research data and the negating effects of standardised annotation comments, or annotation received remotely appears to cause a sense of alienation and disembodiment, which the annotator attempts to ameliorate. So this research finding offers an antidote to technologies

uncritical acceptance in nurse education, where the reduced human contact between annotator and student lessens its influence. This brings me to the last theme.

11.5 The reflective consciousness and slippage

The last research theme, the reflective consciousness and slippage suggests the annotator and student perceived annotation through memory recall, which I suggested is flawed. Annotators have a high level of duty and obligation to nursing, patients and society and their memory recall is influenced by these characteristics. However, I found Ricoeur's three acts of memory recall was a convenient model to categorise the effect of memory in nursing, because in a practical sense what is recalled may change to reinforce a practical or ethical point being made. What I mean by this is that the available store of memories may be adapted to "fit" the situation for the student's benefit. However, unless an annotator is regularly in clinical practice, when referring back to clinical practice, with the best of intentions they are referring to their past and notably, a past refigured practically in the present.

The next kind of memory recall, called ethico-moral, referred to the kind of memories that the annotator or student feel obligated to, such as past experiences, motivational factors, significant others, traditions and social norms, patient experience, and perhaps reinforcing the founding social principles of the NHS motivating them to think socially. However, as I found in the individualism theme of chapter nine, the "defence of nursing" ensures annotators are a mediating force against pressures perceived to affect nurse education. Again, past clinical experience reinforces a world view that requires nursing, and patient care to be defended.

This brings me to the last kind of memory recall, the wounded memory and my exploration of the opposite of the reflective consciousness, the unconscious. This is where the mass of experiences are stored and where the above two kinds of memory dwell, yet the unconscious memory relates to something repressed which may be past experiences, ideas, emotions, thoughts and intentions of both the annotator and student. What is repressed may inform thinking and praxis and what is forgotten may be remembered in fleeting moments, before being forgotten again. However, the three kinds of use and abuse of memory recall depend on the reader negotiating the effects of ordinary, phenomenological and narrative time, which makes any memory recalled a refigured construct (Ricoeur, 2012). Discourse (such as annotation and essay content) has the capacity to remind the reader of the past, and a refusal to accept what is repressed is likely to be repeated. Hence, the transference hypothesis was found to be significant to annotation as a form of discourse, because not all nursing experiences are satisfying, and traumas inflicted on the nurse throughout their career may leave an emotional residue. An emotional residue that may lead to memories being forgotten, repressed, triggered, and finally impact on annotation. Therefore, the wounded memory is important as a reflective opportunity to analyse the intentions of the language used in annotation (and essays) in order to identify emotional issues that need addressing in light of the different types of temporal understanding experienced, which I now discuss.

A difficulty in recalling a remembered clinical experience or relevant theory, which may affect the interpretation of an essay content or annotation, relates to the different kinds of temporal understanding I mentioned previously, such as ordinary, chronological time, phenomenological (consciousness) time and narrative identity (Ricoeur, 1985). In ordinary time, the movement of events before, during and after are subject to the organisation of the text in the tensive system, which I realised I had difficulty with when recalling events, and reading and writing analytically over a sustained period of time. The phenomenological time refers to the empirical use of the senses or imagined in the application of theory to discourse, and this relates to all aspects of the reading and research process. This concept identifies a difficulty with memory recall due to notions of pre-understanding (mimesis1), engagement of the symbolic system as a whole (in mimesis2) and refiguration (mimesis3), or understanding in the last instance of discourse, which may all occur in quick succession. Another literary explanation is the claim of signification when enabling the reader to make sense of discourse as the narrative unfolds, which brings me to narrative identity.

Narrative identity refers to an individual's cognition and sense of ontological identity and this relates to the annotator reading and commenting, the student's constancy of self (see chapter seven) and my decisions as a researcher. The sense of the self, according to Ricoeur, is unconsciously forgotten in the moment of distraction and when the mind's inner voice asks the question "how do I make sense of this?" When reading, the constancy of the self mediates between self-knowledge and knowledge obtained from others (either observed or experienced) to identify with. The constancy of the self refers to a person absorbing the literary and professional ideas of others (idem) and a reader constantly projecting their identity onto others. What temporal understanding is lacking, according to Heidegger, is understanding the connectedness of life between birth and death, and what he called stretching-along. The distance between the two is the stretching-along awareness of the lived experience, and if the reader (student, annotator, research participants, me, you the reader) were to think about the variety of our own personal and professional experiences, it may be easy to see how stretching-along can impact on conscious and unconscious actions. However, these flaws are rarely considered when a memory is recalled and refigured in the latest moment of realisation. Therefore, the theme's contribution to gaps in the literature about annotation relates to temporal understanding being implied yet not explicitly addressed in the findings of the literature review, which I discusss next.

Despite being focussed on handwritten annotation, Ball et al's (2009) research study made the point that distance learning has increased the use of digital annotation back to students. With the arrival of digital annotation within higher education impacted on students in a number of ways. For example, a research extract in this theme reported a student whose child hood memories of receiving handwritten annotation on her essays was a phenomenological experience. As a child the student had looked for cues from the tone of the handwritten annotation to judge whether their work was good enough and if they were liked or disliked. The student reported they could sense the mood and tone of the annotator by the style of their handwriting, whether it was angled, appeared hurried, or use single word tropes as explored in the research extracts "no!" and "what" (see chapter 10.4). The visual impact and rich texture of handwritten annotation contrasted with digital annotation which in many ways reduces the visual cues to communication.

Ball's (2009) research study identified the annotator's negative and irritable tone and one research participant stating "...you can see irritability in lecturers marking.... lots of scribbles and writing is quite shocking really; it looks aggressive..." (p. 120). What was not developed in the research study however, was why this was the case. Therefore, the theme of the reflective consciousness and slippage contributes to fill this gap in the literature through Ricoeur's (2006) use and abuse of memory recall. For example, in relation to ethico-moral memory recall views may change due to the development of new perspectives gained through newer experiences to affect the original memory. This kind of memory parallels the theme of individualism explored in chapter nine and being in a position of authority in nurse education appearing to evoke a nostalgic view about society, the individual's experience of nursing and how to annotate. The practical memory and everyday ability to change perceptions about what has been remembered in order to apply it into the present is a common resource for

nurse lecturers. However, it means the past memory may change in relation to the demands of the present. I will now explore the limitations and strengths of the thesis and research methodology.

11.6 Part 2: Limitations of the study and research methodology

Ricoeur's philosophy was not written with the research process in mind, so I had to chose an appropriate method for organising the research data (see chapter six) to then apply his textual hermeneutic phenomenology (Smythe et al., 2008, see chapter 5.6). Like many doctoral research studies (see chapter 5.6), I used van Manen's (1997) three step model to organise the data and simply adapted his three step model for this purpose. In order to ensure consistent application of Ricoeurean theory to the research findings I later realised van Manen's theory was itself adapted from mimesis1-3 and the hermeneutic circle. However, if I had used mimesis1-3 in such a manner it may have confused its analytical application to the research findings. In short this was an example of mimesis3 and realisation.

Ricoeur's genius suited my purpose because his mastery over the philosophical subject made his work far reaching and applicable. An obvious limitation of this research study has to start with me, the researcher because I could only take Ricoeur's word for some of his reconceptualisations, presented as they were written in a rigorous and scholastic manner. The consequence of using Ricoeur's work as a methodology was an act of imitation, of plot, his ideas and style, variety of topics he wrote about and his interest in temporal action. As discussed already in relation to textual hermeneutics, interpretation involves the constancy of the self within the hermeneutic circle and in this sense any strength and limitation of this thesis starts and ends with me. Other researchers, given the same experiences and research data, may interpret something else which makes my contribution unique and phenomenological. Perhaps it is not important, nor possible to fully understand what any author (or student, or annotator) thought and felt about when they were writing, because, it is in the reader's interpretation where literature is brought to life.

Another reason why I chose Ricoeur's textual hermeneutics was due to its linguistic appeal, to understand ontology and the lived experience (Regan, 2012d; 2015). Ricoeur, like Gadamer aimed to address the meaning of ontology that Heidegger (2003) had begun to analyse when stating language is the house of being. However, Heidegger's technological discourse led to criticism that he made the concept and language more technical (Jackson, 1999; Regan, 2015). My problem was similar to Jackson's (1999) criticism of Heidegger when suggesting "…*language (as) the house of Being… (became) a prison house (of Being)* …" (p. 14) and I realised language constrained an interpretation of Ricoeur's work for me too. This was because I was a nurse not a philosopher and yet my research task was to make Ricoeur's theory applicable to nurse education. However, making the unfamiliar become familiar is a characteristic of hermeneutics and its application to annotation and nurse education is therefore original. I will discuss more about the concept of misunderstanding as a limitation and strength of the thesis and methodology.

Distinguishing the truth from what is false leads a researcher to the possibility of mistaking something and misunderstanding it. This means the possibility of failing to understand, and not knowing, are magnified when not recognising a mistake has been made. Therefore, misrecognition relates to the research themes and may lead to ambiguities such as: self-deception, disappointment and lack of confidence. This may result in a fear of making an error, which pressurises the researcher to understand "something" and "anything" other than uncertainty. This brings me to an issue I identified as a limitation to this study which relates

to discussion in the literature review (see chapter 3.5.5) I discussed in the research theme of reflective consciousness and slippage. That issue is the affective reading of low self-efficacy readers in relation to high efficacy readers. The issues related to the confidence and high levels of motivation which the high efficacy reader had, in contrast to the low efficacy reader who reacted affectively. However, if there was low efficacy as a reader and student, this was not assessed because the reading ability of the research participants were not identified as a criterion in the research methodology. Therefore, the link between low efficacy reading and affective reading could be surmised but not concluded from my research findings. Further research on annotation and nurse education may identify reading ability and the perception of failure rather than actual failure (see chapter 10.8.2 and the "prepare to fail" research extract).

A surprising finding in the thesis was the ability for a student to "read" the writing style of the annotator in order to predict their mood and constructiveness of the feedback comments (see chapter 10.8.2). The student had experience of red pen annotation and the spidery handwriting that filled him or her with dread. This corporeal experience had been negated somehow by the introduction of digital annotation but it does reinforce the psycho-somatic nature of written discourse, to evoke a reaction in the reader. Reader low self-efficacy appears to be a contributing factor to how discourse, in particular annotation is perceived and this appears to relate to cognitive resources that a more sophisticated reader would develop. When I say cognitive resources, I mean the ability to refer to a wide range of literature from which to assess the importance and relevance of a point that is being made. This relates to the theme of rhetoric because I had to try and understand it whilst making the same mistakes in discourse the theory related to. Again, this referred to meta-cognitive understanding but there are linguistic traps. For example, the process of distinguishing something to be true or false means a researcher could mistake the intentions and actions of themselves in the process. As

the research themes suggest, a lack of a reflection may lead to the misunderstanding of others intentions and actions which may impact on the research themes. With this potential for error in negotiating the boundaries of understanding, the methodological choice of Ricoeur's textual hermeneutics requires a brief comment. All I have written about Ricoeur's work attempts to gain entry into his textual hermeneutics and its wide range of application: from life, death, good, evil, religion, politics, social policy, linguistics, and philosophy to name but a few. In contrast, a nurse cum lecturer attempting to understand the alien, technical terminology of Ricoeur's work and research methodology may find what they want to find, because they are searching for the familiar to understand the unfamiliar. In other words, my interpretation and my instinct could have been misdirected and therefore I would not know until I read something new in the future that indicated my error (mimesis3). This is a process of mutual recognition, of oneself and another. Language exists to communicate the lived experience to the self and another and misrecognition and misunderstanding need to occur before understanding is achieved. This means misrecognition is at the centre of recognition and new meaning negotiated through a surplus of meaning.

11.7 Conclusion

In part one of this chapter I examined the research themes original contribution to current knowledge and following the chronology of the thesis discussed what I have learnt through the research process. Temporal understanding and the hermeneutic circle were discussed throughout and I identify gaps in the literature the research themes contribute to. The original contributions to current knowledge are identifying in annotation: self-reference in discourse, the hermeneutic self, the persuasive and rhetorical nature of annotation, the "defence of nursing" from technology and pressures in society that impact on nursing integrity. Lastly, the unconsciousness of the annotator and student were also explored as a contribution to the

literature. The original contribution to exploring the meaning of annotation in nurse education is the synthesis of Ricoeur's textual hermeneutics, and for the first time offering a phenomenological perspective which adds to current knowledge. In part two, I examine the strengths and weaknesses of the study and limitations of the work. I discussed recognition, misrecognition, the hermeneutic self and mutual recognition. By exploring the meaning of annotation in nurse education some of the phenomenon now identified in the thesis can now be read, analysed and implemented into practice by annotators and students alike.

Chapter Twelve

Recommendations and implication for practice

12.1 Introduction

This chapter discusses recommendations and implications for nurse education practice.

12.2 Recommendations

In reply to the initiating questions, what could textual hermeneutics add to annotation and what meaning it has, the first recommendation is to acknowledge the lack of theory informing annotation. Therefore, annotators should now be able to identify the various processes involved in annotation such as: the interpretive bias of the student, and annotator, the use of rhetoric and all of its sub themes, the ethico moral motivations behind annotation and nurse education. Lastly, the conscious and unconscious mental processes that underpin annotation are now both theoretical and evidence based.

12.3 Training

QAA (2012b, p. 15, indicator 4) suggests everyone involved in effective student learning are appropriately qualified, supported and developed. Currently annotators are generally ill prepared and uninformed about annotation theory and the evidence of its effectiveness. The implications for practice from this thesis suggest annotators may benefit from training on different aspects of annotation practice, such as linguistics and interpretation. Therefore, post graduate teacher training should include theory and evidence from annotation research.

12.4 Transferability

In the context of this thesis nursing lecturers were known to collaborate with the assessment, teaching and learning for professional education programmes of midwives, counselling and psychotherapy, occupational health, medicine, physiotherapy, homeopathic and complementary therapies, paramedic and operating department practitioners. This is reinforced by the fact nursing lecturers' work collaboratively within multi professional teams, both pre and post registration suggesting that there is mutuality and shared purpose due to the collegiality, the NHS and mutual benefits as citizens.

12.5 Formative and summative annotation

The QAA (2012a; 2012b) suggested assessed work with written comments and the return of annotation in the marginalia or end comments on an essay to the student. The QAA (2012a, p. 14) suggest feedback should be continual and timely and include dialogue and engagement between the lecturer and student. A recommendation from this thesis is therefore for the annotator to engage actively both at the formative and summative stage of assessment. The QAA guidelines suggest there is little difference in the timeliness of annotation however the research findings suggest the most productive annotation feedback is formative, in person and over time.

12.6 Promoting student reflection

The research theme of the reflective consciousness and slippage identifies students' perception of annotation contrasts between positive and negative. The annotation feedback in one mental health student interviewee evoked what could be described as an atypical reaction. Atypical in that the interviewee was alone in disclosing the emotional effects of the annotation received. Students' pasts are relatively unknown and undisclosed and the impact

of childhood experiences and deciding to become a nurse are important. The past experiences may or may not be dealt with cathartically. Therefore, annotation is one possibility for reflection and when a student perceives annotation to be of significance, whether good or bad, they may benefit from reflecting on the processes involved in order to make sense of any evoked feelings. The promotion of theory to reflection would also move from being overly descriptive to be theoretically informed.

12.7 Discursive versus scientific essays

Due to the implementation of the Bologna process (1999) there is less likelihood of selfdisclosure in essays that promote a scientific approach to nursing, in contrast to discursive and reflective essays (see chapter 2.6). Essays that promote an objectification of knowledge and nursing allow the student to hide in the text. Therefore, what the nurse really thinks and did remain undeveloped and unchallenged. This issue was explored in the research theme the reflective consciousness and slippage and promoting reflective practice for students (see recommendation 12.6) during their training would challenge any objectification of patients, and identify through self-disclosure students with the potential to do harm. Therefore, there is a need during nurse education for balanced assessment with discursive essays and opportunities to learn from annotation.

12.8 Promoting annotator reflection

The QAA (2012b, p. 14, indicator 2) suggests effective teaching and learning support should ensure that staff reflect on their practice from a variety of sources and maintain a practical focus. The policy guidelines do not mention annotation but there is implied concern for it as a teaching and learning supportive method. The lack of detail therefore means there is little or no focus on annotation specifically. QAA (2012b, p. 21, indicator 9) also reinforces the use of personal professional development to reflect on practice. This thesis has highlighted the issue of a lack of reflective practice during clinical nursing experience could possibly impinge on higher education practice because this is the workforce lecturers are directly recruited from (see chapters 1.2.3 to 1.2.5). As the research theme the reflective consciousness and slippage identifies, the issue of unconscious messages, subconscious triggers can be projected through annotation. Therefore, annotation comments may offer the annotator a valuable insight into their thinking processes in particular any slippage. The existing peer observations process, which generally does not focus on annotation as a reflective resource, is therefore a reflective opportunity. Annotation could also be added to the list of potential observed practice.

12.9 Technology

The QAA (2012b, p. 18, indicator 6) reinforces the use of technology to enable student learning including a virtual learning environment. The policy makes no suggestion regarding a need to be critical of its pervasiveness in professional education such as nursing. In order to counteract the risks of reduced face to face contact in supervision, annotators should make every effort to meet a student in person before and after the essay has been submitted to discuss annotation further.

The reasons for the above recommendations are found in chapters seven to ten which examine the impact of annotation, whether digital or handwritten. Due to the advances in technology the use of digital annotation appears to have largely replaced the use of handwritten annotation because students submit their essays online, they are marked online and read online. This issue about technology has of course benefits and risks. Benefits are ease of access for distance learning, reading and writing annotation, the use of anti-plagiarism tools, data collection, data sharing and being able to track something has been read. The risks however are that the uncritical acceptance of technology within higher education risks devaluing what was once valued, and that is the need for personal, face to face contact and developing rapport and influence of a student through a professional educational relationship. This persuasive relationship was examined in the research themes of rhetoric and a person centred theme of individualism reinforcing the notion of the shared experience when communicating professional attitudes of nursing and patient care. The importance of annotation followed by a face to face meeting is more likely to improve the clarity of discourse because any misunderstanding can be further clarified.

12.9.1 Implications for further research

The impact of digital and handwritten annotation would indicate that the former is less communicative and the latter more complex than originally thought. In the latter, one student could identify the tone and mood of the annotator before she read the annotation and this indicates perhaps more traditional forms of communication still have a lot to offer on a symbolic level than digital forms of communication. For example, when reading a student's paper essay, one could sometimes smell coffee, cigarette smoke, damp or even a curry. Sometimes the pages would be stained with a coffee mug ring or have thumb prints on them. Now all extraneous information is removed and sanitised in the pursuit of technological progress. The technological implications for nursing and annotation however, may require further research because of their potential to promote alienation and a disembodied experience.

12.9.2 Recommendations and implication for practice

The recommendations and implications for practice presented in this chapter are related to current QAA policy guidelines, discussion in the thesis and the research themes. The recommendations suggest practice can be improved if annotation is reflected upon by all involved because annotation has the potential to support teaching and learning if it is viewed in less generic terms from other forms of feedback. However, annotation, due to the research themes discussed in this thesis, is unlike other forms of feedback because of its unique situation on the page of the student's essay.

References list

Allan, H.T., Smith, P., & O'Driscoll, M. (2011). Experiences of supernumerary status and the hidden curriculum in nursing: a new twist in the theory practice gap? *Journal of Clinical Nursing*, 20, 847-855.

Angelo, T.A. (1995). Reassessing (and defining) assessment, AAHE Bulletin, 48(2), 7-9.

Antonym. (n.d). Oxford dictionary online. Retrieved from http://oxforddictionaries.com/definition/english/antonym?q=antonym.

Annells, M. (1996). Hermeneutic phenomenology: Philosophical perspectives and current use in nursing Research. *Journal of Advanced Nursing*, 23(4), 705-713. DOI: 10.1111/j.1365-2648. 1996.tb00041.x.

Annells, M. (1999). Evaluating phenomenology: Usefulness, quality and philosophical foundations. *Nurse Researcher*, *6*, (3), 5-19.

Aristotle. (1991). *The art of rhetoric*. Translated by Lawson-Tancred, H. Penguin Classics: London.

Aristotle. (1997). *Poetics*. Dover Thrift. Re-publication of Butcher, S. H. (1895). *Poetics; Aristotle's theory of poetry and fine art*. Macmillan co; London.

Aristotle. (n.d). *Politics: The Athenian constitution*. Translated and edited by Warrington, J. London: Heron Books.

Art.(n.d).OxfordEnglishdictionary.http://www.oed.com/view/Entry/11125?rskey=4gQHKJ&result=1#eid.

Baker, N. (2008). The experiential counselling primer. Ross-On-Wye, UK: PCCS Books.

Ball, E. (2009). A participatory action research study on handwritten annotation feedback and its impact on staff and students. *Systematic Practice in Action Research*, 22, 111-124.

Ball, E. (2010). Annotation an effective device for student feedback: A critical review of the literature. *Nurse Education in Practice*, 7(3):295–307.

Ball, S. J. (2012). Performativity, commodification and commitment: An I-spy guide to the neoliberal University. *British Journal of Educational Studies*, 60(1), 17-28.

Ball, E. Franks, H. Jenkins, J. McGrath, M., & Leigh, J. (2009). Annotation is a valuable tool to enhance learning and assessment in student essays. *Nurse Education Today*, 29:284–291.

Ball, E., & Regan, P. (2013). "How, when, why": A comparison of two action research methods to examine the hidden tones in annotation feedback. *Canadian Journal of Action Research*, *14*(2), 2013, 39-50.

Bandura, A. (1977). Self-efficacy: Towards a unifying theory of behavioral change. *Psychological Review*, 84, 191-215.

Bandura, A. (1997). Self-efficacy: The exercise of control. New York: W.H. Freeman.

Barney, S. A. (1991). Annotation and its texts. Barney, S. A (ed.). Oxford University Press: New York.

Barthes, R. (1964). *Elements of semiology*. New York, Hill and Wang.

Barthes, R. (1973) Mythologies. Paris, Seuil.

Barretti, M. (2004). What do we know about the professional socialisation of our students? *Journal of Social Work Education*, 40(2), 255-283.

Benner, P., Sutphen, M., Leonard, V., & Day, L. (2010). *Educating nurses: A call for radical transformation*. Jossey-Bass; California.

Benstock, S. (1983). At the margin of discourse: Footnotes in the fictional text. Shari Benstock Source: *Modern Language Association*, 98(2), 204-225.

Berg, L., Brämberg, E. B., Carlström, E., Öhlén, J. (2014). Tensions during a process of change –Implementation of the Bologna educational reform in Swedish specialist nursing programmes. *Journal of Nursing Education and Practice*, 4(5), 64-72. DOI: 10.5430/jnep.v4n5p64.

Bernay, R. (2012). *Mindfulness and the beginning teacher*. Unpublished doctoral thesis.http://aut.researchgateway.ac.nz/bitstream/handle/10292/4497/BernayR.pdf?sequence= 3.

Bishop, V. (2008). Clinical governance and clinical supervision; Protecting standards of care. *Journal of Research in Nursing*, *13*, 3-5.

Boell, S. K., & Cecez-Kecmanovic, D. (2010). Literature reviews and the hermeneutic circle. *Australian Academic Research Libraries*, *41*(2), 129-144.

Boud, D. (2007). Reframing assessment as if learning were important. *In*: Boud D, Falchikov, N. (eds). *Rethinking assessment in higher education: Learning for the longer term*. Routledge, London, p. 14-25.

Bologna process. (1999). Bologna declaration of 19th June 1999: Joint declaration by the
EuropeanEuropeanMinistersofeducation.http://www.ehea.info/Uploads/Declarations/BOLOGNA_DECLARATION1.pdf.

Boud, D. (2010). Assessment 2020: Seven propositions for assessment reform in higher education. Sydney: Australian Learning and Teaching Council.

Butterworth, T., Bell, L., Jackson, C., & Pajnkihar, M. (2008). Wicked spell or magic bullet? A review of the clinical supervision literature 2001–2007. *Nurse Education Today*, 28, 264-272.

Changeaux, J. P., & Ricoeur, P. (2000). What makes us think? A neuro-scientist and a philosopher argue about ethics, human nature and the brain. Princeton University Press. Princeton; New Jersey.

Clark, S. H. (1990). Paul Ricoeur. Routledge: London.

Clifford, C. (1993). The clinical role of the nurse teacher in the UK. *Journal of Advanced Nursing*. 18(2), 281-289.

Clifford, C. (1996). Nurse teacher's clinical work: a survey report. *Journal of Advanced Nursing*, 23(5), 603-610.

College of Occupational Therapists (2005). *College of Occupational therapists: Code of ethics and professional conduct*. British Association of Occupational Therapists. Retrieved from http://hsc.uwe.ac.uk/practicesupport/Data/Sites/1/GalleryImages/Occupational-Therapy/Occupational% 20Therapy_College% 200f% 20Occupational% 20Therapists% 20Code % 200f% 20Ethics% 20and% 20Professional% 20Conduct_2005.pdf

Connors, R.J., & Lunsford, A.A. (2006). Teachers' rhetorical comments on student papers. *In.* Straub, R. (2006). *Key works on teacher responses: An anthology*. Straub, R.(ed). Boynton/Cook Publishers: Heinemann, Portsmouth, New Hampshire.

Corben, V. (1999). Misusing phenomenology in nursing research: Identifying the issues. Nurse Researcher, 6(3), 52-66.

Cresswell, J. W. (2007). *Qualitative inquiry and research design; Choosing among five approaches.* 2nd edition. Sage, California.

Crowther, S. A. (2014). *Sacred joy at birth; A hermeneutic phenomenology study*. Unpublished PhD thesis. http://aut.researchgateway.ac.nz/bitstream/handle/10292/7071/CrowtherS.pdf?sequence=3.

Dark.(n.d)OxfordEnglishdictionary.http://www.oed.com/view/Entry/47295?rskey=VjtPpC&result=7#.

Department of Health. (2009). *Transforming community services: Enabling new patterns of provision*. London: Department of Health.

Department for Business, Innovation and Skills. (2011). *Higher education; Students at the heart of the system*. Retrieved from http://www.scribd.com/doc/58879994/11-944-WP-Students-at-Heart.

Department of Health (2012.) Compassion in practice: Nursing, midwifery and care staff.Ourvisionandstrategy.https://www.england.nhs.uk/wp-content/uploads/2012/12/compassion-in-practice.pdf.

Derrida, J. (1982). *Margins of philosophy translation and annotation*. Chicago University Press. The Harvester press Limited, Sussex.

Derrida, J. (1988). *The ear of the other: Otobiography, transference and translation*. Translated by Kamuf, P. Bison book. University of Nebraska Press; Lincoln.

Derrida, J. (1991). This is not an oral footnote. *In*. Barney, S. A. (1991). *Annotation and its texts*. Barney, S. A (ed.). Oxford University Press: New York.

de Saussure, F. (1983). *Course in general linguistics*. (Eds.). Bally, C., & Sechehaye, A. Translated and annotated by Harris, R. Duckworth: London.

de Witt, L., & Ploeg, J. (2006). Critical appraisal of rigour in interpretive phenomenological nursing research. *Journal of Advanced Nursing*, 55, (2), 215-229.

Dewey, J. (1997). How we think. Dover publications: New York.

Dilthey, W. (1988). Introduction to the human sciences: An attempt to lay a foundation for the study of society and history. Translated by Betanzos, R. J. Wayne State University Press; Detroit, Michigan.

DiYanni, R. (2002). One hundred great essays. Addison-Wesley, New York.

Elton, L., & Johnston, B. (2002). *Assessment in Universities: a critical review of research,* LTSN Generic Centre. http://ltsnpsy.york.ac.uk/docs/pdf/p20030617_elton_johnston-assessment_in_universities_a_critical_view_o.pdf.

Elton, L. (2010). Complexity theory-an approach to assessment that can enhance learning and-more generally-could transform university management. *Assessment and Evaluation in Higher Education*, 35(5), 637-646.

Evans, A. M., Pereira, D. A., & Parker, J. M. (2009). Discourse anxiety and transference in nursing practice: The subject of knowledge. *Nursing Inquiry*, *16*(3), 251-260.

Feito, A., & Donahue, P. (2008). Minding the gap annotation as preparation for discussion. *Arts and Humanities in Higher Education*, 7(3):295-307.

Fischer, S. R. (2005). A history of writing. Reaktion Books. The Netherlands.

Freud, S. (1915). The unconscious: The standard edition of the complete psychological works of Sigmund Freud, Volume XIV (1914–1916): *On the history of the psychoanalytic movement, papers on metapsychology and other works.* 159-215.

Freud, S. (1961). The future of an illusion. New York: W.W. Norton & Company.

Freud, S. (1997). Interpretation of dreams. First published 1899. MacMillan. 1913.

Freud, A. (1992). The ego and the mechanisms of defence. Karnac Books.

Freud, S. (2005). *The unconscious*. First published 1915. Translated by Frankland, G. Penguin books: London.
Foxcroft, D., & Cole, N. (2003). Organisational infrastructures to promote evidence based nursing practice. *Cochrane Database of Systematic Reviews*, (4): CD002212.

Gadamer, H. G. (1994). *Heidegger's ways*. Translated by Stanley, J.W. State University of New York: Albany.

Gadamer, H. G. (2004a). Truth and method, 2nd edition, Sheed and Ward Stagbooks.

Gadamer, H. G. (2004b). *Philosophical hermeneutics*. Translated and edited by Linge, D. E. 2nd edition, University of California Press.

Galilei, G. (1932 [1613]). Istoria e Dimostrazioni intorno alle macchie solari. *In*: Favaro, A., (ed.), *Edizione nazionale delle opere di Galileo Galilei*, vol. 5. Firenze: G. Barbèra, pp. 73-240.

Gardner, L., & Rolfe, G. (2013). Essaying the essay: Nursing scholarship and the hegemony of the laboratory. *Nurse Education Today*, *33*, 31-35. doi:10.1016/j.nedt.2011.10.009.

Gelter, H. (2003). Why is reflective thinking uncommon? *Reflective Practice*. 4 (3), 337-344. doi: 10.1080/1462394032000112237.

Giddings, L. S., & Wood, P. J. (2001). Phenomenology-telling a story: An interview with Liz Smythe. *Nursing Praxis in New Zealand*, *17*(3),15-25.

Giles, D. L (2008). *Exploring the teacher-student relationship in teacher education: A hermeneutic phenomenological inquiry*. Unpublished PhD thesis.http://www.taosinstitute.net/Websites/taos/images/ResourcesNoteworthy/GilesD.pdf.

Good, G. (1988). The observing self: Rediscovering the essay. Routledge, London.

Grbich, C. (2013). Qualitative data analysis: An introduction. Sage: London.

Griffin, J. (2006). *Value judgement: Improving our ethical beliefs*. Oxford University Press: Oxford.

Handley, K., Szwelnik, A., Ujma, D., Lawrence, L, Millar, J., & Price, M. (2007). *When less is more: Students' experiences of assessment feedback.* 1st July, 3rd Higher Education Academy annual conference, Engaging students in Higher Education.

Hare, R. M. (1952). The language of morals. Oxford University Press: Oxford.

Hawkins, P., & Shohet, R. (2012). *Supervision in the helping professions*. (4th ed.). Maidenhead, Berks: Open University Press.

Hays, R., & Gay, S. (2011). Reflection or 'pre-reflection': What are we actually measuring in reflective practice? *Medical Education*, 45(2), 116-118, doi: 10.1111/j.1365-2923.2010.03813.x.

Hegel, G. W. F. (1988). *Phenomenology of spirit*. Trans Miller, A. V. Oxford University Press; London.

Hegel, G. W. F. (2008). *Outlines of the philosophy of right*. Translated by Knox, T. M. Oxford University Press:

Heidegger, M. (1977). *The question concerning technology and other essays*. Translated by Lovitt, W. New York: W. Harper Perennial.

Heidegger, M. (1882). On the way to language. Translated by Hertz, P. D. Harper: San Francisco.

Heidegger, M. (2003). *Being and time* (21st ed.). Translated by Macquarrie, J., & Robinson, E. Oxford: Blackwell Science. (Original work 1927/1962).

Hein, S. F., & Austin, W. J. (2001). Empirical and hermeneutic approaches to phenomenological research in psychology: A comparison. *Psychological Methods*, 6(1), 3-17.

Henderson, V. (1966). The nature of nursing: A definition and its implications for practice, research and education. New York: Macmillan.

Henige, D. (2001). Being fair to the hounds: The function and practice of annotation. *History in Africa*, 28, 95-127.

Henige, D. (2002). Being fair to the hounds: The function and practice of annotation, 11. *History in Africa*, 29, 63-88.

Holden, P., & Littlewood, J. (1991). Anthropology and nursing. Routledge: London.

Homonymy. (n.d). Oxford Online dictionary. Retrieved from http://oxforddictionaries.com/definition/english/homonym?q=homonymy#homonym_14.

Husserl, E. (2008). *Logical investigations: Vol. 1.* (1st edition 1900-01) International Library of Philosophy. Translated by Findlay, J. N. From the 2nd 1913 German edition. Routledge: New York.

Husserl, E. (2014). *Ideas 1*. First published 1913. Translated by Dahlstrom, D. O. Hackett Publishing Company: Indianapolis. Indiana.

Husserl, E. (1965). *Phenomenology and the crisis of philosophy*. 1st edition. Translated, notes and introduction by Lauer, Q. Harper Torchbooks: New Work.

Hyland, K. & Hyland, F. (2006). Feedback on second language students' writing. *Language Teaching*. 2, 83-101. doi.org/10.1017/SO261444806003399.

Iser, W. (2006). The reading process: A phenomenological approach. *In* Richter, D. (ed.) *The critical tradition: Classic texts and contemporary trends*, pp. 956-68. New York: Bedford/St Martins.

Identity(n.d.)1b.OxfordEnglishdictionary.http://www.oed.com/view/Entry/91004?redirectedFrom=identity+#eid.

It.(n.d)OxfordEnglishdictionary.http://www.oed.com/view/Entry/100246?rskey=Ovc4uH&result=2#eid.

Jackson, H. J. (2001). *Marginalia: Readers writing in books*. New Haven: Yale University Press.

Jhagroo, J. R. (2011). A hermeneutic phenomenological study of the lived experiences of immigrant students in their mathematics classrooms at a secondary school in Auckland, New Zealand. Unpublished doctoral thesis. http://aut.researchgateway.ac.nz/bitstream/handle/10292/4410/JhagrooJR.pdf?sequence=3.

Juwah, C., Macfarlane-Dick, D., Matthew, B., Nicol, D., Ross, D., & Smith, B. (2004). *Enhancing student learning through effective formative feedback*. Retrieved from http://www.palatine.ac.uk/files/919.pdf.

Kant, I. (1963). Fundamental principles of the metaphysics of ethics: Extracted from Kant's cririque of practical reason and other works on the theory of ethics. Translated by Kingsmill Abbott, T. (10th ed.). Longmans, Green and Co: London.

Kant, I. (1993). *Grounding for the metaphysics of morals: On a supposed right to lie because of philanthropic concerns.* Cambridge: Hackett.

Kant, I. (1999). The metaphysics of morals. In *The Cambridge edition of the works of Immanuel Kant: Practical Philosophy*, trans. by Gregor, M. Cambridge: Cambridge University Press.

Kaplan, D. M. (2006). Paul Ricoeur and the philosophy of technology. *Journal of French Philosophy*, *16*, 1 and 2, 42-56. http://jffp.org/ojs/index.php/jffp/article/viewFile/182/178.

Keall, R., Clayton, J. M., & Butow, P. (2014). How do Australian palliative care nurses address existential and spiritual concerns? Facilitators, barriers and strategies. *Journal of Clinical Nursing*, 1-9. doi: 10.1111/jocn.12566.

Kearney, R. (1996). Paul Ricoeur: The hermeneutics of action. Sage publications: London.

Kemp, P. (1996a). Introduction. In. Ricoeur, P. (1996). The hermeneutics of action. ed. Kearney, R. Sage: London.

Kemp, P. (1996b). Ricoeur between Heidegger and Lévinas original affirmation between ontological attestation and ethical injunction. *In* Schmidt, L. H. (ed.). (1996). *Paul Ricoeur: In the conflict of interpretations*. Aarhus Universitet: Denmark.

Kimmel, A. J. (1988). *Ethics and values in applied social research*. Applied social research Series, Volume 12. Sage: California.

King, V., Garcia-Perez, A., Graham, R., Jones, C., Tickle, A., & Wilson, L. (2014). Collaborative reflections on using island maps to express new lecturers' academic identity, *Reflective Practice: International and Multidisciplinary Perspectives*, *15*(2), 252-267, DOI: 10.1080/14623943.2014.883311.

Kitchenham, B. (2004). *Procedures for performing systematic reviews*. Keele, Eversleigh: Keele University and NICTA, Technical Report.

Knight, P (2004). Assessment for learning in higher education. 3rd edition. Routledge-Falmer, London.

Knoblauch, C. H., & Brannon, L. (1984). *Rhetorical traditions and the teaching of writing*. Boynton Cook Publishers, New Jersey, United States.

Knoblauch, C.H., & Brannon, L. (2006a). Teacher commentary on student writing. Chapter 5. *In* Straub, R. (2006). *Key works on teacher responses: An anthology*. Straub, R. (ed). Boynton/Cook Publishers: Heinemann, Portsmouth, New Hampshire.

Knoblauch, C.H., & Brannon, L. (2006b). The Emperor (still has no clothes). Introduction. *In.* Straub, R. (2006). *Key works on teacher responses: An anthology*. Straub, R. (ed). Boynton/Cook Publishers: Heinemann, Portsmouth, New Hampshire.

Knowles, M. S. (1973). *The adult learner: A neglected species*. Gulf Publishing Company. Houston, Texas.

Knowles, M. S. et al (1984). Andragogy in action. Applying modern principles of adult education, San Francisco: Jossey Bass.

Kshetri, N. (2013). Privacy and security issues in cloud computing: The role of institutions and institutional evolution. *Telecommunications Policy*, *37*(4-5), 372-386.

Lacan, J. (2008). My teaching. 1st edition. Verso books.

Labyrinth. (n.d). 2.a. Oxford English dictionary. http://www.oed.com/view/Entry/104763?rskey=pf1545&result=1&isAdvanced=false#eid

Liu, K. (2006). Annotation as an index to critical writing. Urban Education, 41:192-207.

Loughran, J. J. (2002). Effective reflective practice: In search of meaning in learning about teaching. *Journal of Teacher Education*, *53*(1), 33-43, doi: 10.1177/0022487102053001004.

Lopez, V., Fischer, I., Leigh, M.C., Larkin, D., & Webster, S. (2014). Spirituality, religiosity, and personal beliefs of Australian undergraduate nursing students. *Journal of Transcultural Nursing*, 25: 395-402. doi:10.1177/1043659614523469.

Lunsford, R. F., & Straub, R. (2006). Twelve readers reading. Chapter 10. *In* Straub, R. (2006). *Key works on teacher responses: An anthology*. Straub, R. (ed). Boynton/Cook Publishers: Heinemann, Portsmouth, New Hampshire.

Lyons, J. O. (1967). "Pale fire" and the fine art of annotation. A special number devoted to Vladimir Nabokov. *Wisconsin Studies in Contemporary Literature*, 8(2), 242-249.

Madison, G. (1988). Method in interpretation. In *The Hermeneutics of post-modernity: Figures and themes*. Bloomington. IN: Indiana University Press, 25-39.

Maloney, S., Tai, J., Lo, K., Molloy, E., & Ilic, D. (2013). Honesty in critically reflective essays: An analysis of student practice. *Advances in Health Sciences Education*, 18(4), 617-626.

Mandel, W. M. (1965). Annotation, scholarship, and ignorance. *Slavic Review*, 24(4), 714-721.

Marcel, G. (1965). *Being and having*. 2ndedition. Translated by Black. A & C. Fontana: London.

Marshall, C. C. (1997). Annotation: From paper books to the digital library. *In Proceedings of Digital Library*, Philadelphia, PA, USA, July 23-26. ACM Press, New York, NY, 131-140.

Marshall, C. C. (1998a). *Toward an ecology of hypertext annotation*. In *Proceedings of the Ninth ACM Conference on Hypertext and Hypermedia, Hypertext, 98,* 40-48.

Marshall, C. M. (1998b). *The future of annotation in a digital (paper) World*. Presented at the 35th Annual GLSLIS clinic. Successes and failures of Digital Libraries, June 20-24, University of Illinois at Urbana–Champaign, March 24. 1-20.

Morecambe Bay Investigation (2015). *Report of the Morecambe Bay investigation*. https://www.gov.uk/government/uploads/system/uploads/attachment_data/file/408480/47487 _MBI_Accessible_v0.1.pdf.

Morrissey, J., & Traibe, R. (2001). Parallel process in supervision. *Counselling Psychology Quarterly*, 14(2), 103-110.

Morse, J. M. (1991). *Qualitative nursing research: A contemporary dialogue*. ed. Morse, J.M. Sage, Newbury Park: CA.

Morse, J.M. (2003). The significance of standards. *Qualitative Health Research*, 13, 1187-1188.

McColly, W. (1965). *The dimensions of composition annotation*. State University of New York Press, Oswego, NY.

McGuire, M. C. (1961). Can I do what I think I ought not? Where has Hare gone wrong? *Mind*. LXX (279): 400-404.doi: 10.1093/mind/LXX.279.400.

McMahon, T., Barrett, T., & O'Neill, G. (2007). Using observation of teaching to improve quality: Finding your way through the muddle of competing conceptions, confusion of practice and mutually exclusive intentions. *Teaching in Higher Education*, *12*(4) 499-511. doi: 10.1080/13562510701415607.

McNiff, J. (2013). Action research: Principles and practice. 3rd edition. Abingdon, Routledge.

Narrative. (2013). Oxford Online dictionary. Retrieved from http://oxforddictionaries.com/definition/english/narrative?q=narrative. *National Health Service and Community Care Act.* (1990). The Stationary Office, London. http://www.legislation.gov.uk/ukpga/1990/19/contents.

Nietzsche, F. W. (2007). *Twilight of the idols with the antichrist and ecce homo*. Wordsworth editions. Hertfordshire.

Nichols, S. G. (1991). On the sociology of medieval manuscript annotation. *In* Barney, S. A. (ed). *Annotation and its texts*. Oxford University Press: New York.

Nohrnberg, J. C. (1991). Justifying narrative: Commentary within biblical storytelling. *In*. Barney, S. A. (ed). *Annotation and its texts*. Oxford University Press: New York.

Novak, E., Razzouk, R., & Johnson, T. E. (2012). The educational use of social annotation tools in higher education: A literature review. *Internet and Higher Education*, *15*, 39-49. doi:10.1016/j.iheduc.2011.09.002.

Nursing and Midwifery Order 2001. (2002). HMSO. http://www.legislation.gov.uk/uksi/2002/253/pdfs/uksi_20020253_en.pdf.

Nursing and Midwifery Council. (2008a). *Standards to support learning and assessment in practice: NMC standards for mentors, practice teachers and teachers.* http://www.nmc.org.uk/globalassets/siteDocuments/NMC-Publications/NMC-Standards-to-support-learning-assessment.pdf.

Nursing and Midwifery Council. (2008b). The code: Standards of conduct, performance and
ethicsfornursesandmidwives.http://www.nmc.org.uk/globalassets/sitedocuments/standards/the-code-a4-20100406.pdf.

Nursing and Midwifery Council. (2010). *Standards for pre-registration nursing education*. Portland place, London. http://www.nmc.org.uk/globalassets/sitedocuments/nmc-publications/standards-for-pre-registration-nursing-education-16082010.pdf.

Nursing and Midwifery Council. (2011). *Standards to support learning and assessment in practice: Preparation for mentors, practice teachers and teachers*. Portland place, London. http://www.nmc.org.uk/globalassets/sitedocuments/nmc-publications/nmc-standards-to-support-learning-assessment.pdf.

Nursing and Midwifery Council. (2013). *The quality assurance framework for nursing and midwifery education and local supervising authorities for midwifery*. Portland place, London.

Nursing and Midwifery Council. (2015a). *The code: Professional standards of practice and behaviour for nurses and midwives*. Revised Code. http://www.nmc-uk.org/Documents/NMC-Publications/revised-new-NMC-Code.pdf.

Nursing and Midwifery Council. (2015b). *Guidance: Fitness to practice legislation guidance*.http://www.nmc.org.uk/globalassets/siteDocuments/Legislation/NMC-fitness-to-practise-legislation-guidance-notes-31012012.pdf.

Ought.(n.d).OxfordEnglishdictionary.http://www.oed.com/search?searchType=dictionary&q=%E2%80%9COught%E2%80%9D+&_searchBtn=Search.

Ousey, K., & Gallagher, P. (2010). The clinical credibility of nurse educators: Time the debate was put to rest. *Nursing Education Today*, *30*(7), 662-665.

Paddy, A. (2010). *The phenomenon of being and older experienced health professional*. Unpublished PhD thesis. http://aut.researchgateway.ac.nz/bitstream/handle/10292/1032/PaddyA.pdf?sequence=3.

Paley, J. (2005). Phenomenology as rhetoric. Nursing Inquiry, 12, (2), 106-116.

Palmer, R. E. (1969). *Hermeneutics; interpretation theory in Schleiermacher, Dilthey, Heidegger and Gadamer.* Evanston, Illinois, North Western University Press.

Parliamentary and Health Service Ombudsman. (2011). *Care and compassion? Report of the health service ombudsman on ten investigations into NHS care of older people*. Fourth report of the Health Service Commissioner for England Session 2010-2011. The House of Commons, London. The Stationary Office.

Paterson, M., & Higgs, J. (2005). Using hermeneutics as a qualitative research approach in professional practice. *The Qualitative Report*, *10*(2), 339-357.

Perception.(n.d).OxfordEnglishdictionary.http://www.oed.com/view/Entry/140560?redirectedFrom=perception#eid.

Personcentred.(n.d).OxfordEnglishdictionary.P5b,http://www.oed.com/view/Entry/141476?redirectedFrom=person+centred#eid30952118

Pincus, D., Freeman, F., & Modell, A. (2007). A neurobiological model of perception; Considerations for transference. *Psychoanalytic Psychology*, 24(4), 623-640.

Porter-O'Donnell, C. (2004). Beyond the yellow highlighter: Teaching annotation skills to improve reading comprehension. *English Journal*, *93*(5), 81-88.

Prat-Sala, M., & Redford, P. (2010). The interplay between motivation, self-efficacy, and approaches to studying. *British Journal of Educational Psychology*, *80*, 283-305. doi: 10.1348/000709909X480563.

Predicate.(n.d).OxfordEnglishdictionary.http://www.oed.com/view/Entry/149843?rskey=ikJY04&result=1&isAdvanced=false#eid.

Price, L., Hastie, L., Duffy, K., Ness, V., & McCallum, J. (2011). Supporting students in clinical practice: Pre-registration nursing students' views on the role of the lecturer. *Nurse Education Today*, *31*, 780-784. doi:10.1016/j.nedt.2011.04.009.

Projection. (2013). Oxford dictionary online. http://www.oxforddictionaries.com/definition/english/projection?q=projection.

Quality Assurance Agency for Higher Education. (2012a). *UK Quality Code for higher education*. Part B: Assuring and enhancing academic quality. Chapter Three. http://www.qaa.ac.uk/assuring-standards-and-quality/the-quality-code/quality-code-part-b.

Quality Assurance Agency for Higher Education. (2012b). Understanding assessment: Its role in safeguarding academic standards and quality in higher education. A guide for early career staff.

http://www.qaa.ac.uk/Publications/InformationAndGuidance/Documents/understanding-assessment-second-edition.pdf.

Question.(n.d).OxfordEnglishdictionary.http://www.oed.com/view/Entry/156343?rskey=LmfJFz&result=1#eid.dictionary.

Raj, S. R. (2013). A study of the perception of university lecturers on the use of reflective teaching practice as part of peer review process. *Journal of Educational Chronicle*, 4(1), 9-18.

Rawls, J. (1971). A theory of justice. Cambridge: Harvard University Press. *In* Ricoeur, P. (2000). *The just*. Translated by Pellauer, D. The University of Chicago Press: Chicago.

Rawls, J. (1980). Kantian constructivism in moral theory. September. *The Journal of Philosophy*, 77 (9), 515-572.

Reed, K. (2008). *Resituating the meaning of occupation in the context of living*. Unpublished Doctoral thesis.

http://aut.researchgateway.ac.nz/bitstream/handle/10292/398/ReedK.pdf?sequence=4.

Redding, P. (2013). Georg Wilhelm Friedrich Hegel. *The Stanford encyclopedia of philosophy* (Winter 2013 Edition), Zalta, E. N. (ed.). http://plato.stanford.edu/archives/win2013/entries/hegel/.

Regan, P. (2007). Interpreting the language of reflective practice. *Reflective Practice*, 8(1), 109-122.

Regan, P. (2008). Reflection; How far how deep. Reflective Practice, 9(2), 219-229.

Regan, P. (2010). Read between the lines: The emancipatory nature of formative annotative feedback on draft essays. *Systematic Practice and Action Research*, 23(6), 453-466.

Regan, P., & Ball, E. (2010). Liberating the NHS: Commissioning, outsourcing and a new politics debate. *British Journal of Community Nursing*, 15(10), 503-505.

Regan, P. (2012a). Listening visits to mothers' suffering with mild to moderate peri-natal depression; Critical issues of clinical supervision, training and health visiting capacity to tackle health inequalities. *Primary Health Care*, 1(1), 2-5. Open Access 1:107. doi:10.4172/phcoa.1000107.

Regan, P. (2012b). Reflective insights on group clinical supervision: Understanding transference in the nursing context. *Reflective Practice: International and Multidisciplinary Perspectives*, *13*(5), 679-691. doi: 10.1080/14623943.2012.697880.

Regan, P. (2012c). Hans-Georg Gadamer's philosophical hermeneutics: Concepts of reading, understanding and interpretation. *META: Research in Hermeneutics, Phenomenology and Practical Philosophy*, *4* (2), 286-303.

Regan, P. (2012d). Reflections of specialist community public health nurse practice development: Transformative learning through Mezirow's reflective discourse. *International Practice Development Journal*, 2 (9), 9-15.

Regan, P., & Ball, E. (2013). Breastfeeding mothers' experiences: The ghost in the machine. *Qualitative Health Research*, 23(5), 679-688. doi: 10.1177/1049732313481641.

Regan, P. (2015). This thinking lacks a language: Heidegger and Gadamer's question of being. *META: Research in Hermeneutics, Phenomenology, and Practical Philosophy,* VII (2), 376-394.

Regan, P., & Ball, E. (2016). An unhealthy Alliance. British Journal of Nursing, 25(8), p. 426.

Ricoeur, P. (1965). *History and truth*. Translated by Kelbley, C. A. Evanston: Northwestern University Press.

Ricoeur, P. (1969). The symbolism of evil. Translated by Buchanan, E. Beacon Press: Boston.

Ricoeur, P. (1970). *Freud and philosophy; An essay on interpretation*. Translated by Savage, D. 1st edition. Yale University Press; New Haven.

Ricoeur, P. (1974). Metaphor and the main problem of hermeneutics. *New Literary History*, 6(1), 95-110.

Ricouer, P. (1976). *Interpretation theory: Discourse and the surplus of meaning*. The Texan Christian University Press: Fort Wirth: Texas.

Ricoeur, P. (1978). The metaphorical process as cognition, imagination, and feeling. *Critical Inquiry*, *5*(1), 143-159.

Ricoeur, P. (1985). *Time and narrative. Volume 2*. Translated by McLaughin, K. Pellaur, D. Chicago, University of Chicago Press, London.

Ricoeur, P. (1986). Fallible man. 2nd edition. New York: Fordham University Press.

Ricoeur, P. (1988). *Time and narrative. Volume 3*. Translated by Blamey, K., & Pellaur, D. University of Chicago Press, London.

Ricoeur, P. (1990). *Time and narrative. Volume 1*. Translated by McLaughin, K., & Pellaur, D. University of Chicago Press, London.

Ricoeur, P. (1994). *Oneself as another*. Translated by Blamey, K. Chicago University Press: Chicago.

Ricoeur, P. (1996a). The hermeneutics of action. ed Kearney, R. Sage: London.

Ricoeur, P. (1996b). *A key to Edmund Husserl's Ideas1*. (ed.) Vandervelde, P. Translated by Harris, B. & Bouchard Spurlock, J. Marquette, University Press: Milwauke.

Ricoeur, P. (1998a). *Hermeneutics and the human sciences*. Edited and translated by Thompson, J. B. 15th edition. Cambridge University Press, Cambridge.

Ricoeur, P. (1998b). *Critique and conviction*: Conversations with Azouvis, F & de Laurnay, M. Polity Press: Cambridge.

Ricoeur, P. (1999). *Husserl: Analysis of his phenomenology*. 6th (edn.). Translated by Ballard, E.G., & Embree, L.E. North western University Press: Evanston. Illinois.

Ricoeur, P. (2000). *The just*. Translated by Pellauer, D. The University of Chicago Press: Chicago.

Ricoeur, P. (2003). *Rule of metaphor: The creation of meaning in language*. First published 1975 in Paris. Translated by McLaughin, K., & Costello, J. Routledge Classics, London.

Ricoeur, P. (2006). *Memory, history and forgetting*. Translated by Blamey, K., & Pellaur, D. The University of Chicago Press; London.

Ricoeur, P. (2007). *The course of recognition*. Translated by Pellauer, D. Harvard University Press.

Ricoeur, P. (2008). *From text to action*. Translated by Blamey, K., & Thompson, J. B. Continuum Press: London.

Ricoeur, P. (2009). *On translation: Thinking in action*. Translated by Brennan, E. Routledge Books: Padstow, Cornwall.

Ricoeur, P. (2012). *On psychoanalysis: Writings and lectures volume 1*. Translated by Pellauer, D. Polity Press: Cambridge.

Ricoeur, P. (2013). Being, essence and substance in Plato and Aristotle. Polity Press: Cambridge, UK.

Rickman, H.P. (1979). Wilhelm Dilthey: Pioneer of the human studies. Paul Elak Ltd: London.

Risjord, R. (2014). Nursing and human freedom. *Nursing Philosophy*, 15, 35-45. doi: 10.1111/nup.12026

Rolfe, G. (2014). Editorial: Educating the good for nothing student. *Journal of Clinical Nursing*, 23, 1459-1460, doi: 10.1111/jocn.12556.

Rose, E. (2009). *Nihilism: The root of the revolution of the modern age* (2nd ed.). Platina, CA: St. Herman of Alaska Brotherhood.

Rossouw, G. J. (2009). *Therapists' experience of working with suicidal clients*. Unpublished PhD.http://aut.researchgateway.ac.nz/bitstream/handle/10292/841/RossouwG.pdf?sequence= 3.

Rubin, D. C. (1995). *Memory in oral traditions: The cognitive psychology of epic, ballads, and counting-out rhymes.* Oxford University Press: New York.

Self.(n.d).OxfordEnglishdictionary.http://www.oed.com/view/Entry/175090?result=3&rskey=gwuJss&.

Seneca (2004). Letters from a Stoic: Epistulae morales ad lucilium. Campbell, R. (ed and trans): Penguin Classics.

Schön, D. A. (1983). *The Reflective practitioner: How professionals think in action*. Basic Books: United States.

Schön, D. A. (1987). Educating the reflective practitioner: Towards a new design for teaching and learning in the professions. San Francisco: Jossey Bass.

Schmidt, L.H. [ed]. (1996). Paul Ricoeur: In the conflict of interpretations. Arhus: Denmark.

Skorga, P., & Young, C. F. (2014). Happy anniversary Cochrane collaboration. Editorial. *Journal of Advanced Nursing*. 1-2. doi: 10.1111/jan.1231.

Slights, W.W. E. (1992). Marginall notes that spoile the text: Scriptural annotation in the English Renaissance. *Huntington Library Quarterly*, 55(2), 255-278.

Sommers, N. (2006). Responding to student writing. Chapter 19. In Straub, R. (2006). Key works on teacher responses: An anthology. Straub, R. (ed). Boynton/Cook Publishers: Heinemann, Portsmouth, New Hampshire.

Steven, A., Magnusson, C., Smith, P., & Pearson, P. H. (2014). Patient safety in nursing education: Contexts, tensions and feeling safe to learn. *Nurse Education Today*, *34*(2), 277-284.doi.org/10.1016/j.nedt.2013.04.025.

Straub, R. (2006). *Key works on teacher responses: An anthology*. Straub, R. (ed). Boynton/Cook Publishers: Heinemann, Portsmouth, New Hampshire.

Strauss, C. L. (1972). Structural anthropology. Penguin books: Harmondsworth, Middlesex.

Smith, P., & Allan, T. E. (2010). "We should be able to bear our patients in our teaching in some way:" Theoretical perspectives on how nurse teachers manage their emotions to negotiate the split between education and caring practice. *Nurse Education Today*, *30*, 218-223.doi:10.1016/j.nedt.2009.12.015.

Smythe, E. A., Ironside, P. M., Sims, S. L., Swenson, M. M., & Spence, D. G. (2008). Doing Heideggerian hermeneutic research: A discussion paper. *International Journal of Nursing Studies*, *45*(9), 1389-1397.

Sutton, D. (2008). *Recovery as the re-fabrication of everyday life exploring the meaning of doing for people recovering from mental illness*. Unpublished PhD thesis. http://aut.researchgateway.ac.nz/bitstream/handle/10292/662/SuttonD.pdf?sequence=4.

Synonym. (n.d). Oxford dictionary online. Retrieved from http://oxforddictionaries.com/definition/english/synonym?q=synonym.

Swan, J.S., & Hull, A.M. (2007). The cognitive behavioural analysis system of psychotherapy: A new psychotherapy for chronic depression. *Advances in Psychiatric Treatment*, 13, 458–469.

The Mid Staffordshire NHS Foundation Trust Public Inquiry. (2013). Report of the Mid Staffordshire NHS foundation Trust public inquiry executive summary. House of Commons. Chaired by Robert Francis, QC. Retrieved http://www.midstaffspublicinquiry.com/sites/default/files/report/Executive%20summary.pdf.

Thorne, S. (2000). Data analysis in qualitative research. *Evidence- Based Nursing*, *3*, 68-70.

Tone.(n.d).OxfordEnglishdictionary.http://www.oed.com/view/Entry/203150?rskey=w7H5CG&result=1#eid.dictionary.

Totton, N. (2005). *New dimensions in body psychotherapy*. Berkshire: Open University Press. McGraw-Hill.

Trope.(n.d).OxfordEnglishdictionary.http://www.oed.com/view/Entry/206679?rskey=BF4VMj&result=1#eid.

Urban, A. M. (2012). *Nurses and their work in hospitals: Ruled by embedded ideologies and moving discourses.* http://ourspace.uregina.ca/bitstream/handle/10294/3553/Urban_Ann-Marie_188003945_PhD_EDUC_Spring2012.pdf;jsessionid=207844003EF4172DAF7DD4E 16A175A46?sequence=1.

van Manen, M. (1997). *Researching the lived experience: Human science for an action sensitive pedagogy*. (2nd ed). London, ON: The Althouse Press.

Violence.(n.d).OxfordEnglishdictionary,http://www.oed.com/view/Entry/223638?rskey=NqftoC&result=1#eid.

Walker, T. (2008). The meaning of being in dilemma in paediatric practice: A
phenomenological study.UnpublishedPhDthesis.http://aut.researchgateway.ac.nz/bitstream/handle/10292/405/WaterT.pdf?sequence=1.

Walker, H. R. (2014). Stress, burnout and addiction in the nursing profession. Xlibris. Bloomington: Indiana.

Whitehead, L. (2004). Enhancing the quality of hermeneutic research: Decision trail. *Journal of Advanced Nursing*, 45(5), 512-518. doi: 10.1046/j.1365-2648.2003.02934.x.

Wilson, L. (2013). Welcome on board: Designing support interventions to meet the real needs of new part-time lecturers. *In*. Beaton, F., & Gilbert, A. (Eds.), *Developing effective part-time teachers in higher education: New approaches to professional development*, 117-133. Abingdon: Routledge.

Wolfe, J. (2000). The effects of annotations on student readers and writers. *DL '00: Proceedings of the Fifth ACM Conference on Digital Libraries*. NewYork: Association for Computing Machinery. 19-26.

Wolfe, J. (2001). *Pedagogical uses of annotations and annotation technologies*. *PhD dissertation*. University of Texas at Austin.

Wolfe, J. L. (2002). Marginal pedagogy: How annotated texts affect writing-from-source text. *Writing Communication*, *19*(2): 297-333

Wolfe, J. L., & Nuewirth, C.M. (2001). From the margins to the centre: The future of annotation. *Journal of Business and Technical Communication*, 15(3), 333-371.

Woodbridge, B. M. (1922). What is annotation? *The Modern Language Journal*. 6, (5), 293-295.http://www.jstor.org/stable/313886?origin=JSTOR-pdf&seq=1#page_scan_tab_contents.

Young, C. M. (2011). *The experience of burnout in case loading midwives: An interpretive phenomenological study*. Unpublished PhD thesis. http://aut.researchgateway.ac.nz/bitstream/handle/10292/2447/YoungC.pdf?sequence=3.

Appendix 1: An example of the hermeneutic circle and the literature review. Annotated

notes made over time when reading Feito and Donahue's (2008) (see chapter 3.6 to 3.6.1).

Salvations , parting prenomenar - (what is this?) Description, naming prenomenar - (what is the constraints) Description, naming prenomenar - (what is the const (Scalvation" In The annotatia Minding the Gap Re-reading annotation Jain readup Annotation as preparation for discussion of relevent text (150, Riccen, Salvapris Daalue) JOSÉ ALFONSO FEITO Futuer notes made Saint Mary's College of California, USA oftext PATRICIA DONAHUE Lafayette College, USA Jaton ABSTRACT This research project examines classroom discussion in its relationship to reading as made visible, through the practice of textual annotation. In order to develop a (2005 rich description of student reading/discussion processes, we targeted multiple DEFractitions? undergraduate seminars at a liberal arts college as they encountered the first two of text Acts of Shakespeare's King Lear. We collected triangulated data from these class sessions including targeted reading surveys, student reading annotations, naturalistic observation of real-time seminar discussion behavior, and student reflections. - Not know - p how Our analysis of the students' annotations relies upon reading theorist Wolfgang Aser's conceptions of interpretive gap, consistency building, and individual repertogetstaf toile. Our discussion considers the theoretical implications of this local, in-depth -anbig data for the broader analysis of student reading and discussion practices. Bathes. fedium annotation, difficulty, discussion, pedagogy, reading, student readers KEYWORDS connotatio North - historicoel ficoers remotioness MOST DISCUSSION-BASED COURSES - first-year seminars, capstone (resonance courses, writing courses, numerous courses in the Humanities and Social Sciences, the occasional course in the Natural Sciences or Engineering levip -Moundestad asachd make significant use of assigned texts. Such 'readings' create opportunities for to elect two -Mis-Interput students to confront provocative and difficult ideas, to build upon them, reader prov - unation challenge them. Of course, dialogical work of this kind does not just happen, but must be carefully prepared for. And one way in which students prepare aboutsonet ready themselves for intellectual engagement - or so the story goes - is by writing Atlas for Strategies hemselves into texts through annotation. By circling words, registering chiscussion of -Annatate-bradnet/highlight -play with words Arts & Humanities in Higher Education purlduf C 2008, SAGE PUBLICATIONS, Los Angeles, London, New Delhi, Singapore and Washington DC. ISSN 1474-0222 the chilffoult - id pro runder and p VOL 7(3) 295-307 DOI: 10.1177/14744 perior readept this prooss dees. [296] text Gid - trusher a sogaring Howdors the -testing drospaus 681 reference - Slow dether reade Simons that November 18, 2010 down difficulty Seen as an approximity Satures try not to do it all atorie on distantly reading taxt -HyperDITY - two knidsor Components



lecosin Fra

Product ?

d use of

the readers throuldge or

Meaning, underfanding

of gaps in

P Die - + post

meteolicr

Feito & Donahue: Annotation as preparation for discussion

essay, suffice it to say that both, although in different ways, draw attention to the role played by readers in the process of meaning-production. Iser is concerned less with matters of authorial intention or expression (although he does not ignore the fact that texts are written by authors who have ideas in mind), or with issues of cultural or historical determination (although he also recognizes the status of language as social instrument) than he is with the nature of the reading process. How do readers read? What are the moves they make? What do they identify as important? These are the kinds of question that interest him, and they are also the kinds of question that interested us. In brief, reading, for Iser, is not a process of passive assimilation; it is foremost a process, a process in which texts emerge in the dynamic interaction between what can be understood as residing 'on the page' and what can be understood as being put into play by readers who carry with them a rich repertoire of assumptions, beliefs, habits of mind, and literary knowledge. As Iser explains, reading 'animates' the text: while 'the written text imposes certain limits on its unwritten implications . . . at the same time these implications [are] worked out by the reader's imagination' (Iser, 2006: 956-7). Iser's theory of reading grants to readers considerable interpretive autonomy. However, he insists that a text includes features that readers must recognize: his is not a theory of 'anything goes'. Terms that appear throughout his work include 'creative', 'dynamic', 'interplay', 'anticipation', 'retrospection', 'gaps', 'repertoire'

While Iser's study is rich, produced over decades and revised throughout his lifetime, a selection we found particularly applicable to our study is 'The Reading Process: a Phenomenological Approach' (which appeared originally in *The Implied Reader* – Iser, 1978). While the analysis of the reading process it offers is quite complex, the ideas that we have found most applicable to our project can be summarized with minimal distortion. Iser begins with the basic component of a work, its sentences. He defines those sentences as making 'statements, claims, or observations' or conveying 'information'. However, while sentences must be plumbed for what they appear to say (or not say), sentences only partially constitute the text *qua* text: 'they remain only elements and are not to be identified with the full text itself'. The text only begins to emerge when a reader 'climbs aboard', when readers employ their imagination in discerning or deciding how the linguistic elements connect or relate to each other (Iser, 2006: 957, 955).

Following Husserl, Iser explains that one way in which readers establish connections is by forming expectations, expectations which, he notes, are 'scarcely ever fulfilled in truly literary texts' (which is why interpretation is an interminable process). However, while expectations are always being formed, they are also always being interrupted, thwarted, undermined. They can lead us 'off in unexpected directions' (Iser, 2006: 959). Thus for the reading

[299]

Downloaded from ahh sagepub com by guest on November 18, 2010

ctatus

a SOCIAL

(5 Shuneed

2



lecog 100

the notes, are full contractions interpretation is always being interpretations interpr

dynamic to keep going, readers must not only form expectations about what lies ahead; they must also always look backwards to determine where and how a particular pattern may have emerged. Reading is, therefore, a process of retrospection and anticipation. Another way in which Iser discusses this process is in terms of 'gaps'. A work contains a huge number of moments of disconnect or difficulty which readers must contend with in some way in order to keep the reading process in motion. Importantly, for Iser, different readers will fill in these gaps in different ways, since they bring with them their 'own particular history of experience', their own 'consciousness', their 'own outlook' (Iser, 2006: 958). This tool-bag of possibilities for gap filling that readers bring with them is referred to by Iser as a 'repertoire'.

While Iser builds on this idea of a 'repertoire' throughout his career, for our purposes here we will subdivide it into two major categories: an individual repertoire and a literary repertoire. The individual repertoire consists of personal background (gender, ethnicity, race, class, experiences), assumptions (or ideologies), value systems, acquired knowledge (so that who readers are inflects how they read), while the literary repertoire consists of ideas about literary structures, devices, histories (so that what readers learn about also inflects how they read: the more sophisticated their education, the more nuanced their identification of gaps and the richer their strategies for negotiating them). Of course, it can be difficult, at times even impossible, to distinguish between an individual and a literary repertoire, since Iser would say that the processes of interpreting texts are acquired and not innate. But his larger point seems to be that there are ideas about literature and methods for studying them that are specific to the disciplinary study of literature itself and can be separated, even arbitrarily, from the general experiential base that a reader has acquired - or has been socialized into - throughout his or her life. As Iser explains, while every new text read by readers represents an addition to their repertoire in general, readers do not just passively assimilate a text but read in ways that either confirm or challenge their repertoire. The more sophisticated the reader, the more likely he or she will be willing to challenge their repertoire.

Let us end this summary by mentioning one expectation that Iser believes most readers share: unity. As he explains, readers group together certain aspects of a text 'to form the consistency that the reader will always be in search of' (Iser, 2006: 965). Readers engage in 'consistency building', attempting to establish a <u>unified meaning</u> even if that means ignoring anything that might threaten it. Iser's elevation of unity to near universal status has been a source of complaint about his work, especially in the wake of deconstructive theory, which argues otherwise. But whatever the theoretical legitimacy of this drive to unify, the fact remains that at least in the USA, most high school English

fanitier -fanitier -ghove what mon fuseater a can Downloaded from a [300] a consistency bu oaded from ahh.sagepub.com by guest on November 18, 2010

Moments er disconnect and - Disferen Neades Fillintu gerps in





aspectsi

Feito & Donahue: Annotation as preparation for discussion

classrooms emphasize a thin formalism in which textual unity is indeed a powerful expectation. The recognition of this pedagogical bias highlights the special importance of Iser's work to an analysis of student reading. As we will discuss in the following sections, it is possible to classify student annotations in terms of identification of gaps, negotiation of gaps through repertoire referencing, and efforts to build consistent viewpoints. Such classification not only provided us with a way to manage our data, it also highlighted the ways in which student readers implicitly theorize the work of reading.

A TAXONOMY OF ANNOTATIONS

Using Iser, we have organized the annotations into four major groups: trackings, identification of gaps, individual repertoire and literary repertoire We are also currently studying the ways in which individual readers attempt to 'consistency build' as they navigate their way through a particular text, but we are not providing that research here. However, we do identify a few annotations that could be understood as performing this function. Provided here are examples of each category type in addition to preliminary commentary.

This category represents the effort of readers to register what they are reading in some way, either through underlining, highlighting, questioning, or paraphrases. Especially in paraphrases, such efforts represent a reader's desire to translate what is being read into familiar terms. But any mark or encapsulation might serve to help readers 'climb aboard the text', acquire a point of entry in order to begin to construct a reading. In these extracts, we have provided examples of paraphrases, followed with line citations indicating their approximate position in the text:

'Lear divided his kingdom in three for his daughters' (I. i. 37-8).

'Regan tells Lear to ask for Goneril's forgiveness' (II. iv. 144-50).

'kicks knights out' (I. iv. 231-44).

TRACKINGS

'Lear angry at Kent' (I. i. 160).

We have also placed in this category comments that suggest a reader's frustration at not being able to track the narrative. For instance, one student's only annotations were repeated questions marks in the margins until he finally, at I. ii. 104, wrote in bold capitals across the page: 'WHAT IS GOING ON??!?' After that, he stopped all notes and his survey responses indicated that he had stopped all reading there as well.

2 maibleto tractites namatues



bicceuss voluctrie retarie

Feito & Donahue: Annotation as preparation for discussion

'This is like the story of Edmund and his father, he thinks his father is too old to be in power' (I. iv. 247).

'love = business arrangements' (I. i. 63-8).

'quickly turns on his son. Why so scared of Edgar and not suspicious of Edm? Money? Title? heir?' (II. i. 57).

'maybe a young's heart understands love best?' (I. i. 108).

CINDIVIDUAL REPERTOIRE

Athere > borne bythe OKTERS ofter

This category represents annotations that suggest that certain moments have attracted the reader's attention because of concerns or issues that have personal significance for the reader. They frequently involve establishing blame or comparative fault.

'such a liar! He is in it for himself and gaining wealth and power' (I. ii, 69-70).

'daughters only want power, greedy!' (I. ii. 312).

'dismisses daughters very easily, going to end up alone' (II. iv. 218-30).

the interaction e internet intended for subsequent discussion purposes: In some cases, these types of annotation take the form of questions clearly

'what blinds Lear from seeing true love?' (I. ii. 286-8).

'Is it wrong for Goneril and Regan to do this to their father?' (II. iv. 272-3).

"Who should we sympathize with? Lear or daughters?" (I. v. 1).

'What are Ed's true motives/goals? Wealth or recognition be of bastard status?' (I. ii. 182).

Who is more at fault, Goneril for not respecting Lear or Lear for thinking he is in control of everything?' (I. iv. 258).

The last three examples display questions with embedded either/or comparisons. They may emerge from a personal repertoire that includes moral 'debates' as a familiar form of discussion. These annotations may be informed by this conceptualization of 'discussion as debate'. (See Tannen, 2000.)

LITERARY REPERTOIRE

This category suggests a language for literary understanding that these students have brought with them, a language that they employ as 'natural' and 'obvious'. Evidence of a literary repertoire can be found in the use of 'literary'

terms: the conundrem to the Seerch for hidden nearing is-'hubris' (I. i. 2: & 3 de-construided? 'similes' (II. ii. 69-75). - A pre-supportion that there is a search to be in verted - A pre-supportion that the author is jancare of holden Neoin - What is left when ar 2) unavione - A procemption betweet is found is the revelopment of the truth - A presumption that both the individual repetione of the author & noder connerge when both vory have very different repationes indeed

desoubl

procesa

+ autor

Contantora

pre

bacs

ess ct

'rhyming speech' (I. i. 256-67).

'youth vs elderly motif' (I. ii. 181-2).

'Fool's sayings foreshadow what is to come' (I. iv. 115-24).

the recognition of allusiveness:

'Bible reference' (I. iv. 284-5).

and the identification of 'lessons':

'Lear's lesson: modesty' (I. iv. 114-24).

'Love not a product of flattery' (I. i. 77-9).

It can also be found in a kind of reductive moralizing where the students believe that the 'moral' of a speech is something easily reduced to a cliché, or two:

CONCLUSIONS

'action over words' (I. i. 77–8).

- what do they

'words mightier than sword' (II. iv. 158).

'don't jump to conclusions' (I. ii. 85-6).



We began this article with a mention of the importance of textual annotation to class discussion. That is the larger consideration this project was initially designed to examine, but we soon realized that before we could confidently establish a meaningful relationship between annotation and discussion, we first needed to classify the annotation data in terms of interpretive negotiation. To be sure, this will not be the only time when we will need to take a step 'behind' in our thinking. For example, to identify what kinds of annotations lead to productive discussion, we will need to articulate the qualities of a 'productive' discussion (pedagogy, like interpretation, is truly a process interminable). Is discussion most 'productive' when students are enabling each other to 'climb aboard' a text, or when they are sharing components of their personal repertoire? Is discussion most productive when students identify gaps which they can then collectively fill, or when they put into play a pre-existing language for the 'processing' of literary texts? In addition, we should mention that our initial exploratory data contained tallies of discussion participation, but the coding schemes were not calibrated to the theoretical model that we ultimately used to interpret the annotations. In the future, we hope to collect real-time discussion observations (accompanied with preparatory student annotations) that specifically address questions that arise from the reader response model that we worked with in this article.

affecting

e - exist and pre

Downloaded from ahh sagepub com by guest on November 18, 2010

[304]

Feito & Donahue: Anotation as preparation for discussion

neutrally private it

Shoved annotative.

While there are many, many possibilities that we still need to 'climb aboard' ourselves, let us at least gesture in a particular direction: the importance of the difference between private and collective reading. As suggested earlier, when Iser speaks of reading, he seems to have in mind a relatively private process, carried out in isolation. When students annotate their texts, it is likely that they are reading in private (although we do not know for sure). But we do know that when students begin to articulate for others in the classroom what they have read, they are creating a new type of interpretive encounter: they are reading in public. It is likely that what students assimilate through public discussion will influence not only their rereading of texts (we have collected data on rereading which we will discuss in future essays) but also how they annotate texts: what they see and how they classify their percepindrided tions. This possibility is significant to us for several reasons. First of all, as we have already indicated, it suggests that what Iser names 'individual' repertoire may not be personal at all, but may consist of 'social talk' or the kind of imbricated discourse that Mikhail Bahktin (1982) emphasizes in his work on the dialogical (at the end of the category of 'individual repertoire', we have placed some examples of such 'social talk'). Second, it also suggests that for pedagogical study a meaningful addition to Iser's categories might be 'collective repertoire': a category that would foreground the circulation of language(s) Tiliberal during discussion and the appearance of such language(s) in the 'private' work of annotation. A 'collective repertoire' may even be extended to include as large a group as an institution itself (and groups even larger). We have already mentioned that St. Mary's (the site of the research) is a small liberal arts college, but it is also a religious college where one would expect moral considerations to be highlighted: as we saw in our annotations, this is precisely C engo the case. In addition, as freshmen, these students have all been initiated into these seminars with a semester of classical Greek works, emphasizing the Platonic dialogues. We saw indications of this part of their collective reper-Consoleret toire in annotated questions dealing with the definition of abstract qualities interneta and socras pressures exceled to (e.g. 'what is true devotion?' at I. ii. 182). As we hope we have demonstrated, even in a study in its early stages, an assurp effort to 'mind the annotative gap' - to consider what is both said and not wat is the said when texts are studied – opens up a rich array of possibilities. What is initially framed as a study of a particular reading practice becomes transformed into a study of the practice of reading itself, of the buried assumptions that teachers as well as students bring to the collective enterprise, of the ultimately public nature of the work of meaning making. Our particular DASEIN MITSEIN research process represents one possible common trajectory through which (HEINEGGER) theory informs and permeates this kind of enterprise. As we more systematically investigate it, our students' work may demand that we further explicate hered Shared I not whent is thapent / not [305] repetorive US Social repetorive 1 Sound and netscuid/ felt / not felt / Seen windcaded from ahr sagepus com by guest on November 18, 2010 Re . This and these bat if it remains unsaid expected fallant from Say you what p back to the whiter, and because Sandre poss social impact (Sudert)

/

Appendix 2: An example of an edited transcript with lecturer 8

- Α Annotation is one of the most powerful teaching aides we've got, but it has to be used at the right time, before and during provision – it should be continuous. But I think it's like a hit and run, you get an essay and it comes via computer, grade mark whatever, and you put your asinine comments in there and you are firing all sorts in there - if you've marked 15, you've seen the same reference 13 times and suddenly your tone changes and your grammar changes. Your frustration comes out on this poor unknown students' feedback. You hand the thing back in and it goes back to the student and he opens it up - "whoa, what have they said about it, I'm crestfallen here". His mate might say "hey, who marked you?" "(anon)" "He said this about my feedback, you should see what he said about mine, I'm fed up with this place, I'm going to leave, I've had enough of this place". "Why don't you go and talk to him?" "Well he said at the bottom to go in and see him but I don't feel like it because if he says this on paper (I've heard he's a bit of a hard marker) what's he going to be like?" Whereas, if they come in or if I had used feedback properly, or if I hadn't started to personalise the feedback to that extent, I've lost all objectivity rather than being neutral I've become frustrated. There is a danger when you use feedback, you can invariably become quite hurtful and depersonalised and out of context. The problem is that it requires us to put a bit of effort in with the students, that's what I'm here for.
- Q Does it make a difference having that as a working philosophy for your practice, making a difference through teaching and learning to nurses to develop and transform and think out of the box and see them as individuals as a priority, but isn't there an odd parallel with the NHS and organisations as it is, which sees people as numbers and so does higher education, but you are making it a personal experience?

"What are we here for? Do we care for the person, the product, the system or do we care for the person within the system? What is the answer, I think of the person first and that drives my practice... Think about the Ombudsman and Francis report, they are just the tip of the iceberg and I want to instil in our nursing students the need to be advocates, be critical and courageous. That's what I've done for the majority of my career so that's probably what my philosophy on care is. It should be individually tailored care and not generic and you could almost draw those parallels with annotation feedback ... so I can draw a lot of parallels from my clinical practice. A lot of lecturers say (I've heard them saying this) "I'm not going to make any notes; you make them. Person centred, student learning like the old way, we can learn in theory. It's a learning journey, it's a process, how can you tell if they've captured what's required? Think about the Ombudsman and Francis report, they are just the tip of the iceberg and I want to instil in our nursing students the need to be advocates, be critical and courageous. That's what I've done for the majority of my career so that's probably what my philosophy on care is. It should be individually tailored care and not generic and you could almost draw those parallels with annotation feedback where I use pat sentences, generic feedback for everybody, so I can draw a lot of parallels from my clinical practice.... What I care about is, the student comes in and they sit down, if they are a bit woolly at the beginning of the session when they leave, they understand. My parting question to the student leaving this office is "do you understand what we've been talking about and have you got enough records?" A lot of lecturers say (I've heard them saying this) "I'm not going to make any notes; you make them". Person centred, student learning like the old way, we can learn in theory. It's a learning journey, it's a process, how can you tell if they've captured what's required?

А

... They came in and I got them to talk. It took about 3¹/₄ hours for her to work through the essay. I had time and she had time and we went to about six o'clock and because I had time she understood. I said for her to take my feedback and take my annotation, go away and come back, and she did. By the process of three or four supervision sessions, I annotated and re-annotated and then we went through them together. I didn't do a great deal but I just gave her permission to verbalise what she wanted to write and I just showed her what to write. She didn't get an outstandingly high mark but she did show an outstandingly high understanding and every time she came in she had a folder full of the essays that we'd written in chronological order. I could see by reading, (like handing over a baton), as we had written stuff as I'd given her permission and examples of how to write - she developed to a point (by about the third session) where she was off on her own and she felt confident because she'd seen me. I actually thought she was going to complain about the mark but she thanked me so much for the time and effort I had put in. No matter where she is at, whether she is a Chief Executive or working two or three days a week she will treat people with respect because we treated her well and supported her. It doesn't matter about the system, it's just me and her both sitting together to talk about the annotation, like a focus for a conversation. We annotated. I think there is a link there and it's what goes with the annotation, the investment in the student and it has to be done with the student. Right there face to face, or live on the end of the phone -I've done it with a webcam before and that does work or the other end or Skype[©] and held up the annotation to the camera so they can see it. But it's really person centred student learning like the old way... is a learning journey, it's a process, and how can you tell if they've captured what's required.

- Q Another key thing is your efforts to make a difference to the student nurses training, they will then qualify and make a difference and impact on other students who do their training and it's almost like reinforcing their principles in nursing - caring and compassion and all of the six c's which count and actually not only talking the talk but walking it and demonstrating by keeping time, effort and interest in that individual.
- А Firstly, do they demonstrate an understanding of the topic - are they safe? For my mind, even if it's badly written, that's an academic issue which we can deal with and I might call them in or refer them straight to their personal tutor if it's an academic issue. I'm asking if they have a grasp of the concept and do they understand and not only grasp but a safe grasp. Over the past couple years of marking, there have been a lot of essays where they weren't particularly academically outstanding in that the syntax was wrong, the grammar and punctuation was wrong, but the level of knowledge and understanding was wonderful. I've also read some wonderful works, technically precise but from a nursing perspective, completely imprecise. So, is it relevant to the module learning outcomes and the aspect of nursing that it's testing it on? That's what I'm looking for. The level of understanding, and can I be confident (not just as a lecturer, but as a professional person) that if I left you alone on a Saturday afternoon or Sunday morning in charge of a shift with two healthcare assistants and a first year student nurse, are those people safe in your hands? That's my benchmark because if not, we will do it again and hopefully you will come and see me for supervision. So I challenge on that basis - are you safe and are you a comfortable practitioner? The rest of it we can work on. No one died of a misplaced comment in a reference list. What you might do, if you misplaced that comment, if you talked about clinical values, blood pressures, that's what I'm interested in.

- Q So when you read it...you ask...what they thought and therefore it's a positive experience and productive.
- A I tell the students that when I mark I will be objective. The feedback I give the students is not warm and it's not cold, it's completely neutral. I pull out the good points and I pull out the bad points. A lot of people write things in like "thank you for the assignment on nursing care Mary, (I don't personalise) the focus is purely factual, you explore the care of someone, you highlighted several salient issues which is this, this and this and the areas which you identified as particular interest in point in relation to this. However, there are a few areas where you may wish to focus on and I have noted the following items for your benefit. Same as the reference, I pick out every single reference and error. The reason for this (it's only worth 10 marks) is precision. Precision doesn't bother me but it's to show the students that we should be precise. If I've dropped them marks because of this, yes I've got the threshold of safety, where they could have improved their academic writing too because its 40 per cent safe, but I'd like them to see where they can improve. They might never improve, they might never.
- Q Back to the issue of care the extra care and effort (sometimes 6 times, sometimes 3) you put in means you actually do care to get them through, so what triggers your motivation to put the extra mile in and get them to pass?
- A They've asked for help. I've a duty of care to the people who I care for. I'm not going to apply a paternalistic blanket over all the students. But if someone asks me for help, I'm not going to fob them off. I say "yes, come in, you want my help but I want an hour of your time" I came into nurse education to try to improve my profession and the way I do that is to show these people that it can be done. And by inspiring one or two, that's the way to do it. I actually genuinely care about them so if they know I

care about them then they might care about other people. That's what helps me to sleep at night because if I start to contemplate the other then I won't sleep and that's when we lose faith and its game over.... I don't write it for the students, that's the whole point in my annotation, they are writing it. What I'm doing is using some sophisticated communication techniques. That's what you pay me for because someone interviewed me and thought I might have the skills and abilities (although I don't sound like it) to inspire and take the profession forward. I do it in a blunt way, I communicate and talk to them and use the skills I've used with people in A & E departments, in clinics, on clinical ward, in an operating theatre (whose anxious) and I talk and use those skills - eye contact and body posture, recognising boundaries, feeling empathy, intonation of voice – and I'll tune in to the student. Because I should be able to do that rapidly as a senior nurse, and what I'm doing there is hooking into a student in an individual basis and I want to know what they're like, what their fears are and then I probe. I probe and I probe. I say "how are you?" and they say "oh, I'm ok" ... Same with the student, I go under the skin and try to bring out their hidden academic genius or underpinning knowledge, it's about safety.

*It's like a **dark art** because no one teaches you what to do...maybe the annotation is just a veneer...not the real issue here but it involves capturing the thoughts I had about the essay at the time of reading and if we do meet the use of time when sitting with the student. Reading back the annotation with the student helps them to make changes and we can talk about what the annotation meant to them and me. We can use the time to discuss the temporal nature of reading and understanding the annotation within the narrative of the essay. We can come to some sort of agreement of its meaning. That's why annotation without the student being there to make the changes to the essay content is pointless...the dark art is in the issue of time when the student reads to write for the essay, then reads their interpretation of what they have written, then reading the parts and the whole. Then you add the same process for the annotator, drawing from time and experience and then get them both together in a room to discuss what essentially is both their understanding of the narrative at that moment in time...clearly interpretation changes in time.

Q

**Could you clarify what you meant when saying that annotation is like a "dark art," that no one teaches you what to do, and that annotation maybe just a veneer? You mentioned safety before that and I wonder what you meant.

А They write it not me. I never say you should do this or do this, I might make suggestions "what about this?" I might pose a question that triggers a response but they offer the answer. I offer them a way of writing it in a slightly different way and that takes time and that takes skill which is what we should be able to do. But I don't think we should cast the student adrift because not everybody knows the mysterious art of moving from the different levels of description to critical analysis. It's about putting the time in with students and if you ask someone if they know how somebody can write something if they clearly don't know how to write. We get paid a lot money and it's a craft, an art, to do what we do and we should use the skills. Yes, caring is part of it but the skills should be to communicate with the student for the betterment of them, to improve their standard of life and their standard of living, health of the student. Life and wellbeing. We've all been awake at 3am, panicking and not getting it, we've spent days looking out of the window with mind block and not being able to write a sentence, watching westerns or playing video games which relax the mind. We, down the line, have these problems, what are these people like?

- Q ***When you are with a student, it's about finding out what's within them, perhaps understanding is another aspect. So it's finding their understanding and the meaning of the words they are using or how can they get the message more clearly or succinctly on the page. What is in them, but you are trying to get it out of them.
- A ...Bringing the computer up and writing things down, we capture it. The annotation is just capturing the discussion at the time. Because someone has said they are not good with words, nobody has sat down and shown them, and asked "tell me what this says" or "tell me what you read here". But it requires pen and an ink cartridge and a student next to you. It's a powerful medium on many levels but I don't think people give it any thought as to the significance. I have always done. I will not supervise without writing something down as it's like a snapshot, its capturing that one point in time, knowledge and understanding at that time.
- Q It's also promoting collegiality, avoiding the perception of authority from a decontextualized feedback comment. You are with them and modelling good eye contact and empathy – all the things you mentioned – therefore it sounds like a very powerful process.
- A ****We are unbelievably powerful people and we have the power to inspire, enthuse, intimidate, frighten, de-motivate and destroy or not care. We can do that with our looks (because we are all experts in communication), what we say and certainly what we write and I have become aware that annotation can carry that attitude. I have been a senior nurse and I came into education because I wanted to change the standard of healthcare practice and when I need to be, (and I have been) I make some harsh decisions, but not ruthless, and I've removed students from this programme and sleep nights, because it's for the right reasons, for their benefit or for the benefit of the service. We are unbelievably powerful people but when the students come in here, I

see this as "we are all nurses together; the difference is that I have my registration... Every student that comes in susses out that they will get a posh cup and now I've none left. "They say "what, you are offering me a coffee?" Again, role modelling. In a few years' time you might walk in and you say to the most junior person in the organisation, "do you want me to make a cup of tea?" It's more than that....

- A Because I am caring for the student nurse, I want to inspire and care for them. My approach is the same "how are you, how many tissues do we go through in this office?" ... Pre-registration, post registration, I've not got a problem with, it's the same we are shaping minds here and trying to get them over the hurdle. I might do it with one or two of them that might not have been able to do it. I care, whatever care is, and it's a nebulous concept but care is offering time to students and treating them as human beings. It's allowing them the opportunity to develop. I'm capturing their thoughts and also a bit of mine on the way but they have to provide the material to work with. My role is to do something with that. But I cannot give them me. I wouldn't want to be me. You couldn't work with a room full of me.
- Q I appreciate you giving me the time to interview you and your thoughts are very well considered and will be very useful.

Appendix 3: Research notes from chapter seven

Stars from the taken for granted Notes a Soutence use of the woord "It's stated ١ the reaction to the word, which sprand Rylisation led to thinkip why it had such a readion andotor? while did they that that statiga sand & Sen with fue was 'in was 'not was be? why did other can next uber may ignored it & accepted it? what is more to F) starting The test of honor outres is to 'Play' with a word is order to gren appavalues to now possibulities. The I try out to wood " ot " appared to be linked to the student grapplup to a) interpret the message file reading text and I white es Levelopies that knowledge within an essay. At thisself used the phrase thing-in-itsde to refer to hav sanding appears to an individual, and what appears' IS an abet/phananan (eutity or a word) idea / cancept. : It is if we more to consider a funct the reader substitutes words, ideap felt recognife and groupping what a concept means from that "Thing"'s what is being read. Plans tells about self references and interpretip some trang through substitute some undertanding by remaining "in" Centerie or asay and refer that dated to their am Sentenie or asay and refer that dated to their am Selt - Itself Selt - Itselt inc nump bey head is the dated indhemeaning of it is substituted by the word in the meaning is not yet fully graspid. It's This is a possibility when meaning is not yet fully graspid. Oc is it likely that they the sholent had or is it likely to is it they the sholent had Sevie underfording of the aderence of what they are reading ? Yes Hagadagy - Do, if wants aselans ad ready solf effragy is law. they it is likely that ressay carted will not befully when by doseotie menlege, then it will be descriptive o Exportional and more rescuart with the students thit view. Hone (I) is fullingly to be a process of underlandy that eventially because better deudged and end meaningful.

2 - Ind frequency was noted soldy to canced hew many timer pre wood () Stanked & Southing and to law Self efficary. The frequency Suggest F was the vanatrie affe for sens Auchuts . Hanaver ducto the Andert reading (free dolect) as a thing interpretation is always related to their first and this velales it Seems to self undertending - The to the process of Gracing the maan of smarking and the inpersentation Ships from mis- to madestanding was gradents noty Nearan in the this and temporally grapphy with many. It the further to the this and temporally grapphy 40-55 (It prequency el. to the flag. t effan 55-75 4 efficienty 1-3rd what may be - Lipoal play on trand & meanip) and by coss. REMOVE Atmpinitself Cauter P R) 1.23-1-2.5 could more to discussion Read Self Cousi. (The O goin - replacede for 7+9 annadors distiked (17) so it was their save. Why would they take exception? Plassiby becomes ip sandsand station a Sentence 3 noth speak uc -what It are they wooking to? Is is the loss topse of Southerle, or "It is of syntetically aut of ferror for Same, IF) sprifies a naming issue - what is mout by Substitute a topic far fie word (it), what does (it growify then an an uncarectary lend? Objectificate = Samethip un gracpable/hot or subject of Thing in dealf - noemena) = transautantel. Ttself in the thinp - V the self grapping to make save of the thip beopread, fue object Gracion units Concortiation ; that is knewlage Nevisop wethink p accepted than' monetically ind a forgette.

"Shi(iP) - why is this significent? Firsty because it 3) is an innuedlook reaction it seems to the connected or reading of the word. They state starting the sentence with the word it is par state, lados flaw, is inspecifie (T) and noties the text read body (it) remunds the annotator what it sands like, pethaps if substituted for the wood (it). The annotator play's an (it) (shit Mywe and perhaps some to student to used addivisedly. Nursip, even when we Hen about is a physical process, physically writing about Sandtrup physical and toological. The annotator mary be refaired to the endodmant issue where (F) is abledition and different, not specific and named? (Bon/NEGLECT on one level (iF) when not used to voter to saveance is unatured, difectifying and devogations. Sancturer its possion at the start of a sentence many. So the Context May be Substituted in appropriately and the students use of it ill-Considered a Judged. In other woods havener appropriate the word is 1 to Save (it) may signify something regative, something have disembodied. This is a remote possibility as the essay will have been read and the castext noted. However if the daity daity of discease in the easy is poor then the with still and demanshorting learning has occurred mean not be obrians. demanshorting learning has occurred mean not be obrians. Because the clarify of diserance intractes understanding to poor danity misurdestending and it may be more manifest to poor danity misurdestending and it may be more manifest and self contented as undestanding is filtered though what the Student thanks they know beloart the Subject matter. Analogic phroph the hermaneut cride? Matter, materies (morph the nermenentic Conde". Starting with the student - what is thosen about the subject? What is known about history of ideas? felated to the delet what is known about history of ideas? felated to the delet being need? Pre-concerned ideas a steating neading, being need? Pre-concerned ideas a steating neading. is that greep, superfrained greep temporally informed. Many essays are written with minimal development for Many essays are written with minimal development for Hurs mitted greep hating interandetten superconcerd. this hitsal grass mating interpretation superficeral and brited, undeveloped by a wider sauge of neading. This

Appendix 4: Research notes from chapter eight

Rhetoner J. (1) 0027 Saying it well with poor" underling the adop possianen, canning through the pane of logic, reason and publishing. The phrase appears to encapsulate the underlying philosophy of acadomic activities and relates directly to - endence based practice (EBP) So is divically relevent. If there is a lack of estop to personate someone of concession then it is tikely the personagles does not home the knalledge, facts or skills to use suasony discourse . Or perhaps their argument is weak because they are not even commed of its monit. This also refers to writing begauge the reades reads to be informed, to gether informative to than interpret its relevance to an argument. So ; fthey do not undertand ulattuces are reading and its volumence fully there is the tiklihood of winity bidly, with a lack of clainty then with further reizien clarity is imprised ? yes for the studente -yes for me as gdent! The whole there includes to the same errors I have made, still make and will make in the - future. It is about interpretection raventerpretet If formes 15-23 releasing annotators is eves which veter to rhotonic (Structure, plot, initation, flow, coherence, enors/connections) Marity of whiting then Surely it relates to my and whet of 7 His Obrans because when receive feedback my write is imprised becaused home to think more (

P-2) so the trong is J'me identified a thene and J'mstill writig / thinking / interpretig and matting the same errors as the students the anicotatois are writing about ! ficoes Dijanni branne. academic had brance branne. Since reading about rhotonit I can see it everywhere - In the media, in comedy, in politics, " well Said but where's the endance?" In my an writig, my teading procession my ain annatation back to students. I can see it in my colleaguer wak, in neetings and in everyday conversations. That's why when I read "it" I realised they ware not "saying it well" with poop, lacked clainty, coherence and offen without endence (use of the hterative. That makes writing about thethis even more difficult because as i'm writing/ interprotion the research data scanses the rhotonized errors. I do not understand, I am canfred by the theory, I don't know what Ricoec- wears all offlicture ! IMITATION (Minusis) instation DF what is being read, such as how an argument is arganited; Studice, cadence, key parts, sign posts, couring, use of lope / probability / open uphangars of understanding? Two annotator has a part to plany be cause they will be minted / model accordinic of chinical behavers - tors something & gard, what is reinforced, why it is relevent - Feedback offersthe Student à plimpre of something potessadien, trained person extollip the vitues of nursing

"/Temperatity) lived time / distanciator bas understanding changes in time. what does' momessel-3 mean? How does it relate to the research extrads? The analogy of water in aglass seems relevant 3 to me. Abit cheesy perhaps but it seems to make More souse when I think of it in that wang - the glass is the structure holding the knowledge (M.I) m1 - when full = movers 2 - rdeas gained through M3 reading. After time the knowledge dissolves/evoporates to reduce the pund in the glass - Tasting hypothass, more reading, nous roloes & new understanding -Fillop in the gaps. Minesis 3 - the latest realisation at the time, in the Howent: Therefore M3 is in pepetral metric between MZ? (Sections 8.51-8.5.3). - this whole process is proverlied in (all)aspects of the thess - chapter 3 literative version, disgunptue nessorch notubdology, writing chapters 1=2, the nesearch notubdology, writing chapters 1=2, the interviews, figures 15-23/24 table (12), my pre-prod concerned ideas, my chriccal practice, replactions, person publications, reading hermonents, aspecially Ricoens, reading, revising, revising the theory in a parpetral temportal publics, Even writights theres, re-writig overtime, apresfereback, apter years charges their relevence, their meaning, their Petercey / poignancy. My undustanting (misudestady muchosthe rememente circle, which inherently is temporal. . The used monossi-3 the research proces in chapter six - cinfact all of the theses. . Find it very difficult to clout fy (prosent the Stages - Rehaps M3 is the most durious, Perhaps in an asquirent H2 > H3 Canbe viewed? TEMPORAL STATE OF FLUX

KAODNIT Define it debtend Research extracts "A daik at and the use of metaplan. What is metaphor?" The lectures (8) was the most eloquant and thapplitful about annotation with a style of feading which actively used annotation by Strip with a student, asking questions, and then "Shard-p" them how to improve the > The learner demanstrates in the interview minness 2+3 - they articulate their new and write telling of thestory begin to identify new rectisichers. lectures even states how reprosing it is to be interviewood. Therefore, this is a good example of MI-3, and in their realisation standation being a " date at", with an awareness of their position, power, the verser of annotation, perhops a "thin" verses? The annotators in spiring, transformation attitude to annotation was inspiring to me too. At the end of it Sfelt transformed too, Just like one of their Students. I then have it felt, I could fed their entirem about teading and leaving, and an appressedies about annotation Joarld only unonto envelate: Thereforend's see in the Lectures attitude to annotation that when a date and ", they also meant annotation to be the opposite, to Thinwate, to mapine, to show; to guide and aspire to Smething Temporel State of Aux new; smetring brought to light. I The theneer is the opposite of the whide attitude lecture 3 projects (See appedix 3) which seeps mother very beings. The never is what some lectures preject, but these 3 meanspatiness meanspirit ness (See drapter 10) Lecturer (8) is aware of that That the tone that annotated is instructive and ... an art, that can be dark, or has the potential to bring to light H2-M3 -(Dendeptions in the dapter-

(P.1) Indrividuation 09/14 Rogentan - Individuality, percenticatived necs, the uniqueness of every person. The figures 19-23 (table 1 identity the annotected - modelling (MB) person conductives, moral-ethical integrity, moulded by and shoped by the professions standing in sociecity. - But what do we near by percencerchedness / individuely? - where door it come from ? why is it a fundamental purinciple) - Do we neally bonow individuality when in reality it many be Just Thetonic? Why have Standardised careplans Stendards, policies that, by their generalisation still protend to think (believe / which person cooled ress ? why and Frencis report happen - why the need for 6c's and remember nuses about patient regults, choice, dignity? MUTUALITY It stats with the self - self contradiness - If you treat ones with repeat, if we pende a culture, orgentecter of case, health provision the trade off is when I'm II, when my family are il we have systems in place to help is there annues moral methodien to help others must sincely subcarcaraly near - I'm staning you have to care, we all have apartoplay - and whe tomer of the collective for TI Sexpect the same .. - collective individualition - the comman good / surrival - Human Lights chare (1943) NHS 1948 - 1087 wars 2 autr- rithingni agendas. y - Individualitar / uninesalion - fights to life / no to Aure - Social Cartract, Justice, Equity, Farness, althuism, (Rausis) loci criedie resources to help socreting a indudicent - Social guods - distribution of social guods - health. Safety, Security, utilities, education. - Annetestox promote burnon nogets i individualian/ innovation skey to a clear moral purpose. * lottres of health (4 collectric undertending of right

There is a dear marad importing to "do no hanni," for P2) evidence based processes to do as little horan as possible I and if boun does occur / expected then that is offset by the benefits. P.g Side - Efects from medication.

- because nursing is a purfession in Law (Nursup Order) then there is a Sense that it Should epitonize the trust Placed in it to gelf regulate, identify (Fires to preather issues, id - educational & competencies & ensure the public are as safe as possible - what do the terms peom)
 Tudividualish is collectricen? The safedy of the group, ' Of Society is important and in collectricen, as seen in
 - China as Community States, Sees the group as the prienty - the individual 15 Secondary-
- Individualism pais is the individual which ensures Self interest is a priority - Yet it is the collectre of Individuals with a collectrist idealist sevenan has the capacity to promote both. If there make a Subjugeded imprucified, includeded, M individual them if this state were the non frantice collectre good is markened is all were the non frantice collectre good is markened. Eg N: Konea. Paradoxical instruct where the negatives the Individual is enstruid in Law then a californity or wes.
- Month Does the notion of includuality of nursice, being dust, fair, about lawfull to sponstructly of nursice, being dust, fair, theating people as the would like to be theated (nurtidity) Hence, Self Sering muticality - but promoty what is good, our dust, our obligetian to prosence what is good. The common Good is the good of the group r mendally the good of the individual, a mutical trade of of bene fill. Jo if there is harm to perfect, headed of bedrayal of thust is a duescedy mades on anisotics because of bedrayal, because if if can happen to other, if can happen to us. Therefore, the annotator as a nurse, in the promote of individuality, is the guardien of the proople of Justice for

398

one and all'

(P.3) Individualismy

towers, the what also drives the retree of Individualian) Perhaps a district, or an expectation that all institutions given the electricates arctionsed permission, will at some point betray that trust, because of Selfichness of Serve individuals on the (P.3) Islact to athem of individuals which makes the institution itself the priority. Bothere is need for continued inglance, Self consorting and Consorting of Institutions?

- Individuality & heipmass (empowerment?, Petraps the annotator "thraws" that mistitutions by their very being tend to say one thing, mean one thip yet do another. Phepone VS readility. The annotator as a nuse leather aims petrops to promote the Socred Cartoach, ideals, mellbeing, quality of life though the source of communality. Or yes the mathematical is impationed but public service for the canna good is a way of modelly self sacrifree, keeping a hand in, an influence as institutional wedientimes that may excitedly come to dehunantse individuals. Hence the critican aims to louice the institution theneadies and influence those who wield more priver and activity are withoes and to be instight or commuted are withoes and dudgement.

- I realize is the a strong expensive of person-controdiess because despite the great positivo impacts of the ptrs, in my expensive I have had to defend the notion of my-Salf against forces that are nother toxic. e.g. disempanement and institutions. As a hearth institution I realized the caustant charge P.4 Individuation

Thank appendiced resulted in a decade of practice development expensive and grass voors, bottomy Change to anorcome what could be a toxic environment. I remarked thistop as a health visitor, whose whe it is to effect positive charge in others, "what happens when those employed to empower others feel disquemponend?" What forces are afact to obsempamer? Change which is undemocratic, innegotiated, unconvinting, charge that appears to challenge the very being of the NHS, is a deallarge to individuation. Why? Because if the voting publics here's are not requested or sayplit and Callet agendas are in force, then it is not the (allethie good that is the new prainty, but the profits of kivate health cave, profiteding over the Suffer of others, Hence, anandedor who identifier with particel influences outher NHS may agrice that the public should be trusted to - I think the annotator will be very awave of the NOTS of INTSTANDAUS to delectify and so make the NOTS of INTSTANDAUS to delectify and so make it their pricinity to reinforce the basic principles of Nights and Societal reportabilities. The next generation of nuses are mailded by the last wa Fransitiand process of annotation feedback. Petraper the annotator is also reinforcip, or qualitating their our motives and influence on the putersia? Can Rawls (Ricoeur make Sense afit) The Just is a starting point I think

literature and extracts Consciensvog the annotator seams to have a degree office at times. These is a phenomenan that needs to to understand - what's perforted, what's perceived and what are the likely possibilities -? A second to method if tone can be either projected and meant as it can be i Sometrup that is felt by the receiver, for whatever reason, their past, their memoria, their recall, their Ncterbly menter has the perception and meaning of words and influence on their view of their resonance. The Students who headed endrauely to the annotation waterates they have unresched memories, someting transferred from the past by semetting triggered with prosect. The annotator too will have emotion, thought, tregended by their reflectnic undertailipoptier past expensions - thence, Something can be projected which the annotates themselves may ar may not be awake of (awaveness not withe or - before awaveness to consciences / huncouscions / pre-consciences relater to temperal understanding, how a mancary is processed, recalled and for when purpose Environmental - for teadup and bearing - the annotables mananies Can be used to good effect, in fact a practical form of ameniony poil, informage, intriting, Judgig, proceeding the relevance of though, how though are linked and impact on each other. - This momony poor, is a poor of interpretive bias or inderstanding timber theory to preaction more readily aspectelly & reading award a Sublect or since Save this was expended.

P.2 Reflective Conscitutionesr know! I think what I are begininged to behieve Semetica is that the uniter word is evocatic and thigger (\mathbf{I}) an endoard readien in the reader . Whether it ? 3 the essay content artic annotatory, or the annotably Construct on the Student - both are evocated by the paner; Syndodion Of language. -than can would hind Samane? By the physical great processes that are re-peated, (2007)New- manssin Hers - Pincers et at. memory recall retain to sensory experience - Swell, Fuch, etc. nonony Read transference - repetition of affect related to thiggers. up ycle of recally -As a phenomenon it that thete as it becomes expensioned when it becomes manifest (fum latent sleeping model to (3) be "seen" posseded. (foultes, therdeaper) - Transferences 1932+ 2003 (porallel proces parallel proces 19834 - pulsedire (dont freedor of feelings, thought, activity, and recurrent patterns of communication & behaviours. what many have been forgotten is the affect related to a monony = transference > Entriely individual and unique to part associations affective womany, Repetition of believers / affect int present related to past experience - metaphor may identify more belief systems - symbolised it languge Whent is Frankference? Friendran term "Repetitué expensive of feelups, attindes) defences towards something in the present, uncanspranety displaced for the part "(Swan & Hull, 2007) - ... fast childhood expensions generally (Friend) . Students intervenced (2)

(+3) Reflectrone Consciencess (T) What is canter transferrence? Any reactions to Je something evocative, conscious on forgetten which may 2.2 Something evocative, conscious on forgetten which may 2.2 be misor tepreted whess identified. be missintepreted whess identified. Tone à annétation, evocature words of annétation concop(is in love a content way enact fromstemence or CT Back to Pinaus et al (2007) Question re. any paresponse duplicates a pRIOR recourse - Procus et al research rematarces transferers as a phenomena in Olfactory studies. Olfaction A as a phenomena in Olfactory studies. Olfaction A has emotional responses, carstructure of meaning 1 ordered has emotional responses, carstructure of process thinking learning related to memory. Feedback process thinking learning related to memory, reasoning on exposure is antinued and evolving depending on exposure (neural pruning?). Transferrence, they say needs (neural pruning?). Transferrence, they say needs to occurs as a "precept of another person" (p.634) two is not beneverent defaction - smell is a higher ordered process. In rabbut Studies Smell + reword occur twoych nasal alfacts y receptors and memory - organize patters of activity, in humans, suggerted to be remembered to impact on So - what does Pricoer Suy about transference, menony etc which I can relate to annotation? Babuses OF memory Predical Rebanded and (3) etnico-monal, - why cibuses (auses? Re-call Betwico-monal, - why cibuses (auses? Re-interf - Wounded ideiles to transference Itunt marthentic - Wounded ideiles to transference Itunt marthentic - Practical - Adapterble use of monory which occurs in - Practical - Adapterble use of monory which experience - ethico-moral - Individualitan princide and music All 3 notonly relate to the research partagents but also to myself as a nusse, lecturer, researcher. How do I recall events, proceive (are foot in the part, are foot in the As we "walk" towards the fitne there processes Shift - In lant (forward / back / sodewards - Replection is the key to aux eness